Enabling concentrating solar power in Australia: An investigation of the benefits and potential role of concentrating solar power and non-conventional fuel hybrid plants in Australia's transition to a low-carbon energy future

Juergen Heinz Martin Peterseim

Institute for Sustainable Futures University of Technology, Sydney

Thesis submitted for the PhD in Sustainable Futures

September 2014

STATEMENT OF ORIGINAL AUTHORSHIP

I certify that the work in this thesis has not previously been submitted for a degree and nor has it been submitted as part of the requirements for a degree.

I also certify that the thesis is an original piece of research written by me, except where noted in the text. Any help that I have received in my research work and in the preparation of the thesis itself has been acknowledged. In addition, I certify that all information sources and literature used are indicated in the thesis.

Signature of candidate:

Juergen Heinz Martin Peterseim

I dedicate this thesis to love and hope: love for my wife Anja and my children Lola and Leon, and hope for a bright future for concentrating solar power in Australia.

ACKNOWLEDGEMENTS

Many people supported me in my research and in my personal life throughout this threeyear journey. This includes family, friends, supervisors and colleagues, and it would take too long to thank them all individually. In particular I want to thank my wife Anja and my children Lola and Leon for their support as they provided welcome distractions that took my mind off specific research issues. This gave me time to reconsider and in many cases improve my ideas. I know that during this time I did not always have as much time for my family as I wished, and I want to thank Anja for her understanding and efforts to give me time for my research. Without her support this thesis would not exist.

I want to acknowledge my supervisors Prof. Stuart White, Prof. Udo Hellwig and Dr. Amir Tadros for their continual support, their willingness to help and their optimism. I want to thank Prof. Stuart White for his deep insights into sustainability and energy research. His availability, even at short notice, and his detailed feedback allowed me to solve problems quickly and stay on track. I want to thank Prof. Udo Hellwig, not only for his valuable technical support during my candidature, but also for his mentorship since I entered the energy sector in 2003. I don't think I would be in the position I am in today if it had not been for his support. I also want to thank him for the opportunity to work for him part-time during this research project. This was a significant help for me during my second period of time as a student. Last but certainly not least, I want to thank Dr. Amir Tadros for this for valuable insights into techno-economic modelling. His detailed understanding of power plants in the context of the Australian electricity market was invaluable.

Various people within and outside the university contributed to specific aspects of this research. In particular I want to thank Dr. Deborah O'Connell and Dr. Alexander Herr from CSIRO Ecosystems Sciences, and Sarah Miller from CSIRO Energy Technology, for their contribution to the resource assessment, which is a key component of this work. I also want to thank Frank Klostermann, formerly of Thiess Services Pty. Ltd., for his insights into the industry, and for the financial contribution to the Swanbank case study. I sincerely hope that this proposal will eventually turn into an actual power station. I also want to thank various students and staff at ISF for their friendship, support, insightful discussion and time, which not only helped with the thesis but also contributed to my personal development.

Last but not least I want to thank the University of Technology, Sydney for the confidence they showed in me by awarding me a UTS President's Scholarship and two travel funds. Without the scholarship I would not have been able to do this research and the travel funds allowed me to attend relevant international conferences.

| Statement | of orig | inal authorship | ii | |
|--------------------|--------------------|--|------|--|
| Acknowledgementsiv | | | | |
| Table of cor | Table of contentsv | | | |
| List of Figur | List of Figures | | | |
| List of Table | List of Tables | | | |
| List of publi | catior | IS | xii | |
| List of abbro | eviatio | ons | xiv | |
| Abstract | | | xv | |
| Foreword | | | xvii | |
| 1 Introducti | ion | | 1 | |
| 2 Literature | e revie | w | 4 | |
| 2.1 7 | The Au | stralian electricity market | 4 | |
| 2.1.1 | Curi | ent status | 5 | |
| 2.1.2 | Trar | nsition to a low carbon future | 8 | |
| 2.2 (| Conce | ntrating solar power plants | 13 | |
| 2.2.1 | Hist | ory and technology development | 13 | |
| 2.2.2 | Out | look | 17 | |
| 2.2.3 | Aus | tralian market | 20 | |
| 2.3 N | Non-co | onventional fuels for power generation | 23 | |
| 2.3.1 | Ene | rgy from waste | 23 | |
| 2.3.2 | Ene | rgy from biomass | 28 | |
| 2.3.3 | Aus | tralian market | 30 | |
| 2.4 0 | CSP hy | brid benefits & challenges | 34 | |
| 2.5 (| CSP hy | brid plants | 38 | |
| 2.5.1 | CSP | -Natural gas | 38 | |
| 2.5 | .1.1 | Commercial references | 38 | |
| 2.5 | .1.2 | Concepts | 40 | |
| 2.5.2 | CSP | -Coal | 42 | |
| 2.5 | .2.1 | Commercial references | 42 | |
| 2.5 | .2.2 | Concepts | 43 | |
| 2.5.3 | CSP | -non-conventional fuels | 44 | |
| 2.5 | .3.1 | Commercial references | 44 | |
| 2.5 | .3.2 | Concepts | 45 | |

TABLE OF CONTENTS

| 2.5.4 | CSP | P-geothermal | 47 | | |
|-------------|---|--|----------|--|--|
| 2.6 | Introd | uction to transition management and transition theory | 49 | | |
| 2.7 | 2.7 Conclusion from the literature review and research gaps | | | | |
| 3 Research | desigr | n | 57 | | |
| 3.1 | 3.1 Research questions | | | | |
| 3.2 | Metho | ods | 60 | | |
| 3.2.1 | Wo | rkshops and interviews | 60 | | |
| 3.2 | 2.1.1 | CSP technology selection | 61 | | |
| 3.2 | 2.1.2 | Implementation barriers | 61 | | |
| 3.2.2 | Mo | delling | 62 | | |
| 3.2 | 2.2.1 | Techno-economic modelling | 62 | | |
| 3.2 | .2.2 | GIS modelling | 63 | | |
| 3.2.3 | Mul | Iti-criteria decision-making | 64 | | |
| 3.2.4 | Case | e studies | 65 | | |
| 3.3 | Theore | etical framework | 67 | | |
| 3.4 | Resear | rch ethics | 72 | | |
| 4 Results a | nd disc | cussion | 73 | | |
| 4.1 | CSP hy | vbrid categories and energy source combinations | 73 | | |
| 4.2 | CSP-nc | on-conventional fuel potential and plant area identification in Aust | ralia 82 | | |
| 4.3 | CSP te | chnology selection | 97 | | |
| 4.4 | Techno | o-economic optimisation | 111 | | |
| 4.4.1 | Ider | ntical steam parameter from CSP and biomass components | 111 | | |
| 4.4.2 | Exte | ernal CSP steam superheating with biomass | 124 | | |
| 4.4.3 | Futu | ure CSP–EfB and CSP–EfW hybrid plants | 135 | | |
| 4.5 | Impler | nentation barriers | 138 | | |
| 4.5.1 | Sign | nificant Australian barriers to CSP | 139 | | |
| 4.5 | 5.1.1 | Social barriers | 140 | | |
| 4.5 | 5.1.2 | Technical barriers | 142 | | |
| 4.5 | .1.3 | Environmental barriers | 145 | | |
| 4.5 | 5.1.4 | Economic barriers | 146 | | |
| 4.5 | 5.1.5 | Policy barriers | 148 | | |
| 4.5.2 | Rati | ing results | 151 | | |
| 4.5 | 5.2.1 | CSP-only plants | 153 | | |
| 4.5 | 5.2.2 | CSP hybrid plants | 154 | | |
| 4.5 | 5.2.3 | CSP-only versus hybrid rating differences | 155 | | |

| 4.5.3 | Disc | ussion | 157 |
|-------------|---------|---|-----|
| 4.5.4 | Con | clusions | 159 |
| 4.6 | Case st | udies | 160 |
| 4.6.1 | Swa | nbank, CSP-multiple feedstock hybrid | |
| 4.6 | 5.1.1 | SolarPACES 2012 conference paper | |
| 4.6 | 5.1.2 | Environmental analysis | |
| 4.6 | 5.1.3 | Hybrid versus CSP-only cost comparison | 175 |
| 4.6 | 5.1.4 | Socio-economic benefits | 175 |
| 4.6 | 5.1.5 | Economic and socio-ecological renewal at Swanbank | 178 |
| 4.6.2 | Grif | ith, CSP-single feedstock hybrid | |
| 4.6 | 5.2.1 | SolarPACES 2013 conference paper | 181 |
| 4.6 | 5.2.2 | Socio-economic benefits | 192 |
| 4.7 | CSP hy | brids as a pathway to a low carbon future | 194 |
| 5 Future re | esearch | | 206 |
| 6 Conclusio | ons | | 209 |
| References | s | | 214 |
| Appendix . | | | 236 |
| | | | |

LIST OF FIGURES

| Figure 1: Australia's electricity generation 2011-12 by energy source (Bureau of Resources |
|---|
| and Energy Economics 2013); Other includes oil, bioenergy, solar PV, and multi-fuel |
| fired power plants |
| Figure 2: Changes in electricity generation and emissions in the NEM (pitt&sherry 2014) 6 |
| Figure 3: Changes in electricity generation by fuel type in the NEM (pitt&sherry 2014) 6 |
| Figure 4: Australian electricity generation from renewable energy (Bureau of Resources and |
| Energy Economics 2013)7 |
| Figure 5: Electricity price indices for households and businesses, Australia (Bureau of |
| Resources and Energy Economics 2013)8 |
| Figure 6: Australia's emissions trends, 1990 to 2020 (Department of Climate Change and |
| Energy Efficiency 2012)9 |
| Figure 7: Forecast renewable energy investment – value of construction and capacity added |
| (Macromonitor 2013) 11 |
| Figure 8: Share of electricity generation by energy type, prepared with data from Syed |
| (2012) |
| Figure 9: Al Meadi pumping station using the five parabolic troughs with direct steam |
| generation (Stinnesbeck 1914)13 |
| Figure 10: One of three solar tower systems of the 392 MWe Ivanpah power station, USA 15 |
| Figure 11: Annual electricity capacities and generation for CSP and PV from 2011-18 |
| (International Energy Agency 2013)17 |
| Figure 12: Tariff and levelised cost of energy development above DNI level; Percentage |
| compared to reference plant in Spain with a DNI of 2,084 kWh/m²/a at 100 per cent |
| (AT Kearney & ESTELA 2010)19 |
| Figure 13: CSP project pipeline by technology in per cent of total CSP projects as per 1 $^{ m st}$ |
| March 2013 (SBC Energy Institute 2013) 20 |
| Figure 14: Direct normal irradiation for potential global CSP sites (Trieb et al. 2009)21 |
| Figure 15: Energy from waste conversion technologies (Kaltschmitt 1998) |
| Figure 16: a: Waste incineration plant Bullerdeich in Hamburg in 1896 (Vehlow 2004) and b: |
| modern Energy from Waste plant in Tokyo, Japan (right) |
| Figure 17: Rate of recycling versus incineration with energy recovery of municipal waste, |
| 2005 for the EU (European Environment Agency 2007) |
| Figure 18: CSP power top-up (left) or fuel saver (right) option |

| Figure 19: a:75 MWe equivalent CSP steam boost to Martin Next Generation power station |
|---|
| in the USA (Florida Power & Light Company 2010) and b: 100 MWe Shams One plant |
| (Goebel & Luque 2012) |
| Figure 20: a: 228 MWe solar tower ISCC concept (Peterseim et al. 2012c) and b: 4.6 MWe |
| Solugas tower in Spain (Quero et al. 2013)41 |
| Figure 21: Kogan Creek Solar Boost project under construction as per October 2013 43 |
| Figure 22: First CSP-biomass hybrid plant in Spain, 22.5 MWe Termosolar Borges, Spain 45 |
| Figure 23: Schematic diagram of the hybrid solar-geothermal power plant (Zhou, Doroodchi |
| & Moghtaderi 2013) 48 |
| Figure 24: Transition management cycle (Loorbach 2010)50 |
| Figure 25: Multi-level perspective (Geels 2002) |
| Figure 26: Emerging technical trajectory carried by local projects (Geels & Raven 2006, p. |
| 379) |
| Figure 27: Research outline 57 |
| Figure 28: Research structure showing the sequence of the research components (black |
| arrows) and the information flow (dotted arrows)58 |
| Figure 29: Case study locations in Ipswich, Queensland, and Griffith New South Wales 66 |
| Figure 30: Multi-level perspective on transitions (Geels & Schot 2007, p. 401) |
| Figure 31: Reconfiguration pathway (Geels & Schot 2007)71 |
| Figure 32: Potential efficiency increase (black line) and range of cost impact (red dotted |
| lines) based on future steam parameters for a 100 MWe (net) CSP-EfB hybrid plant |
| with air cooling at Mildura, Australia136 |
| Figure 33: Participant breakdown for implementation barrier ranking |
| Figure 34: Barrier rating results for CSP-only (orange) and CSP hybrid plants (green) 152 |
| Figure 35: Barrier category ratings for CSP-only plants |
| Figure 36: Barrier category ratings for CSP hybrid plants |
| Figure 37: Detrimental CSP implementation cycle with intervention option. Adapted from |
| Effendi and Courvisanos (2012)158 |
| Figure 38: Swanbank site with the proposed CSP–EfB and CSP–EfW hybrid plant (CSP = |
| yellow and EfB/EfW = green), the existing coal fired Swanbank B power plant and the |
| existing gas fired Swanbank E power plant (red squares), and landfill (blue polygon) |
| |
| Figure 39: Vision for an eco-industrial transition – map and concept, Baumann et al. (2012) |
| |
| Figure 40: Potential site for the CSP–EfB hybrid plant near Griffith |

| Figure 41: Multi-level perspective for different renewable energy technologies in the |
|---|
| Australian electricity generation market; Blue = hydro, grey = wind, green = biomass, |
| yellow = PV, orange = CSP, and red = others; Adapted from Geels & Schott (2007) 200 |
| Figure 42: Multi-level perspective for CSP-only and hybrid technologies in the Australian |
| electricity generation market. Adapted from Geels & Schott (2007) |
| Figure 43: Possible reconfiguration pathway for the implementation of CSP technologies in |
| the Australian electricity generation market; Red squares = CSP add-ons to existing |
| power plants, green triangles = new CSP hybrid plants, and yellow pentagons = new |
| CSP-only plants. Adapted from Geels & Schott (2007) |
| Figure 44: Overlay of DNI (PIRSA Spatial Information Services 2009) with mine sites in |
| Australia (Geoscience Australia 2010) 207 |

LIST OF TABLES

| Table 1: Biomass generation capacity in MWe per state and fuel in 2009 (Stucley et al. |
|--|
| 2012) updated with the recently commissioned 36 MWe Mackay plant using bagasse |
| in Queensland (Biomass Power & Thermal Magazine 2011) |
| Table 2: Ranking differences between CSP-only and hybrid plants showing total average and |
| group averages for researcher (ΔR), owners/operators (ΔO), consultants (ΔC), |
| technology provider (Δ TP) and government (Δ G)156 |
| Table 3: Swanbank power plant cost breakdown and investment distribution |
| Table 4: Griffith power plant cost breakdown and investment distribution |

LIST OF PUBLICATIONS

Relevant publications:

- Peterseim J.H., Herr, A., Miller, S., White, S., O'Connell, D.A., 2014, Concentrating solar power/alternative fuel hybrid plants: Annual electricity potential and ideal areas in Australia, *Energy*, vol. 68, pp. 698-711.
- Peterseim, J.H., Hellwig, U., Tadros, A., White, S., 2014. Hybridisation optimization of concentrating solar thermal and biomass power generation facilities. *Solar Energy*, vol. 99, 203–214.
- Peterseim, J. H., Tadros, A., Hellwig, U., White, S., 2014. Increasing the efficiency of parabolic trough plants using thermal oil through external superheating with biomass, *Energy Conversion and Management*, vol. 77, pp. 784–793
- Peterseim, J.H., White, S., Tadros, A., Hellwig, U., 2014. Concentrating solar power hybrid plants - enabling cost effective synergies. *Renewable Energy*, vol. 67, pp. 178-185.
- Peterseim, J.H., White, S., Tadros, A., Hellwig, U., 2013. Concentrated solar power hybrid plants, which technologies are best suited for hybridisation? Renewable Energy, vol. 57, 520–532.
- Peterseim, J.H., Tadros, A., White, S., Hellwig, U., Landler, J., Galang, K., 2013. Solar tower-biomass hybrid plants – maximizing plant performance. *Energy Procedia*, vol. 49, no. SolarPACES 2013 conference special, pp. 1197–1206
- Peterseim, J.H., Tadros, A., White, S., Hellwig, U., Klostermann, F., 2012.
 Concentrated solar power / Energy from Waste hybrid plants creating synergies.
 In: *SolarPACES Conference*, Marrakech.
- Peterseim, J.H., White, S., Hellwig, U., Tadros, A., Vanz, E., 2012. Pre-feasibility study for a multi-fuel / concentrated solar power hybrid plant at Swanbank, QLD.
 Prepared for Thiess Services Pty Ltd by the Institute for Sustainable Futures, University of Technology, Sydney, Unpublished report.

Other publications:

- Peterseim, J.H., Tadros, A., Hellwig, U., White, S., 2013. Integrated solar combined cycle plants using solar towers with thermal storage to increase plant performance. In: ASME Power Conference, Boston.
- Peterseim, J.H., Hellwig, U., Endrullat, K., 2013. Parallel flow boiler designs to minimise erosion and corrosion from dust loaded flue gases. In: ASME Power Conference, Boston.
- Rutovitz, J., Peterseim, J., Elliston, B., Harris, S., Mohr, S., Lovegrove, K., Want, A., Langham, E., MacGill, I., 2013. Breaking the solar gridlock. Potential benefits of installing concentrating solar thermal power at constrained locations in the NEM. Prepared for the Australian Solar Thermal Energy Association (AUSTELA) by the Institute for Sustainable Futures, UTS, Sydney.
- Peterseim, J.H., White, S., Tadros, A., Hellwig, U., 2012. Integrated Solar Combined Cycle plants using solar power towers to optimise plant performance. In: *SolarPACES Conference*, Marrakech.
- Peterseim, J.H., Hellwig, U., Guthikonda, M., Widera, P., 2012. Quick start-up auxiliary boiler/heater – optimizing solar thermal plant performance. In: *SolarPACES Conference*, Marrakech.
- Baumann, C., Asker, S., Giurco, D., Peterseim, J.H., White, S., 2012. ECO-INDUSTRIAL TRANSITION' A vision for economic and socio-ecological renewal at Swanbank. Prepared for Thiess Services Pty Ltd by the Institute for Sustainable Futures, University of Technology, Sydney, Australia.
- Peterseim, J.H., 2012. Energy Efficiency Opportunity Assessment at Tarong Power Station. Prepared for the Department of Resources, Energy and Tourism by the Institute for Sustainable Futures, Sydney, Unpublished report.
- Memary, R., Giurco, D., Prior, T.D., Mason, L. M., Mudd, G.M., Peterseim, J.H., 2011. Clean energy and mining - future synergies. In: *Second International Future Mining Conference*. The AusIMM (The Mineral Institute), Sydney, Australia.
- Peterseim, J.H., Hellwig, U., 2011. Water circulation calculation for Concentrated Solar Thermal Plants. In: *SolarPACES Conference*, Granada.

LIST OF ABBREVIATIONS

| AHP | analytical hierarchy process | MLP | multi-level perspective |
|-----------------|-----------------------------------|------|---------------------------------|
| ASTRI | Australian solar thermal research | MSW | municipal solid waste |
| | initiative | MW | megawatt |
| AU\$ | Australian dollar | MWh | megawatt hour |
| b | billion | MWth | Megawatt thermal |
| CapEx | capital expenditure | NEM | national electricity market |
| CO ₂ | carbon dioxide | OpEx | operational expenditure |
| CSP | concentrating solar power | PPA | power purchase agreement |
| DNI | direct normal irradiance | PV | photovoltaic |
| EfB | energy from biomass | R&D | research and development |
| EfW | energy from waste | RDF | refused derived fuels |
| EPC | engineering, procurement and | RECs | renewable energy certificates |
| | construction | RET | renewable energy target |
| GIS | geographic information system | SEGS | solar energy generation systems |
| GWh | gigawatt hour | SNM | strategic niche management |
| h | hours | SRF | solid recovered fuels |
| HRSG | heat recovery steam generator | t | tonnes |
| ISCC | integrated solar combined cycle | t/h | tonnes per hour |
| kW | kilowatt | TES | thermal energy storage |
| kWh | kilowatt hour | TWh | terrawatt hour |
| LCOE | levelised cost of electricity | US\$ | U.S. dollar |
| m | million | | |
| | | | |

ABSTRACT

After decades of stability the Australian electricity market is undergoing changes. Current government targets aim to reduce greenhouse gas emissions by 5% and raise renewable electricity production to 45 TWh by 2020. In addition, increases to natural gas prices, aging generation assets and falling electricity demand have had an impact in recent years.

Uncertainties exist around current policies, including the carbon pricing mechanism and the renewable energy target, but in light of Australian and international ambitions to lower greenhouse gas emissions the deployment of renewable energy technologies is essential. In recent years wind and photovoltaic installations have shown the highest renewable energy growth rates while concentrating solar power has struggled, despite Australia having some of the best natural resources for concentrating solar power in the world and some selected government funding. Reasons for the slow uptake include the comparatively high cost and lack of financial incentives. While technology costs are expected to decrease by up to 40% by 2020 through deployment as well as research and development, other cost reduction options have to be identified to promote short-term implementation in electricity markets such as Australia where the wholesale cost is low. To overcome the cost problem and to address other relevant implementation barriers this research analyses the hybridisation of concentrating solar power with biomass and waste feedstocks.

The results of this research include:

- a recommendation for a categorisation system for CSP hybrid plants based on the degree of interconnection of the plant components
- the availability of combined resources to generate up to 33.5 TWh per year and abate 27 million tonnes CO₂ annually
- an analysis of the most suitable CSP technologies for hybridisation
- a technology comparison showing CSP cost reductions through hybridisation of up to 40%
- the identification of cost differences of up to 31% between different hybrid concepts
- an analysis showing that the current economic and policy settings are the most significant implementation barriers
- two case studies with different biomass and waste feedstocks requiring power purchase agreements of AU\$ 100-155/MWh.

Based on the various benefits of concentrating solar power hybrid plants, this research analyses the potential role of this technological pairing in Australia's transition to a low carbon energy future. The research concludes that concentrating solar power hybrid plants, not only hybridised with biomass and waste feedstocks, can immediately enable a lower cost deployment of concentrating solar power facilities in Australia. The technology, deployment and operation of the first hybrid installations would provide market participants with valuable lessons and would have the potential to reconfigure the electricity market towards more sustainable generation. This could help promote the development of future low-cost concentrating solar power plants in Australia.

FOREWORD

When I started considering a PhD candidature in 2010 I already had a few potential topics in mind that derived from observations I had made since entering the energy business in 2003. I worked as an industrial engineer in several areas, including project management and business development, for the German boiler design companies La Mont-Kessel GmbH & Co. KG and ERK Eckrohrkessel GmbH. This allowed me to develop a detailed understanding of current issues with solid, liquid and gaseous fuel fired water tube boiler systems, and of their impact on power plant efficiency, reliability and cost. My early focus was on energy from biomass and from waste systems as well as work on compact boiler and heat exchanger systems. After moving to Australia in 2007 I continued work in these fields but also expanded into heat recovery and natural gas fired boilers.

The good resource for solar energy in Australia, and my interest in Rankine cycle systems, shifted my attention to concentrating solar power. The technology was immediately appealing due to its futuristic appearance, its low carbon intensity, and the availability of mature equipment for most of the plant. In late 2010 I was awarded a UTS scholarship and since commencing this research in March 2011 my interest in concentrating solar power has continued to grow. The work I have done for my PhD has enabled me to expand my knowledge not only through theoretical work, such as a literature review and thermo-economic modelling, but also through the exchange of ideas and cooperation with industry partners, both those I had known previously and others I have met during the last three years.

I sincerely hope that this thesis will contribute to the deployment of concentrating solar power plants in Australia and I am looking forward to further engaging with the technology for the foreseeable future.

1 INTRODUCTION

After decades of stability the Australian electricity market is undergoing changes due to a recently introduced carbon pricing mechanism, slowly rising renewable energy penetration, energy efficiency measures, rising natural gas prices, falling electricity demand in recent years, and an aging fleet of coal fired power plants. Despite current uncertainties around the carbon pricing and the renewable energy mechanisms the market is likely to continue its transition due to the increasing certainty on anthropogenic climate change and the need to respond to it, e.g. existing greenhouse gas abatement targets for 2020 and 2050, continuously falling costs for renewable energy technologies and rising natural gas prices.

In 2010–11 10% of Australia's electricity came from renewable sources. This was significantly below the world average of 20% but the country does have the resources, such as wind, solar, biomass, and sites suitable for hydro and tidal generation, to be a leader in renewable energy. Of particular interest for this research is solar radiation, for which Australia has one of the best resources in the world. However, renewable energy uptake has been slow in the past as the country has vast and easily accessible fossil fuel resources, which has led to low wholesale electricity prices, particularly from coal fired power generation. In such an economic environment renewable energy technologies struggle to enter a market where fossil fuels are subsidised, externality costs are not fully accounted for, and renewable energy incentives are limited. This is a particular problem for concentrating solar power (CSP) which is currently a more costly alternative than wind and photovoltaic (PV) power generation. However, CSP is of strategic importance for Australia as it is a renewable energy technology with proven energy storage capabilities that can balance other intermittent renewable energy technologies, such as wind and PV. The reason for the higher CSP cost is its limited global deployment of 4 GWe in 2013 compared to 128 GW for PV and 321 GWe for wind. Therefore, the technology providers have less experience and access to economies of scale. While there are utility-scale reference plants in operation, the industry is still at an early stage of technology implementation.

A variety of different CSP technologies exist. Parabolic trough plants are the most mature but their cycle efficiency is limited. The first commercial plants commenced operations in the late 1970s to early 1980s and current installations reach capacities of 280 MWe with many units having thermal energy storage (TES). Commercial solar tower and Fresnel plants entered the market some 20 years later and have a higher efficiency and cost reduction potential but their development has not yet reached the maturity of parabolic trough systems. However, current solar tower installations reach capacities of 130 MWe. They are the CSP technology with the largest TES capacities, having a plant with 15 h TES capacity in operation and a plant with a 17.5 h capacity about to be constructed. Currently, operational Fresnel plant capacities are limited to 30 MWe but a 2x 125 MWe unit is under construction. Unlike the aforementioned CSP technologies, TES has not been incorporated in Fresnel plants but is being investigated. Currently, both solar tower and Fresnel technologies are demonstrating their technical and economic capabilities, and if the recently and soon-to-be-commissioned plants meet industry expectations both will gain a larger market share.

Technology improvements and continuous deployment are expected to lower plant costs by up to 40% by 2020 but this estimate involves uncertainties around the availability of government support. Moreover, learning from technology deployment, and incorporating the resulting developments into new plants is a time-consuming process. One option to fast-track the 2020 CSP cost reduction target is the hybridisation of CSP with other energy sources. In other countries first reference plants with up to 75 MWe CSP energy equivalent already operate successfully with natural gas, coal and biomass. The success of these plants confirms the cost benefits that CSP hybrids provide. In Australia one small 9.3 MWth CSP retrofit is already operational at the 2 GWe Liddell coal-fired power station while another 44 MWe retrofit is under construction at the 750 MWe Kogan Creek power station. Additionally, some new CSP hybrids with natural gas and biomass are being investigated. In the past only two CSP projects have been awarded significant state and federal government funding but the developers could not secure the remaining funds they needed and had to withdraw their proposals, which highlights the cost problem CSP faces in entering the Australian electricity market. The cost of CSP hybrids is lowered relative to CSP-only plants due to the joint use of plant equipment, such as steam turbines and condensers. In addition, hybrid plants involve less financial risk as they can have much smaller capacities than CSP-only plants which if they are to be economically viable, must be large enough to achieve economies-of-scale. In addition lower-cost CSP hybrids also encounter lower social, technical, environmental and policy implementation barriers. Despite these and other benefits the hybridisation of CSP with fossil fuel generation has to be considered carefully to ensure that such plants provide net environmental benefits, are not used to justify the lifetime expansions of old generation assets, and avoid fossil fuel technology lock-in. While acknowledging that project-dependent CSP-fossil fuel hybrids can have a positive net environmental impact, this research investigates CSP hybrids which use energy from biomass (EfB) and from waste (EfW) as such hybrids are a low carbon intensity power

generation option which is able to provide the learnings the CSP industry needs while providing energy at a competitive cost with high environmental benefits.

To analyse the potential role CSP–EfB and CSP–EfW hybrid plants can have in Australia's transition to a low-carbon future, a detailed understanding of the available technologies, technical concepts, resources and implementation barriers is required. It is unlikely that one niche technology, such as CSP-EfB and CSP-EfW hybrids, can significantly change the current Australian electricity market towards environmental sustainability. Therefore other CSP hybrid options, described in Section 4.1, are considered as well. To understand the combined CSP, EfB and EfW resources, and to identify the total potential for such hybrids, a detailed geospatial analysis considering various technical, economic and environmental constraints was carried out and the results are presented in Section 4.2. Currently, various technologies are available which can be hybridised in many different ways. Identifying the most suitable CSP technologies for hybridisation, not only with EfB and EfW but also with natural gas and coal plants, is the aim of Section 4.3 while the identification of the currently best high solar share CSP–EfB and CSP–EfW hybrid concepts is the aim of Section 4.4. It has been observed in the current electricity market that the uptake of CSP has so far been very limited and therefore implementation barriers must exist. The process of identifying these barriers and assessing their importance for CSP-only and CSP hybrid plants is discussed in Section 4.5, which also examines the perspectives that the various stakeholders in energy projects have about these barriers. To assess the benefits that CSP-EfB and CSP-EfW hybrid plants can provide in actual situations, two case studies were designed and analysed and the results are shown in Section 4.6. The first case study investigates a CSP-multi-fuel plant in Queensland which uses various biomass and waste feedstocks and the second is on a CSP-single-fuel plant in New South Wales which uses straw as the biomass feedstock.

Based on the various findings in Sections 4.1 to 4.6, Section 4.7 analyses the potential role CSP–EfB and CSP–EfW hybrid plants can play in the implementation of CSP technologies in Australia's low cost wholesale electricity market. To broaden the impact of CSP–EfB and CSP–EfW hybrid concepts, a brief discussion of other CSP hybrid options is provided. The potential role of CSP–EfB and CSP–EfW hybrid plants is analysed with the transition management framework. This framework is well suited as it can support the description and analysis of the various aspects of a transition process. It has been used to describe a variety of transition processes, including energy transition research, in other countries worldwide. The technical integration concepts in this work are widely transferable to other Rankine cycle hybrid systems, and this increases the potential for CSP hybrid plants to contribute to the energy transition.

2 LITERATURE REVIEW

This chapter aims to provide a comprehensive overview of the relevant literature for this research. It covers technologies, economics, policy, and the transition management research framework. This literature review is based on journal papers and reports from reputable agencies but to incorporate the latest developments it also includes online references from institutional sources and technology providers. Without them, an up-to-date view of CSP, EfB and EfW developments is not possible.

2.1 The Australian electricity market

The scientific consensus on anthropogenic climate change is overwhelming (Cook et al. 2013) and according to the fifth assessment by the Intergovernmental Panel on Climate Change the future impact of anthropogenic climate change on the Australian environment and economy is expected to be significant, e.g. ecosystem vulnerability, rainfall changes, higher frequency/intensity of extreme climatic events, and risks to coastal infrastructure and agricultural production (Reisinger et al. 2014). Electricity generation is the largest source of greenhouse gas emissions in Australia (Department of Climate Change and Energy Efficiency 2012) and the support of this industry is essential to lower future greenhouse gas emissions, which is the key contributor to anthropogenic climate change. Various renewable energy resources are available in Australia and these have the potential to provide low greenhouse gas emission electricity.

The Australian electricity market is a concentrated one, with a few vertically integrated companies providing generation, transmission, and distribution capacity. Large capacity centralised power stations dominate the power generation portfolio with only some distributed renewable energy and natural gas fired co- and trigeneration facilities in operation. Three transmission–distribution networks exist: the National Electricity Market (NEM) which stretches along the eastern seaboard from Queensland to South Australia, and the South West Interconnect and North West Interconnect Systems in Western Australia. The NEM covers the majority of population and industry in Australia and is by far the biggest electricity network. Remote developments, such as cities like Alice Springs or mine sites, are powered by off-grid power systems, mostly using natural gas or diesel.

This section provides an introduction into this market by highlighting its current status and the policies in place to meet the targets which have been set for 2020 and for 2050.

2.1.1 Current status

Australia has abundant fossil and renewable energy resources (Bureau of Resources and Energy Economics 2013) but the low cost of fossil fuels, the failure of market prices to reflect external costs, and subsidies to the fossil fuel industry (Riedy 2007) have favoured the use of fossil fuels over renewables. The abundance of Australia's coal resources are reflected in their 69% share of the nation's 2011–12 electricity production of 254.7 TWh (see Figure 1). Historically, coal has had an even higher share, over 75% of annual electricity generation (Australian Bureau of Agricultural and Resource Economics 2011), but the combination of the recently introduced carbon pricing mechanism (Australian Government 2011), falling electricity demand in recent years (pitt&sherry 2014) and the advanced age of the majority of Australia's power stations led to the recent mothballing or closure of some coal fired units, such as 2x 350 MWe units at Tarong power station (Hyslop 2013) and 1,400 MWe Munmorah power station (Harvey 2012), and hence lower annual generation and greenhouse gas emissions. . Figure 2 shows the electricity generation and emission developments in the NEM from June 2006 to December 2013. Figure 3 shows the decrease in greenhouse gas emissions in recent years caused by natural gas and wind generation displacing higher carbon intensity coal fired generation. For the period 2013–14 electricity demand in the NEM is expected to further decrease by 1.3% (Australian Energy Market Operator 2014). Across Australia natural gas fired generation increased in 201-12, and with 20% of annual electricity it was the second-largest fuel source for power generation. Natural gas provided twice as much electricity as all renewable energy sources combined (Bureau of Resources and Energy Economics 2013).



Figure 1: Australia's electricity generation 2011-12 by energy source (Bureau of Resources and Energy Economics 2013); Other includes oil, bioenergy, solar PV, and multi-fuel fired power plants



Figure 2: Changes in electricity generation and emissions in the NEM (pitt&sherry 2014)



Figure 3: Changes in electricity generation by fuel type in the NEM (pitt&sherry 2014)

Figure 4 shows that the dominant contributor, today and historically, to Australia's 10% annual renewable electricity was hydro power which provided two-thirds of the nation's renewably generated electricity in 2010–11. Wind is the second-most significant source, followed by biomass, biogas and PV (Bureau of Resources and Energy Economics 2013). No commercial CSP plants were operational in 2010–11 and currently only a small CSP boost at the Liddell coal fired power plant is operational. Despite excellent renewable resources, annual renewable electricity generation has increased only moderately since 2001 and in 2011 it was still well below the world average of 20% (International Energy Agency 2013). However, the percentage of electricity from renewable sources in the NEM increased to 14.3% in 2013 (pitt&sherry 2014). National data and a breakdown by source are not yet publicly available for 2013. The strongest renewable energy growth since 2004–05 came from wind and PV with wind generation increasing by 31% and PV generation increasing by 41% (Bureau of Resources and Energy Economics 2013).



Figure 4: Australian electricity generation from renewable energy (Bureau of Resources and Energy Economics 2013)

A variety of policy, economic, and socio-cultural barriers, which affect not only renewable energy use but also energy efficiency and demand management, are responsible for the comparatively slow uptake of renewable electricity (Byrnes et al. 2013; Dunstan, Ross & Ghiotto 2011; Mckenzie & Howes 2006). While similar barriers exist elsewhere in the world (Moomaw et al. 2011) Australia's abundance of fossil fuels has historically resulted in policy bias against renewable energy (Schläpfer 2009). In fact a recent study (Jackson 2012) found that in 2008 Australia ranked 16th out of 19 G20 countries in its ability to deal with a low carbon emissions world. It was the only G20 country whose ability to cope with a low carbon emissions world decreased since 1995. Reasons for Australia's low score include the small renewable energy contribution and the country's high per capita emissions. However, the recent introduction of the carbon pricing mechanism is likely to have improved this poor ranking.

Historically, electricity prices have been low in Australia and they are still low compared to other OECD countries (Bureau of Resources and Energy Economics 2013). This is one of the main reasons for the absence of CSP plants as they currently have higher levelised costs of electricity (LCOE) than wind or biomass. However, since 2006 business and particularly households have experienced above average electricity price rises (see Figure 5), with the main cost components being distribution (42%), wholesale (24%), retail (13%), transmission (8%), and the large-scale renewable energy target (7%) (Australian Energy Market Commission 2011). From 2013 to 2016 household electricity prices are expected to stabilise

in most states except for Queensland and the Northern Territory where further increases of 8.6% and 6.9% are expected (Australian Energy Market Commission 2013). Despite the significant retail electricity price rises, wholesale electricity prices have not increased sufficiently, currently around AU\$ 50/MWh (RepuTex 2013), to make CSP and other renewable plants viable. This is the case even with the inclusion of the current AU\$ 30/MWh price premium for renewable energy certificates (RECs) (RepuTex, 2013) and a carbon emission price of AU\$24.15/t CO2 in 2013–14 (Australian Government 2011). The proposed repeal of the carbon pricing mechanism by the recently elected Liberal government (The Parliament of the Commonwealth of Australia 2013b) is expected to lower average wholesale electricity prices to AU\$ 31/MWh in 2014 (RepuTex, 2013) which would slow renewable energy growth by reducing the financial incentive and adding market uncertainty around the development of new projects.



Figure 5: Electricity price indices for households and businesses, Australia (Bureau of Resources and Energy Economics 2013)

2.1.2 Transition to a low carbon future¹

According to a recent Australian survey 80% of the interviewees accept that climate change is happening and that human activity significantly contributes to it (Leviston et al. 2014). Despite the high acceptance climate change does not rank high as an important social or environmental concern (Leviston et al. 2014), since issues such as employment, health, water shortages and pollution are considered more important and immediate. Globally, climate change is expected to increasingly impact on the social world including the displacement of millions of people, human and food security, violent conflict and human

¹ The current political and legislative environment in relation to climate change and emissions reduction is a very dynamic area in Australia and the information provided are valid until the submission date of this thesis, the 9th April 2014.

health (Barnett & Adger 2007; Nordås & Gleditsch 2007; Reuveny 2007). It will be increasingly difficult for Australia to isolate itself from these developments.

The Australian government's commitment to climate change mirrors the public acceptance and its priority ranking shown in the aforementioned survey. The current government policy acknowledges climate change is occurring but their actual commitments to lower greenhouse gas emissions are weaker than that of many comparable countries, such as the USA and the UK (Climate Change Authority 2013). Australia is committed to reducing its greenhouse gas emissions by 5% by 2020, compared to the year 2000, and the previous government was prepared, and had bipartisan support, to increase this target to 25% if both developed and developing nations in the international community agreed on more ambitious goals (see Figure 6) (Department of Climate Change and Energy Efficiency 2012). The policy of the recently elected government is not yet known and announcements are expected in the lead-up to the 2015 UN climate conference in Paris (Hunt 2014).



Figure 6: Australia's emissions trends, 1990 to 2020 (Department of Climate Change and Energy Efficiency 2012)

Currently, the two main mechanisms to attain the 2020 greenhouse gas emissions reduction target are the Renewable Energy Target (RET) of 45 TWh by 2020 (Climate Change Authority 2012) and the carbon pricing mechanism (Australian Government 2011). The RET consists of 41 TWh electricity from large-scale plants and 4 TWh from small scale renewable energy installations (Climate Change Authority 2012). The policy is currently under review (Hunt & Macfarlane 2014) and this might result in a lower 2020 target which is adding to investment uncertainty. Australia's current carbon pricing mechanism has a fixed price period without a pollution cap until 1st July 2015 when it is to transform to a flexible market price with a pollution cap on annual greenhouse gas emissions thereafter

(Australian Government 2011). However, the recently elected Liberal government plans to replace this policy with their "direct action" policy. The Emissions Reduction Fund is a key part of the policy and consists of a reverse auction process to buy back the lowest-cost abatement options (Commonwealth of Australia 2013). The Australian Renewable Energy Agency and the Clean Energy Finance Corporation are two entities providing funding for renewable energy projects (Australian Renewable Energy Agency 2014; Clean Energy Finance Corporation 2014) but the latter is slated to be shut down by the recently elected government (The Parliament of the Commonwealth of Australia 2013a).

The current policy uncertainties around the proposed repeal of the carbon pricing mechanism and the simultaneous commissioning of several liquefied natural gas export facilities, linking Australia to world natural gas prices, could see annual coal generation in 2014 return to more than 80% of electricity generation (RepuTex 2013). Average natural gas prices are expected to rise from a current average of AU\$ 6/GJ to AU\$ 9/GJ in 2020. Electricity generation from natural gas without a carbon pricing mechanism could therefore halve over that period (RepuTex 2013). A first sign of this trend actually occurring is the mothballing of a large natural gas fired power plant and the replacement of its capacity through the re-commissioning of an old coal fired unit (Hepworth 2014). However, a recent construction forecast from 2014–18 for utilities in Australia (Macromonitor 2013) predicts ongoing deployment of natural gas (3,100 MWe) and wind capacity (3,838 MWe) but no new coal capacity. The analysis upon which this forecast is based includes off-grid infrastructure projects which rely heavily on natural gas for power generation and process heat. The repeal of the carbon pricing mechanism would slow large-scale wind generation growth from an expected 12% annual electricity contribution to 7.5% in 2020 with the generation difference provided by coal (RepuTex 2013). Other forms of renewable energy deployment are expected to grow by 308 MWe from 2014 to 2018² (Macromonitor 2013). This growth is significantly slower than the growth in wind generation. Figure 7 shows a high annual investment of around AU\$ 2.3b in renewable energy capacity from 2013 to 2015 but continuously falling investment for the period through to 2018. With the committed 2013–15 investment new renewable capacity is expected to peak at 1,550 MW in 2016 and fall sharply to below 600 MWe annually in 2017 and 2018.

² The Macromonitor report only investigates utility scale plants and does not include residential and commercial PV installations.



Figure 7: Forecast renewable energy investment – value of construction and capacity added (Macromonitor 2013)

Renewable energy deployment until 2020 will depend to a large extent on policy changes in 2014 and 2015. It is possible that there will be significant growth due to the carbon pricing mechanism in combination with the RET and rising natural gas prices making renewables a more attractive option. Or there may be marginal growth due to the repeal of the carbon pricing mechanism and the weakening of the RET. Hence investment uncertainties are high at this point in time.

The strategic transition to a low carbon future is supported by the previous Australian Government through their 80% carbon emission reduction target by 2050 compared to 2000 emission levels (Department of Climate Change and Energy Efficiency 2012). The position of the recently elected Liberal government is not yet known. However, apart from an analysis of the possible Australian energy mix by 2049–50 (Syed 2012) a detailed policy proposal for the transition process beyond 2020 is missing. Several institutions have proposed significantly different pathways to reduce greenhouse gas emission by 2050. These pathways include: the use of 100% renewable electricity (Elliston, MacGill & Diesendorf 2013; Wright et al. 2011); the rapid expansion of natural gas (Australian Petroleum Production and Exploration Association 2010); and the implementation of low emissions coal technologies (Australian Coal Assocation 2014).

According to Syed (2012) a mix of all of the above will occur by 2050 with natural gas and coal providing 49% of annual generation and renewable energy sources providing 51% (see Figure 8). The use of brown coal will decrease significantly to 5 TWh in 2034–35. Brown coal is not part of the predicted 2049–50 electricity mix, and neither is oil. The share of natural

gas is expected to increase over the same period with the strongest growth occurring from 2034–35 to 2049–50. Wind and solar are each expected to surpass hydro as sources of renewable electricity by 2034–35 but their growth patterns are different. While wind is expected to grow strongly through to 2034–35 and only marginally from there to 2049–50 solar is expected to grow more slowly until 2034–35 and increase its market share more strongly from there to 2049–50. It is also expected that geothermal will grow continuously to 2049–50 but will remain below 10% of market share while bioenergy will provide only a small amount of renewable electricity.

While the Syed (2012) report has a robust scientific method, some of the assumptions, such as an increase in electricity demand from 254.7 TWh in 2011–12 to 377 TWh in 2049-50, make its results prone to error. The basis for the expected electricity increase includes continuous population and economic growth but even though both have occurred from 2010 to 2013, electricity demand decreased as shown in Figure 2 on page 6. Also the actual cost of fossil fuels, renewable energy generation and energy efficiency technologies might differ to the assumptions, and this would also affect the overall result. Predicting technology costs, market changes and technology implementation three decades ahead is a difficult task with various unavoidable uncertainties and while the details of the energy transition to 2050 are uncertain it is clear that a transition is necessary and this will require substantial changes to the current electricity market. Financial incentives will be needed in the near future to implement renewable and low carbon intensity energy technologies, and further research will be needed to develop the power technologies for the future.



Figure 8: Share of electricity generation by energy type, prepared with data from Syed (2012)

2.2 Concentrating solar power plants

Over the last century CSP has developed from theoretical concepts and first test units to a utility scale power generation option. This chapter outlines this development to date and future growth trajectories worldwide. It also provides as an update on CSP in Australia.

2.2.1 History and technology development

The development of line and point focusing systems started in the 19th century with the work of pioneers including Augustin Mouchot's research on dishes, John Ericsson's research on parabolic troughs and William Adams's research on solar towers (Ragheb 2011). However, the first commercial CSP plant, Solar Engine One shown in Figure 9, was not installed until 1912. It was developed in Egypt to pump irrigation water from the river Nile to a nearby farming community (New York Times 1916; Stinnesbeck 1914). The plant used five parabolic troughs with direct steam generation to operate a 100 horsepower pump and it used pressurised hot water storage to continue operation into the night (New York Times 1916; Stinnesbeck 1914). This CSP plant could compete with coal-fired generation in regions where coal prices exceeded 10 German Marks per tonne and it was easier to operate than a coal boiler (Stinnesbeck 1914). The British and German governments planned further units for their colonies (New York Times 1916) but World War I led to the dismantling of the plant and the abandonment of plans for further installations. Low prices and good availability of fossil fuel after World War I stopped CSP developments for the next several decades.



Figure 9: Al Meadi pumping station using the five parabolic troughs with direct steam generation (Stinnesbeck 1914)

In the 1970s the first commercial CSP plants were designed for electricity generation as a reaction to the oil crises and the subsequent search for energy alternatives. The first of nine Solar Energy Generation Systems (SEGS), using the parabolic trough technology, was a

14 MWe plant commissioned in 1984. Plant sizes rose incrementally to 80 MWe with all nine units having a combined capacity of 354 MWe in 1990 (Skumanich 2011). However, history repeated itself and decreasing oil prices in the 1990s stopped the construction of further commercial plants until 2006 when Spain introduced Royal Decree 436/2004 (Ministerio de Economia 2004) which offered plant operators a fixed feed-in tariff price or a feed-in premium on top of the wholesale electricity price. In addition to the established parabolic trough technology with thermal oil a variety of new technologies emerged on the commercial CSP market. These included solar tower and Fresnel systems, working fluids such as molten salts, and direct steam generation.

Due to the long operational experience with SEGS, in 2012 parabolic trough systems using thermal oil as the primary working fluid had a global market share of 94% of deployed capacity (IRENA 2012) and they remain the dominant CSP technology. Many commercial reference plants exist with and without thermal energy storage (TES) (IRENA 2012) and recently plants have reached capacities of 280 MWe (Abengoa Solar 2013b). To overcome the efficiency limits dictated by accelerated thermal oil degradation at temperatures above 400 °C, some suppliers offer parabolic trough systems with molten salt and direct steam generation. The first small commercial reference plants using this technology are already in operation. Examples include the 5 MWe Priolo Gargallo plant with molten salt in Italy (Falchetta et al. 2009) and the 5 MWe Kanchanaburi plant with direct steam generation in Thailand (Krüger et al. 2012).

Currently, solar tower systems are not considered to be as mature as conventional parabolic trough plants but since the early 1980s several experimental systems have been field tested in Russia, Italy, Spain, Japan, France, and the United States (Meinecke, Bohn & Becker 1995). These include the 10 MWe Solar One and Two in the USA and the 5 MWe SPP-5 in Russia. The PS10 solar tower in Spain began operating in 2006. It is able to generate 10 MWe (net), and it was the first commercial tower plant to commence operation (Solúcar 2006). Further commercial projects with steam and molten salt as the primary working fluid commenced operation or are under construction in Spain, the USA and South Africa. They include the 5 MWe Sun Sierra plant (eSolar Inc. 2013), the 20 MWe PS20 plant (Abengoa Solar 2014b), the 20 MWe Gemasolar plant (García & Roberto Calvo 2012), 50 MWe Khi Solar One (Abengoa Solar 2013a), the 110 MWe Crescent Dunes plant (SolarReserve LLC 2012), and the 392 MWe Ivanpah plant (BrightSource Energy Inc. 2013) shown in Figure 10. Not all plants have TES but many do, with full-load capacities ranging from 2 to 15 h. A new 110 MWe plant with 17.5 h TES is scheduled to commence construction later this year in Chile (Abengoa Solar 2014a).



Figure 10: One of three solar tower systems of the 392 MWe Ivanpah power station, USA³

In addition to Rankine cycles, solar towers can be combined with Brayton cycles to raise cycle efficiency and reduce cost. Experimental systems exist for high-temperature air applications, such as the 230 kilowatt (kWe) SOLGATE project in Spain (ORMAT et al. 2005) and the recently commissioned 4.6 MWe Solugas project in Spain (Quero et al. 2013). However, high-temperature air is also used to operate conventional Rankine cycle systems (e.g. 1.5 MWe Solair/HiTRec project in Germany) (Koll et al. 2009). Also under development are experimental solar towers for hydrogen production through steam reforming of natural gas with the largest prototype being a 300 kWe installation in Australia (Stein et al. 2009). Similarly, a 1 MWth solar tower prototype to convert solid biomass into synthesis gas was built in 2010 and is currently being tested in the USA (Service 2009). As seen from the aforementioned examples solar tower technology is versatile, and with the recent commissioning of several utility scale plants, particularly Ivanpah and Crescent Dunes, they can demonstrate their technical and commercial expectations. If these plants are successful the technology is very likely to gain a significant CSP market share.

Like solar tower plants, Fresnel is a CSP technology that is currently proving its technical benefits and commercial viability. Some commercial plants are already in operation, such as the 5 MWe Kimberlina in the USA (NREL 2014b) and 30 MWe Puerto Errado in Spain (Novatec Solar GmbH 2012c). In addition, larger units are under construction. These include the 44 MWe Kogan Creek Solar Boost hybrid in Australia (CS Energy 2011) and the 2x 125 MWe Reliance project in India (AREVA Solar 2012a). The market share of Fresnel plants is

³ www.brightsourceenergy.com/image-downloads, accessed 04 April 2014

even smaller than that of solar towers (IRENA 2012) and the plants currently in operation or under construction are crucial for establishing the technology as a credible CSP alternative. Currently, Fresnel plants are only offered commercially with direct steam generation but recently research has started into the use of molten salts to simplify the integration of TES (AREVA Solar 2014; Novatec Solar GmbH 2013).

The simplicity of the Fresnel collectors raises expectations for significant cost reductions but the technology has the disadvantage that TES is not yet commercially available. However, this might change in the near future with the successful testing of molten salts and ongoing TES developments involving phase change materials for direct steam systems (Bahl et al. 2011; Kuravi et al. 2013; Laing et al. 2012; Nithyanandam, Pitchumani & Mathur 2014; Yin, Ding & Yang 2014).

Predominantly, parabolic dishes are used in conjunction with Stirling engines, for example at the 50 kWe plant in Saudi Arabia (Noyes 1990), and concentrating PV systems, for example at a 1.5 MWe plant in Australia (Silex Systems Ltd 2013). However, some dish systems are designed for Rankine cycle operation and the 4.9 MWe Solar Plant 1, commissioned in 1984 in the US, was the largest plant in the world using a primary molten salt and secondary superheated steam cycle (Stolpe 1986). Technical problems saw the unit converted to a solar-diesel hybrid plant and the construction of further units was halted (Larson & West 1996). A smaller 25 kWe plant commenced operation in Australia in 1983 (Kaneff 1991) but was dismantled in 1994 with the reflector surface further used for concentrating PV testing (Gordon 2001). Despite having the highest concentration ratio and the potential for excellent conversion efficiencies (Lovegrove & Luzzi 2002) dish systems have not played a relevant role in recent commercial CSP projects. The reasons for their very limited uptake are likely to be the small number of reference plants worldwide, concerns about plant complexity through extensive hot and cold working fluid piping, the lack of established construction companies offering such plants with all standard guarantees, and high finance costs associated with technologies which have not been widely tested. At the Australian National University a 500 m² big dish designed to operate a Rankine cycle is being tested and commercialised (Lovegrove, Burgess & Pye 2011) but so far no complete demonstration plant has been built. A state grant for a commercial project was awarded but later withdrawn (Edwards 2013; Regional Development Australia 2011).

The new CSP technologies emerging in commercial projects combined with innovative TES systems have the potential to lower LCOE significantly but require ongoing deployment to convince financiers to invest in them at levels comparable to current levels of investment in conventional parabolic trough plants. The recent revival and technological diversification of

commercial CSP plants has enabled the CSP industry to increase its cumulative capacity from 354 MWe in 1990 to 4 GWe in 2013 (International Energy Agency 2013). The capacity of individual plants has also risen, with the largest plant to date being the 280 MWe Solana (Abengoa Solar 2013b) and the 390 MWe Ivanpah plants (BrightSource Energy Inc. 2013) in the US. Both of these plants were commissioned recently.

2.2.2 Outlook

In its recent renewable energy projections the International Energy Agency (2013) estimates an increase in cumulative CSP deployment from 4 GWe in 2013 to 12 GWe in 2018. While tripling the installed capacity is a good outlook for the industry, it is worth putting this in perspective with PV installations, which are expected to grow by a slightly smaller factor of 2.4 but from a 2013 basis of 128 GWe (see Figure 11). Wind generation, with a cumulative deployment of 321 GWe in 2013, is even stronger. It is apparent that in the solar market PV has a much stronger share than CSP but it is also apparent that the expected 2018 capacity factor for CSP of 32% (34 TWh generation with 12 GW deployed capacity) is much higher than for PV of 14% (368 TWh with 308 GWe). This is an expected increase from an average CSP capacity factor of 26% in 2013 (9 TWh with 4 GW), which indicates that the availability of mature TES is an important distinction to PV and it will play an increasingly important role.



Figure 11: Annual electricity capacities and generation for CSP and PV from 2011-18 (International Energy Agency 2013)

In recent times the significant cost reductions of PV systems have put pressure on CSP and the decision to switch the first 500 MWe phase Blythe parabolic trough project in the USA

to PV is the most prominent example of the competition with PV so far (Beetz 2012). Even larger PV projects with capacities ranging from 700 to 4,000 MWe are being developed (Sempra U.S. Gas & Power 2012; Willis 2014), and in 2013 83% of new solar projects in the USA were PV (GTM Research & Solar Energy Industries Association 2013). This shows that TES is not yet valued sufficiently because its cost are still high and because fossil powered assets, particularly natural gas, can quickly offset PV shortfalls. However, when considering higher penetration renewable energy scenarios CSP is complementary to PV as intermittent renewable sources require dispatchable renewable sources to ensure stable grid operation (Denholm & Mehos 2011; Elliston, Diesendorf & MacGill 2012).

In order to remain competitive, the CSP industry must not only reduce the solar field cost but also the TES cost. A variety of research and development (R&D) activities aim to improve TES using sensible heat, latent heat and thermochemical concepts (Kuravi et al. 2013). Sensible heat storage systems raise the temperature of a medium, such as molten salt, and the amount of energy stored depends strongly on the medium's heat capacity. The storage medium does not change its phase and it is currently the dominant TES technology. Latent heat storage is a nearly isothermal process. It has the potential to store significantly more energy than sensible storage systems of the same temperature range because the storage capacity of latent heat systems depends not only on the storage medium's specific heat but also on the enthalpy of phase change. Thermochemical storage systems use the heat from the solar field to drive reversible chemical reactions. Currently they are the least mature storage technology but they have the potetnial to store more energy than the two aforementioned technologies. It is expected that through R&D and technology deployment TES costs will decrease from AU\$ 90/MWth in 2011 to AU\$ 22/MWth in 2020 (Hinkley et al. 2011). Continuous CSP deployment could lower the 2011 LCOE range of US\$ 160-270/MWh to US\$ 100-140/MWh by 2020 for parabolic trough systems and to US\$ 80-160/MWh for solar tower systems (IRENA 2012). Obviously, these are high-level predictions, which depend on the actual CSP deployment and the technological progress made by 2020, and more specifically on the solar resource (see direct normal irradiance (DNI) impact in Figure 12), the plant location, local labour costs, CSP technology, finance conditions, and plant and TES capacity.

In recent years CSP deployment has been driven by government incentives, such as feed-in tariffs in Spain (Ministerio de Economia 2004) and loan guarantees in the USA (U.S. Department of Energy 2013a). New technologies require support to enter a market but this comes with the risk that costs will not decrease as quickly as possible anticipated or that governments will shift focus and support mechanisms will disappear. The latter happened
in 2012 in Spain when feed-in tariffs for CSP were retrospectively reduced by Royal Decree 1/2012 (Jeafatura del Estado 2012) with the result that the fastest growing CSP market worldwide disappeared almost overnight. Similarly, in the USA loan guarantees enabled the construction of the largest CSP plants worldwide but with these plants being completed or moving closer to completion no new plants are under construction. Other markets exist to compensate for these shortfalls and ensure continuous CSP growth. Examples include the Middle East and North Africa region target of 37.5 GWe by 2032 (Bryden, Riahi & Zissler 2013), India's 20 GWe target by 2022 (Ministry of New and Renewable Energy 2011), and China's 3 GWe target by 2020 (Lluna 2011). However these targets also rely on government support mechanisms which can disappear quickly. In summary future CSP deployment is not easy to predict as unforeseen events can occur and unfortunately the ambitious 2020 target of 30 GWe predicted by AT Kearney Consulting in 2010 is unlikely to be realised.



Figure 12: Tariff and levelised cost of energy development above DNI level; Percentage compared to reference plant in Spain with a DNI of 2,084 kWh/m²/a at 100 per cent (AT Kearney & ESTELA 2010)

In addition to learning from deployment, R&D programs can reduce LCOE and currently various projects worldwide aim to do this. Two significant projects are the Sunshot Initiative from the USA with an LCOE target of US\$ 60/MWh (SunShot Initiative 2012) and the Australian Solar Thermal Research Initiative (ASTRI) which aims to reduce LCOE to AU\$ 120/MWh (CSIRO 2013) by 2020. Both investigate novel concepts, including supercritical CO₂ cycles, to significantly increase cycle efficiency. While the ASTRI goal is ambitious it is within the range of other studies (IRENA 2012). However, the Sunshot goal seems very challenging and a recent assessment concluded that the probability of achieving it is very low. However, a 50% chance exists to reach US\$ 120/MWh (Ho, Mehos & Wagner 2014).

Finance is particularly relevant for CSP projects and developers prefer large projects to gain economy-of-scale benefits. However, this has the consequence that project budgets can easily reach hundreds of millions or even a billion dollars. For example the 50 MWe Andasol plant in Spain had a budget of \notin 300m (Solar Millennium AG 2006), 280 MWe Solana plant has a budget of US\$ 2b (NREL 2014d), and the 392 MWe Ivanpah plant had a budget of US\$ 2.2b (NREL 2014a). The consequences of combining large project budgets with less mature technologies are high finance risk premiums, and this is one reason for the dominant market share of parabolic trough plants despite their lower efficiency compared to solar tower and Fresnel systems (see Figure 13). The recent commissioning of new solar tower plants and the growing confidence in the technology has meant that in 2013 there were more plans for tower projects than for parabolic trough projects. However, planned parabolic trough plants are still very significant and the technology is likely to retain a dominant role in regards to deployed capacity for years to come (see Figure 13). Fresnel, and particularly dish systems, will have to mature significantly to reach a higher market penetration in the future.





2.2.3 Australian market

Australia has one of the best solar resources in the world (see Figure 14), and according to a recent CSP study for Australia (Lovegrove et al. 2012) the potential for off-grid CSP (plants <10 MWe) is 100 MWe, for medium-scale CSP (10-20 MWe plants) it is 720 MWe, and for

utility-scale CSP connected to the current grid (>20 MWe plants) it is for 3-4 GWe. However, the current situation is very different as Australia has been a very slow adopter of CSP with a variety of reports investigating ideal sites and costs (Cameron & Crompton 2008; Clifton & Boruff 2010; Dawson & Schlyter 2012; Feeney et al. 2010; Lovegrove et al. 2012; Parsons Brinckerhoff Australia Pty Ltd 2010; Rutovitz et al. 2013; Wright et al. 2011) but no commercial stand-alone plants under construction or in operation. The only two commercial developments are two Fresnel systems attached to coal fired generators. These are the operational 18.3 MWth system at the 2 GWe Liddell power station, comprising a 9 MWth field build in 2008 and a 9.3 WMth field built in 2012 (AREVA Solar 2012b; CSP World 2014; Novatec Solar GmbH 2012a), and the 44 MWe equivalent system under construction at the 750 MWe Kogan Creek power station (CS Energy 2011).



Figure 14: Direct normal irradiation for potential global CSP sites (Trieb et al. 2009)

Attempts were made by former state and federal governments to build commercial CSP plants through the allocation of AU\$ 464m for a 250 MWe Fresnel plant in Queensland (AREVA Solar 2011) and AU\$ 60m for a 40 MWe SolarOasis big dish plant in South Australia (Regional Development Australia 2011). However, neither could not secure the remaining finance and were subsequently withdrawn (Edwards 2013; Kelly 2012). The only standalone CSP plant to operate in Australia for an extended period (between 1983 and 1994) was the 25 kWe White Cliffs project which used dishes to operate a steam engine (Gordon 2001). A larger 100 kWe system at Meekathara in Western Australia commenced operation in 1982, using small troughs to operate an organic Rankine cycle, but was only operational for a short period (Hellweg 1983). In 2011 a 3 MWe multi-tower test plant was set up (Lloyd Energy Systems 2011) and another 1.2 MWe multi-tower plant is currently under construction (ABC 2013; CSP World 2013).

Options to increase the financial viability of CSP in the current electricity market exist and include placing plants in grid-constrained locations, thereby deferring or even offsetting investment in transmission infrastructure (Rutovitz et al. 2013) and off-grid applications where they compete with costly diesel or natural gas fuelled plants.

Low wholesale electricity prices, fossil fuel subsidies, the lack of a stable support mechanisms, and the absence of full externality costing for fossil generators mentioned in Section 2.1.1 are key reasons for the absence of commercial CSP plants in Australia. Currently, only hybrid plants seem viable due to lower costs, relative to CSP-only plants, achieved through the joint use of plant equipment, such as steam turbines and condensers. The AU\$ 104.7m for the 44 MWe equivalent CSP retrofit to the Kogan Creek power station (CS Energy 2011) represents a significantly lower specific investment (AU\$ 2.4m/MWe) than the AU\$ 200m for the 53 MWe PV plant (AU\$ 3.8m/MWe) about to be constructed in Broken Hill (Boisvert 2013; Hobson & Winsbury 2013). Therefore CSP hybrids seem to offer a promising transition technology to allow CSP to enter the Australian market, build local learning experience and increase investment confidence. Further to the two current CSP retrofits plants, new hybrid projects are currently being investigated with natural gas, such as 30 MWe Collinsville proposal (Australian Renewable Energy Agency 2013a), and with biomass/waste feedstocks such as the 35.5 MWe Swanbank proposal (Ipswich City Council 2012; Peterseim et al. 2012b). Also a recent study on the retrofit of CSP to existing fossil fuel plants identified nine opportunities with natural gas and seven with coal (Meehan 2013).

To lower the future cost of CSP the Australian government is investing in R&D programs to progress cutting edge technologies. Projects include the recent AU\$ 87.3m ASTRI development (CSIRO 2013), test facilities for the big dish in Canberra (Lovegrove, Burgess & Pye 2011), and solar tower test facilities in Newcastle (CSIRO 2012).

2.3 Non-conventional fuels for power generation

In this research the term 'non-conventional fuels' refers to waste feedstocks, such as wood waste and refuse-derived fuels, and solid biomass feedstocks, such as wood, straw and bagasse. These are used widely worldwide but despite their benefits, such as renewable energy generation and landfill diversion, they are also controversial in regards to debates about food versus fuel and material versus energy recovery.

A variety of thermochemical, physiochemical and biochemical technologies are available to convert biomass and waste feedstocks to thermal energy (see Figure 15). However, the focus of this research is the hybridisation of solid materials with CSP in Rankine cycle systems and the literature review is therefore limited to direct combustion, gasification, plasma, and pyrolysis treatment options.



Figure 15: Energy from waste conversion technologies (Kaltschmitt 1998)

2.3.1 Energy from waste

Permanent open fires burning waste were reported in Jerusalem as early as 1,000 B.C. but the first industrial Energy from Waste (EfW) facility was built in London in 1870 to guarantee the destruction of germs in municipal waste streams (Vehlow 2004). Other cities in Europe followed by building plants able to provide electricity and/or heat (e.g. EfW plants in Hamburg in 1896 – shown in Figure 16a – and Copenhagen in 1903). It is worth mentioning that these facilities were accompanied by the construction of recycling plants for paper, textiles, leather glass and metal (Vehlow 2004).

Since the commissioning of these first plants EfW technologies have improved significantly in regards to cycle efficiency, from 15–21% in the 1980s to over 30% in recent units (Gohlke 2008), and emissions (e.g. dioxin and furan emissions fell by 99% between 1990 and 2005 (Stevenson 2007). These improvements were possible through a variety of technologies including higher steam parameters, steam reheating, optimised combustion, minimised parasitic losses, and high efficiency baghouse filters and flue gas scrubbing.

Today almost 2,200 commercial EfW plants operate worldwide (ecoprog 2013) many of them in the middle of densely populated cities, such as Paris, London, Berlin, and Tokyo (as shown in Figure 16b), where waste materials are created and electricity and process heat is needed. In 2006 EfW plants produced 46 TWh of electricity and an equal amount of process heat worldwide (Themelis 2006) and by 2017 another 180 plants are expected to be operational (ecoprog 2013).



Figure 16: a: Waste incineration plant Bullerdeich in Hamburg in 1896 (Vehlow 2004) and b: modern Energy from Waste plant in Tokyo, Japan $(right)^4$

Prior to using waste materials for energy recovery, higher priority has to be given to waste prevention, reuse and recycling. Despite debates about the competition of waste recycling with EfW, Figure 17 shows that typically countries with EfW facilities have higher recycling rates than countries without (European Environment Agency 2007). This supports the finding made by German Federal Environment Environment Agency that:

waste incineration does not have a negative impact on waste prevention. Its primary task is the safe and proper disposal of wastes not avoided and not recycled. Waste incineration thus delivers the disposal security which remains necessary in a

⁴ Image courtesy Logan Mirto

recycling economy based on material and resource efficiency (German Federal Environment Agency 2008, p. 5).

Countries with recycling facilites have the ability to recover recyclable materials and use the remaining material streams for energy recovery rather than disposal in landfills. For example in 2005 Germany introduced a landfill ban on waste materials with an organic content of more than 3% and since then the majority of municipal solid waste has been recycled, non-recyclable materials have been used in EfW plants providing electricity and heat, and less than 1% of municipal solid waste has gone to landfill (European Environmental Agency 2009). In the future new recycling technologies are likely to enable higher recycling rates and reduce the amount of material available for EfW plants, but for the foreseeable future different EfW technologies, including landfill gas and aerobic and anaerobic digestion systems, will continue to play an important role in global waste management strategies.





Feedstocks used predominantly in current EfW facilities include municipal solid waste (MSW), refuse derived fuel (RDF), sewage sludge, tyres and wood waste. Especially equipped plants are able to accept hazardous waste streams, such as chemical or medical waste, without exceeding the stringent emission limits set by the European Union (European Parliament and Council 2010) and other authorities. While MSW streams can have a biogenic content of up to 50% (Gohlke & Spliethoff 2007), the use of wood waste and RDF is of particular interest as wood waste is a fully renewable fuel and RDF does maximise recycling efforts, has a significant renewable component and is more consistent than MSW. This has positive flow-on effects in the power plant including better combustion performance, higher efficiency and simplified flue gas cleaning (Chang, Chen & Chang

1998). The recovery of wood waste and RDF from municipal, commercial and industrial waste streams is well established with many units in commercial operation. RDF is not a precisely defined fuel and therefore the term 'solid recovered fuels' was introduced. Solid recovered fuels have defined quality specifications in regards to material origin and properties (Rotter et al. 2011). However, the term RDF is still widely used and many literature sources do not yet use solid recovered fuels.

Currently, a variety of thermal conversion technologies in different stages of technological maturity are available. They include combustion, gasification, plasma and pyrolysis, as well as mixtures of these (Lamers et al. 2013). Currently, grate combustion systems dominate the EfW market with around 2,000 references worldwide, with plant capacities from 0.3-150 MWth, various fuel types, significant operational experience, and a variety of technology suppliers (Lamers et al. 2013; Spliethoff 2010). To raise feedstock conversion efficiency some systems have been hybridised with natural gas through external EfW steam superheating in dedicated superheaters, such as 28 MWe Holstebro plant in Denmark (Andersen, Oksen & Olesen 1992; Babcock & Wilcox Vølund A/S 2008), or in adjacent combined cycle plants. These include the 88 MWth Mainz plant in Germany (Entsorgungsgesellschaft Mainz mbH 2007) and a 93.2 MWe Bilbao plant Spain (Gohlke & Spliethoff 2007). Some fluidised bed systems exist for MSW, such as an 800 t/d plant in Cixi China (Huang, Chi & Themelis 2013) and RDF, such as the 30 MW cogeneration plant in Eisenhüttenstadt Germany (EnBW Energie Baden-Württemberg AG 2009). However, despite the size of these reference plants and their technical benefits, such as improved efficiency and reduced excess air requirements, the smaller number of reference plants leads to a lower maturity than grate systems.

Gasification systems offer technological and environmental benefits over combustion systems as some technologies are able to produce a synthesis gas that is compliant with gas engine and turbine specifications, a vitrified non-leaching slag without additional treatment, improved waste volume reduction, and lower air requirements for the conversion process (Arena 2012; Lamers et al. 2013). EfW gasification technologies have to be separated into gasification/combustion and gasification technologies (Lamers et al. 2013). Gasification/combustion, also known as staged combustion, uses gasifiers to produce a raw gas that is burnt in a boiler connected to a conventional Rankine cycle power plant. Gasification technologies, also referred to as true gasification, clean the produced raw gas to a synthesised gas suitable for high efficiency conversion in gas engines and turbines. A variety of commercial gasification/combustion plants operate on RDF worldwide, including a 50 MWe Lahti plant in Finland (Lamers et al. 2013), a 33 MWe Albano plant in Italy (JFE Engineering 2013), a 20 MWe Fukuyama plant in Japan (JFE Engineering 2011), and a 6.7 MWe Greve plant in Italy (Granatstein 2003). There are fewer gasification plants with gas engines or turbines. Most of them are in Japan, and many of them use the Thermoselect technology. These include the 1.5 MWe Chiba plant and the almost twice as large Mizushima plant which provides synthesis gas to an adjacent processing plant (Sumio, Masuto & Fumihiro 2004). The Thermoselect technology is a combination of pyrolysis followed by high temperature gasification at over 1,200°C, which decomposes the MSW, RDF and other waste feedstocks into a synthesis gas (consisting mainly of hydrogen and carbon monoxide) as well as vitreous slag and metals which can be recycled or used for construction purposes (Sumio, Masuto & Fumihiro 2004). A 6.5 MWe plant using a different pyrolysis/gasification technology was commissioned in March 2013 in the U.K. (Reece 2013) but long-term operational data is not yet available. In total less than 100 waste gasification units have been built worldwide and not all of them have been commercial successes (Lamers et al. 2013), which is a key reason why gasification systems do not currently have a larger market share.

Plasma gasification further raises the process temperature to over 2,000°C by using an electric arc and thereby converting the feedstock predominantly to hydrogen and some carbon monoxide suitable for gas engine and turbine operation. This process also produces vitreous slag. A reference plant exists for asbestos treatment in France (Europlasma Group 2014) and the largest plant yet, producing 12 MWe, commenced operation in 2012 also in France (Waste Management World 2012). Globally around 15 units, including pilot plants, exist with the majority of them being in Japan (Lamers et al. 2013). Plasma systems are beneficial in regards to high synthesis gas quality and waste volume reduction but the high capital (CapEx) and operational expenditures (OpEx) as well as plant complexity have limited technology uptake to date.

In pyrolysis plants the feedstock, typically RDF, is heated to above 300°C in the absence of oxygen and converts to synthesis gas, oil, char, and recyclable materials. The oil and synthesis gas is suitable for high efficiency power generation in gas engine and turbine plants and the metals are recyclable. The first large test unit, in Fürth Germany, was unable to achieve a continuous operation while another unit in the USA never met the air emission standards and was decommissioned in 2009 (Lamers et al. 2013). However, due to higher efficiency expectations around 25 plants were built worldwide but the failures in the first units damaged the technology's reputation significantly and little information is available about the other reference plants (Lamers et al. 2013).

Technology distribution in the EfW market is similar to CSP where the less efficient but mature parabolic trough technology using thermal oil as the primary working fluid dominates the market, just as grate combustion systems do in the EfW market. Efforts are being made to increase the efficiency of combustion systems through the use of fluidised beds combustion systems, not dissimilar to the development of direct steam or molten salt parabolic trough systems. The more efficient gasification technologies are able to raise cycle efficiency, further reduce emissions, and produce valuable secondary products, such as fuels. This is comparable to the way in which solar tower and Fresnel systems are currently proving their techno-economic viability and are increasing their market share. Currently pyrolysis systems have, despite technical benefits such as char production, the least market penetration and are comparable to CSP dish systems, which also offer high technical potential but have had little commercial success to date. The future development of EfW technologies will strongly depend on the technical and commercial success of recent reference plants as well as government priorities in regards to electricity, heat, material recovery, and fuel production from waste materials, as well as emissions.

2.3.2 Energy from biomass

Since the industrial revolution biomass has been used for power generation purposes and in 2013 biomass was, after hydro and wind, the third-most important source of renewable electricity worldwide with an output of 396 TWh (International Energy Agency 2013). Despite currently having a smaller installed capacity than PV, it is projected that by 2018, biomass plants will generate 560 TWh – significantly more electricity than PV which is expected to provide 368 TWh (International Energy Agency 2013). The reason for this is the high capacity factor of biomass plants reaching 8,000 operational hours per year or 91.3%.

Feedstocks for energy from biomass (EfB) are versatile. They include wood, bark, bagasse, agricultural crops (e.g. straw and rice husk), and energy crops (e.g. miscanthus). With some of these fuels potentially competing with food production (Baffes 2013; O'Connell et al. 2009) the use of algae is being investigated more closely for electricity and fuel production (Jonker & Faaij 2013; Rashid et al. 2013; Resurreccion et al. 2012) and while several test facilities exist no commercial units are operational yet.

Depending on fuel, the capacity and conversion technology cycle efficiencies for current EfB plants vary from 15–35% (Stucley et al. 2012). Biomass integrated combined cycle plants using gasification or pyrolysis have the potential to raise cycle efficiency to over 40% through the use of a Brayton cycle and a bottoming Rankine cycle but no reference plants exist yet.

The capacities of EfB plants range from small industrial heating systems to the world's largest plant, the Polaniec power plant in Poland, which has an output of 205 MWe (GDF SUEZ 2013). However, the comparatively low calorific value of biomass compared to black coal or natural gas leads to high transport costs, which usually limits plant size to 30 MWe. Larger plants are only feasible when co-located with saw, pulp and paper, and sugar processing facilities, or when using wood pellets and briquettes to increase the feedstock's calorific value. Of increasing interest worldwide is biomass co-firing in utility scale power plants as it is a low-cost option for the integration of renewable energy (Lempp 2013). Case studies have already been investigated in Australia (Meehan 2013). Worldwide, several units blend solid biomass and coal prior to combustion such as in the 590 MWe Avedore power station in Denmark (Stucley et al. 2012), or they gasify biomass and burn the raw gas in parallel to the main fuel, such as in the 560 MWe Vaskiluodon Voima Oy power station in Finland (Metso Corporation 2011). The choice of the co-firing system depends on the feedstock quantity and quality as well as the design of the coal plant (Lempp 2013; Tadros et al. 2009).

The conversion technologies for EfB are essentially the same as for EfW plants except that: a) plasma technologies are not used, as EfB plants do not have to cope with significant plastic compounds; and b) that gasification and pyrolysis is more advanced for biomass feedstocks because it is less heterogeneous than waste feedstocks. Grate combustion systems are the ones most established in EfB plants (Spliethoff 2010) but a variety of fluidised bed systems are in operation with benefits in regards to cycle efficiency and emissions (e.g. 20 MWe Königs Wusterhausen plant in Germany) (Lahoda, Arndt & Hanstein 2006). Also a variety of commercial gasification/combustion references exists worldwide with capacities reaching 140 MWth (Metso Corporation 2011) and gasification engine systems with capacities of 6 MWe (Spectrum Magazine 2009). Unlike EfW pyrolysis the production of biochar is an important development as it can be used for carbon sequestration, for soil enhancement, and as a renewable feedstock for metallurgical processes (Garcia-Perez, Lewis & Kruger 2010; Sohi et al. 2009). A variety of demonstration plants exist, such as the Somersby facility in Australia which is capable of processing 300 kg of feedstock per hour to produce biochar and 200 kW electricity (Pacific Pyrolysis 2010). However no commercial plants are operational yet. The variety of possible products is promising for markets in the future but current demonstration facilities have to further prove the technical viability of this approach before the first commercial units can test the technical and economic expectations.

The future for EfB strongly depends on the sustainable production of biomass feedstocks without impacting food production. Options exist to avoid this conflict, including the colocation of food and energy production or the use of waste materials from agriculture for energy generation. However, the right policy settings are required to ensure the existence of these conditions.

2.3.3 Australian market

Municipal solid, commercial, industrial, and construction and demolition waste streams going to landfill in 2006–07 amounted to 21 million tonnes (t), which equals 48% of total waste generation (Environment Protection and Heritage Council 2010). In Australia the recycling rate for MSW was 40% in 2006–07 (Environment Protection and Heritage Council 2010) which is similar to Sweden's in 2005 (see Figure 17 on page 25), with the noticeable difference that Sweden landfilled less than 5% of its MSW and used the majority for energy recovery. In Australia 60% of MSW was landfilled without any energy recovery. Considering that in Australia up to 72% of this waste stream consists of organic materials (Environment Protection and Heritage Council 2010), the potential for biochemical treatment of putrescible waste (Stucley et al. 2012), and for thermal treatment plants using wood waste and RDF is significant. Some plants for biochemical treatment already exist in Australia (Stucley et al. 2012).

In 1972 the Waterloo waste incinerator commenced operation in Sydney but it was decommissioned in 1993 due to public protests which were due in part to the plant's emissions being 12 to 38 times larger than permitted emission limits on various occasions (Sharp et al. 2004). The plant sparked protests from the beginning and opposition increased due to a plant failure in 1975 which caused it to shut down for several months (Sharp et al. 2004). Due to this experience the reputation of EfW in Australia was harmed and currently only pulp and paper mills operate EfW cogeneration facilities using rejects from their processes as well as wood waste (e.g. Visy Paper) (Visy Industries Australia 2012). One designated RDF production facility exists in South Australia (Watson & Peterseim 2010) which supplies feedstock for co-firing in a cement kiln (Adelaide Brighton 2007). Recently, a proposal for a 15 MWe (gross) RDF gasification/combustion plant in Port Hedland, Western Australia, received environmental approval (Environmental Protection Authority 2013) and others are being developed, such as a 75 MWe proposal for Tumut in New South Wales (Kitney 2011). However, no project has yet commenced construction.

In recent years almost all states have conducted studies or prepared policies for EfW facilities, including the Australian Capital Territory (Lokuge & Loftus 2010), Western

Australia (Fanning 2013), South Australia (Warren et al. 2013), Victoria (Environmental Protection Agency Victoria 2013), and New South Wales (Environmental Protection Agency NSW 2013). These are steps towards the implementation of such facilities in the future.

Based on the experience of the Waterloo incinerator in Sydney the only option to obtain community acceptance for new EfW plants in Australia is early community engagement, compliance with stringent international emission limits (e.g. the Environmental Protection Agency recent NSW (2013) adopts the EU emission limits in its recent EfW draft policy), and the use of materials downstream of a recycling process. Currently, the wholesale electricity price in the NEM and the REC price are insufficient to make EfW plants economically viable in Australia. For these plants to be profitable they would require another revenue source which could be provided through waste disposal levies. These levies would also encourage the diversion of waste from landfill.

In Australia sufficient lignocellulosic biomass is available from current agricultural and forestry production systems to annually generate 35 TWh of electricity (Farine et al. 2011), which is equal to 14% of the 2011–12 Australian electricity generation of 254.7 TWh (Bureau of Resources and Energy Economics 2013). Considering the integration of new eucalypt plantings incorporated into agricultural landscapes this potential could increase by another 20.2 TWh/year (Farine et al. 2011). In 2012 the total installed capacity of wood waste and bagasse power plants was 573 MWe (see Table 1) of which 206 MWe has commenced operation since 2001 (Biomass Power & Thermal Magazine 2011; Stucley et al. 2012). Australia's projected bioelectricity generation for 2012–13 was 2 TWh (Bureau of Resources and Energy Economics 2013) and assuming the fuel mix in Table 1 as well as identical capacity factors, bagasse and wood waste could have generated 1.3 TWh, of which 0.9 TWh could have been produced from bagasse in Queensland alone. Comparing the current electricity generation and the potential from current agricultural and forestry production systems highlights the significant growth potential for EfB in Australia. However, the electricity production from all bioenergy sources in 2034–35 is predicted to be only 7 TWh (Bureau of Resources and Energy Economics 2013).

Due to the high carbon intensity of Australia's generation assets, particularly coal, currently available lignocellulosic biomass feedstocks are best used for electricity generation rather than for biofuels. Using biomass for electricity generation could yield 30 million tonnes CO_2 equivalent greenhouse gas mitigation, whereas if it is used to produce biofuels it would only provide 9 million tonnes CO_2 equivalent greenhouse gas mitigation (Farine et al. 2011). Assuming increasing renewable and natural gas fired generation in the future, this benefit

will decrease over time but the significant greenhouse gas mitigation difference is likely to favour biomass for electricity generation over biofuels for many years to come.

| Table 1: Biomass generation capacity in MWe per state and fuel in 2009 (Stucley et al. 2012) updated |
|--|
| with the recently commissioned 36 MWe Mackay plant using bagasse in Queensland (Biomass Power |
| & Thermal Magazine 2011) |

| State | Biogas | Bagasse | Wood waste | Other bioenergy⁵ | Total |
|------------------------------|--------|---------|------------|------------------|-------|
| New South Wales ⁶ | 73 | 81 | 42 | 3 | 199 |
| Victoria | 80 | - | - | 34 | 114 |
| Queensland | 19 | 413 | 15 | 4 | 415 |
| South Australia | 22 | - | 10 | - | 32 |
| Western Australia | 27 | 6 | 6 | 63 | 102 |
| Tasmania | 4 | - | - | - | 4 |
| Northern Territory | 1 | - | - | - | 1 |
| Total | 226 | 500 | 73 | 104 | 903 |

With the low average wholesale electricity price in Australia biomass co-firing has been investigated as a low-cost approach, LCOE below AU\$ 50/MWh (Meehan 2013), to integrating renewable energy with coal plants. Tests were carried out at several power stations in New South Wales and Queensland (Meehan 2013) and while Delta Electricity and Macquarie Generation co-fired low levels of biomass materials commercially for several years they stopped due to issues around fuel availability and economics (McEvilly, Abeysuriya & Dix 2011). Plans were made to replace up to 20% of coal consumption with biomass at Wallerawang but the necessary technical changes required have not been made yet (McEvilly, Abeysuriya & Dix 2011). Currently, only smaller units, such as the 114 MWe Worsley multi-fuel plant in Western Australia and the 55 MWth Tumut plant in New South Wales, co-fire biomass to provide electricity and process heat (Meehan 2013).

Biomass pyrolysis is being discussed in Australia as an option to generate electricity and simultaneously biochar to increase soil carbon. Agriculture and horticulture are important business sectors in Australia and improving soil carbon would be beneficial, particularly with some waste materials from these industries, such as wood waste and stubble, being suitable for biochar production (Cox et al. 2012). Since 2006 a test facility has been operational in New South Wales (Pacific Pyrolysis 2010). In 2011 the Victorian Government

⁵ Unspecified biomass and biodiesel

⁶ Includes the Australian Capital Territory

allocated AU\$ 4.5m to co-fund a commercial biochar plant in Melbourne (Manning 2011), and in 2012 the federal government allocated AU\$ 4.3m for another commercial facility in New South Wales (Saunders 2012). Unfortunately, the construction of commercial plants has not commenced yet and little information has been available.

Similar to EfW, if new EfB plants were to be constructed they would require another revenue source in addition to the income from current wholesale electricity and RECs. Alternatively, wholesale electricity and REC prices would need to be higher. Process heat is an attractive revenue stream and the existing cogeneration plants adjacent to sugar as well as pulp and paper processing plants benefit from the generation of both electricity and heat. This should be a priority for the development of future EfB plants. Continuous feedstock supply is crucial for EfB projects, and this requires a well-managed supply chain to ensure operation throughout the seasons and potentially periods of drought. However, these issues have been successfully managed in existing plants in Australia and overseas and the comparatively low cost of EfB, compared to solar, geothermal and offshore wind, as well as its high capacity factor make it a promising renewable energy source to contribute to Australia's 2020 renewable energy target.

2.4 CSP hybrid benefits & challenges

Currently, CSP is considered a higher cost renewable energy source than wind or PV (International Energy Agency 2013) but its ability to be integrated into existing natural gas and coal power plants has significant cost reduction potential and studies have confirmed that several power plants in Australia and the USA meet hybridisation criteria, such as high DNI and high available plant capacity (Meehan 2013; Turchi et al. 2011). However, significant cost reductions are possible not only in retrofit scenarios but also in new CSP hybrid plants using fossil and renewable fuels. For example CSP-natural gas hybrids with an LCOE of US\$ 168/MWh compared to US\$ 180/MWh for CSP-only plants (Turchi & Ma 2014) and CSP-biomass hybrids which can be economic at AU\$ 145/MWh compared to AU\$ 200/MWh for CSP-only plants (Peterseim et al. 2014a). Such reductions are highly important to increase the CSP market share as they enable the reduction or even elimination of government incentives. In addition they make the projects more attractive to private investors.

Due to the need for economies-of-scale to attain lower specific costs (IRENA 2012) capacities for most CSP-only plants are between 10 and 280 MWe to ensure commercially viability. Large capacities coupled with technology and TES-dependent specific costs of AU\$ 4.6-7.4m/MWe (Lovegrove et al. 2012) result in hundreds of millions or even billiondollar project budgets. For example the 50 MWe Andasol plant in Spain has a budget of € 300m (Solar Millennium AG 2006), 280 MWe Solana plant has a budget of US\$ 2b (NREL 2014d), and the 392 MWe Ivanpah plants has a budget of US\$ 2.2b (NREL, 2013b). Such large projects are not only difficult to finance but also incur a higher risk premium than kilowatt-scale PV systems. However, CSP hybrids can have significantly smaller capacities, and this reduces budget size and financial risk as shown in the 9.3 MWth CSP feedwater heating augmentation system at Liddell power station in Australia (Novatec Solar GmbH 2012a). Without CSP expertise, power plant owners and financiers are likely to favour technologies they know over CSP when deciding on new generation assets, assuming similar financial conditions. Initially building smaller CSP plants would allow the developers, and also equipment suppliers, to gain local CSP expertise without committing too much capital. The establishment of smaller plants will also enable plants to upskill staff in parallel to day-to-day business, and to build the confidence needed to subsequently operate larger CSP installations (Peterseim et al. 2014d).

Compared to lower cost but also lower average capacity factors for wind and PV generation, 20% for PV and 30% for wind (Hearps & McConnel 2011), CSP can reach high capacity factors by incorporating TES, e.g. 74% demonstrated in Gemasolar plant (SENER

2014). However, TES was at AU\$ 90/kWh (thermal) in 2011 CapEx intensive (Hinkley et al. 2011) and hybrid systems can lower plant costs by reducing or even avoiding high cost TES investment. With TES costs expected to decrease to AU\$ 22/kWh (thermal) by the end of this decade (Hinkley et al. 2011) this advantage of CSP hybrid developments will decrease but not disappear due to the inherently lower cost and lower complexity of integrating a EfB, EfW or natural gas steam system. Additionally, CSP hybrid plants can meet variable daily electricity demand with the host plant operating constantly and the CSP component providing additional capacity during the day when electricity demand and prices are typically significantly higher, as demonstrated by two plant proposals in Australia, the 30 MWe CSP–EfB hybrid for Griffith (Peterseim et al. 2014c) and the 35.5 MWe CSP-multiple feedstock hybrid for Swanbank (Peterseim et al. 2012a).

Typically, CSP-only plants require a DNI of at least 2,000 kWh/m²/year to be commercially viable but CSP hybrid plants have been built in locations with only 1,800 kWh/m²/year (Morell 2012) and have been considered at even lower DNI levels of as low as 1,691 kWh/m²/year (Pérez & Torres 2010). Moving CSP out of remote arid desert/semi desert regions closer to agricultural/urban regions expands potential CSP sites, reduces transmission losses, potentially avoids new and capital intensive transmission infrastructure, and enables access to back-up fuel sources (e.g. agricultural and urban waste feedstocks). Being close to load centres also increases opportunities for highly efficient cogeneration as well as industrial heating/cooling applications. The use of biomass and waste feedstocks in larger hybrid plants increases, albeit involving some part-load operation, the plant's cycle efficiency compared to a EfB- and EfW-only facilities and thereby increases annual generation (Peterseim et al. 2014a).

All power projects create employment during construction and operation but renewable energy systems have higher socio-economic benefits than fossil fuel powered plants (Caldés et al. 2009; Wei, Patadia & Kammen 2010). The employment benefits are very valuable to the local communities but they also contribute to operational cost. Efforts are being made to minimise capital and operational cost, as in fossil fuel fired power plants, and in the future the employees required, personnel per MWh, to operate renewable energy plants is likely to decrease with more learning experience. Like CSP-only plants, CSP–EfB and CSP– EfW hybrids can increase direct and indirect employment through ongoing feedstock collection, processing and transport. For EfB facilities this is far more significant than the direct employment needed to operate the actual power plant (Thornley, Rogers & Huang 2008). This also broadens community involvement and increases long lasting acceptance of a power generation asset. In addition to primary benefits, such as renewable energy or local employment, CSP–EfB and CSP–EfW hybrids can provide secondary benefits such as bushfire hazard reduction through biomass collection and landfill diversion.

Despite the numerous benefits of CSP hybrid plants they face some additional challenges compared to CSP-only installations. A potential challenge is the perceived technology risk associated with new power plant concepts and the limited number of CSP hybrid plants already in operation and currently under construction (see Section 2.5). Although individual CSP components have been commercially proven, the various hybrid concepts, when first proposed, will incur higher financing costs due to the need for risk mitigation. In addition to the limited operational expertise with CSP hybrid plants, environmental approval processes and technical standards might be additional and time consuming obstacles for first local reference plants.

Resource security is another key consideration as CSP hybrid plants not only require a sufficiently high DNI but also need a continuously available second energy source, such as natural gas or biomass. Also these resources have to be secured contractually, which is a significant challenge over the circa 30 year power plant lifetime. Identifying sites that fulfil the resource criterion in addition to standard requirements, such as transmission infrastructure, water supply, and road infrastructure is not trivial and increases the work required for project development. Moreover, current transmission infrastructure in Australia is designed for centralised power generation (Dopita & Williamson 2009). This makes the grid connection of distributed generators, which comprises most renewable energy systems including CSP–EfB and CSP–EfW hybrid plants, more complicated and costly compared to utility scale CSP-only plants with capacities of more than 100 MWe.

Despite the aforementioned cost reductions of CSP hybrid plants their electricity costs would still exceed the combination of the current average wholesale electricity price of AU\$ 50/MWh (RepuTex 2013), which includes a carbon price of AU\$ 24.15/t CO₂ (Australian Government 2011), and the REC price of AU\$ 30/MWh (RepuTex 2013). Economic viability is a significant barrier for countries with low wholesale electricity prices, such as Australia (Bureau of Resources and Energy Economics 2013) and plant siting is key. It is important to identify locations where network constraints exist and competing energy sources have higher costs.

Apart from technical and economic challenges, socio-cultural factors have to be considered too. Proposals for new gas and coal fired power plants with a CSP contribution and CSP retrofits to existing fossil fuel plants can face community opposition as the CSP component could be perceived as "greenwashing" of fossil electricity generation. The hybridisation of fossil fuel and CSP power generation is currently seen as a cost efficient option for the transition to a low carbon future but it needs to be borne in mind that such plants could also be used to unnecessarily extend the transition process. This is a particular concern when the CSP contribution to the overall hybrid plant output is small.

In addition, the use of controversial fuels, such as coal seam gas in such plants is likely to cause community concern. For CSP–EfB and CSP–EfW hybrids, community acceptance issues could arise from concerns around ongoing debates about fuel vs. food, material vs. energy recovery, native forests for bioenergy, and land use changes. These points can be addressed through appropriate feedstock selection (Farine et al., 2011) and require active community engagement strategies.

2.5 CSP hybrid plants

A variety of CSP hybrid plants exist and are being investigated. To cover the diversity of the field this section provides an overview of CSP–EfB and CSP–EfW and other CSP hybrid options. These include fossil and renewable hybrids with industrial- to utility-scale power generation capacities. Currently, most CSP hybrids involve the use of fossil fuels but fully renewable hybrids exist and are under development.

Figure 18 shows the two main CSP integration options: a power top-up and a fuel saver. Fuel saver options are ideally used in carbon intensive power generation assets such as coal fired power stations, while the power top-up option provides additional capacity during daytime periods of high demand and high prices. The decision about whether to adopt the power top-up or the fuel saver model depends not only on the fuel type but also on local market conditions, such as electricity demand, price structure, carbon pricing, power plant operation, and the overall renewable energy contribution to the network.





2.5.1 CSP-Natural gas

Currently, most CSP hybrid plants in the world use natural gas as it is a promising option to further lower the carbon intensity of natural gas fired power plants while also reducing consumption of an increasingly expensive fuel. The CSP contribution to the annual generation of current reference plants is typically small but technical concepts for high solar share CSP hybrids exist.

2.5.1.1 Commercial references

Several integrated solar combined cycle (ISCC) plants are already in operation worldwide with CSP equivalents of 75 MWe at the FPL Next Generation plant in the USA (Florida Power & Light Company 2010) (see Figure 19a), 20 MWe at the Hassi R'Mel plant in Algeria

(Abengoa 2013), 20 MWe at the Ain-Beni-Mathar plant in Morocco (Abengoa 2013), and 19.2 MWe at the Kuraymat plant in Egypt (Brakmann et al. 2009). These plants use parabolic through systems with thermal oil to provide high pressure saturated steam to the heat recovery steam generators (HRSG) of the combined cycle power plants for further superheating and have no TES. The only exception is the 5 MWe Archimede ISCC plant in Italy, which has TES and uses parabolic troughs with molten salt to generate steam parameters identical to the HRSG of the combined cycle plant (Falchetta et al. 2009). More ISCC plants are under construction in Mexico (Abengoa 2013), the USA (U.S. Department of Energy 2013b) and Canada (SkyFuel 2012).

In current ISCC plants, the CSP portion of the total plant capacity is usually relatively small at less than 10%. The only CSP-natural gas hybrid plant with a CSP share of more than 80% of installed capacity is the 100 MWe Shams One plant in the United Arab Emirates (Goebel & Luque 2012) shown in Figure 19b. The Shams One plant consists of a conventional parabolic trough with thermal oil and it has natural gas fired external superheaters to increase the 380 °C CSP steam to 540 °C thereby increasing cycle efficiency (Goebel & Luque 2012). The higher efficiency reduces the solar field size which is the most capital intensive component of a CSP plant.



Figure 19: a:75 MWe equivalent CSP steam boost to Martin Next Generation power station in the USA (Florida Power & Light Company 2010) and b: 100 MWe Shams One plant (Goebel & Luque 2012)

Almost all CSP plants have a natural gas fired heaters but these are not designed to provide full plant capacity or continuous operation. Rather, they are for plant start-up, generation stabilisation, and emergency operation. They are therefore not considered CSP hybrids. However, full capacity gas fired boilers or heaters in parallel to the CSP component is a technical option, and such a unit is in fact installed in the Shams One plant to ensure supply availability at all times (Goebel & Luque 2012). However, the natural gas conversion efficiency of these systems is slightly lower than in a combined cycle plant. To limit gas consumption in back-up systems, Spain limited the natural gas contribution to 15% and the Shams One plant is expected to provide only 4% of annual generation through the gas fired heater system (Goebel & Luque 2012).

2.5.1.2 Concepts

In addition to the reference plants available, a variety of other concepts exist for CSP natural gas hybrid Rankine cycle plants. Several studies analyse and outline ways of optimising the currently dominant ISCC concept with parabolic trough plants using thermal oil to provide saturated steam to the HRSG's high pressure steam cycle (Baghernejad & Yaghoubi 2010; Dersch et al. 2004; Horn, Führing & Rheinländer 2004; Kelly, Herrmann & Hale 2001; Ugolini, Zachary & Park 2009). To increase the annual CSP contribution, adaptations were suggested to include steam reheating from parabolic trough (Kane et al. 2000) and Fresnel fields (Siva Reddy, Kaushik & Tyagi 2012). To reduce the investment needed and to simplify CSP integration, parabolic trough systems with direct steam generation are being investigated to provide saturated steam to the high pressure steam cycle of the HRSG (Montes et al. 2011) and to provide superheated steam directly to the steam turbine (Nezammahalleh, Farhadi & Tanhaemami 2010). Similarly, the use of CO₂ as the heat transfer fluid in the solar field (Cau, Cocco & Tola 2012) and the use of CO₂-steam cycles (Gou, Cai & Hong 2007) have been investigated but both technologies still require significant R&D work before commercialisation.

Further concepts using solar towers with direct steam and molten salt were investigated as ways to provide high pressure superheated steam to the steam turbine including CSP steam reheating (Peterseim et al. 2012c, 2013a; Ugolini, Zachary & Park 2009). One such example is shown in Figure 20a and apart from providing high cycle efficiencies and high solar shares when incorporating TES, the concept allows the independent operation of the CSP and natural gas plants. A recent comparison of ISCC plants with parabolic trough and solar tower systems confirmed that solar towers have a higher power production and efficiency potential than trough systems (Franchini et al. 2013). To maximise overall cycle efficiency and provide drinking water, ISCC concepts coupled with seawater desalination have been investigated (Alrobaei 2008).

Most ISCC designs have a low CSP contribution to annual generation due to a) the lack of TES and b) cycle efficiency reductions at times when CSP is not contributing to steam turbine operation. One CSP hybrid concept with a greater than 50% CSP share of annual generation uses a gas turbine in parallel with a parabolic trough plant, either with thermal oil or molten salt, with the exhaust energy from the gas turbine being recovered by the solar field's primary working fluid cycle (Turchi & Erbes 2011; Turchi & Ma 2014). The

concept is being analysed with and without TES and has the benefit of very flexible power generation due to the quick start-up capabilities of gas turbines and the highly efficient use of natural gas in the gas turbine and heat recovery systems.

Unlike the use of CSP and natural gas in Rankine cycle plants, extensive research into solarised Brayton cycles is ongoing due to their high efficiency and lower cost prospects. Rather than thermal oil, molten salt or direct steam systems Brayton cycle applications heat air after the compressor part of the gas turbine to 800–1,200 °C (Ávila-Marín 2011) and re-inject it with natural gas into the gas turbine's combustion chamber. The first detailed studies go back to the 1990s with continuous concept improvements and prototypes (Ávila-Marín 2011) until in 2012 the first pre-commercial 4.6 MWe Solugas tower, shown in Figure 20b, commenced operation in Spain (Quero et al. 2013). Currently, prototype and demonstration plants operate as open cycle units but the aim is to increase efficiency by adding a bottoming Rankine cycle and TES.



Figure 20: a: 228 MWe solar tower ISCC concept (Peterseim et al. 2012c) and b: 4.6 MWe Solugas tower in Spain (Quero et al. 2013)

A different approach to integrating CSP with natural gas is the injection of steam from a CSP system into the gas turbine's combustion chamber (Livshits & Kribus 2012). In plants that do not use CSP this is a common technology for improving the performance of open and combined cycle plants. It works by taking steam from the HRSG. CSP could also be used to cool the inlet air of gas turbines via absorption chillers to avoid performance degradation at high ambient temperatures (Bellac & Destefanis 1999; Popov 2014). This concept is also common in non-CSP plants and it works by using steam from the HRSG to operate the absorption chiller, or by using electricity from the generator to operate mechanical chillers. Currently, the least mature but technically promising CSP-natural gas hybrid concept is the thermochemical conversion of natural gas to a higher calorific value synthesis gas,

predominantly consisting of hydrogen. Research on this concept commenced in the 1980s and since then steam reforming of natural gas has been successfully demonstrated in a 300 kWe gas turbine test facility in Australia (Stein et al. 2009). However, with CSP only providing the heat for the thermochemical process, the solar energy content of the synthesis gas is limited to 26% (Stein et al. 2009). Solar tower and dish systems have been favoured in thermochemical concepts due to their ability to provide high temperatures and therefore high conversion efficiency. However the greater maturity of parabolic trough systems has led to recent thermochemical investigations of this collector type (Bianchini, Pellegrini & Saccani 2013). Mature thermochemical CSP-natural gas hybrids would allow the use of highly efficient combined cycle plants and lower cost thermochemical energy rather than sensible heat storage.

2.5.2 CSP-Coal

The hybridisation of CSP with coal, particularly in retrofits, is widely discussed as a low-cost option for reducing the greenhouse gas intensity of these assets while simultaneously progressing knowledge about the construction and operation of CSP (Jamel, Abd Rahman & Shamsuddin 2013; Meehan 2013; Siros et al. 2012; Turchi et al. 2011). However, CSP retrofits have to be considered carefully on a project-by-project basis as the CSP contribution to the annual generation is typically small and the renewable asset could be used as a pretext for extending the operating life of the station. Section 4.1 discusses this contentious issue in more detail based on two reference plants in Australia.

2.5.2.1 Commercial references

Currently, the only operational CSP-coal hybrid plant worldwide is the 9.3 MWth feedwater heating augmentation facility at the 2 GWe Liddell power station in Australia (Macquarie Generation 2012; Novatec Solar GmbH 2012a). The solar field is providing saturated steam at 55 bar to the coal plant's high pressure feedwater heater (Lovegrove et al. 2012; Novatec Solar GmbH 2012a). Another 4.4 MWth feedwater heating augmentation system was in operation for one year at the 49 MWe unit 2 of the Cameo power station, in the USA but it has since been decommissioned (Xcel Energy Inc 2010). The largest plant yet, a 44 MWe equivalent solar boost to the 750 MWe Kogan Creek power station – shown in Figure 21 – in Australia is designed to provide steam to the cold reheat line of the power plant. It is expected to come online in 2015 (NREL 2014c). A second project under construction is the 5 MWe equivalent solar boost to the 156 MW Unit 4 of the Sundt power station in the USA

(AREVA Solar 2012c). Except for the Cameo hybrid plant which uses the parabolic trough technology, all of these plants use a Fresnel solar field and no reference plant has TES.



Figure 21: Kogan Creek Solar Boost project under construction as per October 2013⁷

2.5.2.2 Concepts

A variety of concepts exist for the hybridisation of CSP with coal. Most focus on feedwater heating and steam supply into the cold reheat line, with references already in operation and under construction as shown in the previous section. Feedwater heating options for lignite and black coal power plants receive most attention, with several case studies worldwide identifying their carbon emission reduction and low cost potential (Bakos & Tsechelidou 2013; Hu et al. 2010; Meehan 2013; Mills, Lièvre & Morrison 2003; Pierce et al. 2013; Popov 2011; Siros et al. 2012; Suresh, Reddy & Kolar 2010; Turchi et al. 2011). Although the world's largest CSP-coal hybrid under construction at the Kogan Creek power station uses CSP to provide additional steam to the cold reheat line (CS Energy 2011; NREL 2014c) little research has been carried out in this field. However a recent report identified further opportunities for this CSP hybrid option combined with feedwater heating in Australia (Meehan 2013). Two studies on the augmentation of CSP to fossil fuel plants in Australia and the USA concluded that steam from a conventional parabolic trough plant with thermal oil could be integrated before the coal boiler's superheater, while steam from solar towers could be directed straight into the high pressure steam turbine section of subcritical power plants (Turchi et al., 2011, Meehan, 2013). Principally, parabolic trough plants with molten salt, as demonstrated in the Archimede facility in Italy (Falchetta et al. 2009), would also be capable of providing CSP steam to the high pressure steam section turbine of a subcritical coal fired power plant. The use of CSP to provide supercritical steam to a coal fired power station has not been investigated yet.

⁷ http://kogansolarboost.com.au/news/photo-gallery/?album=1&gallery=1, accessed 04 April 2014

A more experimental concept is the possible use of CSP to provide steam to the reboiler of a carbon capture plant in coal fired power stations (Li, Yan & Campana 2012; Mokhtar et al. 2012; Qadir et al. 2013). The aim of this integration option is to minimise the coal plant's output reduction by providing the thermal energy required for the CO₂ separation process with CSP and rather than from the coal fired boiler. A research/industry consortium between CSIRO and Delta Electricity is investigating this option closely and is currently testing a prototype plant in Australia (Australian Renewable Energy Agency 2013b; McGregor et al. 2013).

The use of CSP to support the gasification of coal and the subsequent synthesis gas combustion in a combined cycle power plant is being investigated as a possible means of further reducing the carbon intensity of integrated gasification combined cycle plants (Ng & Lipiński 2012; Ozturk & Dincer 2013; Zedtwitz & Steinfeld 2003). A prototype which heats coal with molten salt prior to gasification is currently being designed in the USA (National Energy Technology Laboratory 2013).

2.5.3 CSP-non-conventional fuels

CSP-non-conventional fuels refers to CSP hybrids which use various solid biomass and waste feedstocks. Such plants are a fully or predominantly renewable power generation option, depending on waste feedstock composition, and while it is a niche market at the moment, the high renewable energy prospects and secondary benefits, such as bushfire hazard reduction through biomass collection and landfill diversion, make the implementation of such facilities attractive.

2.5.3.1 Commercial references

The only commercial CSP–EfB reference in the world is the 22.5 MWe (net) Termosolar Borges plant in Spain (Protermosolar 2012). It is located further north than any other CSP project. Figure 22 shows the plant, which consists of parabolic trough collectors with thermal oil to generate saturated steam at 40 bar and two biomass boilers which subsequently superheat the CSP steam to 520 °C (Morell 2012). The other operational plant is a trigeneration demonstration facility at the University of Phitsanulok in Thailand. It is able to generate up to 50 kWe of electricity and 500 kWth for heating and cooling purposes (Solarlite CSP Technology GmbH 2014). The demonstration plant was completed in 2012 and combines parabolic troughs with direct steam generation and a biomass-fired boiler. In 2012 the Italian utility ENEL in a consortium with a finance and a research organisation announced the construction of a 1-5 MWe CSP–EfB hybrid (ENEA-Press and Media Relations Office 2012) but no information about the progress of this project is available. In the past several other CSP–EfB projects were developed worldwide, such as a 106.8 MWe San Joaquin plant in the USA (California Energy Comission 2008), but no other project has yet commenced construction.



Figure 22: First CSP-biomass hybrid plant in Spain, 22.5 MWe Termosolar Borges, Spain

2.5.3.2 Concepts

In the 1980s the combination of dish systems with biomass, waste and other feedstocks was suggested (McDonald 1986) but the concept was not even trialled at the prototype stage. In 2002 a 12.8 MWe CSP–EfB hybrid plant was investigated in Spain using two biomass gasifiers in combination with a solar tower providing 680 °C air to an HRSG (Romero et al. 2002). However, due to the choice of immature technologies the project did not proceed and the first CSP–EfB hybrid, the aforementioned Termosolar Borges plant in Spain, uses the conventional parabolic trough and biomass combustion technologies to minimise risk. Technology choice is important as financiers and plant operators are typically very risk averse.

A variety of more conventional CSP–EfB and CSP–EfW concepts have been investigated since then, including the use of CSP for feedwater and combustion air preheating to increase power generation but these options only allow a small 1–2% solar contribution (Kaeding 2010; Spliethoff et al. 2010). The generation of identical steam parameters from the CSP, CSP–EfB and CSP–EfW steam generators to enable joint steam turbine operation is an option to provide a significantly higher solar contribution and raise plant capacity. Studies are available which apply such concepts to the use of parabolic troughs (California Energy Comission 2008; Cot et al. 2010; Kaeding 2010; Pérez & Torres 2010; Schnatbaum 2009; Spliethoff et al. 2011; Peterseim et al. 2014c; Romero et al. 2002). The combination of parabolic troughs to provide identical steam parameters to EfB and EfW boilers is not ideal as the comparatively low thermal oil temperatures of less than 400 °C would limit steam temperatures to under 390°C at appropriate pressures, which in turn limits the biomass and

waste feedstock conversion efficiency. Such feedstocks could be converted to \geq 400 °C steam without encountering high-temperature corrosion and ash fusion-related issues inside the EfB and EfW boilers. This includes EfW plants using MSW where most units operate with steam parameters of 380–400 °C at 40 bar while newer units have higher parameters due to corrosion resistant superheater alloys (e.g. an EfW plant in Amsterdam with 440 °C and 130 bar (Gohlke 2008). To maximise biomass and waste feedstock conversion efficiency, Fresnel systems with steam temperatures up to 500 °C (Morin et al. 2011) and solar towers up to 565 °C, as in the Ivanpah plants (NREL 2014a), are better suited for hybridisation (Peterseim et al. 2013b). Dish systems can also reach such temperatures but are significantly less mature.

To improve the efficiency of EfW plants a study investigated the use of CSP to externally superheat the 380 °C steam from an EfW plant to 420 °C at 40 bar using the Fresnel technology (Kaeding 2010). The small temperature increase limits cycle efficiency gains to 0.2% but means that the plant would not require natural gas to maintain a temperature of 420 °C at times when CSP is not available as the steam turbine could continue operation at 380 °C. Some EfW plants already use the external superheater concept with natural gas rather than CSP. There are plants in operation which use superheaters directly fired by natural gas to raise the EfW steam outlet temperature from 412 °C to 522 °C at 65 bar(Andersen, Oksen & Olesen 1992) , and there are also combined cycle plants which raise the EfW steam outlet temperature from 400 °C to 510 °C at 100 bar in the HRSG (Toebes & Beker 1998). It is technically possible to realise similar steam temperature increases with CSP superheaters. However, the significant temperature rise would, even with significant TES, require the use of natural gas at times when CSP in not available.

The CSP-assisted gasification of biomass into synthesis gas is a different process to the hybridisation of CSP with biomass and waste feedstocks in mature Rankine cycle plants. CSP-assisted biomass gasification has recently been investigated in detail (Kruesi, Jovanovic & Steinfeld 2014; Kruesi et al. 2013; Lichty et al. 2010; Piatkowski et al. 2011; Saade et al. 2012; Woodruff & Weimer 2013) and since 2009 a 1 MWth test facility has been in operation in the USA (Service 2009). The benefits of a CSP-biomass-derived synthesis gas would be its use in high efficiency combined cycle plants, thermochemical energy storage, and transport fuel applications. These benefits are identical to those obtained from CSP-assisted steam reforming of natural gas, (see Section 2.5.1.2), with the difference that a CSP-biomass synthesis gas would be produced using 100% renewable energy compared to

the 26% embedded solar energy involved in using CSP to reform natural gas (Stein et al. 2009).

2.5.4 CSP-geothermal

Typically, the net plant efficiency of geothermal plants is lower than that of other thermal energy plants due to the inherently low enthalpy of the geothermal resource (Zhou, Doroodchi & Moghtaderi 2013). Also some geothermal resources are located in arid regions where plants with air-cooling systems would suffer from efficiency losses during high ambient temperature periods. This means CSP is an attractive option for hybridisation with geothermal plants that face these constraints. Several concepts have been investigated for Rankine and Organic Rankine cycle plants using CSP for brine preheating, additional steam generation to maintain output during high ambient temperature periods, and steam superheating (Astolfi et al. 2011; Peterseim et al. 2014d; Rawlins & Ashcroft 2013; Zhou, Doroodchi & Moghtaderi 2013). While no commercial CSP-geothermal hybrid plants have yet been developed, studies are being carried out in the USA (Nelson & Larsen 2013) and Turkey (Kuyumcu et al. 2013) and according to parabolic trough technology provider Skyfuel a 17 MWth test plant is currently under construction at the 34 MWe Stillwater geothermal plant in the USA (Frazier 2013). The benefit of raising the efficiency of a geothermal plant is the reduction of the required number of production and injection wells which are the most expensive part of a geothermal plant, or the increase in plant capacity(Peterseim et al. 2014d).

The use of CSP to further superheat the geothermal steam is particularly promising (see Figure 23) as it has the potential to raise Rankine cycle efficiency significantly, by up to 23% in systems with parabolic troughs and by up to 38% in solar tower systems (Peterseim et al. 2014d), and it has the potential to lower the LCOE from AU\$ 225/MWh to AU\$ 165/MWh (Zhou, Doroodchi & Moghtaderi 2013). To minimise the use of natural gas during times without solar irradiance, substantial TES, such as 15 h, is important in order to continue CSP heat input but currently this also represents a significant cost factor. The small temperature differences of 380°C in the hot to 280°C in the cold tank render parabolic trough plants with large TES less economical than solar towers with 540°C in the hot and 280°C in the cold tank (Peterseim et al. 2014d). The hybridisation of CSP and geothermal energy sources certainly has technical and economic benefits but is has to be seen as a niche market with limited applications worldwide.



Figure 23: Schematic diagram of the hybrid solar-geothermal power plant (Zhou, Doroodchi & Moghtaderi 2013)

2.6 Introduction to transition management and transition theory

Electricity demand is expected to more than double globally between 2010 and 2050 and supplying this demand with renewable and fossil energy sources is expected to require a cumulative investment of up to US\$ 26 trillion, with up to 70% of this investment being for renewable energy plants (Frei et al. 2013). This represents a significant change from the current investment pattern to a future electricity mix. The research field of transition management has been used to analyse such energy transitions, and others, for various countries across the world. First analyses came from the Netherlands where it has been used to understand, design and evaluate transition pathways to a low carbon energy system by analysing the role of public policy (Kern & Smith 2008; Rotmans, Kemp & van Asselt 2001), technical developments, and changes in rules, visions and social networks (Verbong & Geels 2007). Later similar analyses for pathways to low carbon energy systems, considering differences in local policy, infrastructure and society, were investigated for other countries, including the UK (Foxon, Hammond & Pearson 2010), Panama (Lachman 2014), and Israel (Teschner et al. 2012). Transition management can be considered as a "a form of process management against a set of goals chosen by society" (Kemp & Loorbach 2003, p. 12) and Rotman et al. (2001) identified five main characteristics of transition management:

- long-term thinking of at least 25 years
- thinking in more than one domain (multi-domain) with different actors (multiactor) at different scale levels (multi-level)
- a focus on learning and a special learning philosophy (learning-by-doing and doingby-learning)
- encouraging system innovation alongside system improvement
- keeping a large number of possible options (wide playing field).

Due to the inherent uncertainties involved when intervening in complex systems, transition management does not aim to bring about a specific transition; that is, transition management is not committed to a specific future energy system, but instead works towards a transition that "offers collective benefits in an open, exploratory manner" (Rotmans, Kemp & van Asselt 2001, p. 22), for example as a result of a move towards low carbon intensity electricity generation. Consequently, there is unavoidable uncertainty around future transition pathways and so continual monitoring, refinement, and adjustment is required over time. To influence transitions, a cyclical model with four not necessarily sequential components, the transition management cycle shown in Figure 24, was introduced. This model suggests the adoption of the following approach:

(1) structure the problem in question, develop a long-term sustainability vision and establish and organize the transition arena; (2) develop future images, a transition agenda and derive the necessary transition paths; (3) establish and carry out transition experiments and mobilize the resulting transition networks; (4) monitor, evaluate, and learn lessons from the transition experiments and, based on these, make adjustments in the vision, agenda, and coalitions (Loorbach 2010, p. 172).

Due to the transition context, and the actors and problems involved, every transition management process is different but the cycle is *"flexible enough for adaptation but prescriptive enough to be functional in practice"* (Loorbach 2010, p. 172).



Figure 24: Transition management cycle (Loorbach 2010)

Transition management defines a transition as a "process of change where the structural character of a society (or a complex sub-system of society) transforms" (Rotmans, Kemp & van Asselt 2001, p. 16) and the electricity market is such a complex sub-system of society. Additionally, transitions are seen as "a set of connected changes, which reinforce each other but take place in several different areas, such as technology, the economy, institutions, behaviour, culture, ecology and belief systems" (Rotmans, Kemp & van Asselt 2001, p. 16). Throughout the transition process "new products, services, business models, and organizations emerge, partly complementing and partly substituting for existing ones" (Markard, Raven & Truffer 2012, p. 956). Enabling a transition in a certain sub-system of society requires a network of innovative actors to develop, through participatory processes, the required long-term visions, agendas and experiments which in transition management are said to form the "transition arena" (Loorbach 2010).

Traditionally, technical transitions have been used to analyse how new technologies replace existing ones but this approach is insufficient to explain the transformation of a complex sub-system of society. Socio-technical transitions extend the technical dimension by including changes in user practices and institutional structures as well as the consideration of complementary technological and non-technical innovations, such as complementary infrastructures (Markard, Raven & Truffer 2012). In light of climate change, sustainability transitions were introduced to shift socio-technical systems to more sustainable modes of production and consumption (Loorbach 2010; Markard, Raven & Truffer 2012; Schot & Geels 2008) and a key aspect of this is the role of governance to guide the process through incentives and constraints (Loorbach 2007; Smith, Stirling & Berkhout 2005).

To better understand transitions a multi-level perspective (MLP) with three levels has been introduced (see Figure 25), consisting of the socio-technical landscape, also known as the macro level; regimes, also known as meso level; and niches, also known as micro level (Geels 2002; Rip & Kemp 1998; Rotmans, Kemp & van Asselt 2001). The landscape provides an exogenous context and consists of factors that change slowly over decades, such as the natural environment, geo-economic and political developments, political culture, demography and social values. However, in some instances the landscape can also change quickly through shocks, such as wars (van Driel & Schot 2005). The landscape level exerts pressure on both the regime and niche levels. The regime level is based on dominant social, technological, economic, environmental and political structures. It provides stability and includes industry networks, techno-scientific knowledge as well as dominant practices and policies. It has various actors from industry, research and government and they often have different interests. The niche level provides the innovations that can enable transformative changes in technology, society, culture, business and governance. They are developed through small networks or individuals and relate to technologies and practices.



Figure 25: Multi-level perspective (Geels 2002)

Figure 25 shows that the landscape embeds the various regimes while these embed the various niches. Innovations are created "*in the context of existing regimes and landscapes with its specific problems, rules and capabilities* [...] on the basis of knowledge and capabilities and geared to the problems of existing regimes" (Geels 2002, p. 1261). Innovations are initially unstable and require protection against mainstream market selection (Kemp, Schot & Hoogma 1998). According to the MLP, transitions occur through interactions between the three levels in which:

(a) niche-innovations build up internal momentum, through learning processes, price/performance improvements, and support from powerful groups, (b) changes at the landscape level create pressure on the regime and (c) destabilisation of the regime creates windows of opportunity for niche-innovations (Geels & Schott 2007, p. 400).

Hence the success of an innovation is dependent on supportive developments in the regimes as well as the landscape. The concept of strategic niche management (SNM) has been introduced to describe these processes in more detail (Kemp, Schot & Hoogma 1998; Schot & Geels 2008). SNM is "not a technology push approach [but] sustainable development [that] requires interrelated social and technical change" (Schot & Geels 2008, p. 538) and it therefore fits well into the MLP. Being a management tool it is based on

the creation, development and controlled phase-out of protected spaces for the development and use of promising technologies by means of experimentation, with the aim of (1) learning about the desirability of the new technology and (2) enhancing the further development and the rate of application of the new technology (Kemp, Schot & Hoogma 1998, p. 186).

In today's internationally connected environment many niche innovations have to be considered in a global market and therefore "sequences of local projects may gradually add up to an emerging field (niche) at the global level" (Schot & Geels 2008, p. 543). Figure 26 illustrates the concept which can be observed in technology learning processes where projects in one country provide valuable learning in regards to the cost, policy and acceptance of other projects elsewhere and in doing so they help to create technical trajectories.



Figure 26: Emerging technical trajectory carried by local projects (Geels & Raven 2006, p. 379)

Depending on the timing and nature of interactions within the landscape, regime and niche levels, five different transition pathways have been theorised, in addition to a reproduction pathway where there is no external landscape pressure, a situation which allows the regime to reproduce itself (Geels & Schot 2007). While these pathways do not necessarily occur in their pure form, each has a recognisable internal logic:

- 1. Transformation pathway: Moderate landscape pressure does not allow insufficiently developed innovations to break into the regime and the regime actors respond to this by changing the development path and innovation activities. In this pathway new regimes grow out of old ones through adjustments and reorientations and most regime actors survive the transition.
- 2. De-alignment and re-alignment pathway: Significant landscape pressure or a specific shock increases problems in the existing regime and causes its actors to lose faith in it with the consequence that the regime de-aligns. Without sufficiently developed innovations a clear substitute is missing and various innovations emerge in the vacuum. Over time one innovation becomes dominant and the regime re-aligns.
- Technological substitution pathway: Significant landscape pressure or a specific shock increases problems in the existing regime but this time innovations are ready to break through and replace it.
- Reconfiguration pathway: Moderate landscape pressure allows the regime to implement some sufficiently developed symbiotic innovations to solve local problems but it remains mostly unchanged (the transformation pathway). Over

time regime actors explore further combinations, implement more innovations and thereby adjust the basic architecture of the regime.

5. Sequence of pathways: Disruptive changes in the landscape can lead to a sequence of the aforementioned pathways. Initially, regime problems are solvable with adjustments (transformation pathway) but over time problems grow and symbiotic innovations are also implemented (reconfiguration pathway). If this proves insufficient to solve the regime's problems some less developed innovations (dealignment and re-alignment pathway) or developed innovations (substitution pathway) enter and subsequently change the regime. Other sequences with fewer pathways are possible.

Recently, another pathway has been proposed (Papachristos, Sofianos & Adamides 2013) in which at least two stable regimes with landscape and/or internal pressures do not have niche innovations for problem solving and therefore a new system emerges on the fringes of the existing ones.

Transition management is a continuously evolving field with ongoing detailed analyses of the role of government, inter- and intra-level interaction, innovation uptake and other issues. It has been used by academia and industry to understand historic transitions and to influence future transitions and it is frequently used to analyse and facilitate the world's transition to a low carbon future.
2.7 Conclusion from the literature review and research gaps

Globally, various R&D activities have been carried out on the hybridisation of CSP with renewable and fossil energy sources and in Australia several studies identified the best areas for CSP-only and fossil fuel plant retrofit options. However, a clear categorisation of CSP hybrids and a broad overview of preferred areas for CSP hybrids with coal, natural gas, biomass, waste, geothermal and wind sources is missing and Section 4.1 addresses these gaps.

Not all of the available research provides clear guidance on the technology selection process and it can be assumed that some choices are based on the selection of products from associated technology providers and extended research association with a particular technology. Technology maturity and efficiency is one of the selection criteria frequently mentioned but others such as land use, technology complexity and cost reduction potential are also relevant. A transparent comparison and evaluation of the different CSP technologies is important and multi-criteria assessments are ideal for identifying the currently preferred technology for a certain application. Multi-criteria assessments have been used in the energy sector to compare fossil fuel and renewable energy options and Section 4.3 applies the method for the first time to CSP hybrids with biomass, waste, natural gas and coal.

In Australia several studies have identified preferred areas for CSP-only plants and CSP retrofits to existing fossil fuel assets in detail, but a similar assessment for CSP–EFB and CSP–EfW hybrids is missing. Section 4.2 provides the first geospatial analysis of the ideal areas and the energy potential for such plants for Australia. Currently, similar studies are not available in any other country in the world.

A variety of research projects have investigated the hybridisation of CSP with renewable and fossil energy sources but the focus of these studies is on the combination of an individual CSP with an individual renewable or fossil fuel technology. For CSP–EfB and CSP– EfW hybrids, steam temperatures were typically limited to less than 500 °C. This is sufficient for air heating, feedwater heating, cold reheat steam. It is also sufficient for scenarios where the biomass/waste feedstock composition limits steam temperatures to under 500 °C due to high-temperature corrosion and ash fusion related issues, as experienced with wood waste, RDF and MSW. However, it is not ideal for CSP–EfB hybrids using feedstocks that allow higher steam parameters, such as wood chips, straw, and bagasse. This is a shortcoming as the identification of the best CSP hybrid concept requires a broad comparison of technologies and feedstocks, and Section 4.4 provides this comparison for CSP–EfB and CSP–EfW hybrid plants. Worldwide and in Australia, a variety of barriers to the broad implementation of renewable energy technologies, including CSP-only plants, have been investigated but the barrier differences between CSP-only plants and hybrid plants have not been analysed yet. An analysis of the barrier differences is part of this research project and is provided in Section 4.5. Barrier rating differences between different stakeholders in energy projects are also considered.

To test the various results of the individual parts of this research project in actual situations, two case studies are provided in Section 4.6. The first case study is a hybrid power plant in Queensland involving CSP and multiple feedstocks and the second is a CSP– EfB hybrid plant in New South Wales.

This research brings together the different areas of a comprehensive technology assessment, including technology selection, techno-economic optimisation, identification of implementation barriers, resource potential and preferred regions. This method provides the first holistic view of CSP–EfB and CSP–EfW hybrid plants, and provides the basis for an analysis of their potential role in Australia's transition to a low carbon future. This analysis is presented in Section 4.7. The methodology used is also transferable to the hybridisation of CSP with other energy sources, such as natural gas or geothermal power.

3 RESEARCH DESIGN

To provide a comprehensive assessment on the potential for CSP–EfB and CSP–EfW hybrid plants in Australia this project takes a transdisciplinary approach covering technical, economic, environmental, socio-cultural and policy matters. To answer the research questions in Section 3.1 five main components and two case studies were investigated. Figure 27 outlines the components and Figure 28 the structure of this research project.



Figure 27: Research outline

The identification of suitable CSP technologies for hybridisation (see Section 4.3) was the first step as it influenced the other research components. Upon completion of the selection of the most suitable CSP technologies the other research tasks were carried out in parallel and informed the case studies, (see Figure 28). The results of the individual research tasks contributed to the analysis of the potential role of CSP–EfB and CSP–EfW hybrids in Australia's transition to a low-carbon energy future (see Section 4.7). The resource assessment and prospective area assessment (see Section 4.2) is essential for obtaining an understanding of the best sites for the construction of CSP–EfB and CSP–EfW hybrid plants while the techno-economic optimisation (see Section 4.4), is important for identifying the best technology combinations for such plants. Various barriers hinder the uptake of renewable energy technologies and an understanding the most significant ones for CSP–EfB and CSP–EfW hybrid plants (see Section 4.5), is important for future deployment. Without

an understanding of the relevant barriers, even the best techno-economic power plant concept will not progress to the construction stage.

Two case studies (see Section 4.6) with different biomass and waste feedstocks were investigated to assess the feasibility of developing such CSP hybrid plants in Australia.

To broaden the applicability of this research, the hybridisation of CSP with other fuels has been examined in some detail (see Section 4.1). This examination also includes a proposal for the categorisation of CSP hybrid plants to better compare the different hybrid concepts and plants available.

Analysing these different components of the research project required the use of a variety of methods (see Section 3.2) and the identification of a suitable theoretical framework (see Section 3.3) to embed the complete work in the context of Australia's transition to a low carbon future.



Figure 28: Research structure showing the sequence of the research components (black arrows) and the information flow (dotted arrows)

3.1 Research questions

The various areas of this project derive from the broad range of research questions it aims to answer. The main question is: What role can CSP–EfB and CSP–EfW hybrid plants play in Australia's transition to a low carbon future? This question is important because lowering the cost of implementing renewable energy is a key priority for government and industry. Section 4.7 addresses this question by investigating potential pathways for CSP-only and hybrid plants from niche technologies to being part of the mainstream energy market. To confidently answer this overarching question, a subset of more specific technical, economic, environmental, social, and policy questions have to be answered. These are:

- What other CSP hybrid options exist with renewable sources? Section 4.1 investigates the hybridisation of CSP with geothermal and wind resources and provides maps with the preferred regions for such plants.
- Where are the ideal regions to build CSP–EfB and CSP–EfW hybrid plants in Australia? Section 4.2 identifies the preferred regions for different CSP-nonconventional fuel hybrids across Australia.
- What is the annual electricity potential of CSP–EfB and CSP–EfW hybrid plants in Australia? Section 4.2 identifies the annual electricity potential for CSP hybrids using single biomass/waste feedstocks or multiple feedstocks. The case studies in Section 4.6 provide details for two specific sites.
- What is the greenhouse gas mitigation potential of CSP–EfB and CSP–EfW hybrid plants in Australia? Section 4.2 identifies the greenhouse gas mitigation potential for CSP hybrids using single biomass/waste feedstocks or multiple feedstocks and the case studies in Section 4.6 provide details for two specific sites.
- Which CSP technologies are ideally suited for hybridisation with EfB and EfW plants? Section 4.3 investigates various CSP technologies in regards to their suitability for integration at different points in a Rankine cycle power plant.
- What are the best concepts for combining CSP with EfB and EfW systems? Section
 4.4 analyses two different high solar share hybrid concepts with various CSP, EfB and EfW technology combinations.
- What are the cost differences between CSP–EfB and CSP–EfW hybrids and CSP-only plants in Australia? Section 4.4 identifies the cost differences for various CSP hybrid options and the case studies in Section 4.6 provide examples for two specific sites.
- What are the key barriers to the development of CSP–EfB and CSP–EfW hybrid plants in Australia? Section 4.5 addresses various barriers and compares them with the barriers for CSP-only plants.

3.2 Methods

In a transdisciplinary research project a variety of areas have to be investigated and a mixed methods approach was adopted which involved both quantitative and qualitative data. The methods applied include thermodynamic, economic, and geospatial modelling, multicriteria decision-making, and workshops and interviews. This Section provides an overview of the methods used and further details are provided in the published papers reproduced in Chapter 4.

3.2.1 Workshops and interviews

To obtain a detailed understanding of a complex topic such as CSP, and to consider the various perspectives different stakeholders have in their capacities as researchers, financiers, consultants, operators, technology providers, and government representatives, it is important to directly engage with these stakeholders to avoid the generation of data that misrepresents reality. It is generally assumed that the accuracy of group judgements is greater than the accuracy of individual judgements (Sniezek & Henry 1989). Therefore two workshops were organised as part of this research project to get the judgement of a large and experienced group. The groups had 49 and 58 participants, and they discussed which CSP technologies are currently the most suitable for hybridisation (see Section 4.3), and what barriers hinder the development of CSP in Australia (see Section 4.5). Interviews were subsequently conducted with the small number of people who had other commitments on the day of the workshop.

In both workshops various pre-selected options were presented to the participants and they had the opportunity to discuss, comment and add to them before rating them from 1 to 9 (Saaty & Ozdemir 2003), with 9 being 'of extreme importance' and 1 being 'of no importance'. Questionnaires were provided for the rating process and an example for the technology selection process is provided in Section 4.3 and one for the implementation barriers in the appendix. The results of the individual ratings were aggregated to provide group and total averages, which in addition to the total ratings provide an understanding of the differences in opinion among the various stakeholders of energy projects. The process for the face-to-face and phone interviews after both workshops was identical to the workshops themselves. The various options were presented to the interviewees, they were given the opportunity to comment, add to the options, and then rate them individually. The University of Technology Sydney's Human Research Ethics Committee reviewed and

approved the design and conduct of the workshop and interview activities and Section 3.4 provides for more details about the ethics approval process.

3.2.1.1 CSP technology selection

For the CSP technology selection process, quantitative data, such as steam parameters and cycle efficiencies, and qualitative data, such as people's individual ratings, were combined using the analytical hierarchy process (see Section 3.2.3) to create the scores for the various technologies assessed. The quantitative data derived from the literature, and calculations were undertaken as part of this thesis, while the qualitative data were gathered during a workshop with 49 participants on 21 July 2011 and during three subsequent interviews. Because of the numerous quantitative data a concurrent embedded approach was taken that *"has a primary method that guides the project and a secondary database that provides a supporting role in the procedures"* (Creswell 2009, p. 214). The secondary database is qualitative and is based on the participants' judgements. Despite the qualitative data being secondary they were crucial in the identification of the best CSP technologies for hybridisation. A more detailed description of the workshop method and the results can be found in Section 4.3.

3.2.1.2 Implementation barriers

To identify the most significant CSP implementation barriers the STEEP analysis method was applied as it examines the social, technical, environmental, economic, and policy aspects of a particular problem or question (Fleisher & Bensoussan 2003). The method is widely used in academia and industry to study developments in a particular environment, such as an organisation, a state or in this case, a country.

The barrier ranking occurred during an expert workshop on 24 June 2013 with 52 professional participants and was complemented by six interviews. Unlike the CSP technology ranking process, this ranking was based purely on qualitative information from the participants' assessments of the different barriers.

A set of 25 pre-selected social, technical, environmental, economic and policy barriers was presented to the audience with some specific examples for CSP-only and hybrid plants. After the presentation the pre-selected barriers were discussed and the participants were invited to add more barriers. Seven barriers relevant to social, technical, economic and policy matters were added and the participants rated all barriers individually for CSP-only and hybrid plants. Of the questionnaires 45 were accepted for further analysis, with the remaining not being filled in properly or not returned. After the workshop all individual ratings were aggregated to identify: a) the most significant individual barriers; b) most significant barrier categories; c) a rating of the differences between participants; and d) barrier differences for CSP-only and hybrid plants. These results can be found in Section 4.5.

3.2.2 Modelling

Thermodynamic, economic, and geospatial modelling tools were used to identify the best techno-economic concepts, the best resource potential, and the most promising areas for CSP–EfB and CSP–EfW hybrid plants in Australia. To minimise the risk of the modelling being inaccurate only commercially proven software packages were used and competent partners were involved where necessary.

3.2.2.1 Techno-economic modelling

Accurate and reproducible techno-economic modelling is a key aspect of this research project and the use of a commercial software package for Rankine cycle modelling was therefore important. In this project the commercial software Thermoflex⁸, versions 22 and 23, were used to identify the best currently available techno-economic CSP-EfB and CSP-EfW hybrid concepts (Section 4.4), the plant efficiencies and costs for the resource assessment (Section 4.2), and the detailed Rankine cycle designs for the case studies (Section 4.6). The software is well established, has been available since 1995, and is widely used in academia and industry to model actual natural gas, coal, nuclear, biomass, waste, geothermal and CSP Rankine cycle plants. Thermoflex is capable of providing design-point heat balances, off-design performance degradations, physical equipment sizes and cost estimates for parabolic trough, linear Fresnel and solar tower plants. Various working fluids can be used, including water-steam, molten salts, and thermal oils, and the modelling of hybrid plants is possible through more than 180 standardised component elements, including boilers, furnaces, gasifiers, feedwater heaters, flow mixers and splitters, and pumps. Other software packages have similar technical capabilities but do not provide power plant and component costs. The integrated cost estimation is unique to Thermoflex and is based on the physical design of the individual plant components as well as countryspecific cost factors. To identify the best CSP-EfB and CSP-EfW concepts, robust cost estimations are important and the ability to make such estimates was, among its technical capabilities, a key reason for selecting the software for this research project.

To ensure that the Thermoflex costing is accurate for the Australian market different CSP Rankine cycle plant models were created and validated against published power plant investment data, such as € 300m for the 50 MWe Andasol II parabolic trough plant with 7.5 h of thermal storage in Spain (Solar Millennium AG 2006), AU\$ 104.7m for the 44 MWe solar boost to Kogan Creek power station in Australia (CS Energy 2011), and AU\$ 120m for

⁸ www.thermoflow.com, accessed 04 April 2014

36 MWe EfB cogeneration plant at Mackay Sugar in Australia (Biomass Power & Thermal Magazine 2011). The Australian default cost factors in Thermoflex were adapted to match the published power plant investment amounts before changing the models to the concepts investigated in Sections 4.2, 4.4, and 4.6. Biomass boilers are not a standard Thermoflex element and had to be built from individual components, such as furnaces, superheaters, evaporators, economisers, and air heaters. The cost estimation is therefore less reliable and to improve cost accuracy prices were obtained for this equipment from the ERK Eckrohrkessel GmbH⁹ in Germany, which has actual EfB and EfW experience from over 1,000 reference plants worldwide.

To incorporate future CSP cost reductions from current plant deployment, the solar field investment was lowered by 10% to reflect 2015 pricing based on a reasonable learning curve (IRENA 2012). Further details of the various Thermoflex modelling assumptions are described in the published papers reproduced in Sections 4.2, 4.4, and 4.6.

3.2.2.2 GIS modelling

Geospatial modelling is widely used in the energy sector globally to identify the best sites and resources for new fossil fuel power plants, and for plants using renewable sources including wind (Baban & Parry 2001; Hossain, Sinha & Kishore 2011; Janke 2010), biomass (Bryan, Ward & Hobbs 2008; Viana et al. 2010), and CSP (Cameron & Crompton 2008; Charabi & Gastli 2010; Clifton & Boruff 2010; Dawson & Schlyter 2012; Feeney et al. 2010; Gastli, Charabi & Zekri 2010; Parsons Brinckerhoff Australia Pty Ltd 2010; Rutovitz et al. 2013) It is also used to identify suitable sites for CSP retrofits to existing fossil fuel power plants (Meehan 2013; Turchi et al. 2011).

Various proven software packages are available on the market, and to analyse the CSP–EfB and CSP–EfW hybrid potential and the best areas for their development (see Section 4.2) the public domain software R¹⁰ provided the capacity for the spatial biomass and waste feedstock analysis and ArcMap¹¹ enabled the production of maps. R is capable of classical statistical analysis, and linear and nonlinear modelling and is widely used for statistical computing. ArcMap is one of the world's leading Geographic Information Systems (GIS) and is able to analyse a range of geospatial information and produce specialised maps based on the combination of various layers. The reason for using R for the spatial biomass analysis and not ArcMap, even though it is capable of the required tasks, is R's ability to more

⁹ www.eckrohrkessel.com, accessed 04 April 2014

¹⁰ www.r-project.org, accessed 04 April 2014

¹¹ www.esri.com, accessed 04 April 2014

quickly and reliably process large datasets as well as simpler programming due to a library of algorithms.

To identify the resource potential and prospective areas for CSP and CSP hybrid plants in Australia, multiple geospatial information sets such as topography, road and rail infrastructure, transmission lines, daily average DNI, biomass production, and population estimates were combined. This information was complemented with data on power plant efficiencies and costs derived from Thermoflex modelling, for the various CSP–EfB and CSP–EfW hybrid plant capacities and feedstocks. In addition several technical, economic, and environmental constraints (e.g. no plants in DNI areas <18 MJ/m²/day, feedstock within 50 km radius around the power lines to limit transport distance and allow access to the electricity grid, and no energy recovery from native forest as well as recyclable waste materials) were applied to provide an accurate assessment of the current potential. Other relevant constraints and further details are listed in the published paper reproduced in Section 4.2.

Obtaining reliable information on the availability of different biomass resources across Australia is a complex task and therefore a collaboration with CSIRO Ecosystems Sciences and CSIRO Energy Technology was formed as they had carried out several individual geospatial analyses on biomass and CSP resource mapping. This research combined the data from the individual CSIRO studies and used this information in conjunction with further research to identify potential sites for CSP–EfB and CSP–EfW hybrid plant development. The collaboration proved successful and it provided all the skills needed for this work, such as geospatial and power plant modelling expertise, and the ability to introduce novel modelling improvements, such as the use of varying efficiencies based on power plant capacity and feedstock.

3.2.3 Multi-criteria decision-making

Multi-criteria decision-making is widely used to analyse complex problems and several models exist, ranging from the simple weighted sum model to the more complex technique for order preference by similarity to ideal solution (Triantaphyllou 2000).

The Analytical Hierarchy Process (AHP) is one multi-criteria decision-making tool that provides a comprehensive and rational framework for structuring complex decision-making problems and comparing the different alternatives. The method is widely used in academia and industry to compare fossil fuel sources with renewable energy sources, such as best electricity generation options in Jordan (Akash, Mamlook & Mohsen 1999), and more recently different CSP technologies, for example for CSP plants in India (Nixon, Dey & Davies 2010). To compare the different CSP technologies currently available in regards to their suitability for hybridisation the AHP was used to integrate the relevant quantitative and qualitative information.

The AHP decomposes a complex problem into several sub-problems, such as risk, economics, feasibility, and environmental impact, and in doing so it provides an easy-tounderstand path for multi-dimensional decision-making (Triantaphyllou 2000). The decomposition of the problem is done by identifying the main criteria and sub-criteria relevant to the problem and ranking them in a hierarchy. To compare different alternatives the AHP can use precise technical, economic and other data (quantitative information) as well as the decision-makers' personal judgments (qualitative information). In this research quantitative data were merged with the workshop participants' 1–9 criteria ratings (see Section 3.2.1) to derive a total score for each CSP technology option. The publication reproduced in Section 4.3 provides further details on the technology ranking process.

3.2.4 Case studies

Experiments and case studies are suitable tools for analysing an assumption within a research project (Yin 2009). Experiments or prototypes are not suitable for this research work as the testing of a CSP–EfB and CSP–EfW hybrid demonstration plant would require a multimillion-dollar investment and is in fact not necessary, due to advanced modelling software being available (see Section 3.2.2.1). Case studies, on the other hand, are very suitable for this research project to "gain a better understanding of the whole by focusing on a key part" (Gerring 2007, p. 29). One key benefit of the method is the "depths of the analysis it offers. One may think of depth as referring to the detail, richness, completeness, wholeness, or the degree of variance in the outcome that is accounted for by an explanation. [...] Case studies are thus rightly identified "holistic" analysis makes the case study method very suitable for this research as relevant aspects from various fields can be considered, e.g. technical, economic, environmental and socio-economic. Case studies are often used in various research fields, including business, political science, the humanities, and psychology, and represent a well-accepted research method (Gerring 2007).

However, a few concerns have been raised in regards to case studies in academic research, particularly in regard to the assertions that they provide little basis for scientific investigation, that they potentially lack or rigour, and that they require a significant amount of time to provide relevant details (Yin 2009). To avoid these issues the case studies in this research were designed to analyse technical, economic, environmental, socio-economic

and architectural considerations. This broad analysis provided a strong basis for investigation. The risk of a lack of academic rigour was addressed by subjecting the research results to review prior to publication or prior to presentation at well regarded conferences in the CSP field (see Section 4.6). Also the feedback from various conference participants helped to bring academic rigour to the case studies. Time management is essential in every project and investigating various research aspects in a case study is indeed time consuming. Due to the author's previous project management experience, the use of commercial modelling software, and good site information, time requirements were manageable. The first case study took longer than the second as it involved the case study design, which was in large part transferable to the second case study (e.g. both case studies involved thermodynamic and economic modelling as well as greenhouse gas abatement calculations).

In this research project two case studies, shown Figure 29, were investigated. The first is a 35.5 MWe CSP-multiple feedstock hybrid plant with biomass and RDF feedstocks at the Swanbank landfill in the Ipswich local government area in Queensland (Section 4.6.1). The second is a 30 MWe CSP–EfB hybrid plant using straw at Griffith in New South Wales (Section 4.6.2). The aim of both case studies was to demonstrate that rather than being merely technically possible, CSP–EfB and CSP–EfW hybrid plants are viable business propositions for Australia and can provide sustainable energy. To achieve this technical, economic, environmental, socio-economic characteristics of the plants were analysed and comparisons provided to CSP-only plants.



Figure 29: Case study locations in Ipswich, Queensland, and Griffith New South Wales

3.3 Theoretical framework

To analyse the integration of CSP through hybridisation into the socio-technical regime of the Australian electricity market, transition management is used as it can support the description and analysis of the various aspects of the transition process. An introduction to this concept is provided in Section 2.6. The theoretical framework for this research includes the overarching transition management framework along with two key models for analysing and managing technological transitions: the multi-level perspective (MLP) and strategic niche management (SNM). Potential transition pathways based on the MLP (Geels & Schot 2007) are also identified.

In the context of this research the four components of the transition management cycle (Loorbach 2010) are considered as follows:

- Problem structuration, long-term sustainability vision and transition arena: The problem of low CSP uptake is addressed through the analysis of the various benefits CSP hybrids provide with a long-term vision in which CSP supplies a significant share of Australia's electricity by 2050. The transition arena consists of some innovative actors in the Australian electricity market who see CSP as an important technology for Australia, such as technology suppliers, utilities, researchers and the government.
- 2. <u>Development of future images, a transition agenda and transition pathway:</u> The future image in this research is one of a low carbon electricity market and due to CSP's unique technological characteristics, such as mature energy storage, and the excellent solar resource available in Australia, it is a valuable part of the transition agenda. The transition path considered is the reconfiguration pathway.
- 3. Establish and carry out transition experiments and mobilise the resulting transition <u>networks</u>: First transition experiments already exist with CSP retrofits to coal fired power plants. In most cases such symbiotic innovations will be implemented first but their potential is limited and CSP–EfB and CSP–EfW hybrid plants are among the next lower-cost niche innovations. Transition networks have already been formed with various actors having a vision of the future electricity market and the technologies needed to reach it.
- 4. <u>Monitor, evaluate, and learn lessons from the experiments and make adjustments</u> <u>in the vision, agenda, and coalitions:</u> Because there are few CSP references in Australia, monitoring, evaluation and lessons are limited and international projects are important for identifying trends. However, the local expertise is very valuable to identify country-specific factors, such as cost differences and time taken for

approval processes. With first visions built predominantly on information from international projects, particularly in regards to plant cost, local information is essential for adjusting the vision, agenda and coalitions where necessary.

Currently, CSP can be considered a niche technology that has started to enter sociotechnical regimes elsewhere in the world, such as Spain and the US. While these projects provide valuable learnings for current and future projects, in accordance with Figure 26 on page 53, the technology still requires protection from mainstream market selection, such as financial support for R&D and project implementation, given that the full externality costs of fossil fuel electricity generators are not yet considered.

In the light of transitioning Australia to a low carbon future and CSP being a niche technology the MLP with an SNM approach was chosen for this research. According to SNM *"sustainable innovation journeys can be facilitated by creating technological niches, i.e. protected spaces that allow the experimentation with the co-evolution of technology, user practices, and regulatory structures"* (Schot & Geels 2008, p. 537). Additionally, SNM aims to develop particular types of innovations which are *"(1) socially desirable innovations serving long-term goals such as sustainability, (2) radical novelties that face a mismatch with regard to existing infrastructure, user practices, regulations, etc."* (Schot & Geels 2008, p. 539). While protected spaces, which phase out over time are required to allow innovations to evolve, some controlled selection pressure is also required to allow the emergence of those innovations with the highest probability of successfully entering the regime (Schot & Geels 2008). Selection pressures include technological superiority, market preferences and lobbying associations.

Originally, the MLP was used to analyse historic transitions, such as the change from sailing ships to steam ships (Geels 2002), but has also been used more recently to design future transition pathways, such as pathways to a low carbon electricity system in the UK (Foxon, Hammond & Pearson 2010). Its three-level structure is very well suited to mapping relevant aspects of the current Australian electricity market and to analysing how CSP hybrids can be part of the transition process. SNM practices for the implementation of renewable energy systems can be observed by various governments, including Australia's, by providing financial incentives to ensure their deployment. Incentives include research grants and additional revenue streams. These practices protect niche technologies in their initial stages against mainstream selection pressures and some government programs (e.g. Germany's renewable energy law) have even incorporated decreasing revenue streams over time to allow for future cost reductions (The Federal Ministry for the Environment Nature Conservation and Nuclear Safety 2012). Consistent with Australia's current renewable

energy policies such as the RET, and other incentives such as the current carbon pricing mechanism, SNM is a suitable framework for considering the implementation of CSP systems through hybridisation. This will be analysed in more detail in Section 4.7 based on five steps from Kemp et al. (1998) that enable governments to influence the transition process. These are: *"the choice of technology, the selection of an experiment, the set-up of the experiment, scaling up the experiment and the breakdown of protection by means of policy"* (Kemp, Schot & Hoogma 1998, p. 186).

Early MLP and particularly SNM work was critiqued for a focus on a bottom-up process where niche expansions lead to regime shifts, as shown in Figure 25 on page 51. However, recent work (Geels 2011; Schot & Geels 2008) addresses this and other critiques by acknowledging that in addition to the importance of niche innovations "*they can only diffuse more widely if they link up with ongoing processes at regime and landscape levels*" (Schot & Geels 2008, p. 539). Hence it is acknowledged that both landscape and regime changes and pressures are essential to successfully transform a socio-technical regime. The extended MLP (Geels & Schot 2007) shown in Figure 30 includes the different regime actors, indicates the interactions between the three levels in more detail, and shows that with time a dominant niche innovation evolves due to the selection processes. The relevant Australian landscape, together with regime and niche characteristics and pressures for CSP– EfB and CSP–EfW hybrid plants, are discussed in this context in Section 4.7.

According to Geels and Schot (2007), if they are to lead to a transition CSP–EfB and CSP– EfW hybrid plants have to meet two criteria. These criteria are related to: 1) the nature of the interactions between landscape, regime and niche; and 2) the timing of the innovation's development. The nature of the interaction refers to niche innovations and landscape developments reinforcing the regime, which stabilises it and discentivises transitions, or to niche innovations and landscape developments disrupting the regime, which applies pressure and incentivises transition. The timing refers to the ability of an innovation to take advantage of landscape and regime pressures. This ability is dependent of the degree of maturity of an innovation. Technological maturity is debatable and researchers, technology suppliers, plant operators and financiers will all have different perspectives. Geels & Schot (2007) introduced four criteria which they argue need to be met before an innovation can be considered to be mature: "(a) learning processes have stabilised in a dominant design, (b) powerful actors have joined the support network, (c) price/ performance improvements have occurred and there are strong expectations of further improvements and (d) the innovation is used in market niches, which cumulatively amount to more than 5% market share" (Geels & Schot (2007), p. 405). CSP–EfB and CSP– EfW hybrids are analysed based on these criteria in Section 4.7.



Increasing structuration of activities in local practices

Figure 30: Multi-level perspective on transitions (Geels & Schot 2007, p. 401)

However, to analyse the prospective CSP implementation pathway in more detail the reconfiguration pathway is chosen from the various transition pathways mentioned in Section 2.6. Figure 31 shows this pathway, as proposed by Geels & Schott (2007). This pathway is based on moderate landscape pressure on the regime leading to the implementation of some sufficiently developed symbiotic innovations to solve local problems. The regime remains mostly unchanged, as in the transformation pathway, however over time regime actors explore further combinations, implement more innovations and in doing so they adjust the basic architecture of the regime.

This pathway is the most suitable for this research as through hybridisation CSP can provide symbiotic innovations to the current Australian electricity regime, such as CSP retrofits to existing power stations. With regime actors such as utilities and financiers exploring further CSP options, new CSP–EfB and CSP–EfW hybrid plants and also CSP-natural gas hybrid plants could enter the regime and over time adjust its basic architecture to allow the broad uptake of CSP technologies in the future.



Figure 31: Reconfiguration pathway (Geels & Schot 2007)

Transition management also uses additional research theories and methods, such as systems thinking, action research and case study method (Loorbach 2007; Rotmans & Loorbach 2008) and requires qualitative input. According to Rotmans, Kemp & van Asselt (2001) qualitative measures are essential to achieve real transition. Stakeholder engagement, as emphasised by transition management, can also be found in action research where "the relationship between researcher and researched (the other-the object) is seen as an interactive and linguistic relationship, characterised by joint action, joint involvement and shared responsibility" (Ottosson 2003, p. 90). Action research tools have been used in transition management (Loorbach 2007).

To include the various aspects of CSP–EfB and CSP–EfW hybrid plants, this research project includes quantitative technical, economic and environmental data as well as qualitative input from various stakeholders. A form of action research can be found in this research since strong stakeholder engagement was an important feature throughout various parts of project. These stakeholders included workshop participants, CSIRO Ecosystems Sciences, CSIRO Energy Technology and Thiess Services Pty. Ltd.

3.4 Research ethics

To obtain feedback on CSP technologies and barriers from active stakeholders in Australia's energy market, this research project engaged with them through workshops and interviews. At the Univerity of Technology, Sydney this engagement required ethics approval from the Research Ethics Committee to:

- minimise risk and harm for research participants and researchers
- protect the confidentiality and privacy of the participants
- ensure that pre-existing relationships do not affect the research
- guarantee that data collection, analysis, storage, and disposal are in accordance with UTS guidelines.

These issues were addressed in a comprehensive application for ethics approval which was lodged on 17 May and granted on 5 July 2011. The approval letter is provided in the appendix. Measures to mitigate the aforementioned risks included consent forms from the participants, no publication of individual but only aggregated results, only references to stakeholder groups and not individuals, secure data storage and destruction of hard copies after five years.

4 RESULTS AND DISCUSSION

This chapter provides the analyses for: the CSP hybrid categorisation, further CSP hybrid configurations, CSP technology selection, CSP–EfB and CSP–EfW resources, technoeconomic optimisation, implementation barriers, the case studies, and the role CSP hybrids can play in Australia's transition to a low carbon future. It is structured differently to a conventional thesis as most of the individual results and discussions have already been published and these publications are incorporated in the various Sections of this Chapter. Therefore the Sections also include a description of the methods that is more detailed than the outline provided in Section 3.2 and the answers to most research questions are provided in the publications.

4.1 CSP hybrid categories and energy source combinations

The following publication introduces a categorisation of CSP hybrid plants based on the degree of interconnection of the plant components. It also answers the research question about other CSP renewable energy hybrid options and addresses the hybridisation of CSP with natural gas and coal. The publication aims to demonstrate that in addition to CSP–EfB and CSP–EfW hybrid plants, other hybrid configurations are possible in areas of Australia where multiple energy sources are available in the same location.

It should be highlighted that the discussion about the CSP boost at the Liddell coal fired power station in Section 4.1 in the paper refers to the recently commissioned 9.3 MWth solar field (Novatec Solar GmbH 2012b) and not the initial demonstration plant (Mills, Lièvre & Morrison 2003). The initial demonstration plant was built in 2004 and in fact very valuable as it demonstrated the Fresnel technology for the first time at an industrial scale and provided important testing for further design improvements. Since then significantly larger CSP plants, incorporating the learnings from the demonstration project, have been built and are under construction by successors of the original company Solar Heat and Power Pty. Ltd., including the 8 MWth CSP boost to the Liddell power plant in Australia (AREVA Solar 2012b), the 5 MWe Kimberlina plant in the USA (NREL 2014b), the 44 MWe equivalent Kogan Creek Solar Boost in Australia (CS Energy 2011), and the 2x 125 MWe Reliance power plant in India (AREVA Solar 2012a). The demonstration plant was therefore an important milestone in CSP technology development and deployment.

The discussion a CSP-natural gas hybrid plants in Section 4.2 of the publication does not consider "greenwashing" issues as the CSP-coal section does. However, CSP systems in new and existing natural gas fired plants should also have a significant CSP share and need to be evaluated based on their net environmental benefits or possible technology advancements.

Renewable Energy 67 (2014) 178-185

Contents lists available at ScienceDirect

Renewable Energy

journal homepage: www.elsevier.com/locate/renene

Concentrating solar power hybrid plants – Enabling cost effective synergies



^a Institute for Sustainable Futures, University of Technology Sydney, Level 11, UTS Building 10, 235 Jones Street, Ultimo NSW 2007, Australia ^b Laing O'Rourke Australia, Australia

^c ERK Eckrohrkessel GmbH, Germany

ARTICLE INFO

Article history: Received 15 October 2013 Accepted 14 November 2013 Available online 8 December 2013

Keywords: Concentrating solar power Solar hybrid plants Technology comparison Geothermal Wind

ABSTRACT

This paper categorises different concentrating solar power (CSP) hybrid options into light, medium and strong hybrids and discusses the combination of CSP with coal, natural gas, biomass and waste materials, geothermal, and wind. The degree of hybridisation depends on the interconnection of the plant components. Light hybrids create only limited synergies, such as the joint use of a substation, and their cost reduction potential is therefore limited, while strong hybrids share major plant components, such as steam turbine and condenser, and can better match their energy output with electricity pricing.

The hybridisation options for CSP with different energy sources are plentiful ranging from feedwater heating, reheat steam, live steam to steam superheating with some options better suited for a specific energy source combination than others. The synergies created in hybrid plants can lead to cost reductions of 50%, better energy dispatchability as well as revenue maximisation.

Several CSP hybrid studies exist for coal, natural gas and biomass but these are often investigating a specific hybrid concept. This paper considers several options at a higher level and also includes geothermal and wind which is novel.

While the paper focuses on Australia the approach taken and concepts discussed are transferable to other countries.

© 2013 Elsevier Ltd. All rights reserved.

1. Introduction

Despite an excellent solar resource and some state and federal programs for concentrating solar power (CSP) there is no commercial standalone CSP plant in operation in Australia yet. Some proposals secured significant state/federal funding, such as AU\$464m for 250 MWe SolarDawn and AU\$60m for 40 MWe SolarOasis projects, but neither was able to secure the remaining funding and had its support subsequently withdrawn [1,2]. Policy uncertainty and CSP's comparatively high investment compared to other renewable energy sources, such as wind and biomass, in a traditionally low wholesale cost electricity market are key reasons for the poor success rate.

Different to standalone CSP plants hybrid plants are being built in Australia with one plant already operational at Liddell, New South Wales [3], and another one under construction at Kogan Creek, Queensland [4]. Several studies investigate hybrids with gas [5,6] and even biomass [7] but none have yet been built.

CSP hybrids are well established worldwide, predominantly with natural gas [8–10] but also biomass [11], and provide lower cost benefits through the joint use of plant equipment, such as steam turbine, condenser and feedwater equipment, and better energy dispatchability as the host plant can provide electricity during times when CSP is not operating. Both aspects could help the Australian CSP industry to start grow their market share, ramping up manufacturing capabilities as well as gaining relevant project implementation experience. The wind and PV industries realised learning curve cost reductions of 15-20% when doubling cumulative deployment [12] and the CSP business could benefit similarly. At the same time plant operators and financiers would become more familiar with the variety of CSP technologies and their specific benefits. Another benefit of hybrid projects is the smaller investment, which reduces financial risk, and would particularly help newly developed CSP technologies to prove their capabilities. Due to a minimum CSP plant efficiency scale of 10-100 MWe for standalone systems such projects incur a higher financial risk over kWe scale PV systems. However, CSP add-on references exist at sizes of only 9 MWth [3].

Typically, standalone CSP plants require a direct normal irradiance (DNI) levels >20 MJ/m²/day but the lower specific investment





CrossMark

^{*} Corresponding author. Tel.: +61 2 9514 4878.

E-mail address: juergenheinzmartin.peterseim@uts.edu.au (J.H. Peterseim).

^{0960-1481/\$ -} see front matter © 2013 Elsevier Ltd. All rights reserved. http://dx.doi.org/10.1016/j.renene.2013.11.037

of hybrid systems allows the implementation in lower DNI areas which are often closer to load centres. The first CSP-biomass plant in Spain is located in 18 MJ/m²/day DNI area [13] and other studies consider regions with even lower DNI levels of 17 MJ/m²/day [14].

Natural gas, biomass and waste materials as well as geothermal are obvious energy sources to hybridise with CSP as such plants can share Rankine cycle equipment but also the integration of wind has benefits in terms of joint use of plant infrastructure and energy dispatchability.

While CSP hybrids are being discussed broadly in academia and industry with several studies on different integration options for CSP with a specific fuel, this paper categorises hybrid plants and discusses not only one energy source but several.

2. Methods

The technical and economic information provided in this paper derive from the literature and own modelling work using Thermoflex version 23.0. The software is widely used in academia and industry to model actual gas, biomass, waste, coal, geothermal and CSP plants.

The maps provided are based on publically available information which is combined using a commercial imaging software. References to the different energy sources, DNI, and transmission infrastructure are provided in each relevant section. Sites with a DNI $> 18 \text{ MJ/m}^2/\text{day}$ are considered suitable for CSP hybrids as commercial plants already operate in such DNI areas [13].

3. Hybrid categories

With a variety of possible CSP hybridisation options it is sensible to categorize these to better assess the degree of dependence between CSP and the other power generation component. The following three categories are suggested to do this.

3.1. Light hybrid synergies

Light CSP hybrids only share minimal plant infrastructure and the operation of the different assets does not depend on each other, such as CSP and a wind farm jointly using the switchyard and substation. This results in only minimal cost savings unless both plants can share the cost for building new transmission infrastructure, which would be a major benefit as transmission lines are capital intensive.

Despite the minimal interaction of both generation assets one plant could use certain equipment from the other to enhance its energy dispatchability/limit curtailment if this equipment is not being used at its maximum capacity at this point of time. One example would be a wind farm topping up the thermal storage of a CSP plant in winter to store some of its electricity via electric heaters in molten salt for dispatch at higher electricity prices at a different time of the day.

3.2. Medium hybrid synergies

Medium CSP hybrids are physically connected with each other and share major equipment, such as the steam turbine or condenser. However, the CSP component requires the operation of the typically much larger host plant while the host plant can operate without the CSP component. Currently, this is the dominant form of CSP hybrids where the CSP plant provides feedwater heating [3], cold reheat steam [4], or a saturated/superheated steam boost [8,9,15]. These concepts exist for CSP with coal and natural gas.

Due to the joint use of capital intensive plant infrastructure cost reductions can be significant but the solar share in these concepts is typically below 10% of the installed plant capacity.

3.3. Strong hybrid synergies

Strong CSP hybrid plants are physically connected with each other, share major equipment and have a significantly higher solar share, >30%, compared to medium CSP hybrids. CSP plants using another fuel to further superheat its steam fall in this category, such as 100 MWe SHAMS One using natural gas [16] and 22.5 MWe Termosolar Borges using biomass [13]. Alternatively the CSP component can be used to raise the steam parameters of low temperature renewable energy sources, such as geothermal plants [17]. The external superheating concept increases plant efficiency and hence lowers cost. Strong hybrids also include plants where CSP and the other energy source provide similar quantities of high temperature/pressure steam parameters to the joint turbine, e.g. 30 MWe CSP hybrid study for Griffith, Australia [18].

Strong hybrids can reduce specific plant investment significantly while enabling solar shares of 30–90%.

4. Hybridisation options

Several hybridisation options exist to integrate CSP into the Rankine cycle of another power plant. They include feedwater heating, reheat steam and high pressure/temperature live steam, see Fig. 1, and the most suitable CSP technologies for these options with biomass, waste, natural gas and coal have been assessed recently [19]. All three options have been implemented in reference plants worldwide, e.g. Liddell [3] and Kogan Creek in Australia [4], Martin Next Generation in the USA [8] and Archimede in Italy [15].

Using CSP to externally superheat low temperature steam is another option which has not been realised yet, but the concept is discussed in more detail in Section 4.4.

4.1. CSP-coal

Several studies investigate the different CSP integration options with utility scale coal fired power stations [20–22] and from a technical and commercial perspective they are all sensible with positive results in terms of CSP cost reductions and conversion efficiencies, e.g. specific investment of AU\$2.4/MWe for Kogan Creek solar boost [4]. However, when proposing CSP retrofits to coal fired power plants other criteria, such as age and remaining lifetime of the power station, have to be considered carefully to assess the real greenhouse gas abatement potential and avoid "greenwashing". With coal fired power stations having a design lifetime of circa 40 years and CSP plants requiring up to 20 years to amortise, coal plants older than 10-15 years should not be retrofitted as extending its operation only for a short period of time can offset the avoided greenhouse gas emissions from the CSP add-on due to the typically small solar contribution, well below 10% of the plant capacity.

The recently commissioned 9.3 MWth CSP feedwater boost at the 2 GW Liddell power station is an interesting example as it reduces greenhouse gas emissions by approximately 5000 t/a [3], which is equivalent to only 2.8 h of full-load operation of the coal station with its carbon intensity of 892 kg/MWh or 0.064% of the stations 2011–12 annual emissions from 8764 GWh of electricity generated [23]. There is no doubt modern power stations exist, typically supercritical, where a solar boost makes economic and environmental sense, such as 44 MWe solar boost to 750 MWe Kogan Creek power station [4], as these latest generation plants, Kogan Creek was commissioned in 2007, are most likely to operate for the next 25–30 years. However, other projects might add a small renewable energy component to an older unit to justify operation beyond the initial design lifetime. The recent CSP feedwater booster at the Liddell power station could be such a case as



Fig. 1. Simplified schematic of CSP integration options into host plants. (1) Feedwater heating; (2) high pressure steam; and (3) cold reheat line [19].

the coal units were commissioned in 1971 [23] and the CSP add-on is very small.

4.2. CSP-natural gas

Currently, most CSP hybrids operate as integrated solar combined cycle (ISCC) plants in combination with combined cycle gas turbine (CCGT) plants and references exist in the US [8], Morocco [10] and Egypt [9]. These plants use the CSP component to provide additional saturated steam to the high pressure drum of the heat recovery steam generator where the high temperature gas turbine exhaust superheats the combined steam quantity. Feedwater heating would be another option but has not been realised yet.

The CSP component could also provide similar steam parameter as the HRSG to the high pressure steam turbine and the first plant is in Priolo Gargallo, Sicily [15]. This 5 MWe reference uses parabolic trough with molten salt to provide up to 538 °C steam to the 760 MWe CCGT's steam turbine. Different to ISCC's with parabolic troughs an ISCC solar tower concept has been investigated but no reference plants exist yet [6].

Almost all CSP plants have a natural gas fired start-/back-up heater but these heaters are not designed to provide full-plant capacity or continuous operation but rather for plant start-up, generation stabilisation and emergency operation. However, a full capacity gas fired boiler in parallel to the CSP component is a technical option but would result in a low natural gas conversion efficiency compared to a CCGT plant. It is therefore not a recommended option.

For new plants the combination of CSP and natural gas has the potential to reduce LCOE from US\$243/MWh to US\$175/MWh [24]. This 38% cost reduction is significant and could fast-track CSP deployment in low price wholesale electricity markets such as Australia. Retrofits of CSP to existing open or combined cycle gas turbine plants can realise even lower LCOE, e.g. US\$150/MWh [25]. Despite the use of a fossil fuel and depending on the ISCC concept the solar contribution can be >50% [24] while the use of natural gas

to further superheat the CSP steam, such as in the SHAMS One plant, has the potential to achieve solar contributions >80% [16].

The potential for ISCC plants in Australia is high as several natural gas fired power stations already operate/are proposed in suitable DNI areas, see Fig. 2. Particularly promising is the Pilbara region in Western Australia and the Mount Isa region in Queens-land where gas prices and direct normal irradiance (DNI) levels are higher than elsewhere. However, several mining sites outside these regions qualify as well. ISCC plants do not exist yet but are investigated at different stages, e.g. 30 MWe Collinsville project [5] and 228 MWe ISCC in Port Hedland [6].

4.3. CSP-biomass and waste materials

Some early efforts to combine CSP with biomass or waste materials were discussed briefly in the 1980s with dish systems [29]. However, for several reasons no plants were built and it took another 25 years before construction of the first commercial CSP– biomass hybrid plant commenced near Lleida, ca. 150 km west of Barcelona, Spain [30]. The 22.5 MWe Termosolar Borges plant came online in late 2012 [11], is located further north than any other CSP project in Spain and uses the mature parabolic trough technology with thermal oil [31].

Several other studies investigated the hybridisation of parabolic trough plants with biomass [14,32–34] but no other project has yet commenced construction. Alternatively, Fresnel has been investigated for hybridisation with biomass and waste materials [35–37]. The benefit of using Fresnel would be steam temperatures of up to 500 °C [38] and subsequent higher conversion efficiencies. However, no reference plants exist yet either. For CSP–biomass hybrid plants with steam parameters, >500 °C and >100 bar, solar towers are best suited with direct steam systems preferred for plants without thermal storage and molten salt systems for plants with thermal storage. Solar towers with volumetric air receivers have been investigated as well [39] but it is likely that securing finance for such a project is more complicated due to the limited reference situation.



Fig. 2. Overlay of DNI [26] with operating natural gas plants (left) and proposed natural gas plants (right) [27] and transmission infrastructure [28].

While most studies consider the generation of identical steam parameters from biomass and CSP some publications investigate the use of CSP for air and feedwater heating as well as external steam superheating from an energy from waste plant [40]. However, none of the concepts considers CSP steam temperatures >430 °C which limits the conversion efficiency.

A recent 30 MWe CSP—biomass hybrid study in Australia shows a specific investment of AU\$4.2m/MWe for the biomass and AU\$7mMWe for CSP component, a combined specific investment of AU\$5.6m/MWe. A standalone CSP plant with the same annual generation (26 MWe solar tower with 15 h thermal storage) would require AU\$11.2/MWe including the network connection [18]. This equals a significant 50% cost reduction and other CSP—biomass studies indicate similarly significant cost reductions through hybridisation [14]. This difference is likely to decrease with CSP prices expected to decrease in the future but will not reach the cost competitiveness of a CSP—biomass hybrid due the inherently higher labour requirements for the manufacturing and installation of CSP systems compared to a biomass power station.

In Australia several agriculture regions exist with sufficiently high DNI levels for CSP systems, see Fig. 3. Several potential areas exist in Queensland, New South Wales, Victoria, South and Western Australia that have a similar or better DNI than the Lleida region in Spain. Also CSP—waste hybrid plants could be considered as waste materials downstream a recycling process are low cost and cause other environmental problems, such as leachate or fugitive emissions in landfills. To obtain some economy-of-scale benefits a minimum of 5 MWe is recommended, which would require a population of \geq 4080 people assuming the 2011 Australian average electricity consumption of 10.7 MWh per capita [41]. Several such cities exist, such as Mildura in Victoria or Griffith in New South Wales.

4.4. CSP-geothermal

Currently, both standalone geothermal and CSP are considered higher cost renewable energy sources, compared to wind or biomass, with high capital requirements for components such as solar field and production/injection wells. The combination of both resources provides not only the potential to share plant equipment, such as steam turbine or condenser, but also significant efficiency and investment improvements.

The net plant efficiency of geothermal plants is 8-10% comparatively low due to the low steam enthalpy from Australian resources. Raising the cycle efficiency is desirable and would reduce the number of capital intensive production and injection wells, the most expensive part of a geothermal plant or increase plant capacity with the same brine flow. Several CSP integration options



Fig. 3. Overlay of DNI with suitable biomass land use areas (left) [42] and population distribution (right) [43] with transmission infrastructure.



Fig. 4. Process diagram of a geothermal plant with CSP post-flashing steam superheating.

exist ranging from feedwater heating, the generation of identical steam parameters and CSP steam superheating. To limit changes to current geothermal plant concepts and to maximize cycle efficiency CSP is best used to further superheat the "low" temperature geothermal steam, typically between 150 and 200 °C at 6-10 bar, to 380 °C with a parabolic trough and to 525 °C with a molten salt solar tower system. Fig. 4 explains the geothermal-solar tower hybrid concept with the production and injection wells (1-2), the brine-water/steam heat exchanger (8), flash tanks (7, 28), solar tower with thermal storage (10), CSP steam superheater (16), steam turbine (4–6), generator (G1), condenser (3), feedwater heating system (24), as well as brine (15), molten salt (18) and feedwater pumps (9, 12). Steam superheating requires only around 20% of the total energy input which minimises the CSP component and its cost. Significant thermal storage, such as 15 h, increases the solar share and reduces the use of natural gas for external superheating during the night/CSP system maintenance. Typically, natural gas and other fossil fuels are expensive in remote locations but an additional external superheater system is required for back-up purposes.

Based on the brine flow in Fig. 4 the net plant capacity of a geothermal only plant would be 6.3 MWe with a net cycle efficiency of 10.2% and a specific investment of AU\$20.5m/MWe. The external steam superheating with a parabolic trough system would result in 8.4 MWe, 12.5% net cycle efficiency, and AU\$22.1m/MWe while the solar tower option would yield 9.9 MWe, 14.1% net cycle efficiency, and AU\$16.8m/MWe. The integration of a parabolic trough system can provide technical benefits but the high thermal storage cost, due to low molten salt temperature differences of 380 °C in hot and 280 °C in cold tank, actually increases the specific investment. Smaller thermal storage capacities would lower the specific investment but increase the use of natural gas for steam superheating. The higher temperature difference of the thermal storage system in a solar tower (540 °Cin hot and 280 °C in cold tank) lowers storage cost and makes the geothermal-solar tower hybrid economically more viable and also reaches the best technical performance.

Complementary to CSP steam superheating the generation of additional steam through the CSP component could be considered to compensate the geothermal output reduction during the day due to higher ambient temperatures. An ambient temperature increase from 20 °C to 45 °C can, depending on condenser design, reduce plant output by around 20%.

Australia has high potential for CSP—geothermal hybrid plants with the best resources overlapping in Queensland, New South Wales and South Australia, see Fig. 5. Particularly, promising are Longreach in Queensland and Leigh Creek in South Australia as transmission infrastructure is already available.

4.5. CSP-wind

The combination of CSP with wind, currently one of the lowest cost renewable energy sources, has been investigated for Texas, US, and results show that the combination of CSP with wind farms provides some benefits in terms of load matching [45]. With wind generation typically being lower during the day than at night the addition of CSP provides further daytime and evening power to match the demand profile. Fig. 6 shows the generation profile for wind only and different CSP—wind configurations.

In addition to the co-location of CSP and wind both forms of generation could be more integrated by adding electric resistance heaters into the CSP plant's thermal storage system to store wind energy during low price and dispatch during high price periods. Electric heaters are almost 100% efficient and a mature technology with thousands of references in various industries. Typically, electricity prices are lower at night and some of the wind energy could be used to fully charge the thermal storage for dispatch during the morning electricity peak. During winter the CSP plants can rarely charge their thermal storage to 100%, due to less daily solar irradiance, which makes it more complicated to dispatch electricity for all of the attractive evening period unless the plant has a very large thermal storage system. With the electric heater option wind could fully charge the thermal storage and provide electricity during this attractive period.



Fig. 5. Overlay of geothermal resources >200 °C [44] in DNI areas >19.1 MJ/m²/day with transmission infrastructure.



Fig. 6. Diurnal mismatch of wind speed with utility electrical loading as height increases (left) and annual utility loading compared to different ratios of wind farm to CSP rated generation in 2004 [45].

However, assuming a 99% electric heater efficiency and 39% CSP plant cycle efficiency the electricity price difference between charging and discharging the thermal storage would have to be >260%. This is a significant price difference but such events occur frequently, for example dispatch interval prices of AU\$40–60/ MWh between 0 and 6 AM and AU\$110–180/MWh between 5 and 10 PM [46], and could improve the economic viability of CSP and wind generation. Additionally, areas with network congestion, such as South Australia [47], could reduce curtailment by temporarily storing wind generation.

To minimize renewable energy losses and investment the electric heater capacity should not exceed 5–10% of the CSP plant capacity. The aim should not be to use wind to fully charge but only top-up the CSP thermal storage to cover high electricity price periods better and increase financial viability.

Some wind farms and CSP plants are already co-located but not integrated, such as the Dólarwind farm close to the Andasol I–III CSP plants in Spain, and several sites in South and Western Australia have potential for CSP–wind hybrids, see Fig. 7. Particularly promising are locations north of Port Augusta (South Australia) and between Perth and Geraldton (Western Australia) as the DNI is high and transmission structure available.

5. Conclusions

Several CSP hybrid options with different energy sources are discussed in this paper and results show that depending on plant concept and energy source combination, the cost of hybrids can be up to 50% lower than for standalone CSP plants. Also CSP hybrids can lower the cost of other renewable energy sources, such as geothermal, by



Fig. 7. Overlay of wind resources >7 m/s [48] in suitable DNI areas $>19.1 \text{ M}/\text{m}^2/\text{day}$ with transmission infrastructure.

significantly increasing cycle efficiency. These benefits paired with the potential to realise small CSP installations, rather than several hundred million or even billion dollar projects, allow financiers and operators to understand the different CSP technologies better and finance larger hybrid and standalone systems in the future.

Categorising CSP hybrids into light, medium and strong synergy systems enables a clear understanding of the different options and degrees of interconnection. While energy sources sharing Rankine cycle components with CSP, such as natural gas and biomass, have an inherently higher potential for synergies, and therefore cost reductions, even the combination with wind provide benefits in terms of revenue optimisation by matching electricity output with market prices.

Considering that CSP is currently a high cost renewable energy source its hybridisation with other fuels, ideally renewables, can fast-track the implementation of CSP systems in Australia, allow the growth of a local industry, test different CSP technologies and therewith help transition Australia's electricity mix from coal domination to a mix of renewable energy sources.

References

- Edwards V. Axing of \$60m solar grant floors Whyalla. The Australian 11-Jun-2013. Sydney, Australia.
- [2] Kelly J. Darkness falls on \$1.2bn SolarDawn project. The Australian 13-Nov-2012:4. Sydney, Australia.
- [3] Novatec Solar GmbH. Novatec Solar's Australian fuel-saver commences operation [Online]. Available: http://transfield.com.au/mediareleases/ 20121024_Novatec%20Solar%20Australian%20Fuel%20Saver%20Commences% 20Operation.pdf; 2012 [accessed 15.10.2013].
- [4] CS Energy. Work starts on Kogan Creek Solar Boost Project [Online]. Available: http://kogansolarboost.com.au/news/work-starts-on-kogan-creek-solarboost-project/; 2011 [accessed 15.10.2013].
- [5] Australian Renewable Energy Agency. ARENA projects solar [Online]. Available: www.arena.gov.au/programs/projects/index.html; 2012 [accessed 15.10.2013].
- [6] Peterseim JH, White S, Tadros A, Hellwig U. Integrated solar combined cycle plants using solar power towers to optimise plant performance. In: Solar-PACES conference 2012.
- [7] Ipswich City Council. Ipswich supports pioneering eco-precinct concept [Online]. Available: http://www.ipswich.qld.gov.au/about_council/media/view_ release/?action=viewMediaRelease&id=3657; 2012 [accessed 15.10.2013].
- [8] Florida Power & Light Company. FPL's Martin next generation solar energy center [Online]. Available, www.fpl.com/environment/solar/pdf/Martin.pdf; 2010 [accessed 04.10.2013].

- [9] Brakmann G, Mohammad F, Dolejsi M, Wiemann M. "Construction of the ISCC Kuraymat. In: SolarPACES conference 2009.
- [10] The World Bank. Nurturing low carbon economy in Morocco [Online]. Available: http://go.worldbank.org/KIN4DEUC70; 2010 [accessed 15.10.2013].
- [11] Protermosolar. Abantia and Comsa Emte set up the first facility in the world that combines solar thermal and biomass energy [Online]. Available: http:// www.protermosolar.com/saladeprensa/?q=en/content/abantia-and-comsaemte-set-first-facility-world-combines-solar-thermal-and-biomass-energy; 2012 [accessed 15.10.2013].
- [12] IT Power. Realising the potential of concentrating solar power in Australia. Canberra, Australia: IT Power; 2012.
- [13] Morell D. CSP BORGES The World's first CSP plant hybridized with biomass. In: CSP today USA 2012 – sixth concentrated solar thermal power conference 2012.
- [14] Pérez Á, Torres N. Solar parabolic trough biomass hybrid plants: a cost efficient concept suitable for places in low irradiation conditions. In: Solar-PACES conference 2010.
- [15] Falchetta M, Mazzei D, Crescenzi T, Merlo L. Design of the Archimede 5MW molten salt parabolic trough solar plant. In: SolarPACES 2009 2009.
- [16] Goebel O, Luque F. Shams one 100 MW CSP plant in Abu Dhabi update on project status. In: SolarPACES conference 2012.
- [17] Zhou C, Doroodchi E, Moghtaderi B. An in-depth assessment of hybrid solargeothermal power generation. Energy Convers Manag Oct. 2013;74:88–101.
- [18] Peterseim JH, Tadros A, White S, Hellwig U, Landler J, Galang K. Solar towerbiomass hybrid plants – maximizing plant performance. In: SolarPACES conference 2013.
- [19] Peterseim JH, White S, Tadros A, Hellwig U. Concentrated solar power hybrid plants, which technologies are best suited for hybridisation? Renew Energy Sep. 2013;57:520–32.
- [20] Hu E, Yang Y, Nishimura A, Yilmaz F, Kouzani A. Solar thermal aided power generation. Appl Energy 2010;87(9):2881–5.
- [21] Siros F, Philibert C, Le Moullec Y, Tusseau M, Bonnelle D. The value of hybridizing CSP. In: SolarPACES conference 2012.
- [22] Pierce W, Gauché P, von Backström T, Brent AC, Tadros A. A comparison of solar aided power generation (SAPG) and stand-alone concentrating solar power (CSP): a South African case study. Appl Therm Eng Nov. 2013;61(2): 657–62.
- [23] Macquarie Generation. Annual report. Macquarie Generation; 2012.
- [24] Turchi CS, Erbes M. Gas turbine/solar parabolic trough hybrid designs. In: ASME TurboExpo 2011 2011.
- [25] Peterseim JH, Tadros A, Hellwig U, White S. Integrated solar combined cycle plants using solar towers with thermal storage to increase plant performance. In: ASME Power 2013.
- [26] Spatial Information Services. Concentrating solar resource. Adelaide: Primary Industries and Resources South Australia; 2009.
- [27] Department of the Environment Water Heritage and the Arts. Fossil fuel power stations [Online]. Available: http://www.ga.gov.au/fossil_fuel/; 2011 [accessed 15.06.2013].
- [28] Australian Bureau of Agricultural and Resource Economics. Energy in Australia 2011. Canberra, Australia: Australian Bureau of Agricultural and Resource Economics; 2011.
- [29] McDonald CF. A hybrid solar closed-cycle gas turbine combined heat and power plant concept to meet the continuous total energy needs of a small community. J Heat Recover Syst 1986;6(5):399–419.

- [30] Abantia. The installation of the solar field Termosolar Borges, the first plant to combine thermal and biomass, has started [Online]. Available: www.abantia. com/noticias-prensa/en_noticias/28/the-installation-of-the-solar-fieldtermosolar-borges-the-first-plant-to-combine-thermal-and-biomass-hasstarted; 2011 [accessed 07.10.2013].
- [31] Cot A, Amettler A, Vall-Llovera J, Aguilo J, Arque JM. Termosolar Borges: a thermosolar hybrid plant with biomass. In: Third international symposium on energy from biomass and waste 2010.
- [32] Schnatbaum L. Biomass utilization for Co firing in parabolic trough power plants. In: SolarPACES conference 2009.
- [33] Nixon JD, Dey PK, Davies PA. Which is the best solar thermal collection technology for electricity generation in north-west India? Evaluation of options using the analytical hierarchy process. Energy 2010;35(12):5230-40.
 [34] California Energy Commission. San Joaquin solar 1 & 2 power plant licensing
- case. Sacramento, California: California Energy Commission; 2008.
- [35] Peterseim JH, Tadros A, White S, Hellwig U, Klöstermann F. Concentrated solar power/energy from waste hybrid plants – creating synergies. In: SolarPACES conference 2012.
- [36] Rojas M, Hellwig U, Peterseim JH, Harding J. Combining solar thermal power systems with solid fuels in new plants. In: SolarPACES conference 2010.
- [37] Spliethoff H, Kaeding N, Murer M, Alonso-Herranz E, Gohlke O. Combining energy from waste and concentrated solar power: new solutions for sustainable energy generation. In: International solid waste association conference 2010 2010.
- [38] Fluri TP, Lude S, Lam J, Morin G, Paul C, Platzer WJ. Optimization of live steam properties for a linear Fresnel power plant. In: SolarPACES conference 2012.

- [39] Coelho B, Schwarzbözl P, Oliveira A, Mendes A. Biomass and central receiver system (CRS) hybridization: volumetric air CRS and integration of a biomass waste direct burning boiler on steam cycle. Sol Energy Oct. 2012;86(10): 2912–22.
- [40] Kaeding N. Hybridkraftwerk Solar/Abfall: Erstellung und Beurteilung von Konzepten. Munich, Germany: Technische Universitaet Muenchen; 2010.
- [41] The World Bank. Australia electric power consumption (kWh per capita) [Online]. Available: http://data.worldbank.org/country/australia; 2009 [accessed 21.10.2012].
- [42] Department of Agriculture Fisheries and Forestry. Land use of Australia, version 4, 2005/2006 [Online]. Available: http://adl.brs.gov.au/anrdl/ metadata_files/pa_luav4g9abl07811a00.xml; 2010 [accessed 15.10.2013].
- [43] Australian Bureau of Statistics. Year book Australia. Canberra: Australian Bureau of Statistics; 2008.
- [44] Geoscience Australia. Geothermal energy project. Canberra: Geoscience Australia; 2006.
- [45] Vick BD, Moss TA. Adding concentrated solar power plants to wind farms to achieve a good utility electrical load match. Sol Energy June 2013;92:298–312.
- [46] Australian Energy Market Operator. Aggregated price and demand data files [Online]. Available: http://www.aemo.com.au/Electricity/Data/Price-and-Demand/Aggregated-Price-and-Demand-Data-Files; 2013 [accessed 15.10.2013].
- [47] Australian Energy Market Operator. Wind integration in electricity grids work package 5: market simulation studies. AEMO; 2012.
- [48] Department of Primary Industries and Regions. Predicted wind speed at 80 metres above ground level, 1995–2005, Australia. Adelaide: Renewables SA; 2012.

4.2 CSP-non-conventional fuel potential and plant area identification in Australia

The following publication investigates the annual electricity generation potential and ideal areas for CSP–EfB and CSP–EfW hybrid plants in Australia. It answers three research questions:

- Where are the ideal regions to build CSP–EfB and CSP–EfW hybrid plants in Australia?
- What is the annual electricity potential of CSP–EfB and CSP–EfW hybrid plants in Australia?
- What is the greenhouse gas mitigation potential of CSP–EfB and CSP–EfW hybrid plants in Australia?

The geospatial modelling for this work was carried out in cooperation with Dr Deborah O'Connell and Dr Alexander Herr from CSIRO Ecosystems Sciences, and Sarah Miller from CSIRO Energy Technology. Based on the analyses requested by the author, the available geospatial information on biomass, waste and solar resources was combined by Dr Alexander Herr with the power plant cost and efficiencies, provided by the author. Using several jointly developed constraints, the collected information was used to produce maps showing the most promising regions for CSP–EfB and CSP–EfW hybrid plants in Australia. The author thanks Dr Deborah O'Connell, Sarah Miller and Dr Alexander Herr for their cooperation and support during this analysis.

In addition to the detailed CSP–EfB and CSP–EfW resource assessment in this publication, a higher level analysis to identify potential CSP–EfB hybrid plant areas worldwide is provided in the publication in Section 4.4.2.

Energy 68 (2014) 698-711

Contents lists available at ScienceDirect

Energy

journal homepage: www.elsevier.com/locate/energy

Concentrating solar power/alternative fuel hybrid plants: Annual electricity potential and ideal areas in Australia



Available and ne at

Juergen H. Peterseim^{a,*}, Alexander Herr^b, Sarah Miller^c, Stuart White^d, Deborah A. O'Connell^b

^a Institute for Sustainable Futures, University of Technology Sydney, Level 11, UTS Building 10, 235 Jones Street, Ultimo, NSW 2007, Australia
 ^b CSIRO Ecosystem Sciences, GPO Box 284, Crace, ACT 2601, Australia
 ^c CSIRO Energy Technology, Mayfield West, NSW 2304, Australia
 ^d Institute for Sustainable Futures, University of Technology Sydney, 235 Jones Street, Ultimo, NSW 2007, Australia

ARTICLE INFO

Article history: Received 28 October 2013 Received in revised form 13 January 2014 Accepted 18 February 2014 Available online 20 March 2014

Keywords: Bioenergy Concentrating solar power Hybrid plants Resource assessment CO₂ abatement potential Australia

ABSTRACT

Australia's extensive solar resource is underexploited especially in the CSP (concentrating solar power) arena because of the high investment and lack of stable investment incentives. CSP hybrid plants provide an option to improve returns from CSP installations because of lower specific investment. This paper investigates the generation potential and most prospective regions for 5–60 MWe CSP hybrids using forestry residues, bagasse, stubble, wood waste and refuse derived fuels in locations with a direct normal irradiance >18 MJ/m²/day. Different plant efficiencies are used to identify the overall electricity potential for single and multiple feedstocks systems. The EfB (energy from biomass) or EfW (energy from waste) components of the hybrid plants considered are assumed to allow base load operation with the CSP components providing additional capacity during the day.

The total CSP-EfB & EfW hybrid potential in Australia, within 50 km of existing transmission and distribution infrastructure, is 7000 MWe which would require an investment of AU\$ 39.5b to annually generate 33.5 TWh. This is equivalent to 12.8% of all electricity generated in 2008–2009 or 74% of Australia's 2020 renewable energy target. The CO₂ abatement potential of CSP-EfB & EfW hybrids is up to 27 Mt or 4.8% of all 2009–10 CO₂ emissions.

© 2014 Elsevier Ltd. All rights reserved.

1. Introduction

Australia has one of the best solar resources in the world, however the historically low electricity prices compared with most other OECD (Organisation for Economic Co-operation and Development) countries has constrained the broader implementation of CSP (concentrating solar power) and other renewable energy plants. There are currently no standalone CSP plants in Australia and the two significant projects that were offered state/federal funding, being AU\$ 464m for 250 MWe SolarDawn and AU\$ 60m for 40 MWe SolarOasis projects, had their offers withdrawn in 2012 and 2013 as neither project was able to secure the remaining funding [1,2]. While solid biomass and waste feedstocks are

* Corresponding author.

available for power generation [3], few plants operate at commercial scale, total installed capacity of 170 MWe [4], and only one new 36 MWe industrial scale facility in Mackay, Queensland [5], commenced operation in the last four years.

The comparatively high cost of CSP and continuous/significant cost reductions of PV (photovoltaic) systems has in recent times put pressure on CSP. The decision to switch the first 500 MWe phase Blythe (USA) parabolic trough project to PV is the most prominent example of the competition with PV so far and it is likely that the second 500 MWe phase will be PV too [6]. In order to remain competitive, the CSP industry has to further demonstrate the grid value and other benefits of CSP arising from energy dispatchability as well as reduce plant costs. High renewable energy scenarios modelling in Australia identified CSP as a key technology to provide grid stability [7].

The hybridisation of CSP with forestry residues, bagasse, stubble, wood waste and RDF (refuse derived fuels) is one promising option to realise these two objectives and is endorsed globally [8] and in Australia [9]. Such hybridisation, not only with biomass or



Abbreviations: CSP, concentrating solar power; PV, photovoltaic; EfB, energy from biomass; EfW, energy from waste; DNI, direct normal irradiance; AU\$, Australian dollar (AU\$/US\$ = 0.96).

E-mail address: JuergenHeinzMartin.Peterseim@uts.edu.au (J.H. Peterseim).

waste, can provide distributed renewable/low-emission dispatchable power, capacity factors of up to 91%, CSP uptake in DNI (direct normal irradiance) areas lower than the usual $>20 \text{ MJ/m}^2/$ day, and investment reductions up to 28% [10] through the joint use of equipment and avoidance of currently capital intensive thermal storage.

Currently, several CSP hybrid plants operate as solar add-ons to coal and gas plants worldwide. Compared to the deployment of standalone CSP plants around the world, Australia deployed only CSP hybrids. In particular, the 9.3 MWth CSP-coal at Liddell is in operation [11], the 44 MWe CSP-coal at Kogan Creek is under construction [12], and new hybrids under investigation are a 30 MWe CSP-natural gas hybrid at Collinsville [13] and 35.5 MWe CSP-biomass hybrid in Ipswich [14].

In contrast to fossil fuels, forestry residues, bagasse, stubble, and wood waste are renewable resources. Only the non-renewable RDF fraction is fossil derived but consists largely of non-recyclable materials that would otherwise go to landfill. Including avoided fugitive landfill gas emissions EfW and EfB systems can be a greenhouse gas negative form of power generation [15]. Late in 2012, the first commercial scale CSP-EfB hybrid plant, 22.5 MWe Termosolar Borges using parabolic trough technology, commenced operation near Barcelona, Spain [16], which is significantly higher latitude and therefore has a lower solar resource than all other CSP plants in Spain [17]. The Termosolar Borges plant is built in an agricultural area with a DNI of 18 MJ/m²/day and is using forest and agricultural residues [18].

Similar or equivalent regions, in terms of combined biomass production and DNI >18 MJ/m²/day, exist in Australia. Agriculture and forestry in Australia produce significant quantities of biomass and some of it could be used as feedstocks for electricity generation as well as biofuels [3,19,20]. Investigations show that due to Australia's high carbon intensity electricity mix significantly greater greenhouse gas mitigation can be realised by using biomass for electricity generation, 30 Mt CO₂ equivalent, rather than biofuels, 9 Mt CO₂ equivalent [3]. The high solar irradiation in many Australian agriculture/horticulture areas offers the unique possibility to use both energy sources in designated plants. Non-recyclable and renewable waste materials, such as RDF, can be used in CSP hybrid plants, would increase the overall feedstock potential, divert waste from landfill, and reduce fugitive landfill emissions.

Different studies used GIS (geospatial) modelling to identify suitable regions and sites for standalone CSP power plants [21– 26] and CSP retrofits to existing fossil fuel plants [27] in Australia but the objective of this paper is to identify regions in Australia which may be suitable to site CSP-biomass hybrid plants with different power generation capacities. It provides a broad scale assessment of Australia's CSP-EfB & EfW hybrid potential and forms a basis for project developers and researcher to investigate specific sites within the most prospective regions identified. No such work has been undertaken to date in Australia, and the information is a critical pre-commercial step required to underpin future commercial feasibility assessments of CSP plant locations.

2. Methods

This assessment applies thermal analysis and GIS modelling to identify the electricity potential and the most prospective regions for CSP-EfB & EfW hybrid plants. The modelling includes specific technical, environmental and economic constraints as well as performance differences in regards to power plant feedstock and capacity.

2.1. GIS modelling

Globally geospatial/GIS modelling is widely used in the energy sector to identify the best sites and resources for new renewable and fossil power plants, including wind [28–30], biomass [31,32], standalone CSP [21-26,33,34] as well as CSP retrofits to existing fossil fuel plants [27,35]. Various proven software packages are available and for this paper the public domain software R (www. r-project.org) provided the capacity for the spatial biomass analysis and ArcMap (www.esri.com) enabled the map production. The daily average DNI (direct normal irradiance) for Australia was derived from 1995 to 2011 gridded hourly solar exposure data from the Bureau of Meteorology Australia [36]. Road and rail infrastructure stem from topology 250K data [37], and population estimates are based on the 2006 census data [38]. Biomass production and potential availability is based on a recent Australian assessment of biomass for bioenergy [3] and stubble [39,40], while RDF data derived from several publications [41– 43] combined with population estimates. The CSP and biomass resource data/maps in Section 3 derive from the combination of this information

Transmission lines were identified with information from Geoscience Australia [44], Australian Energy Market Operator Australia [45], the Energy in Australia 2011 report [46] and Western Power [47]. The exact GIS locations of transmission lines in Australia are not publicly available and therefore had to be approximated. This assessment considers transmission lines \geq 66 kV as they can technically absorb the output of 5–60 MWe CSP hybrid plants. A 50 km buffer around transmission lines was included in the GIS model as this is a viable transport distance for biomass [3].

To identify plant capacities, annual generation, and investment shown in Section 3, local biomass quantities were combined with the power plant modelling results (Tables 1–4), considering environmental, technical and economic constraints (2.3). Multiplying the locally available feedstocks with conversion efficiencies, capacity factors and costs lead to final investment requirements, e.g.

- stubble availability in a particular area of 42,500 t/a * conversion efficiency of 0.94 MWhe/t (Table 4) = 40,000 MWhe/a biomass generation
- 40,000 MWhe/a/8000 h biomass capacity factor (constraint in 2.3) = 5 MWe biomass capacity
- biomass capacity equals CSP capacity = 10 MWe hybrid plant capacity
- 1577 h/a CSP capacity factor (constraint in 2.3) * 5 MWe CSP capacity = 7885 MWhe/a CSP generation
- 7885 MWhe/a CSP + 40,000 MWhe/a biomass = 47,885 MWhe/ a total generation, and
- 10 MWe hybrid capacity * AU\$ 7.2m/MWe (Table 1) = AU\$ 72m investment.

The Australia wide potential for CSP-biomass hybrids is identified by combining all areas that met the constraints. Additionally,

Table 1

Specific investment data for different CSP hybrid plants sizes in AU\$m/MWe for the $>\!21\!-\!24$ MJ/m²/day DNI category.

| Feedstock | 10 | 20 | 30 | 40 | 50 | 60 |
|----------------------------|-----|-----|-----|-----|-----|-----|
| | MWe | MWe | MWe | MWe | MWe | MWe |
| Forestry residues + CSP | 7.0 | 5.6 | 4.8 | 4.4 | 4.3 | 4.2 |
| Bagasse + CSP | 7.1 | 5.7 | 4.9 | 4.5 | 4.4 | 4.3 |
| Urban wood waste + CSP | 7.5 | 6.0 | 5.2 | 4.7 | 4.6 | 4.5 |
| Refuse Derived Fuels + CSP | 7.6 | 6.1 | 5.3 | 4.8 | 4.7 | 4.6 |
| Stubble + CSP | 7.2 | 5.8 | 5.0 | 4.4 | 4.4 | 4.3 |

Table 2 Calorific values, boiler efficiency, fuel conversion efficiency, and maximum recommended steam temperatures.

| Feedstock | Fuel CV (dry), MJ/kg | Boiler efficiency, % | Fuel conversion efficiency, MWh _{th} /t fuel | Max. Steam temperature, °C |
|----------------------|----------------------------|-------------------------|---|-------------------------------|
| Forestry residues | 19.0 | 90.7 | 4.79 | 540 |
| Bagasse | 17.7 | 90.5 | 4.45 | 540 |
| Urban wood waste | 17.7 | 89.0 | 4.38 | 450 |
| Refuse derived fuels | 18.6 | 85.9 | 4.44 | 430 |
| Stubble | 15.7 | 89.5 | 3.90 | 540 |

the CSP-EfB & EfW hybrid electricity potential is modelled for single and multiple-feedstock plants.

2.2. Power plant efficiency and cost modelling

Thermoflex version 23.0 (www.thermoflow.com) was used to model the different power plant efficiencies and investment. The software is used in academia and industry to model actual parabolic trough, Fresnel and solar tower plants.

The plant sizes considered range from 5 to 60 MWe to realise economy-of-scale benefits and limit feedstock transport distances and emissions. The power plant efficiencies were modelled individually for all feedstock and plant size categories (Table 3) considering different steam parameters, economy-of-scale efficiency changes and feedwater heating arrangements. Plants smaller than 20 MWe are modelled without steam reheating, while larger units include a single reheat. Process models for clean biomass feedstocks, such as forestry residues, only include a baghouse filter while contaminated materials, such as RDF, have additional flue gas scrubbing equipment. All models are based on 25 °C ambient temperature, the use of air cooled condensers, a solar multiple of 1.4 (solar field is oversized by this multiplier to achieve full-load at below design point conditions) and avoidance of flue gas condensation. Thermal storage is not modelled in detail as the EfB/EfW component ensures lower cost power generation, albeit some part-load efficiency losses, than thermal storage at night and during extended cloud coverage.

Both the CSP and the EfB/EfW steam generators are expected to provide superheated steam at 400–540 °C, depending on fuel quality, to individually meet 50% of the plants installed capacity (Figs. 1a and 2). This plant configuration is chosen to a) have a significant CSP share in the hybrid plant, b) allow CSP to capitalise on typically higher electricity prices during the day (Fig. 1b), c) provide additional generation capacity during higher daytime demand, and d) ensure efficient part-load operation at night. Without CSP, the plant's cycle efficiency with only biomass, at a part-load of 50%, would only be 10% lower than the hybrid at full load. The plant cycle efficiency reduction at part-load is not as great as first expected because at night only the steam turbine operates in partload, while the biomass boiler is still operating at full capacity and its maximum design efficiency. Also condenser performance

Table 3

Biomass hybrid plant net cycle efficiencies for varying plant capacities and feedstocks.

| Feedstock | 5 | 10 | 15 | 20 | 25 | 30 |
|----------------------------|-------|-------|-------|-------|-------|-------|
| | MWe | MWe | MWe | MWe | MWe | MWe |
| Forestry residues + CSP | 24.4% | 27.2% | 29.1% | 33.4% | 33.8% | 34.1% |
| Bagasse + CSP | 24.2% | 26.8% | 28.6% | 31.9% | 33.0% | 33.5% |
| Urban wood waste + CSP | 22.9% | 26.0% | 27.2% | 30.8% | 31.6% | 32.0% |
| Refuse Derived Fuels + CSP | 22.6% | 25.5% | 26.5% | 29.4% | 29.8% | 30.3% |
| Stubble + CSP | 24.1% | 26.8% | 28.7% | 32.9% | 33.4% | 33.6% |

increases at night. Operating the steam turbine <50% part-load would lead to significantly higher cycle efficiency reductions. This hybrid concept has already been investigated for two sites in Australia and is at this point in time commercially more attractive than a CSP standalone plant in these locations [48,49].

In comparison to other assessments on the bioelectricity potential, such as Farine et al. (2011) [3], this assessment uses different boiler and power plant efficiencies for the individual feedstock and plant capacities (Tables 2 and 3), resulting in varying conversion efficiencies for the individual fuels (Table 4). The feedstock strongly affects the maximum steam parameters recommended for the EfB/EfW plant (Table 2) as some substances, such as chlorine or potassium, cause high-temperature corrosion and ash fusion related problems on the superheater section of the boiler. Because of the feedstocks different composition it also determines the boiler's maximum efficiency to avoid dew-point corrosion on the boiler tubes.

Single and multiple-feedstock plants have been considered in this assessment. Single-feedstock plants have higher individual cycle efficiencies as the plant can be purpose designed for one fuel. Power plant cycle efficiency increases with capacity (Table 3), which benefits multiple-feedstock plants in any location. In general feedstock availability limits plant size, so multiple feedstock sources enable larger plant capacity. These multiple-feedstock plants capitalise on economy-of-scale benefits, even though they need to be designed for the lowest feedstock quality in the mix.

The Thermoflex cost database is used for the different power plant components to identify the specific plant investment (Table 1). With biomass boilers not being a standard Thermoflex component, industry prices were obtained for this equipment to optimise investment accuracy [50]. Prices consider economy-ofscale differences, different flue gas cleaning options, steam reheating >20 MWe and different DNI levels. Confirmation of the costing has been demonstrated by modelling the 36 MWe EfB project at Mackay Sugar in Queensland and the 44 MWe Solar Boost at Kogan Creek in Queensland. The investment for Mackay Sugar is AU\$ 120m [5] and Solar Boost AU\$ 104.7m [12]. Modelling both projects resulted in a 5% higher investment (4.7% Mackay Sugar and 5.3% Kogan Creek) which is acceptable as not all project data are known. Additionally, both plants are retrofit projects which typically have lower cost through the use of existing infrastructure.

The investment for CSP hybrid plants with an 18% capacity factor (percentage of actual power production of a plant compared to its power output at continuous full-load operation) depends on the quality of the DNI resource and this assessment uses the three different DNI categories shown in Fig. 3. The specific investment shown in Table 1 is the basis for the >21–24 MJ/m²/day category, while the 18–21 MJ/m²/day DNI category requires a 10.3% higher investment. The investment decreases by 4% in the >24 MJ/m²/day area. The reason for the investment differences is the fixed CSP capacity factor, which results in smaller solar fields in higher DNI areas and larger solar field in lower DNI areas. With the solar field

Table 4

Conversion rates (tonnes of feedstock to MWh electricity) for different feedstocks and power plant sizes.

| Feedstock | 5 | 10 | 15 | 20 | 25 | 30 |
|-----------------------------|------|------|------|------|------|------|
| | MWe | MWe | MWe | MWe | MWe | MWe |
| Forestry residues, MWh/t | 1.17 | 1.30 | 1.39 | 1.60 | 1.62 | 1.63 |
| Bagasse, MWh/t | 1.08 | 1.19 | 1.27 | 1.42 | 1.47 | 1.49 |
| Urban wood waste, MWh/t | 1.00 | 1.14 | 1.19 | 1.35 | 1.38 | 1.40 |
| Refuse derived fuels, MWh/t | 1.00 | 1.13 | 1.18 | 1.30 | 1.32 | 1.34 |
| Stubble, MWh/t | 0.94 | 1.05 | 1.12 | 1.28 | 1.30 | 1.31 |



Fig. 1. a) CSP hybrid concept for a 40 MWe plant and b) NSW average weekdays system demand and spot price year ending 2010 [52].

being the most capital intensive component of a CSP plant [51] its size directly affects overall plant investment.

2.3. Constraints

Studies have shown significant differences between the theoretical and actual potential for CSP sites [21–25] and biomass resources [3,20,40] in Australia. To ensure an accurate resource assessment and identification of potential regions for CSP-EfB & EfW hybrids a number of technical, environmental and economic constraints were applied in this study.

2.3.1. Environmental

To minimise the environmental impact of CSP hybrid plants the following environmental constraints apply:

- No energy recovery from recyclable waste materials, only RDF and urban wood waste feedstocks.
- At least 1 t/ha of agricultural residues (crop stubble) in southern cropping regions and 1.5 t/ha in northern cropping regions retained to protect soils from the risks of erosion [39].
- All foliage or branches from forestry residues to be retained on site to assist maintenance of organic matter, and nutrients.
 Forest residues from plantations only.



Fig. 2. Simplified process diagram for a CSP-biomass hybrid plant without steam reheating.

2.3.2. Technical

Various technical constraints in feedstock harvest, processing and conversion were applied considering constraints. They include:

- Capacity factor of 91.3% (8000 h/year) for EfB & EfW and 18% (1.577 h/year) CSP component.
- CSP plants with direct steam generation capabilities from 400 to 540 $^\circ\text{C}.$
- Sufficient EfB/EfW feedstock in a 50 km radius to continuously operate the biomass component of the 5–60 MWe hybrid power plant at full capacity.
- Plant to be located within a 50 km distance to existing transmission infrastructure.
- Facilities not considered in DNI areas <18 MJ/m²/day.
- Stubble cannot practically be cut lower than 12.5 cm [53].

2.3.3. Economic

High level economic constraints were applied in regards to feedstock supply, plant size, and electricity off-take. A detailed economic evaluation is not part of this study but scope for further work requiring detailed local information, such as labor and transport costs. The constraints include:

- Feedstock within approximately 50 km radius around the power lines to limit transport distance of feedstocks and access of CSP facility to electricity grid.
- At least a 5 MWe plant capacity to gain economy-of-scale benefits and maximum 60 MWe.

2.4. Technology selection

To start implementing CSP-EfB & EfW hybrid plants immediately, the use of mature technologies is essential. For this reason grate and fluidised bed combustion system are considered for all solid feedstocks and gasification with syngas firing in a boiler only for clean biomass.

CSP technologies considered to directly generate 400–540 °C steam are linear Fresnel and solar towers. Both technologies have >5 MWe plants in operation and under construction, e.g. 5 MWe Kimberlina Fresnel plant in the U.S [54], 2×125 MWe Reliance Fresnel project in India [54], 5 MWe Sun Sierra tower plant in the U.S [55], and 3×130 MWe Ivanpah tower plant in the U.S [56]. All the solid feedstocks considered can readily achieve steam



Fig. 3. Average daily direct normal irradiance across Australia showing >24–27 MJ/m²/day (high DNI category), >21–24 MJ/m²/day (medium DNI category) and >18–21 MJ/m²/day (low DNI category) for CSP-EfB & EfW plants.

temperatures >400 °C and reducing the EfB/EfW plant's steam temperature would unnecessarily decrease the feedstock conversion efficiency. Therefore, the most mature parabolic trough technology using thermal oil is not considered as the oil's thermal stability limits steam temperatures to 385 °C. Since linear Fresnel steam temperatures at commercial scale are currently limited to a maximum of 500 °C [57] the technology is best suited for CSP hybrids using urban wood waste and RDF [58]. Solar towers are considered the best CSP technology for hybrids using clean biomass feedstocks, such as woodchips or straw [58], with steam temperatures >500 °C.

3. Results

This section outlines the energy resources, electricity and greenhouse gas potential, most prospective regions as well as the economic impact of CSP-EfB & EfW hybrid plants in Australia.

3.1. Solar resource

The solar resource in Australia is among the best in the world [59] with the theoretical potential to provide all of Australia's energy needs. As almost all CSP plants worldwide operate in DNI areas \geq 20 MJ/m²/day, we have assumed that this is a typical requirement of standalone CSP plants. Based on 1995–2011 DNI data by the Bureau of Meteorology [36] a significant part of Australia meets this criterion (Fig. 3). However, the comparatively high capital costs of CSP, traditionally low wholesale electricity prices, the lack of stable investment incentives, and the need to build new transmission infrastructure to access the highest DNI areas has proved a barrier to the construction of plants.

Cost savings through the joint use of equipment allow CSP-EfB & EfW hybrid plants to be considered in areas with a DNI \geq 17 MJ/m²/day [60] and the Termosolar Borges commenced operation in a lower than usual DNI area recently, 18 MJ/m²/day [18]. Compared to standalone CSP plants new hybrids can realise investment reduction in excess of 20% [10,48,61] which offsets the 10.3% investment increase for a larger solar field in a 18–21 MJ/m²/day DNI area. Therefore this assessment considers three DNI categories for CSP-EfB & EfW hybrid plants (Fig. 3), >24–27 MJ/m²/day (high DNI category), >21–24 MJ/m²/day (medium DNI category), and >18–21 MJ/m²/day (low DNI category).

3.2. Biomass feedstock

Biomass production systems can have economic benefits through diversification of farming products and for areas of low agricultural production value. In many parts of the world, including Australia, biomass is contentious in the public domain, with the food versus fuel debate dominating the discussion. This assessment focuses on existing biomass resources and only considers existing crops or waste residues. It uses the approach taken by Farine et al. (2011) [3] which applies simple sustainability constraints to the proportion of biomass removed from current agricultural and forestry production systems. Feedstocks include bagasse, stubble, plantation residues and urban wood waste as in Farine et al. (2011). This assessment includes copper chrome arsenate treated timber as several plants operate successfully on this feedstock worldwide, such as the 20 MWe combined heat and power plant Berlin-Neukölln in Germany [62], and because this material poses environmental problems in landfills, e.g. leachate and methane formation.

The total yearly biomass feedstocks in suitable DNI areas amount to 21,334 kt and have calorific values from 15.7 to 19 MJ/kg (Fig. 4 and Table 5).

3.3. Waste feedstocks

RDF is produced from municipal, commercial and industrial waste streams by removing recyclable materials, such as plastics and metals, inert materials, such as glass and bricks, and organic materials, such as garden and food waste. RDF preparation systems and energy recovery technologies are well established [41], with many commercial plants worldwide. RDF has a renewable energy content of 40–60% [63], a high calorific value between 11.7 and 31.6 MJ/kg [64] and is more consistent than unsorted wastes, which has positive flow-on effects in the power plant, such as better combustion performance [65]. Demand for RDF as a sorted waste product also maintains the waste recycling priority and can divert waste from landfill.

The Australian waste quantities from MSW (municipal solid waste), commercial and industrial waste, and construction and demolition waste streams in 2006–07 amounted to 43,777 kt, of which 52% was recycled with the remaining 48% destined for landfill [66]. Post recycling, one tonne of waste per person was landfilled. Currently, global RDF quantities range from 23 to 55% of the waste stream input [41–43] and this assessment assumes an

Australian average RDF quantity of 0.4 t/person/year. This leads to a total yearly RDF feedstock potential of 2631 kt for suitable DNI areas (Fig. 5 and Table 5).

3.4. Electricity and greenhouse gas abatement potential

Currently, CSP-EfB & EfW hybrid plants within 50 km buffers around existing transmission infrastructure (Fig. 6), have a combined annual electricity potential of 30.8 TWh/year for singlefeedstock and 33.5 TWh/year for multiple-feedstock plants (Table 5). This is 20% higher than only using the biomass resources for power generation. The single-feedstock potential equals 11.8% of all electricity (including off-grid electricity) generated in Australia in 2008–09 [46] and the multiple-feedstock potential equals 12.8%. This also would cover 74% of the Australian Government's 2020 renewable energy target of 45 TWh [67]. When considering only DNI areas >21 MJ/m²/day the potential is with 21.6 TWh for singlefeedstock hybrids and 22.6 TWh for multiple-feedstock hybrids still significant equalling 8.3% and 8.7% of the total 2008–09 Australian electricity.

The high DNI category (>24–27 MJ/m²/day) has with 3 TWh the lowest annual electricity potential, which consists to 98% of CSP-stubble hybrid plants. The annual electricity potential in the medium DNI category (>21–24 MJ/m²/day) is 19.6 TWh, which is the highest of all areas and is also dominated by stubble. The wood



Fig. 4. Potential biomass availability in Australia overlayed with DNI; Stubble (top left), forestry residues (top right), urban wood waste (bottom left) and bagasse (bottom right).

Table 5

Annual feedstock, electricity and CO₂ abatement potential for CSP hybrid plants in Australia.

| | Low DNI category >18—21 MJ/m²/day | Medium DNI category >21–24 MJ/m²/day | High DNI category >24 MJ/m²/day | Total |
|--|--------------------------------------|---|------------------------------------|------------|
| Feedstock, dry t/a | | | | |
| Stubble | 1,800,000 | 11,407,000 | 2,328,000 | 15,535,000 |
| Forestry residues | 2,043,000 | 300,000 | _ | 2,343,000 |
| Wood waste | 2,071,000 | 1,359,000 | 26,000 | 3,456,000 |
| RDF | 1,391,000 | 1,215,000 | 25,000 | 2,631,000 |
| Multiple-feedstock | 8,608,000 | 15,502,000 | 2,447,000 | 26,557,000 |
| Capacity potential, MWe | | | | |
| Stubble + CSP | 450 | 3080 | 630 | 4160 |
| Forestry residues + CSP | 540 | 80 | _ | 620 |
| Wood waste + CSP | 560 | 390 | 6 | 956 |
| RDF + CSP | 370 | 320 | 5 | 695 |
| Total single-feedstock hybrids | 1920 | 3870 | 641 | 6431 |
| Total multiple-feedstock hybrids | 2280 | 4090 | 630 | 7000 |
| Generation potential, MWh/a | | | | |
| Stubble + CSP | 2,155,000 | 14,749,000 | 3,017,000 | 19,921,000 |
| Forestry residues + CSP | 2,586,000 | 384,000 | _ | 2,970,000 |
| Wood waste + CSP | 2,682,000 | 1,868,000 | 29,000 | 4,579,000 |
| RDF + CSP | 1,772,000 | 1,533,000 | 24,000 | 3,329,000 |
| Total single-feedstock hybrids | 9,194,000 | 18,532,000 | 3,070,000 | 30,796,000 |
| Total multiple-feedstock hybrids | 10,918,000 | 19,585,000 | 3,017,000 | 33,520,000 |
| Total investment, AU\$m | | | | |
| Single-feedstock hybrids | 11,600 | 20,500 | 3300 | 35,400 |
| Multiple-feedstock hybrids | 13,800 | 22,300 | 3400 | 39,500 |
| Total CO ₂ abatement potential, t/a | | | | |
| Single-feedstock hybrids | 7,110,000 | 15,048,000 | 2,574,000 | 24,732,000 |
| Multiple-feedstock hybrids | 8,560,000 | 15,933,000 | 2,529,000 | 27,022,000 |

waste and RDF potential is 3.4 TWh. The low DNI category (>18–21 MJ/m²/day) has a lower 10.9 TWh annual electricity potential as it covers neither prime agricultural nor forestry regions within proximity to transmission infrastructure. Here, woody biomass and RDF estimates are lower and stubble contribution is increasing with higher DNI (Fig. 7). Still, the first CSP-biomass hybrid plant in the world is located in a 18 MJ/m²/day DNI area [18], which shows that Australia has an excellent potential for CSP-biomass hybrids in this and higher DNI locations.

The electricity potential for CSP-EfB hybrids using multiple feedstocks is 8.8% higher than the potential for single-feedstock

hybrids. The reasons for this are that multiple-feedstock plants qualify for some locations where a single feedstock would not suffice to operate a minimum 5 MWe CSP-EfB & EfW hybrid plant.

Stubble has the highest potential for CSP hybridisation accounting for 64% of the total electricity potential followed by urban wood waste with 15%, RDF with 11% and forestry residues with 10%.

Bagasse is already in use in Australia for electricity and process heat generation in sugar industry centers, mostly in Queensland [4]. Bioenergy assessments identify an annual electricity potential from bagasse of 2.2 TWh [3], but the material grows predominantly close to the coast where DNI values are lower than further inland.



Fig. 5. Potential refuse derived fuel availability in Australia overlayed with DNI.



Fig. 6. 50 km distance areas around existing transmission lines, overlayed with different DNI categories.

Also the larger area requirement of a CSP-bagasse hybrid facility compared to a standalone bagasse electricity generator makes it more costly where it competes for higher value land. Identifying the few suitable sites would require more localised analysis and is scope for future work.

Considering the average 841 kg CO₂/MWh carbon intensity of Australian generation assets in 2010 [68], the exploitation of the CSP-EfB & EfW hybrid resources could, without consideration of biomass harvest and transport emissions, displace up to 27 million tones CO₂/year (Table 5). This equals 4.8% of all Australian 2009–10 CO₂ emissions [69].

3.5. CSP hybrid plant location potential

CSP-stubble hybrids could be located across large areas of Western Australia, New South Wales and Victoria while hybrids using urban wood waste and RDF are mostly associated with population centers, such as Mildura in Victoria or Perth in Western Australia (Fig. 8). The proximity of RDF and urban wood waste to urban centers makes them promising candidates for multiplefeedstock and cogeneration systems in terms of potential availability of biomass and proximity of demand. From a DNI perspective the most prospective regions for CSPstubble plants are between Griffith and Wagga Wagga in New South Wales, between Mullewa and Moora in Western Australia, and Cunderdin to Merredin in Western Australia with a total potential of >500 MWe each. The New South Wales region could accommodate several 5–55 MWe plants while the Western Australia regions could accommodate several 6–45 MWe. The Mullewa to Moora region has the highest DNI of all three regions. The two largest CSP-stubble hybrid plants with a capacity of 67 MWe and 72 MWe could be built in the north-eastern region of the York peninsula, South Australia.

The most prospective regions for CSP-forestry waste hybrids are around Gympie in Queensland with a total potential of 270 MWe (17–70 MWe plants) and Tumut in New South Wales with a total potential of 150 MWe (6–32 MWe plants). While the capacity potential is higher in Gympie, the DNI is better in the Tumut region. The pulp and paper mill in Tumut is currently investigating a 75 MWe expansion using waste feedstock [70] and a CSP component could further increase capacity or reduce the waste material quantity required.

The Mildura region in Victoria could provide the highest DNI for a CSP-urban wood waste hybrid with a capacity of 6 MWe.




Fig. 8. MWe potential for CSP hybrids using stubble (top left), forestry wood waste (top right), urban wood waste (centre left), RDF (centre right) and multiple-feedstocks (bottom) across Australia.

Significantly larger plants are possible around Perth (up to 150 MWe), Brisbane (up to 115 MWe), Adelaide (up to 80 MWe) and Canberra (up to 40 MWe). With wood waste and RDF often arising in the same regions Mildura has the potential for a 5 MWe CSP-RDF plant, Perth for up to 115 MWe, Brisbane for up to 120 MWe, Adelaide for up to 90 MWe and Canberra up to 45 MWe. Due to the

similar power plant design for wood waste and RDF, such as steam parameters and flue gas cleaning, both feedstocks should be used in multiple-feedstock plants to gain economy-of-scale benefits.

The most prospective regions for CSP multiple-feedstock hybrids in the higher DNI areas are identical to the areas of high stubble production but the generation capacity is slightly lower, circa 5% for Griffith to Wagga Wagga region, as even the mixture of stubble with small quantities of wood waste or RDF requires lower steam parameters to avoid high temperature corrosion issues. Hence, multiple-feedstock plants using feedstock with vastly different combustion properties that affect overall plant efficiency significantly are not recommended for regions with one dominant feedstock. For lower DNI areas CSP multiple-feedstock hybrids are more suitable as there is a variety of feedstocks available, mainly wood waste and RDF. The combination of these feedstocks with CSP can more than double the plant capacity for the potential areas around Perth (up to 270 MWe), Brisbane (up to 240 MWe), Adelaide (up to 175 MWe) and Canberra (up to 90 MWe).

Generally larger CSP power projects are economically more attractive than small ones as their specific investment is lower. In New South Wales, Western Australia, South Australia and Victoria several potential areas exist for 21–40 MWe CSP-stubble hybrid plants and even few 41–72 MWe installations (Fig. 8). CSP-forestry residue hybrids are mainly <25 MWe with only few regions in New South Wales and Queensland having potential for larger units. CSP-wood waste and RDF hybrids close to major urban centers could reach 150 MWe, while plants closer to rural townships would typically remain below 10 MWe. CSP-multiple-feedstock hybrids have with up to 270 MWe the potential for the largest plant sizes around Perth, Brisbane, Adelaide and Canberra but the identification of a specific site with sufficient size at an acceptable land price will be a challenge.

3.6. Economic analysis

The feedstocks used in a hybrid plant affect its investment, particularly in regards to boiler and flue gas cleaning design, and cycle efficiency (Table 1). To obtain a preliminary understanding of the most cost effective way to use the resources potentially available single and multiple-feedstock CSP-EfB & EfW hybrid plants have to be considered. While single-feedstock plants have higher efficiencies, typically less complex flue gas treatment, and a lower specific investment they cannot capitalise on economy-of-scale benefits as much as multiple-feedstock plants as single feedstocks typically arise in smaller quantities than feedstock mixtures. Realising the electricity potential for all feedstocks available in singlefeedstock plants would require a total investment of AU\$ 35.4b while multiple-feedstock plants would require AU\$ 39.5b. The dominant role of stubble makes single-feedstock hybrids, on average, slightly more cost competitive than multiple-feedstock hybrids (AU\$ 5.2m/MWe compared to AU\$ 5.3m/MWe) because they can be efficiency optimised for one feedstock. Plant cost would increase/efficiency decrease when co-firing even small amounts of lower quality feedstock, such as wood waste or RDF.

4. Discussion

Investment into new technology is generally fraud with more uncertainties than proven technology. This is also the case for CSP-EfB & EfW hybrid plants. Uncertainties, such as investment differences in various locations, DNI data quality, alternative use of feedstocks, temporal feedstock availability, and impact of future climate change require consideration. Such issues must be considered carefully on a project by project basis to ensure successful implementation.

4.1. Economic implications

This analysis identified several areas in Australia that have sufficient feedstock and solar radiation for the development of CSP-EfB & EfW hybrid plants, which could reduce electricity related CO₂ emissions by up to 27 million tones/year. However, CO₂ reduction requires implementation of renewable energy technology onground at lowest cost. Due to the defined CSP capacity factor of 18%, higher DNI locations lead to a lower specific investment and electricity cost, so such locations should be preferred when sufficient biomass is available. Additionally, high DNI locations are usually in rural areas and hybrid plants could provide network benefits, such as increased grid stability and investment deferral in new transmission infrastructure. In Australia biomass is available in DNI areas suitable for standalone CSP plants (>20 MJ/m²/day), which is an incentive to build hybrids as the CSP component would not suffer from lower than usual DNI levels.

The second most relevant cost reduction factor is deployment. Additional global CSP deployment is expected to reduce solar field costs by up to 40% by 2020 [71], which in turn could lead to CSP-EfB & EfW plant cost reductions of up to 21% by 2020. Lower costs would lead to more competitive CSP hybrid and standalone power plants in all DNI areas considered.

When assessing the impact of power plant projects on the Australian economy, not only the direct investment benefits should be considered but also the multiplier effect of benefits these projects create in the wider economy. In addition to the power plant the feedstock procurement has to be considered for CSP-EfB & EfW hybrids. A value-added multiplier effect for biomass plants, including feedstock procurement, of 1.42 has been used before [72] and seems also realistic for this assessment. CSP multiplier effects have been also investigated and a value of 2.3 is used in this assessment [73]. This assessment is based on 50% steam turbine capacity from CSP and 50% from EfB/EfW which leads to a combined multiplier of 1.86. Using this multiplier and assuming 29% of the plant investment flowing overseas for equipment manufacturing [73] the realisation of the single-feedstock CSP-EfB & EfW potential could lead to an overall financial stimulus of AU\$ 46.7b and AU\$ 52.2b for multiple-feedstock hybrids. Such an investment equals 3.5% of Australia's GDP in 2010-11 and 4% respectively [74].

4.2. DNI data quality

The annual electricity potential for CSP-EfB & EfW hybrid plants depends on the DNI input data, which tend to vary depending on data provider, time period considered, and data resolution. This assessment used daily average DNI data from the Australian Bureau of Meteorology [36] and the mapping results are similar to publications using data from NASA [9,75], or the German Aerospace Center [59]. Therefore the results are a reliable indication of the general potential but the use of other DNI sources might lead to slightly different results.

Our analysis is focused on regional CSP opportunities. Assessing the feasibility of a specific project requires more detailed DNI data to estimate the hourly, daily, and seasonal radiation variation more precisely. Methods exist to predict hourly data from DNI averages [76,77], but ground measured data are essential to verify the DNI models before committing funding for the construction of a CSP plant.

4.3. Alternative uses of biomass feedstock

While biomass may be available for power generation now, with oil prices generally expected to increase in the future, other industries will divert non-contaminated biomass, such as forestry residues, to higher value added products, e.g. biofuels or biochar. Also large amounts of biomass may already be committed to other purposes than electricity generation, e.g. plywood or paper industry production [20]. Biomass feedstock resource security needs to be considered carefully with long term feedstock supply contracts in place when planning a power station with an expected operational life of at least 25 years.

The low energy density of biomass requires significant feedstock quantities in the vicinity of typically large biofuel facilities which has implications in regards to continuous material availability, for example stubble potentially available for harvest has high annual variations [39], and transport costs, e.g. stubble transport cost increase from AU\$ 8/t for 10 km, AU\$ 12/t for 50 km to AU\$ 18/t for 100 km [20]. Therefore, the cost of transporting biomass resources long distance to centralised production facilities may become prohibitive. Typically, electricity prices are higher in rural areas than main population centers, due to higher transmission losses and maintenance, and distributed electricity generators using the lowest cost locally available biomass/waste feedstocks could provide electricity and avoid transmission losses from centralised generators, which are usually fossil fuel fired [20]. In regions with a sufficiently high DNI, >18 MJ/m²/day, the hybridisation with CSP is promising to gain economy-of-scale benefits, increase the use of local energy sources, and raise local power generation capabilities.

4.4. Temporal availability of biomass and solar resources

Agricultural and horticultural biomass residues as well as DNI vary throughout the year depending on the crop, weather conditions, harvesting cycles and the sun's angle of incidence. With forestry residues, plantation wood, urban wood waste, and RDF being more predictable feedstocks these could be/are already used in plants in Denmark [78] or Spain [79] to compensate the shortfall in less reliable feedstocks, such as stubble.

CSP-stubble hybrids have the highest annual electricity potential in Australia, but stubble potentially available for harvest is highly variable over the years, e.g. approximately 35 million tonnes in 2004 and approximately 7.5 million tonnes in 2003 [39]. Therefore stubble based CSP hybrids require a detailed feedstock availability/variability analysis and a conservative estimate of the continuously available feedstock potential. Rather than opting for a large plant, such as 50 MWe to obtain economy-of-scale benefits, a smaller unit, such as 30 MWe, has a higher probability of sufficient feedstock supply over its lifetime. Storing baled stubble is an alternative to minimise the risk of supply shortages [39].

To ensure continuous supply feedstock storage facilities are required for all biomass feedstocks. Also locating the plant in areas where crop and non-crop residues are available is beneficial as is the use of multiple feedstocks.

A low cost option is to oversize the start-up burners a biomass boiler requires, typically firing natural gas or diesel. Larger burners would allow plant operation at minimum load not only during supply shortages but also during equipment failures, such as conveyer belt breakdown. However, this option may not be competitive in areas where natural gas or diesel is not readily available.

With the DNI varying throughout the year, being higher in summer than winter, the thermal output of CSP plants can vary significantly depending on CSP technology and plant design, e.g. <50% of the design output in winter. Typically, the output of a biomass plant increases slightly in winter, due to lower ambient temperatures, but that is not sufficient to compensate the CSP shortfall. To smooth the annual electricity output, the solar field is typically oversized by up to 40%, which equals a solar multiple of 1.4. The larger field size increases investment but allows a higher annual energy yield therewith offsetting the additional investment without requiring thermal storage.

4.5. Impact of climate change

With climate change projected to increase annual Australian average temperatures by 0.6-1.5 °C by 2030 [80] the annual generation of a power station with air-cooling would decrease as higher temperatures reduce condenser performance. Assuming the predicted 0.6 °C increase the annual electricity output would decrease by 0.2% and 0.5% at 1.5 °C. The annual DNI would have to increase by 1% and 2.8% respectively to compensate these shortfalls. In Australia the surface solar radiation increased from 1994 to 2003 by an unspecified amount [81] but more data are required to reliably identify if the increase.

Another climate change impact on CSP-EfB hybrid plants is rainfall. High potential areas for such plants are in the wheat regions of New South Wales and Western Australia but these are also the regions that experienced very much below average rainfall for the period from 1997 to 2011 [80]. Longer droughts and more periods of heavy rather than normal rainfall can impact biomass production significantly, particularly considering that the rainfall in Australia is already highly variable. With the expectation of higher yearly variability in crop production, long-term feedstock storage is important to dampen the effects of variable feedstock availability.

4.6. Additional feedstock and energy potential

Identifying all potential biomass and waste feedstocks across Australia at a scale suitable for a CSP facility would be time consuming and expensive. It is appropriate to take a staged approach, using a high level reconnaissance approach first, identify prospective regions, and then conduct more detailed studies in the most prospective regions [82]. This assessment focuses on the materials outlined in Table 2 but acknowledges that other suitable feedstocks may exist.

Several agricultural regions exist in high DNI areas of New South Wales, Queensland, Victoria, South and Western Australia, and more detailed resource assessments in the future are likely to reveal additional biomass feedstocks from horticulture, cotton, olive, wine, and other industries. However, the nature of these feedstocks is diverse ranging from traditional biomass, such as wood waste from citrus farms, to exotic materials, such as olive pits or nutshells. This assessment, for example, identified a promising potential for stubble, urban wood waste and RDF for the Mildura region, in northern Victoria, where a biomass project currently under construction uses spent grape marc [83] rather than the regionally more abundant feedstocks. Further potential for a total of 120,000 t/a spent grape marc [84] exists in three DNI suitable sites in South Australia and New South Wales which is sufficient feedstock for three 5.2 MWe (25 GWh/year) CSP hybrid plants. Similarly, 181.500 t of almond shells and hulls were available in 2013 in South Australia and Victoria [85]. The material potentially available from the larger plantations would suffice to generate up to 213 GWh of additional electricity with CSP. With almond residues expected to increase to 238,500 t/a in 2015 [85], the electricity potential would increase to 284 GWh/year. In addition to biomass further CSP-EfW opportunities exist for remote towns and mines, such as 13.5 MWe Boodarie EfW proposal in Western Australia [86]. Adding a CSP component to this plant would allow a total generation capacity of 27 MWe increasing the annual electricity output to 130 GWh. It is very likely that similar opportunities exist across Australia and the potential of these materials has to be assessed on a regional to local basis to gain a suitable picture of feedstock quantities for site specific CSP-EfB & EfW developments.

Agroforestry is another option to increase the potential for CSP-EfB hybrid plant across Australia and in Western Australia more than 12,000 ha of mallee trees have already been planted in narrow belts to enhance erosion and salinity control [87]. The material is suitable for power generation but the existing mallee resource within a 100 km radius of any point is likely to be less than 50,000 t/ a [4], which is not sufficient to operate a small power station. The situation in the Mildura region is similar with regards to material density [88]. However, further agroforestry and biomass production expansions could supply sufficient material for several smaller, 5–10 MWe, distributed generation plants in the Mildura region [31].

Adding thermal storage is another option to increase the electricity potential of CSP-EfB & EfW hybrid plants and might be economically viable at the end of this decade with thermal storage prices expected to decrease from AU\$ 90/kWhth to AU\$ 22/kWhth in 2020 [71]. The well proven 7.5 h thermal storage concept, as applied in the Andasol parabolic trough plants in Spain [89], could increase the CSP capacity factor in hybrid plants to 40% therewith increasing the annual electricity potential to 37 TWh for single and 40.3 TWh for multiple-feedstock hybrids.

4.7. Limitations

This study aims to identify the electricity potential and prospective regions for CSP-EfB & EfW hybrid plants in Australia. It does not provide specific sites for such plants as site selection depends on criteria that have to be addressed in a local study, such as land ownership and other land users. Also the electricity potential of power plants can change with its location, due to grid constraints and ambient temperatures.

Thermal storage in the CSP plant is not considered in this assessment but provides scope for future study as it could increase the electricity generation potential significantly. The optimum thermal storage capacity is site specific, depending on time and lengths of high electricity prices/demand. It is therefore complicated to incorporate this for an Australia wide assessment but the numbers provided outline the general potential of thermal storage.

While this paper provides the investment data these are generalised to cover all of Australia. Identifying the detailed investment and economic viability of a power station requires an understanding of local prices for electricity, fuel, labor, and equipment.

The CO₂ abatement potential CSP-EfB & EfW hybrid plants depends on various factors, such as fuel composition and transport distances. Assessing this nationwide required some generalisation with results that are not as detailed as a local assessment based on site specific data.

In summary this paper provides a high-level assessment of the CSP-EfB & EfW hybrid plant potential in Australia. Identification of the best specific sites for such projects requires further localised studies with more detailed technical and economic assumptions.

5. Conclusions

The potential for CSP-EfB & EfW hybrid plants in Australia is significant with a total electricity potential within 50 km distance around existing transmission lines of up to 33.5 TWh/year, equivalent to 12.8% of all electricity generated in Australia in 2008–09. The potential for CSP hybrids using multiple feedstocks is 8.8% higher than the potential for hybrids using a single feedstock as multiple-feedstock plants qualify for more locations. The average investment in single-feedstock hybrids is slightly lower than in multiple-fuel hybrids as such plants do not require more expensive equipment capable of handling several feedstocks. The feedstock with the largest individual potential is stubble, 64% of total generation, followed by urban wood waste, 15% of total generation.

The majority of identified locations could provide biomass for 5–25 MWe CSP-EfB & EfW hybrid plants. Several regions exist in New South Wales, Victoria, Queensland, Western and South Australia for up to 50 MWe installations while fifteen locations could support even larger capacities.

Considering the average 2010 carbon intensity of Australian generation assets, the exploitation of the CSP-EfB & EfW hybrid resources could displace up to 27 million tones CO₂/year, which equals 4.8% of all Australian 2009–10 CO₂ emissions.

This assessment of feedstocks and most prospective regions for CSP-EfB & EfW hybrid plants provides guidance to project developers and researcher for more detailed local resource and site identification studies in the future. In addition, local biomass feedstocks, such as spent grape marc or almond shells exist, thus increasing the annual electricity yield potential. However, it is not possible to assess these localised feedstocks on a nationwide basis.

CSP-EfB & EfW hybrid plants are suitable for several regions in Australia and there is the potential to realise CSP installations at significantly lower cost than standalone CSP plants. With the first CSP-EfB hybrid in Spain demonstrating the benefits of such hybrids and providing financiers with confidence in the technology, similar systems could be seen in Australia within the next years. However, Australia's electricity production is built on a large and cheap coal resource, resulting in low wholesale electricity prices. Renewable energy, including CSP, has to compete with coal, taking into account investment, operating/maintenance costs, avoided CO₂ emissions, and returns from renewable energy certificates. Thus, future work for CSP developments requires an economic analysis considering these costs and returns from power production.

References

- Kelly J. Darkness falls on \$1.2bn SolarDawn project. Sydney, Australia: The Australian; 13-Nov. p. 4.
- [2] Edwards V. Axing of \$60m solar grant floors Whyalla. Sydney, Australia: The Australian; 11-Jun.
- [3] Farine D, O'Connell DA, John Raison R, May BM, O'Connor MH, Crawford DF, et al. An assessment of biomass for bioelectricity and biofuel, and for greenhouse gas emission reduction in Australia. GCB Bioenergy 2011;4(2):148-75.
- [4] Stucley C, Schuck S, Sims R, Bland J, Marino B, Borowitzka M, et al. Bioenergy in Australia. Sydney; 2012.
- [5] Bagasse power. Biomass Power Therm Mag [Online]. Available: http:// biomassmagazine.com/articles/5247/bagasse-power; 2011 [accessed 10.01.14].
- [6] Beetz B. Whole 1 GW Blythe project to convert to PV. PV Mag [Online]. Available: www.pv-magazine.com/news/details/beitrag/whole-1-gw-blytheproject-to-convert-to-pv_100007604/#axzz2A5Oyy1Jv; 2012 [accessed 10 01.14].
- [7] Elliston B, Diesendorf M, MacGill I. Simulations of scenarios with 100% renewable electricity in the Australian National Electricity Market. Energy Policy Jun. 2012;45:606–13.
- [8] International Energy Agency. Technology roadmap concentrating solar power. Paris: OECD Publishing; 2010.
- [9] IT Power. Realising the potential of concentrating solar power in Australia. Canberra, Australia; 2012.
- [10] Turchi CS, Erbes M. Gas turbine/solar parabolic trough hybrid designs. In: ASME TurboExpo 2011; 2011.
- [11] Novatec Solar GmbH. Novatec Solar's australian fuel-saver commences operation [Online]. Available: www.transfield.com.au/news/MediaReleases/ 20121024_NovatecSolar Australian Fuel Saver Commences Operation.pdf; 2012 [accessed 10.01.14].
- [12] CS Energy. Kogan creek solar boost project [Online]. Available: http:// kogansolarboost.com.au; 2013 [accessed 10.01.14].
- [13] Department of Resources Energy and Tourism. Pathway to solar thermal for coal-fired Collinsville. Canberra; 20-Feb-2013.
- [14] Ipswich City Council. Ipswich supports pioneering eco-precinct concept [Online]. Available: http://www.ipswich.qld.gov.au/about_council/media/ view_release/?id=3657&action=viewMediaRelease; 2012 [accessed 10.01.14].
- [15] Sathaye J, Lucon O, Rahman A, Christensen J, Denton F, Fujino J, et al. Renewable energy in the context of sustainable development. In: IPCC special report on renewable energy sources and climate change mitigation. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press; 2011.

- [16] Protermosolar. Abantia and Comsa Emte set up the first facility in the world that combines solar thermal and biomass energy [Online]. Available: http:// www.protermosolar.com/saladeprensa/?q=en/content/abantia-and-comsaemte-set-first-facility-world-combines-solar-thermal-and-biomass-energy; 2012 [accessed 10.01.14].
- [17] Cot A, Amettler A, Vall-Llovera J, Aguilo J, Arque JM. Termosolar borges: a thermosolar hybrid plant with biomass. Third International Symposium on Energy from biomass and waste; 2010.
- [18] Morell D. CSP BORGES the World's First CSP plant hybridized with biomass. In: CSP Today USA 2012-6th Concentrated solar Thermal power Conference; 2012.
- [19] Rodriguez LC, May B, Herr A, Farine D, O'Connell D. Biofuel excision and the viability of ethanol production in the Green Triangle, Australia. Energy Policy Apr. 2011;39(4):1951–7.
- [20] Rodriguez LC, May B, Herr A, O'Connell D. Biomass assessment and small scale biomass fired electricity generation in the Green Triangle, Australia. Biomass Bioenergy Jul. 2011;35(7):2589–99.
- [21] Parsons Brinckerhoff. Solar power plant pre-feasibility study. Brisbane, Australia; 2008.
- [22] AECOM Australia Pty Ltd. Pre-feasibility study for a solar power precinct. Sydney; 2010.
- [23] Parsons Brinckerhoff Australia Pty Ltd. Queensland concentrated solar power. Pre-feasibility Report December 2009. Brisbane, Australia: Queensland Government, Office of Clean Energy, Department of Employment, Economic Development and Industry; 2010.
- [24] Clifton J, Boruff B. Site options for concentrated solar power generation in the Wheatbelt. Perth; 2010.
- [25] Dawson L, Schlyter P. Less is more: strategic scale site suitability for concentrated solar thermal power in Western Australia. Energy Policy Aug. 2012;47:91–101.
- [26] Rutovitz J, Peterseim J, Elliston B, Harris S, Mohr S, Lovegrove K, et al. Breaking the solar gridlock. Potential benefits of installing concentrating solar thermal power at constrained locations in the NEM. Sydney; 2013.
- [27] Parsons Brinckerhoff. Hybridisation of Fossil fuel Energy generation in Australia public report. Melbourne; 2013.
- [28] Baban SM, Parry T. Developing and applying a GIS-assisted approach to locating wind farms in the UK. Renew Energy Sep. 2001;24(1):59–71.
- [29] Janke JR. Multicriteria GIS modeling of wind and solar farms in Colorado. Renew Energy Oct. 2010;35(10):2228–34.
- [30] Hossain J, Sinha V, Kishore VVN. A GIS based assessment of potential for windfarms in India. Renew Energy Dec. 2011;36(12):3257–67.
- [31] Bryan BA, Ward J, Hobbs T. An assessment of the economic and environmental potential of biomass production in an agricultural region. Land Use Policy Oct. 2008;25(4):533–49.
- [32] Viana H, Cohen WB, Lopes D, Aranha J. Assessment of forest biomass for use as energy. GIS-based analysis of geographical availability and locations of woodfired power plants in Portugal. Appl Energy Aug. 2010;87(8):2551–60.
- [33] Charabi Y, Gastli A. GIS assessment of large CSP plant in Duqum, Oman. Renew Sustain Energy Rev Feb. 2010;14(2):835–41.
- [34] Gastli A, Charabi Y, Zekri S. GIS-based assessment of combined CSP electric power and seawater desalination plant for Duqum-Oman. Renew Sustain Energy Rev 2010;14(2):821–7.
- [35] Turchi C, Langle N, Bedilion R, Libby C. Solar-augment potential of U.S. Fossil-Fired power plants. Golden; 2011.
- [36] Bureau of Meteorology. Australian monthly average hourly global and direct Normal solar exposure gridded data. Melbourne; 2012.
- [37] Geoscience Australia. GEODATA TOPO 250K series 3-packaged. Canberra; 2006.
- [38] Australian Bureau of Statistics. 1209.0.55.002-Mesh blocks digital boundaries, Australia. Canberra; 2006.
- [39] Herr A, O'Connell D, Dunlop M, Unkovich M, Poulton P, Poole M. Second harvest—is there sufficient stubble for biofuel production in Australia? GCB Bioenergy 2012;4(6):654–60.
- [40] Herr A, Dunlop M. Bioenergy in Australia: an improved approach for estimating spatial availability of biomass resources in the agricultural production zones. Biomass Bioenergy May 2011;35(5):2298–305.
- [41] Gendebien A, Leavens A, Blackmore K, Godley A, Lewin K, Whiting KJ, et al. Refuse derived fuel, current practice and perspectives. Brussels: European Commission; 2003.
- [42] Rotter VS, Kost T, Winkler J, Bilitewski B. Material flow analysis of RDFproduction processes. Waste Manag 2004;24(10):1005-21.
- [43] Christensen TH. Solid waste technology and management. 1st ed., vol. 1. New York: John Wiley & Sons; 2010.
- [44] Geoscience Australia. MapConnect [Online]. Available: http://www.ga.gov.au/ topographic-mapping/mapconnect.html; 2012 [accessed 10.01.14].
- [45] Australian Energy Market Operator. National transmission network development plan 2012. Sydney; 2012.
- [46] Australian Bureau of Agricultural and Resource Economics. Energy in Australia 2011. Canberra, Australia; 2011.
- [47] Western Power. Network capacity mapping tool [Online]. Available: http:// www.westernpower.com.au/ldd/ncmtoverview.html; 2013 [accessed 10.01.14].
- [48] Peterseim JH, Tadros A, White S, Hellwig U, Klostermann F. Concentrated solar power/energy from waste hybrid plants – creating synergies. In: SolarPACES Conference; 2012.

- [49] Peterseim J, Tadros A, White S, Hellwig U, Landler J, Galang K. Solar towerbiomass hybrid plants – maximizing plant performance. In: SolarPACES Conference; 2013.
- [50] ERK Eckrohrkessel GmbH. Personal communication with ERK Eckrohrkessel GmbH. Berlin, Germany; 2013.
- [51] IRENA. Renewable Energy technologies: cost analysis series. Abu Dhabi, UAEIn Concentrating solar power, vol. 1 issue 2/5; 2012.
- [52] Lekovic S, Priftakis P, Donoghue K, Rigzin T, Cole S, Pryor B, et al. Electricity gas Australia 2011. Melbourne, Australia: Energy Supply Association of Australia; 2011.
- [53] Herr A, Farine D, Poulton P, Baldock J, Bruce J, Dunlop M, et al. Harvesting stubble for energy in Australia – take it or leave it? Phase II of opportunities for energy efficiency, regional self-sufficiency and energy production for Australian grain growers; 2010.
- [54] AREVA group. A bright future for CSP-AREVA spotlight: CSP projects [Online]. Available: www.areva.com/EN/operations-3642/areva-solar-projects.html; 2011 [accessed 10.01.14].
- [55] eSolar Inc. Sierra SunTower a new Blueprint for solar Energy [Online]. Available: http://www.esolar.com/wp-content/uploads/2013/10/sierra_fact_ sheet.pdf; 2009 [accessed 10.01.14].
- [56] BrightSource Energy Inc. Ivanpah overview [Online]. Available: http:// ivanpahsolar.com/about; 2013 [accessed 10.01.14].
- [57] Morin G, Kirchberger J, Lemmertz N, Mertins M. Operational results and simulation of a superheating fresnel collector. In: SolarPACES Conference; 2012.
- [58] Peterseim JH, White S, Tadros A, Hellwig U. Concentrated solar power hybrid plants, which technologies are best suited for hybridisation? Renew Energy Sep. 2013;57:520–32.
- [59] Trieb F, Schillings C, O'Sullivan M, Pregger T, Hoyer-Klick C. Global potential of concentrating solar power. In: SolarPACES Conference; 2009.
- [60] Pérez Á, Torres N. Solar parabolic trough biomass hybrid plants: a cost efficient concept suitable for places in low irradiation conditions. In: Solar-Paces Conference; 2010.
- [61] Siros F, Philibert C, Le Moullec Y, Tusseau M, Bonnelle D. The value of Hybridizing CSP. In: SolarPACES Conference; 2012.
- [62] RWE AG. Wood-fired cogeneration plant Berlin-Neukölln/Gropiusstadt [Online]. Available: www.rwe.com/web/cms/en/66012/rwe-innogy/technologies/ biomass/wood-fired-cogeneration-plant-berlin-neukoelln-gropiusstadt/; 2012 [accessed 10.01.14].
- [63] BGS eV. Mitverbrennung von g
 ütegesicherten Sekund
 ärbrennstoffen und energetische Verwertung. M
 ünster; 2011.
- [64] van Tubergen J, Glorius T, Waeyenbergh E. Classification of solid recovered fuels. Bruxelles: European Recovered Fuel Organisation; 2005.
- [65] Chang YH, Chen WC, Bin Chang N. Comparative evaluation of RDF and MSW incineration. J Hazard Mater 1998;58(1–3):33–45.
- [66] Environment Protection and Heritage Council. National waste report 2010. Canberra, Australia; 2010.
- [67] Climate Change Authority. Renewable energy target review. Canberra; 2012.
- [68] IEA. CO2 emissions from fuel combustion 2012-Highlights. Paris; 2012.
- [69] Department of Climate Change and Energy Efficiency. Australian National Greenhouse Accounts – National Inventory by Economic Sector 2009–10. Canberra; 2012.
- [70] Kitney D. Anthony Pratt seeks \$300m from government to flick switch on waste-to-power scheme. The Australian, Sydney: The Australian; 23-Nov.
- [71] Hinkley J, Curtin B, Hayward J, Wonhaus A, Boyd R, Grima C, et al. Concentrating solar power-drivers and opportunities for cost-competitive electricity. Sydney, Australia: CSIRO & Aurecon Australia Pty Ltd; 2011.
- [72] Gan J, Smith CT. Co-benefits of utilizing logging residues for bioenergy production: the case for East Texas, USA. Biomass Bioenergy Sep. 2007;31(9):623–30.
- [73] Caldés N, Varela M, Santamaría M, Sáez R. Economic impact of solar thermal electricity deployment in Spain. Energy Policy 2009;37(5):1628–36.
- [74] Australian Bureau of Statistics. Year book Australia 2012-value of goods and services produced by Australian industry. Canberra; 2012.
- [75] Geoscience Australia and ABARE. Australian Energy resource assessment chapter 10 solar energy. Canberra; 2010.
- [76] Bernardos A, Fernández-Peruchena C, Blanco M. Generation of series of high frequency DNI years consistent with annual and monthly long-term averages using measured DNI data. In: SolarPACES Conference; 2013.
- [77] Zhang HL, Baeyens J, Degrève J, Cacères G. Concentrated solar power plants: review and design methodology. Renew Sustain Energy Rev Jun. 2013;22: 466–81.
- [78] Babcock & Wilcox Vølund A/S. Multifuel-fired combined heat and power plant Måbjergværket Måbjerg, Denmark [Online]. Available: www.volund.dk/ Multi_fuel_energy/~/media/Downloads/Brochures-MULTI/Maabjerg-Holstebro-Denmark.ashx; 2008 [accessed 10.01.14].
- [79] Acciona Energía. Biomass plant from straw combustion in Sangüesa [Online]. Available: http://www.acciona-energia.com/media/219313/ACCIONA_ Sanguesabiomassplant_EN.pdf; 2002 [accessed 10.01.14].
- [80] CSIRO and Australian Bureau of Meteorology. State of the climate 2012. Canberra; 2012.
- [81] Wild M. Global dimming and brightening: a review. J Geophys Res Jun. 2009;114:D00D16.
- [82] Herr A, Connell DO, Farine D, Dunlop M, Crimp S, Poole M. Watching grass grow in Australia: is there sufficient production potential for a biofuel industry? Biofuel Bioprod Bioref 2012;(6):257–68.

- [83] Australian Industry Group. \$7.5 million renewable energy plant to be powered by wine industry grape waste; 2012.
- [84] Tarac Technologies Pty Ltd. Tarac technologies [Online]. Available: http:// www.tarac.com.au; 2013 [accessed 10.01.14].
- [85] Brown B. Personal communication between Juergen H. Peterseim and Ben Brown from the Almond Board of Australia regarding almond residue availability in South Australia and Victoria. Berri, South Australia; 2013.
- [86] New Energy Corporation. New Energy is investing in state-of-the-art waste to energy technology in the Pilbara [Online]. Available: http://www. newenergycorp.com.au/projects/pilbara-wa/; 2013 [accessed 10.01.14].
- [87] Bartle JR, Abadi A. Toward sustainable production of second generation bioenergy feedstocks. Energy Fuel Jan. 2010;24(1):2–9.
- [88] Pearl D. Personal communication between Juergen H. Peterseim and Darryl Pearl from the Department of Primary Industries Victoria regarding mallee availability in the Mildura region. Mildura: Department of Primary Industries Victoria; 2013.
- [89] Flagsol GmbH. Andasol projects, Spain [Online]. Available: http://www.flagsol. com/flagsol/english/aaareferences-alt/andasol-projects-spain-alt/andasolprojects-spain.html; 2012 [accessed 10.01.14].

4.3 CSP technology selection

The following publication analyses various currently available CSP technologies in regards to their suitability for hybridisation with Rankine cycle power plants and it identifies the most promising ones for various integration options, not only with biomass and waste feedstocks but also with natural gas and coal. The focus of this research is on the hybridisation of CSP with biomass and waste materials. However, the assessment methodology was applied for natural gas and coal hybrids and therefore the results are shown even though they are not part of this research. This analysis was the first part of the research project to be completed, and it provided important information for the other research components.

An industry engagement workshop on 21 July 2011 at the University of Technology, Sydney was a key part of this analysis and provided not only the CSP technology ratings but also valuable input into the research project from various industry professionals.

Renewable Energy 57 (2013) 520-532

Contents lists available at SciVerse ScienceDirect

Renewable Energy

journal homepage: www.elsevier.com/locate/renene

Review

Concentrated solar power hybrid plants, which technologies are best suited for hybridisation?

Juergen H. Peterseim^{a,*}, Stuart White^b, Amir Tadros^c, Udo Hellwig^d

^a Institute for Sustainable Futures, University of Technology Sydney, Level 11, UTS Building 10, 235 Jones Street, Ultimo NSW 2007, Australia ^b Institute for Sustainable Futures, University of Technology Sydney, Australia ^c Aurecon Australia Pty Ltd, Australia ^d EEW, Educated Captul Company

^d ERK Eckrohrkessel GmbH, Germany

ARTICLE INFO

Article history: Received 5 July 2012 Accepted 14 February 2013 Available online

Keywords: Concentrated solar power Solar hybrid plants Technology comparison Multi-criteria decision making Analytical Hierarchy Process

ABSTRACT

This assessment aims to identify the most suitable concentrated solar power (CSP) technologies to hybridize with Rankine cycle power plants using conventional fuels, such as gas and coal, as well as nonconventional fuels, namely biomass and waste materials. The results derive from quantitative data, such as literature, industry information and own calculations, as well as qualitative data from an expert workshop. To incorporate the variety of technology criteria, quantitative and qualitative data the Analytical Hierarchy Process (AHP) is used as the multi-criteria decision making (MCDM) tool. Only CSP technologies able to directly or indirectly generate steam are compared in regards to feasibility, risk, environmental impact and Levelised Cost of Electricity (LCOE). Different sub-criteria are chosen to consider the most relevant aspects. The study focuses on the suitability of CSP technologies for hybridisation and results obtained are reality checked by comparison with plants already being built/ under construction. The results of this assessment are time dependant and may change with new CSP technologies maturing and prices decreasing in the future.

Key findings of this assessment show that Fresnel systems seem to be the best technology for feedwater preheating, cold reheat steam and <450 °C steam boost applications. Parabolic troughs using thermal oil rank second for all CSP integration scenarios with steam temperatures <380 °C. Generally, for steam temperatures above 450 °C the solar towers with direct steam generation score higher than solar towers using molten salt and the big dish technology. At and above 580 °C the big dish is the only alternative to directly provide high pressure steam.

In addition to a general CSP technology selection for hybridisation the framework of this study could be used to identify the most suitable CSP technology for a specific CSP hybrid project but this requires detailed information for direct normal irradiance, climate conditions, space constraints etc to provide reliable results.

© 2013 Elsevier Ltd. All rights reserved.

1. Introduction

In recent months the CSP industry has faced increasing competition from the photovoltaic (PV) industry as the latter's manufacturing costs are decreasing rapidly due to mass manufacturing and associated learning curves. The conversion of the first 500 MW_e phase, and potentially second, of the 1000 MW_e Blythe Solar Power Project, US from parabolic trough to PV is the most prominent example so far [1]. In order to remain competitive the CSP industry has to emphasise the benefits of energy dispatchability, and also to further reduce plant cost.

Demonstrating the benefits of energy dispatchability through thermal storage is relatively simple as several plants are equipped with 7.5 h [2] to 15 h [3] full-load storage systems. In summer the 15 h storage allows 24/7 base load generation and intermittent load generation in winter. However, thermal storage is still expensive, currently costing around US\$90/kWh_{th} [4]. Through innovation and learning experiences these costs are expected to decrease to US\$22/ kWh_{th} by 2020 [4]. More cost competitive thermal storage technologies would give CSP a unique advantage over PV but are not available yet. Hence, hybridising a CSP steam generator with a host Rankine cycle plant makes sense as the host plant can provide the additional energy required at lower cost until more cost competitive thermal storage is available. Currently, a 7 h thermal storage for a 50 MW_e parabolic trough plant contributes 9% to the required





^{*} Corresponding author. Tel.: +61 2 9514 4636.

E-mail address: JuergenHeinzMartin.Peterseim@uts.edu.au (J.H. Peterseim).

^{0960-1481/\$ –} see front matter \odot 2013 Elsevier Ltd. All rights reserved. http://dx.doi.org/10.1016/j.renene.2013.02.014

investment [5]. Reducing the investment by 9% while decreasing plant complexity and enabling better energy dispatchability is likely to put CSP systems into a better commercial position.

Reducing capital expenditure (CapEx) is more difficult and all technology providers are working intensively on this. Compared to the time consuming re-design of mirrors, support structures, absorber tubes etc a rapid way to reduce CapEx is the hybridisation of CSP steam generators with other Rankine cycle based power generation assets. This could be in the form of retrofits such as the example at Liddell, Australia [6] or new purpose build plants such as Kuraymat, Egypt [7]. The joint use of equipment, such as steam turbine, condenser, building infrastructure and feedwater equipment, can provide significant LCOE reductions, up to 28% [8], which would help the CSP industry to start building more plants, ramping up manufacturing capabilities as well as gaining relevant project implementation experience. The wind and PV industries realised learning curve cost reductions of 15% and 20% when doubling cumulative deployment [9] and the CSP business could benefit similarly. At the same time plant operators and financiers will become more familiar with the variety of CSP technologies and their specific benefits. Another benefit would be projects with smaller investment requirements, which reduces financial risk. This would particularly help newly developed CSP technologies to prove their capabilities. Due to a minimum CSP plant efficiency scale of 10-100 MWe for standalone systems [4] such projects incur a higher financial risk over kWe scale PV systems. However, CSP add-on references exist at sizes of only 9 MW_{th} [6].

In addition to energy dispatchability and cost benefits CSP hybrid plants expand suitable plant locations from usually arid to semi-arid and temperate regions. While CSP standalone plants typically require a direct normal irradiance (DNI) \geq 2000 kWh/m²/year CSP hybrids can be located in areas \geq 1700 kWh/m²/year [10]. This moves CSP out of remote desert type areas closer to load centres and back-up fuel sources such as biomass in agricultural regions and waste materials in urban areas.

A number of plants combining CSP and coal/natural gas are already in operation in Egypt [7] and the US [11], as well as under construction in Australia [12], Iran [13] and Morocco [14]. These projects prove that the hybridisation of CSP with other energy sources has benefits in terms of CapEx and energy dispatchability.

Identifying the ideal CSP technology for a specific power plant location is highly dependant on local conditions, for example DNI and climate, as well as site specific constraints, e.g. land and water availability as well as topography. As it is impossible to assess all suitable locations for CSP hybrid plants in one comparison, this study provides a general multi-criteria based approach to identify the most suitable CSP technologies for hybridization with Rankine cycle power plants using a number of fuels. This has already been done for standalone solar plants [15] but not yet for hybrid configurations. The generic input data provided can be substituted with site specific data allowing the programme to be used for broad as well as specific investigations.

2. Technology ranking process

The technology ranking process used consisted of three consecutive steps:

- Selection of the main criteria and sub-criteria;
- Data collection; and
- Criteria weighting at an expert workshop using the Analytical Hierarchy Process.

2.1. Criteria selection

To make a holistic comparison it is necessary to select criteria assessing the feasibility, risk, environmental impact and LCOE of suitable CSP technologies. Each of these criteria groups is divided into sub-criteria, see Fig. 1, except the LCOE as this information already includes capital, operational and financing expenditures as well as parasitic losses. At the workshop (described in Section 2.4), all criteria were rated individually by the participants.

2.1.1. Feasibility

In this study the feasibility of integrating a CSP technology into a host plant comprises the peak solar to electricity efficiency, suitable operation range as well as the maximum site gradient.

The peak solar to electricity efficiency is chosen as the annual efficiency varies too much with the project location and is therefore not suitable for a general assessment. The annual efficiency would be preferred for a site specific CSP assessment.

The suitable operation range is selected to outline the reference case for each technology (in MW_{th}) with plants in operation and under construction being considered. In the biomass/waste hybrid scenario a 20 MW_{th} CSP contribution is assumed but not all CSP technologies considered have references in this range.

The maximum site gradient is important to assess the locational flexibility. Particularly smaller plants closer to urban areas might not have graded areas for a CSP system. Hence the higher the possible site gradient of a technology the better. The maximum site gradient of 12° [16] is determined by the tipping angle of a water truck cleaning the heliostat or mirror field.

2.1.2. Risk reduction

Risk reduction includes the technical maturity, plant complexity and integration simplicity.

The technology maturity is assessed by the largest plant (in MW_e) with financial approval. Obtaining financial approval for high



Fig. 1. Criteria hierarchy.

CapEx projects is a clear indicator of how comfortable financiers are with a technology in terms of performance, reliability and risk mitigation.

Plant complexity assesses the number of working fluid cycles required. DSG plants require only one cycle while others, such as thermal oil or molten salt, require two. Generally, the lower the number of system cycles the lower the technical, operational and maintenance risk. Specific technical issues, such as two-phase flow related problems in DSG plants, are a reflection of the technology maturity, where parabolic trough (thermal oil) plants score best due to their long track record. However, even these plants still have some technical issues requiring further improvement, e.g. optimizing flexible hoses connecting the absorber tubes to distribution/ collection headers.

The criterion for the integration simplicity is the steam temperature of the CSP technologies. Being able to produce the steam parameters required by the host plant is essential to integrate a CSP steam generator and therefore this is an exclusion criterion. In case a CSP technology is unable to provide the steam parameter required by the host plant it is not further considered.

2.1.3. Environmental impact

The environmental impact reduction comprises land use and cleaning water consumption.

The land use in $m^2/MWh/a$ is important for environmental but also commercial reasons. Particularly, CSP systems close to urbanization have to expect higher land cost and therefore minimise land use.

In relation to the water consumption (litre/MWh_e), only cleaning water for heliostat and mirror washing is considered as the Rankine cycle connected to the host plant is a closed system and blowdown losses are expected to be similar across all CSP technologies.

2.2. Data collection

To identify relevant data for the chosen criteria an extensive literature review is essential. Most of the data given in Section 3 derive from publicly available literature with additional information from industry sources as well as own engineering work. The LCOE obtained were from different sources, are inflation adjusted and converted into US\$.

Technologies not directly or indirectly generating steam to feed into a Rankine cycle are excluded, e.g. Dish-Stirling systems.

2.3. Analytical hierarchy process

The Analytical Hierarchy Process (AHP) provides a comprehensive and rational framework for structuring complex decision problems and comparing the different alternatives considered. The method is widely used in the research and industry world, including assessments comparing fossil fuels with renewable sources [17,18] and different CSP standalone technologies with each other [15].

Because of the significant number of technical, risk, environmental and economic based criteria as well as the use of qualitative and quantitative data the AHP is chosen for this assessment. The AHP decomposes a complex problem into several sub-problems, such as economic and environmental impact, and therewith provides an easy to understand path for multi-dimensional decision making [19]. This is essential to include the different aspects of CSP technologies and was a main reason for selecting the AHP.

The problem decomposition takes place by identifying criteria (main- and sub-criteria) relevant to the problem and organizing them in levels of hierarchy, see Fig. 1. The AHP can use precise criteria data (quantitative information) to compare the different options as well as the decision makers' personal judgements (qualitative information). Subsequently, quantitative and qualitative information are merged to calculate the total score for each option.

As we identified quantitative data for all criteria this study used the original AHP with the following relationship [19] rather than the revised pairwise comparison matrices [20]:

$$\text{AHP score} = \max \sum_{j=1}^{n} a_{ij} w_j,$$

for i = 1,2,3, ..., m, a = alternative and w = criteria weighting.

After merging quantitative and qualitative data the alternative with the maximum score is considered the preferred option. The different criteria and its hierarchy in this assessment are shown in Fig. 1. The qualitative data used are shown in Tables 1\and 2, while the quantitative data are outlined in Tables 3–6. Due to the availability of quantitative data for all criteria of each CSP technology an expert workshop was organized to specifically identify the relevance of the individual criteria.

To accommodate uncertainties in the input data $\pm 10\%$ sensitivity is applied to all results.

2.4. Workshop

To rate the different criteria and sub-criteria from 1 to 9 [21], with 9 being of extreme importance and 1 being of no importance, we organized an expert workshop with 49 professional participants and conducted another 3 interviews. The workshop method was chosen to increase judgement accuracy [22] and subsequent interviews were only conducted for people who had other commitments on the day of the workshop.

After presenting the CSP technologies and criteria, and explaining/discussing the selected quantitative data all participants individually rated the importance of the various CSP criteria for the different host plant scenarios (Section 4). All individual participant ratings were combined and the average for all criteria identified. The form given to the participants for the non-conventional fuel hybrid scenarios is shown in Fig. 2. The same forms with adapted data were provided for the natural gas and coal scenarios. Of all participants 40 result sheets were evaluable, with the remaining not being filled in properly or not returned. Despite this the number of criteria ratings obtained is considerable compared to other workshops with significantly fewer people [15]. The University of Technology Sydney's Human Research Ethics Committee checked and approved the design and conduct of the workshop and interview activities.

| able 1 | | | | | | | |
|--------|--------------|--------------|---------|---------------|------|-------------|-------|
| verage | score of rat | ing criteria | and sco | re difference | s by | participant | group |

| Criteria | All | ΔΟ | ΔTP | ΔC | ΔR | ∆Unknown |
|--------------------------------------|------|-------|-------------|-------|------------|----------|
| Peak solar to electricity efficiency | 6.52 | -0.52 | -0.12 | -0.58 | 0.91 | 0.36 |
| Suitable operation range | 6.85 | 1.05 | -0.08 | -1.45 | -0.33 | -0.02 |
| Maximum site gradient | 4.70 | 0.07 | 0.07 | 0.50 | -0.03 | -0.45 |
| Technology maturity | 6.99 | 0.61 | 0.31 | -0.86 | -0.28 | -0.37 |
| Plant complexity | 6.18 | -0.04 | 0.53 | 0.49 | 0.59 | -1.43 |
| Integration simplicity | 6.78 | 0.05 | 0.42 | -0.38 | 0.50 | -0.78 |
| Land use | 4.82 | 0.12 | 0.55 | 0.52 | 0.14 | -1.28 |
| Cleaning water consumption | 5.19 | -0.76 | 0.68 | -0.59 | 1.33 | -0.69 |
| Criteria groups | | | | | | |
| Feasibility | 6.92 | -0.25 | 0.08 | -0.25 | 0.23 | 0.17 |
| Risk reduction | 6.93 | 0.27 | 0.27 | -0.53 | 0.26 | -0.56 |
| Impact reduction | 4.92 | -0.28 | 0.62 | -0.58 | 0.99 | -0.92 |
| LCOE | 7.98 | 0.45 | -0.12 | 0.02 | 0.40 | -0.78 |

Table 2

Average score of rating criteria for the 10 MW_e biomass/waste scenario and score difference for the 200 MW_e CCGT and 500 MW_e coal plants.

| Criteria | $10 \ \text{MW}_{\text{e}}$ | $\Delta 200 \; \text{MW}_{\text{e}}$ | $\Delta 500 \ \text{MW}_{\text{e}}$ |
|--------------------------------------|-----------------------------|--------------------------------------|-------------------------------------|
| Peak solar to electricity efficiency | 6.28 | 0.27 | 0.45 |
| Suitable operation range | 6.60 | 0.40 | 0.35 |
| Maximum site gradient | 4.45 | 0.27 | 0.48 |
| Technology maturity | 6.33 | 0.90 | 1.10 |
| Plant complexity | 6.33 | -0.35 | -0.10 |
| Integration simplicity | 6.98 | -0.27 | -0.30 |
| Land use | 4.88 | -0.03 | -0.15 |
| Cleaning water consumption | 5.30 | -0.18 | -0.15 |
| Criteria groups | | | |
| Feasibility | 6.80 | 0.13 | 0.23 |
| Risk reduction | 6.73 | 0.13 | 0.50 |
| Impact reduction | 4.90 | 0.05 | 0.00 |
| LCOE | 7.95 | 0.08 | 0.02 |

To cover the variety of aspects of a power project we invited people with different perspectives. They included plant operators, technology provider/EPC contractors, consultants and researchers, participant breakdown see Fig. 3. The operators included companies owning fossil and renewable energy assets, the technology provider comprised companies offering all of the CSP technologies assessed, see Section 3, and the consultants as well as researchers selected had a power generation background. All participants had years and many decades of experience in the energy business and represented well known national as well as international companies.

Due to the varying interest of the workshop participants/interviewees differences were observed in their criteria rating, see Table 1. Generally, operators (Δ O), consultants (Δ C) and researchers (Δ R) ranked the LCOE more important than the technology provider (Δ TP). Feasibility, risk and impact reduction were considered more important by technology providers and researchers while consultants ranked these criteria below average. Operators rated risk reduction above average and feasibility as well as impact reduction below average.

Similarly, rating differences were observed for the different CSP/ host plant scenarios, see Table 2. The importance of a number of criteria increases with larger plant sizes, e.g. peak solar to electricity efficiency, technology maturity and suitable operation range. Interestingly, while participants rated all criteria groups getting more important with increasing plant sizes they considered half of the sub-criteria less relevant at larger scale, e.g. land use, water consumption and plant complexity.

As expected some sub-criteria were considered less important with increasing plant sizes, e.g. land use and integration simplicity. The reason for land use is that larger plants tend to be more remote than smaller facilities where land prices are typically significantly lower. At larger scale integration simplicity is less relevant as Economy-of-Scale benefits allow the installation of additional equipment. Interestingly, the cleaning water consumption for larger CSP installations was considered slightly less relevant. This was unexpected as usually water is scarcer in remote locations.

3. CSP technology selection

A precondition to obtain hybridization synergies is the ability of the CSP technologies to either directly or indirectly generate steam

Table 3

Parabolic trough data for integration of 80 MW_{th} from CSP into a 200 MW_e CCGT power plant.

| Criteria | Unit | Parabolic trough | | |
|---|--|------------------|----------------------|-------------------|
| High pressure CSP steam to CCGT plant | | Molten salt | Synthetic oil | Water-steam |
| Peak solar to electricity conversion efficiency | % | 26.7 [32] | 21.0 [33] | 22.6 ^a |
| Suitable operation range | MW _{th} | 12 [30] | 80 [2] | 34.2 ^a |
| Maximum site gradient | Degrees | 1 [24] | 1 [24] | 1 [24] |
| Technology maturity | Largest plant with financial approval, MWe | 5.0 [30] | 140.0 [28] | 9.0 [29] |
| Plant complexity | System cycles | 2 | 2 | 1 |
| Integration simplicity into water-steam system | Steam temperature of hybrid plant in °C | 535 [30] | 380 [34] | 400 [35] |
| Land use | m ² /MWh/a | 6 ^a | 8 [36] | 7 ^a |
| Cleaning water consumption | litre/MWhe | 60 ^a | 75 [37] | 70 ^a |
| LCOE | US\$/MWh | 199 ^a | 232 ^b [4] | 202 ^a |

^a Calculation by Juergen Peterseim based on information derived from literature and personal communication.

^b Data from references modified to accommodate inflation and adjusted to 2011 US\$ exchange rate.

Table 4

Solar tower data for integration of 80 MW_{th} from CSP into a 200 MW_e CCGT power plant.

| Criteria | Unit | Solar tower | | | | |
|---|--|----------------------|------------------|-----------------------|-------------------|-----------------------|
| High pressure CSP steam to CCGT | | Molten salt | Water-steam | Air | Syngas | Gas turbine |
| Peak solar to electricity conversion efficiency | % | 27.3 ^a | 28.7 [47] | 31.5 ^a | 55.0 ^a | 47.5 [8] |
| Suitable operation range | MW _{th} | 80 [40] | 80 [47] | 3 [41] | 0.4 [42] | 1.2 [48] |
| Maximum site gradient | Degrees | 12 [16] | 12 [16] | 12 [16] | 12 [16] | 12 [16] |
| Technology maturity | Largest plant with financial approval, MW _e | 110 [45] | 130.0 [47] | 1.5 [41] | 0.3 [42] | 0.25 [49] |
| Plant complexity | System cycles | 2 | 1 | 2 | 1 | 2 |
| Integration simplicity into water-steam system | Steam temperature of hybrid plant in °C | 540 ^a | 565 [47] | 565 [49] | 565 ^b | 565 ^b |
| Land use | m²/MWh/a | 12 [36] | 11 ^a | 8 ^a | 6 ^a | 7 ^a |
| Cleaning water consumption | litre/MWhe | 114 ^a | 108 ^a | 98 ^a | 55 ^a | 64 ^a |
| LCOE | US\$/MWh | 235 ^c [4] | 229 ^a | 359 ^c [50] | 126 ^a | 146 ^c [49] |

^a Calculation by Juergen Peterseim based on information derived from literature and personal communication.

^b Solarized gas turbine exhaust would allow the same Rankine cycle steam parameter as conventional gas turbines.

^c Data from references modified to accommodate inflation and adjusted to 2011 US\$ exchange rate.

| Table 5 |
|---------|
|---------|

| Fresnel data for integration of 80 MW_{th} from CSP into a 200 MW_e CCGT power p |
|--|
|--|

| Criteria | Unit | Fresnel | |
|---|---|-------------------|-----------------------|
| High pressure CSP steam to CCGT | | Saturated steam | Superheated steam |
| Peak solar to electricity conversion efficiency | % | 16.9 ^a | 24.0 ^a |
| Suitable operation range | MW _{th} | 80 [6] | 80 [12] |
| Maximum site gradient | Degrees | 3 [25] | 3 [25] |
| Technology maturity | Largest plant with financial approval, MW _e | 30.0 [6] | 125.0 [53] |
| Plant complexity | System cycles | 1 | 1 |
| Integration simplicity into water-steam system | Steam temperature of hybrid plant in °C | 270 [6] | 450 [52] |
| Land use | m²/MWh/a | 6 [36] | 4 ^a |
| Cleaning water consumption | litre/MWhe | 15 [26] | 10 ^a |
| LCOE | US\$/MWh | 279 ^a | 237 ^b [54] |

^a Calculation by Juergen Peterseim based on information derived from literature and personal communication.

^b Data from references modified to accommodate inflation and adjusted to 2011 US\$ exchange rate.

for the host plants Rankine cycle. That's why this study disregards PV, non-concentrating solar power as well as Dish-Stirling configurations. All data given should be considered relevant for the year 2011 with $\pm 10\%$ sensitivity. Results may change over the next years with new CSP systems maturing quickly and proving its expectations.

Low- and high-temperature integration options are assessed considering line and point focussing systems, including parabolic trough, Fresnel, solar tower and big dish. Depending on the CSP technology different working fluids, ranging from thermal oil, water-steam, molten salts, hot air and syngas, are considered.

In total 9 different technology options are compared for host plants using non-conventional fuels and coal. For combined cycle gas turbine (CCGT) plants two additional options are included in the technology comparison.

Most information derive directly from the literature and but some data were calculated using different literature sources, e.g. using the LCOE of US\$232/MWh for parabolic troughs with thermal oil and applying a 14.2% LCOE cost reduction for molten salt [23]. Similarly, the land use of molten salt and DSG parabolic troughs derives from published data for parabolic trough plants with thermal oil and applying the different plant efficiencies, see Table 3. Other data derive from personal communication with industry experts, e.g. maximum site gradient for parabolic trough [24] and Fresnel systems [25] or cleaning water consumption for Fresnel plants [26].

3.1. Parabolic trough

Parabolic troughs using thermal oil are the most mature of all CSP technologies, with around 1.8 GW capacity installed worldwide. This reflects a 94% CSP market share [27]. Operating plants include the 354° MW_e SEGS I-IX, US and 3×50 MW_e Andasol, Spain

facilities. Most of the units currently under construction use Parabolic troughs with thermal oil typically ranging from 50 MW_e, for example Extremadura Solar Complex, Spain, up to 280 MW_e units, e.g. 2 \times 140 MW_e Solana project, US [28].

However, thermal oil degradation limits steam temperatures to around 380 °C, which poses a barrier to raise efficiencies. For this reason direct steam generation (DSG) as well as molten salts are being investigated with small-scale plants currently being erected. The efficiency of such systems could be significantly higher due to steam parameters well above 450 °C and 100 bar. Higher plant efficiencies also reduce the solar field size and therewith cleaning water consumption as well as land use. The largest DSG plant in operation is the 9 MW_e Suphanaburi project in Thailand [29] while a 5 MW_e molten salt plant is built in Priolo Gargallo, Italy [30]. Currently, both technologies are significantly less mature than thermal oil in parabolic troughs but higher efficiency and lower LCOE prospects make them attractive alternatives for future projects.

Integrated Solar Combined Cycle plants using parabolic troughs are already in operation, e.g. the 75 MW_e equivalent CSP contribution to the FPL Next Generation plant, US [11] or the 22 MW_e equivalent CSP contribution to the Kuraymat plant, Egypt [7]. Also feedwater heating has been tested in the Cameo coal fired power station, US [31].

Table 3 shows the data used in the workshop to compare the different parabolic trough technologies for integration of 80 MW_{th} from CSP into a 200 MW_e CCGT power plant [23].

3.2. Solar tower

Currently, solar towers represent only $\sim 4\%$ of CSP plants in Spain [27]. This low percentage increases risk driven financing costs. However, a few units with DSG, such as the 5 MW_e Sierra

Table 6

Dish data for integration of 80 MW_{th} from CSP into a 200 MW_e CCGT power plant.

| Criteria | Unit | Big dish |
|---|--|------------------|
| High pressure CSP steam to CCGT | | Water-steam |
| Peak solar to electricity conversion efficiency | % | 27.0 [59] |
| Suitable operation range | MW _{th} | 18 ^b |
| Maximum site gradient | Degrees | 12 [16] |
| Technology maturity | Largest plant with financial approval, MW _e | 4.9 [60] |
| Plant complexity | System cycles | 1 |
| Integration simplicity into water-steam system | Steam temperature of hybrid plant in °C | 565 [56] |
| Land use | m*/MWh/a | 12 [36] |
| Cleaning water consumption | litre/MWhe | 114 ^b |
| LCOE | US\$/MWh | 252ª [61] |

^a Calculation by Juergen Peterseim based on information derived from literature and personal communication.

^b Data from references modified to accommodate inflation and adjusted to 2011 US\$ exchange rate.

| Criteria | Importance (1-9) | Criteria groups | Importance (1-9) | Participant: |
|---------------------------------|------------------|-----------------------|---------------------|--------------|
| Solar to electricity efficiency | | | | Assumption |
| Suitable operation range | | Feasibility | | - 10MWe so |
| Maximum site gradient | | | | a) 30MWtl |
| Technology maturity | | | | b) 35MWt |
| Plant complexity | | Risk reduction | | c) 38MWt |
| ntegration simplicity | | | | d) 40MWtl |
| Land use | | Imnact radiation | | e) Steam |
| Fotal water consumption | | | | f) Feedwai |
| | | LCOE | | - 20MWth a |

| umptions: | |
|--------------------------|----------------|
| MWe solid fuel power pla | ant |
| 30MWth Clean Biomass | 480°C @ 100bar |
| 35MWth C&D Timber | 450°C @ 80bar |
| 38MWth RDF | 430°C @ 60bar |
| 40MWth MSW | 400°C @ 40bar |
| Steam reheating | 300°C @ 26bar |
| Feedwater preheating | 270°C @ 40bar |
| MWth additional CSP col | ntribution |

| Criteria | Unit | | Parabolic Tro | hgh | | Solar Tower | | Fre | snel | Dish |
|--|---|-------------|---------------|-------------|-------------|-------------|------|--------------------|----------------------|-------------|
| a) Clean Bio | mass | Molten Salt | Synthetic Oil | Water-Steam | Molten Salt | Water-Steam | Air | Saturated Steam | Superheated steam | Water-Steam |
| Peak solar to electricity conversion efficiency | % | 26.7 | 21.0 | 22.6 | 27.3 | 28.7 | 31.5 | 16.9 | 24.4 | 27.0 |
| Suitable operation range | MWth | 12 | 20 | 20 | 20 | 20 | 3 | 20 | 20 | 18 |
| Maximum site gradient | Degrees | 1 | - | ٢ | 12 | 12 | 12 | 3 | 3 | 12 |
| Technology maturity | Largest plant with financial approval, MWe | 5.0 | 140.0 | 0.6 | 110.0 | 130.0 | 1.5 | 30.0 | 125.0 | 4.9 |
| Plant complexity | System cycles | 2 | 2 | 1 | 2 | 1 | 2 | 1 | 1 | |
| Integration simplicity into water- steam system | Steam temperature of hybrid plant in °C | 480 | 380 | 400 | 480 | 480 | 480 | 270 | 450 | 480 |
| Land use | m2/MWh/a | 9 | 8 | 2 | 12 | 11 | 10 | 9 | 4 | 12 |
| Cleaning water consumption | litre/MWhe | 60 | 76 | 20 | 114 | 108 | 98 | 15 | 10 | 114 |
| LCOE | \$/MWh | 199 | 232 | 202 | 235 | 229 | 359 | 279 | 237 | 252 |
| | | | | | | | | | | |
| Criteria | Unit | Ч | arabolic Trou | lgh | | Solar Tower | | Fre | snel | Dish |
| Integration simplicity into v | water-steam system | Molten Salt | Synthetic Oil | Water-Steam | Molten Salt | Water-Steam | Air | Saturated Steam | Superheated steam | Water-Steam |
| b) Construction & Demolition Timber | Steam temperature of hybrid plant in °C | 450 | 380 | 400 | 450 | 450 | 450 | 270 | 450 | 450 |
| c) Refuse Derived Fuel | Steam temperature of hybrid plant in °C | 430 | 380 | 400 | 430 | 430 | 430 | 270 | 430 | 430 |
| d) Municipal Solid Waste | Steam temperature of hybrid plant in °C | 400 | 380 | 400 | 400 | 400 | 400 | 270 | 400 | 400 |
| e) Steam reheating | Steam temperature of hybrid plant in °C | 300 | 300 | 300 | 300 | 300 | 300 | 270 | 300 | 300 |
| f) Feedwater heating | Steam temperature of hybrid plant in °C | 270 | 270 | 270 | 270 | 270 | 270 | 270 | 270 | 270 |

| Criteria | Unit | | arabolic Trou | lgh | | Solar Tower | | Fre | snel | Dish |
|--|--|-------------|---------------|-------------|-------------|-------------|-----|--------------------|----------------------|------------|
| Integration simplicity into | water-steam system | Molten Salt | Synthetic Oil | Water-Steam | Molten Salt | Water-Steam | Air | Saturated Steam | Superheated steam | Water-Stea |
| b) Construction & Demolition Timber | Steam temperature of hybrid plant in °C | 450 | 380 | 400 | 450 | 450 | 450 | 270 | 450 | 450 |
| c) Refuse Derived Fuel | Steam temperature of hybrid plant in °C | 430 | 08£ | 400 | 430 | 430 | 430 | 270 | 430 | 430 |
| d) Municipal Solid Waste | Steam temperature of hybrid plant in °C | 400 | 08£ | 400 | 400 | 400 | 400 | 270 | 400 | 400 |
| e) Steam reheating | Steam temperature of hybrid plant in °C | 300 | 300 | 300 | 300 | 300 | 300 | 270 | 300 | 300 |
| f) Feedwater heating | Steam temperature of hybrid plant in °C | 270 | 270 | 270 | 270 | 270 | 270 | 270 | 270 | 270 |

Fig. 2. Participant's form for CSP integration into a biomass/waste to energy host plant.



Fig. 3. Percentages of people rating the CSP integration scenarios.

SunTower in the US [38], the 11 MW_e PS10 and 20 MW_e PS20 plants in Spain [39], and molten salt, for example 20 MW_e Gemasolar in Spain [40], are already in operation. Significantly smaller tower systems generating high temperature air for Brayton or Rankine cycles, such as the 1.5 MW_e Solair project in Germany [41], and syngas for gas turbine applications, such as the 300 kW_e SOLASYS project in Israel [42], are tested but currently far from commercial reality due to its small scale. However, these systems have the potential to significantly reduce the LCOE of CSP in the future.

The capacity of current solar tower installations does not exceed 20 MW_e, even though the 120 MW_{th} capacity of the Gemasolar receiver could power a larger turbine but instead charges a maximum 15 h full-load thermal storage [40]. However, significantly larger plants are under construction, for example $3 \times DSG$ solar towers totalling 392 MW_e at Ivanpah [43], and the 110 MW_e Crescent Dunes project in the US using molten salt [44,45]. Once these plants come online and prove their technical and commercial expectations the solar tower technology is very likely to gain a significantly bigger CSP market share. Currently, there are no solar tower hybrid plants in operation worldwide despite the concept being investigated [46].

The efficiency differences for the selected tower technologies results in significantly different values for cleaning water and land use as well as LCOE. Syngas and gas/hot air turbine concepts have the highest reduction potential due to the high efficiency potential of combined cycle systems.

Table 4 shows the data used in the workshop to compare the different solar tower technologies for integration of 80 MW_{th} from CSP into a 200 MW_e CCGT power plant. Solar tower options to provide high temperature air for hot air/gas turbine operation as well as syngas generation were only considered for the CCGT scenarios.

3.3. Fresnel

Similar to solar towers Fresnel is a CSP technology that's currently in the stage of proving its commercial viability with commercial plants being under construction. Currently, only ~2% of the CSP plants in Spain use this technology [27]. Suppliers offer Fresnel plants generating saturated steam, such as the 1.4 MW_e and 30 MW_e Puerto Errado plants in Spain [6], and superheated steam, e.g. 5 MW_e Kimberlina plant in the US, 44 MW_e Kogan Creek Solar Boost in Australia and 250 MW_e (2×125 MW_e) Reliance project in India [51]. Fresnel plants are commercially available for steam temperatures of up to 450 °C [52].

The simplicity of the Fresnel system raises expectations for cost reductions but the technology has the disadvantage that thermal storage is not commercially available yet. However, the current lack of monetary value for energy dispatchability, as seen by the conversion of CSP projects to PV [1], does not require thermal storage for a commercially viable project, as demonstrated by the 250 MW_e Reliance project in India [53]. Currently, large/smart electricity networks have the ability to absorb the limited outputs of CSP plants and higher daytime demand coincides with CSP plant operation.

Fresnel plants are already retrofitted to coal fired power stations for feedwater heating, for example at Liddell in Australia [6], and under construction to provide steam into the cold reheat line, e.g. Kogan Creek power station in Australia and Sundt power station in the US [51].

The efficiency of Fresnel systems can vary significantly with saturated steam systems having inherently low cycle efficiencies compared to the latest Fresnel demonstration projects reaching 500 °C [52]. The higher efficiencies impact LCOE as well as land use and cleaning water consumption. It is worth mentioning that Fresnel systems have the lowest cleaning water requirements of all CSP technologies due to robotic cleaning of the flat mirror panels.

Table 5 shows the data used in the workshop to compare the different Fresnel technologies for integration of 80 MW_{th} from CSP into a 200 MW_e CCGT power plant.

3.4. Big dish technology

Predominantly, paraboloidal dishes are used in conjunction with Stirling engines. As this concept does not offer relevant synergies only dish concepts using water-steam are considered. The big dish technology developed by the Australian National University (ANU) [55] fulfils this criterion, demonstrated steam temperatures of 580 °C [56] and is therefore considered in this assessment.

Despite having the highest concentration ratio and the potential for excellent conversion efficiencies [57] big dish systems do not play a relevant role in new CSP construction projects currently. The reasons for the small uptake are likely to be the very limited number of reference plants worldwide as well as high costs associated with premature technologies. At the ANU a 500 m² big dish is currently being tested and the only multi MW_e standalone project under development with a substantial government grant is the 40 MW_e SolarOasis project in South Australia [58].

With the high steam parameters demonstrated, accurate solar tracking and high concentration ratios the annual efficiency potential of these systems is high but the comparatively small thermal capacity of each big dish requires an extensive high temperature/ pressure piping system with associated thermal losses. A precommercial plant, around 5 MW_e, would be required to demonstrate the benefits and test the concept in a commercial environment. The land use of this technology is high compared to line-focussing systems, but in line with molten salt and DSG solar towers, as the dishes have to be spaced out to avoid shading.

Table 6 shows the big dish data used in the workshop for the integration of 80 MW_{th} from CSP into a 200 MW_e CCGT power plant.

4. Host plant scenarios

The host plant scenarios considered cover all relevant fuels, biomass, waste materials, natural gas and coal, as well as several steam integration options, e.g. feedwater heating, cold reheat line, steam boost and superheated steam to the steam turbine's high pressure stage, see Fig. 4. The CCGT scenario has two additional options, syngas and high temperature air for gas/hot air turbines.

Analysing the different conversion technologies for solid fuels, such as combustion or gasification, and gaseous fuels, such as different gas turbine types, is not part of this study and not relevant as only the steam parameters of any given steam generator are of relevance for the integration of the CSP technology.

Depending on fuel and location the CSP field can act as a fuel saver or power top-up to the host plant. Maximizing the equipment usage is desirable to recover the capital spent as quickly as possible. For this reason the ideal scenario for new plants, using low/no emission fuels, is a power top-up while retrofits are likely to be fuel saver applications substituting expensive/CO₂ intensive fuels.

4.1. Non-conventional fuel fired power plants

In this assessment non-conventional fuels include biomass as well as waste materials. The fuels considered are clean biomass, construction & demolition (C&D) timber, refuse derived fuel (RDF) as well as municipal solid waste (MSW). With falling fuel qualities the superheated steam temperature decrease due to high-temperature corrosion inside the boiler and other technical issues, see Table 7. Steam reheating and feedwater heating is considered for all fuels and all scenarios assuming a 10 MW_e power plant with a 20 MW_{th} CSP contribution.

As seen in Table 7 not all CSP technologies can achieve the steam temperatures required by the different host plants. CSP technologies that cannot produce the desired steam temperature are excluded from the assessment.

The hybridisation of CSP with non-conventional fuels is likely to be a niche market, compared to natural gas, but allows CSP plants to transition from arid desert/semi desert regions to agricultural/urban regions. Being closer to load centres reduces transmission losses, can avoid/defer new transmission infrastructure and enables the access to back-up fuel resources, e.g. agricultural and urban waste materials. Using biomass and waste materials does allow, subject to waste material composition, renewable base-load power generation.

Table 7

Suitability of CSP technologies to reach the required steam temperatures for 10 MW_e non-conventional fuel fired power plants.

| Temperature required | Clean biomass, 480 °C | C&D timber, 450 °C | RDF, 430 °C | MSW, 400 °C | Steam reheating, 300 °C | Feedwater heating, 270 °C |
|-------------------------|-----------------------------|--------------------------|----------------|----------------|-------------------------------|---------------------------------|
| Parabolic trough | | | | | | |
| Molten salt | Yes | Yes | Yes | Yes | Yes ^a | Yes ^a |
| Synthetic oil | No | No | No | No | Yes | Yes |
| Water-steam | No | No | No | Yes | Yes | Yes |
| Solar tower | | | | | | |
| Molten salt | Yes | Yes | Yes | Yes | Yes ^a | Yes ^a |
| Water-steam | Yes | Yes | Yes | Yes | Yes ^a | Yes ^a |
| Air | Yes | Yes | Yes | Yes | Yes ^a | Yes ^a |
| Fresnel | | | | | | |
| Saturated steam | No | No | No | No | No | Yes |
| Superheated steam | No | Yes | Yes | Yes | Yes | Yes ^a |
| Big dish | | | | | | |
| Water-steam | Yes | Yes | Yes | Yes | Yes ^a | Yes ^a |

^a Can realise the steam temperature required but is not considered in assessment as systems are not designed for low steam temperature applications.

4.2. Combined cycle gas turbine plants

Principally, the CSP technologies considered in this section are the same as in Sections 4.1 and 4.3 except that CCGT plants can be hybridized with two additional CSP technologies, solar towers providing hot air [48] and syngas [42]. Different to the other CSP technologies these two technologies have a significantly higher efficiency potential, see Table 4, as they can be used in the toping cycle of the CCGT plant and the bottoming Rankine cycle.

The integration of CSP at different steam temperatures is assessed including steam reheating and feedwater heating, see Table 8. The assumed host plant is a 200 MW_e CCGT unit with a 80 MW_{th} CSP contribution for each scenario.

Integrated solar combined cycle (ISCC) plants have significant potential to further reduce CO₂ emissions from natural gas



Fig. 4. Simplified schematic of CSP integration options into host plants; 1) feedwater heating; 2) high-pressure steam; 3) cold reheat line/steam boost.

combustion as many OCGT & CCGT plants operate in high DNI areas, such as Australia, Africa, India, Arabian Peninsula or the US.

4.3. Coal fired power station

Depending on the construction year of coal fired power stations different steam temperatures and pressure apply. While the majority of the older coal fired power plants operate around 540 °C and 160 bar steam, newer installations operate with 580 °C, supercritical or ultra-supercritical conditions. Currently, no CSP technology demonstrated supercritical steam at pilot plant scale and we therefore chose 580 °C, demonstrated by the big dish technology in Australia [56], as the highest steam temperature in this assessment. The other steam temperature scenarios chosen and the ability of the CSP technologies to meet them are listed in Table 9. The assumed coal plant to be retrofitted with 150 MW_{th} from CSP is a 500 MW_e unit.

The retrofit of CO_2 intensive coal fired power station is one promising option to reduce the carbon footprint of these plants for their remaining lifetime. However, the addition of a CSP steam generator should not be used as a justification to extend the originally predicted operational life of old assets or delay the investment in low emission/renewable energy plants. Such behaviour can easily offset the CO_2 emissions avoided by the CSP add-on. Hence careful consideration has to be given to CSP retrofit proposals.

5. Technology ranking results

The technology ranking in this assessment is based on the criteria rating by the workshop participants/interviewees as well as the data identified for the different CSP technologies. Most CSP data used in the assessment remain unchanged for the different integration options except for the steam temperature and the CSP reference plants (20 MW_{th}, 80 MW_{th} and 150 MW_{th}).

5.1. Energy from non-conventional fuel fired power plants

For the integration of a CSP system into a host plant using clean biomass and generating steam at 480 °C the DSG solar tower scores best, see Fig. 5. The molten salt solar tower technology ranks second and is well within the 10% sensitivity margin. The big dish technology ranks third but is just outside the DSG solar tower sensitivity margin. Only 5 of the 9 CSP technologies are able to reach the steam temperature required and the DSG solar tower

Table 8

Suitability of CSP technologies to reach the required steam temperatures for a 200 MW_e CCGT plant.

Table 9

Suitability of CSP technologies to reach the required steam temperatures for a 500 MW_{e} coal fired plant.

| Temperature required | HP steam I, 580 °C | HP steam II, 540 °C | Steam reheating, 360 °C | Feedwater heating, 270 °C |
|----------------------|-----------------------|------------------------|-------------------------------|---------------------------------|
| Parabolic trough | | | | |
| Molten salt | No | No | Yes ^a | Yes ^a |
| Synthetic oil | No | No | Yes | Yes |
| Water-steam | No | No | Yes | Yes |
| Solar tower | | | | |
| Molten salt | No | Yes | Yes ^a | Yes ^a |
| Water-steam | No | Yes | Yes ^a | Yes ^a |
| Air | Yes | Yes | Yes ^a | Yes ^a |
| Fresnel | | | | |
| Saturated steam | No | No | No | Yes |
| Superheated steam | No | No | Yes | Yes ^a |
| Big dish | | | | |
| Water-steam | Yes | Yes | Yes ^a | Yes ^a |

^a Can realise the steam temperature required but is not considered in assessment as systems are not designed for low steam temperature applications.

scores best because of the good reference situation in this plant size and a higher permissible site gradient.

Host plants using C&D timber, RDF and MSW ideally use Fresnel steam generators to provide superheated steam at 400–450 °C to the joint turbine, see Fig. 5. The reasons for the very high score include the low cleaning water requirements through robotic cleaning of the flat mirror panels, the strong reference situation and the compact solar field minimizing land use. The DSG solar tower ranks second followed by the molten salt solar tower technology. However, their scores are significantly lower and outside the \pm 10% Fresnel sensitivity margin. Several other CSP technologies qualify for these host plant scenarios but their total scores are even lower.

Fresnel with superheated steam also scores best for cold reheat steam followed by parabolic trough systems using thermal oil. For solar feedwater heating applications Fresnel with saturated steam ranks best but is closely followed by parabolic troughs using thermal oil. DSG parabolic trough systems rank third and all three technologies are within the $\pm 10\%$ sensitivity margin.

Currently, the only CSP/biomass hybrid plant under construction is the 22.5 MW_e Termosolar Borges project, Spain [62]. The plant is using parabolic troughs with thermal oil providing 375 °C steam to the turbine [63]. Using this temperature in our assessment Fresnel would have been the chosen option followed by parabolic trough with thermal oil. The maturity of the parabolic trough technology is a significant contributor to its total score but Fresnel

| Temperature required | HP steam, 565 °C | High temperature steam boost, 450 °C | Low temperature steam boost, 350 °C | Steam reheating, 300 °C | Feedwater heating, 260 °C |
|-------------------------------|---------------------|---|--|----------------------------|------------------------------|
| Parabolic trough | | | | | |
| Molten salt | No | Yes | Yes ^a | Yes ^a | Yes ^a |
| Synthetic oil | No | No | Yes | Yes | Yes |
| Water-steam | No | No | Yes | Yes | Yes |
| Solar tower | | | | | |
| Molten salt | No | Yes | Yes ^a | Yes ^a | Yes ^a |
| Water-steam | Yes | Yes | Yes ^a | Yes ^a | Yes ^a |
| Air | Yes | Yes | Yes ^a | Yes ^a | Yes ^a |
| Syngas | Yes | Yes ^b | Yes ^b | Yes ^b | Yes ^b |
| Air to air/gas turbine | Yes | Yes ^b | Yes ^b | Yes ^b | Yes ^b |
| Fresnel | | | | | |
| Saturated steam | No | No | No | No | Yes |
| Superheated steam Big dish | No | Yes | Yes | Yes | Yes ^a |
| Water-steam | Yes | Yes | Yes ^a | Yes ^a | Yes ^a |

^a Can realise the steam temperature required but is not considered in assessment as systems are not designed for low steam temperature applications. ^b Technologies developed to be used in a gas/hot air turbine only.



Fig. 5. CSP technology ranking results for the hybridisation with 10 MW_e non-conventional fuels.

scores better because of its decent reference situation in the \leq 20 MW_{th} scale and additional benefits, such as low land and cleaning water use. The poorer Fresnel reference situation during the planning period of the project and the better bankability of parabolic trough (thermal oil) is likely to have given it a competitive edge in the Termosolar Borges decision making process, considering that this is the first CSP/biomass hybrid plant worldwide and risk mitigation was probably a key criterion. It needs to be mentioned that the steam temperature from the biomass used in the Termosolar Borges project could be at least 50 °C higher, without high-temperature corrosion problems inside the biomass boiler, when choosing a Fresnel or solar tower system.

Currently, there are no CSP hybrid plants under construction using C&D timber, RDF or MSW.

5.2. Combined cycle gas turbine plants

For integrating a 565 °C steam source into the high pressure Rankine cycle of a 200 MW_e CCGT plant the DSG solar tower seems the preferred option, see Fig. 6, very closely followed by the syngas solar tower. Despite limited references the syngas solar tower scores well due to the highly efficient syngas use in the gas turbine and associated low LCOE prospects. However, financing a commercial scale plant is likely to be challenging because of limited



Fig. 6. CSP technology ranking results for the hybridisation with a 200 MW_e CCGT plant.



Fig. 7. CSP technology ranking results for the hybridisation with a 500 MW_e coal fired power plant.

references. The DSG solar tower scores slightly better due to the strong reference situation and technology maturity which simplifies project finance at this point in time. The solar tower providing hot air to a solarised GT ranks third because of its high efficiency and low LCOE potential. All three technologies are within the $\pm 10\%$ sensitivity margin.

For the high temperature steam boost at 450 °C Fresnel (superheated steam) qualifies as a viable option and scored best, due to its reference situation combined with low cleaning water and land use. DSG solar towers rank second with molten salt solar towers third, see Fig. 6. However, both technologies are outside Fresnel's $\pm 10\%$ sensitivity margin.

At steam temperatures <380 °C, as in the low temperature steam boost and cold reheat scenarios, Fresnel (superheated steam) ranks best due to the good reference situation and low water/land use. Parabolic trough (thermal oil) is second and DSG parabolic troughs rank third but with significantly lower scores outside Fresnel's \pm 10% sensitivity margin, see Fig. 6. Point focussing systems are not considered as they are designed to provide steam at temperatures >400 °C. For the 260 °C feedwater heating scenario Fresnel generating saturated steam score best closely followed by parabolic troughs with thermal oil. Fresnel with superheated steam is not considered an appropriate technology for such low steam temperatures.

To subject the assessment process to reality we used the CSP technology chosen for the ISCC plant Kuraymat, Egypt. Through thermal oil a parabolic trough field provides saturated steam at 95 bar to the high pressure drum of the heat recovery steam generator [64]. At 95 bar the saturated steam temperature is 307 °C. Using this data in our assessment Fresnel would have been the chosen technology followed by parabolic trough. Using the very limited Fresnel reference situation from 2001 in this assessment, Kuraymat planning commenced this year, results in parabolic trough with thermal oil being the preferred CSP choice. The latest hybrid project at the dual-fuel (coal and gas) Sundt power station in

the US is using Fresnel as the steam boost technology [51]. This demonstrates the ongoing changes in the CSP industry caused by quickly maturing new technologies.

5.3. Coal fired power station

Feeding steam at 580 °C into the high pressure steam turbine of a coal fired power plant is only possible with 2 of the 9 CSP technologies assessed, big dish and solar tower (hot air). The big dish is the preferred technology for this scenario because of its slightly better reference situation, lower plant complexity and lower LCOE, see Fig. 7. The total score of both technologies is comparatively low as the reference situation is very limited, maturity low and hence LCOE higher than for other technologies.

For providing steam at 540 °C, HP steam II scenario, the solar towers with DSG and molten salt qualify for the assessment and score best because of the better reference situation, higher maturity and lower LCOE than the big dish and hot air solar tower. The DSG technology ranks better than molten salt due to the simpler integration but both technologies are within the $\pm 10\%$ sensitivity margin, see Fig. 7.

The steam reheat and feedwater heating results look very similar to the CCGT results in Fig. 6. Steam temperatures are similar and only the sub-criterion for suitable operation range changes the total scores slightly, see Fig. 7.

At the 750 MW_e Kogan Creek power station in Australia a Fresnel plant providing superheated steam to the cold reheat line is currently under construction [12]. According to our assessment the preferred CSP technology to provide 370 °C steam would have been Fresnel with superheated steam too. Kogan Creek power station is an air cooled unit which indicates water scarcity in the area. The low cleaning water consumption as well as land use are likely reasons for the Fresnel selection. Similarly the results are in line with the recent solar feedwater heating project at Liddell power station in Australia which uses Fresnel with saturated steam [6].

6. Conclusions

The hybridisation of CSP steam generators with different Rankine cycle host plants was assessed in this paper using the Analytical Hierarchy Process with quantitative and qualitative data inputs and the results are in line with the hybrid plants in operation/under construction worldwide.

According to this assessment line focussing systems seem to be the ideal choice for lower temperature steam integration (<400 °C), with Fresnel and parabolic trough (thermal oil) being the most relevant. In the >380 to <450 °C steam integration range Fresnel systems score best. While the parabolic trough technology scores well, because of technology maturity, Fresnel scores well at land use and cleaning water consumption.

The high temperature steam integration of CSP (>450 °C) favours point focussing systems with DSG solar towers scoring best. In 2011 they had the best reference situation of all point focussing technologies considered and operating expertise has existed for several years. Solar towers with molten salt are maturing quickly with the Gemasolar plant, Spain, in operation and larger facilities under construction. They are likely to be the favoured high steam temperature technology for CSP hybrid projects including thermal storage.

This assessment includes general characteristics but results are likely to alter slightly using site specific data, such as land availability or water scarcity. The attitude of investors to technology risk is a criterion that can change the ranking process significantly. The model presented can accommodate such site specific changes to provide appropriate guidance for CSP technology selection.

References

- Beetz B. Whole 1 GW Blythe project to convert to PV. Available from: www. [1] pv-magazine.com/news/details/beitrag/whole-1-gw-blythe-project-toconvert-to-pv_100007604/#axzz2E8gWzRJ9; 2012 [accessed 05.03.2013]
- [2] Flagsol GmbH. Andasol projects, Spain. Available from: www.flagsol.com/ flagsol/english/aaareferences-alt/andasol-projects-spain-alt/andasol-projectsspain.html; 2012 [accessed 05.03.2013]. Torresol Energy Investments S.A. Gemasolar. Available from: www.
- [3] torresolenergy.com/TORRESOL/gemasolar-plant/en; 2011 [accessed 05.03.2013].
- Hinkley J, Curtin B, Hayward J, Wonhas A, Boyd R, Grima C, et al. Concentrating [4] solar power - drivers and opportunities for cost-competitive electricity. Sydney, Australia: CSIRO & Aurecon Australia Pty Ltd; 2011.
- International Energy Agency. Technology roadmap concentrating solar [5] power. Paris 2010.
- [6] Novatec Solar GmbH. Projects. Available from: www.novatecsolar.com/8-1-Projects.html; 2012 [accessed 05.03.2013].
- Brakmann G, Mohammad F, Dolejsi M, Wiemann M. Construction of the ISCC [7] Kuraymat, Presented at the SolarPACES 2009, Berlin, Germany 2009,
- [8] Turchi CS. Erbes M. Gas turbine/solar parabolic trough hybrid designs. Presented at the ASME TurboExpo 2011, Vancouver, Canada 2011,
- Hayward JA, Graham PW, Campbell PK. Projections of the future costs of [9] electricity generation technologies: an application of CSIRO's Global and Local Learning Model (GALLM). Newcastle, Australia: CSIRO; 2011.
- [10] Pérez Á, Torres N. Solar parabolic trough biomass hybrid plants: a costefficient concept suitable for places in low irradiation conditions. In: Solar-Paces conference 2010, Perpignan, France 2010.
- [11] Florida Power & Light Company. FPL's Martin next generation solar energy center. Available from: www.fpl.com/environment/solar/pdf/Martin.pdf; 2010 [accessed 05.03.2013].
- Energy CS. Kogan Creek solar boost. Available from: www.kogansolarboost. [12] com.au; 2012 [accessed 05.03.2013].
- Fichtner Solar GmbH. References. Available from: http://www.fichtnersolar. [13] com/fichtnersolar_references.htm#yazd01; 2012 [accessed 05.03.2013]
- [14] The World Bank. Nurturing low carbon economy in Morocco. Available from: http://go.worldbank.org/KIN4DEUC70; 2010 [accessed 05.03.2013].
- [15] Nixon JD, Dey PK, Davies PA. Which is the best solar thermal collection technology for electricity generation in north-west India? Evaluation of options using the analytical hierarchy process. Energy 2010;35:5230-40.
- [16] Volvo Group Australia Pty Ltd. Slope ability of a 8 imes 4 Volvo truck with a water tank capacity of 14,200 litres. pers. comm. 17th of June, Brisbane, Australia 2011.
- [17] Akash BA, Mamlook R, Mohsen MS. Multi-criteria selection of electric power plants using analytical hierarchy process. Electric Power Systems Research 1999;52:29-35.

- [18] Ramanathan R, Ganesh LS. Energy resource allocation incorporating qualitative and quantitative criteria; an integrated model using goal programming and AHP. Socio-economic Planning Sciences 1995;29:197-218.
- Triantaphyllou F. Multi-criteria decision making methods: a comparative study. Dordrecht, The Netherlands: Kluwer Academic Publishers; 2000. [20] Saaty TL. Decision making with the analytic hierarchy process. International
- Journal of Services Sciences 2008;1:83-98. Saaty TL, Ozdemir MS. Why the magic number seven plus or minus two. [21]
- Mathematical and Computer Modelling 2003;38:233-44. [22] Sniezek JA, Henry RA. Accuracy and confidence in group judgment. Organi-
- zational Behavior and Human Decision Processes 1989:43:1-28. [23] Kearney D, Herrmann U, Nava P, Kelly B, Mahoney R, Pacheco J, et al.
- Assessment of a molten salt heat transfer fluid in a parabolic trough solar field. Journal of Solar Energy Engineering 2003;125:170-6. [24]
- Zarza E. Parabolic trough with direct steam generation. pers. comm. 30th of May, Almería, Spain 2011. Novatec Solar GmbH. Permissible site gradient for Fresnel steam generators. [25]
- pers. comm. 20th of May, Karlsruhe, Germany 2011 [26]
- Novatec Solar GmbH. Water consumption of Fresnel plants. pers. comm. 20th of Mav. Karlsruhe, Germany 2011. [27]
- IRENA. Renweable energy technologies: cost analysis series concentrating solar power. Abu Dhabi: International Renewable Energy Agency; 2012.
- [28] Abengoa Solar SA. Plants under construction. Available from: www. abengoasolar.com/corp/web/en/nuestras_plantas/plantas_en_construccion/; 2011 [accessed 05.03.2013].
- [29] Solarlite GmbH. "TS 2" Suphanburi Province, Thailand. Available from: www. solarlite.de/en/project_suphanburi.cfm; 2011 [accessed 05.03.2013].
- [30] Falchetta M, Mazzei D, Crescenzi T, Merlo L. Design of the Archimede 5 MW molten salt parabolic trough solar plant. Presented at the Solar PACES 2009, Berlin, Germany 2009.
- [31] Xcel Energy Inc. Colorado integrated solar project. Available from: www. xcelenergy.com/staticfiles/xe/Corporate/Environment/09-10-204% 20CameoSolarFS[1]%20V%204%20111109.pdf: 2010 [accessed 05.03.2013]
- [32] Falchetta M. Parabolic troughs plants using molten salt. pers. comm. 05th of July, Rome, Ital. y; 2011.
- [33] Philibert C. The present and future use of solar thermal energy as a primary source of energy. Paris, France: International Energy Agency; 2005.
- [34] Zarza E, López C, Cámara A, Martinez A, Burgaleta JI, Martín JC, et al. Almería GDV: the first solar power plant with direct steam generation. In: 14th international SolarPACES symposium, Las Vegas, USA 2008.
- [35] Zarza E, Valenzuela L, León J, Hennecke K, Eck M, Weyers HD, et al. Direct steam generation in parabolic troughs: final results and conclusions of the DISS project. Energy 2004;29:635-44.
- [36] Müller-Steinhagen H, Trieb F. Concentrating solar power: a review of the technology. Ingenia; February 2004. 43-50.
- [37] NREL. Concentrating solar power commercial application study: reducing water consumption of concentrating solar power electricity generation. Report to Congress. Golden, U.S.: U.S. Department of Energy; 2009
- [38] eSolar Inc. Sierra SunTower – a new blueprint for solar energy. Available from: http://www.esolar.com/sierra_fact_sheet.pdf; 2011 [accessed 05.03.2013].
- [39] Abengoa Solar SA. Operating facilities Spain. Available from: http://www. abengoasolar.com/corp/web/en/nuestras_plantas/plantas_en_operacion/ espana/index.html; 2011 [accessed 05.03.2013].
- [40] García E, Calvo R. One year operation experience of Gemasolar plant. Presented at the SolarPACES 2012 conference, Marrakech, Morocco 2012.
- Koll G, Schwarzbozl P, Hennecke K, Hartz T, Schmitz M, Hoffschmidt B. The [41] solar tower Jülich - a research and demonstration plant for central receiver systems. In: 15th SolarPACES conference, Berlin, Germany 2009.
- [42] Stein W, Edwards J, Hinkley J, Sattler C. Natural gas: solar-thermal steam reforming. In: Encyclopedia of electrochemical power sources. Amsterdam: Elsevier; 2009. p. 300-12.
- [43] BrightSource Energy Inc. Ivanpah overview. Available from: http://ivanpahsolar. com/about; 2011 [accessed 05.03.2013].
- Gould WR. SolarReserve's 565 MWt molten salt power towers. Presented at [44] the SolarPACES 2011 conference, Granada, Spain 2011.
- [45] SolarReserve LLC. Crescent dunes. Available from: www.solarreserve.com/ what-we-do/csp-projects/crescent-dunes; 2012 [accessed 05.03.2013].
- [46] Peterseim JH, Tadros A, White S, Hellwig U. Integrated solar combined cycle plants using solar power towers to optimise plant performance. In: Solar-PACES 2012 conference, Marrakech, Morocco 2012
- [47] NREL. Ivanpah solar electric generating station. Available from: www.nrel.gov/ csp/solarpaces/project_detail.cfm/projectID=62; 2012 [accessed 05.03.2013].
- [48] Ávila-Marín AL. Volumetric receivers in solar thermal power plants with central receiver system technology: a review. Solar Energy 2011;85:891-910. [49] ORMAT, CIEMAT, DLR, SOLUCAR, TUMA. EUR 21615-SOLGATE - solar hybrid
- gas turbine electric power system. Luxembourg: European Commission; 2005. Meinecke W, Kiera M, Wehowsky P. 30 MWe PHOEBUS feasibility study: re-
- sults of system engineering. Solar Energy Materials, 1991;24:95-107. [51] AREVA group. AREVA solar. Available from: www.areva.com/EN/solar-220/
- areva-solar.html; 2012 [accessed 05.03.2013]
- Morin G, Mertins M, Kirchberger J, Selig M. SUPERNOVA construction, control & performance of steam superheating linear Fresnel collector. Presented at the SolarPACES Conference 2011, Granada Spain 2011.
- [53] AREVA group, India: AREVA awarded contract by Reliance to build Asia's largest concentrated solar power installation. Available from: www.areva.com/EN/

news-9299/india-areva-awarded-contract-by-reliance-to-build-asia-s-largest-concentrated-solar-power-installation.html; 2012 [accessed 05.03.2013].

- [54] de Lalaing J, Millan R, Görlich F, Bautista E. Potential of solar add-ons to coalfired power plants – evaluation of different combination concepts. Presented at the Power-Gen Europe, Amsterdam, Netherlands 2010.
- [55] Lovegrove K, Burgess G, Pye J. A new 500 m² paraboloidal dish solar concentrator. Solar Energy 2011;85:620–6.
- [56] Siangsukone PL. Transient simulation and modelling of a dish solar thermal power system. PhD, Faculty of Engineering and Information Technology, The Australian National University, Canberra, Australia; 2005.
- [57] Lovegrove K, Luzzi A. Solar thermal power systems, encyclopedia of physical science and technology. 3rd ed., vol. 15. San Diego: Academic Press; 2002.
- [58] Australian Renewable Energy Agency. ARENA projects. Available from: www. arena.gov.au/programs/projects/index.html; 2012 [accessed 05.03.2013].
 [59] Lovegrove K, Taumoefolau T, Paitoonsurikarn S, Siangsukone P, Burgess G,
- [59] Lovegrove K, Taumoefolau T, Paitoonsurikarn S, Siangsukone P, Burgess G, Luzzi A, et al. Paraboloidal dish solar concentrators for multimegawat

power generation. Presented at the Solar World Congress, Goteborg Sweden 2003.

- [60] Stolpe GW. Solar plant 1: the formula for success. United States: Alternative Sources Energy; 1986.
- [61] Rabl A. High temperature solar thermal power. In: Second Sede Boqer symposium on solar electricity production. Israel: Ben-Gurion University of the Negev; 1987.
- [62] Abantia. Borges installed solar thermal power generation turbine, a turning point for its implementation. Available from: www.abantia.com/noticiasprensa/en_noticias/28/the-installation-of-the-solar-field-termosolar-borgesthe-first-plant-to-combine-thermal-and-biomass-has-started; 2012 [accessed 05.03.2013].
- [63] Cot A, Amettler A, Vall-Llovera J, Aguilo J, Arque JM. Termosolar Borges: a thermosolar hybrid plant with biomass. In: Third international symposium on energy from biomass and waste, Venice, Italy 2010.
- [64] NEM BV. Al Kuraymat Egypt. Available from: www.nem-group.com/EN/ projects/solar_power/6/al_kuraymat/8/; 2011 [accessed 05.03.2013].

4.4 Techno-economic optimisation

The following publications analyse different CSP–EfB and CSP–EfW hybridisation options in regards to their technical, economic and environmental performance. Various CSP, EfB and EfW technologies are considered to answer the following research questions:

- What are the best concepts for combining CSP with EfB and EfW systems?
- What are the cost differences between CSP–EfB and CSP–EfW hybrids and CSP-only plants in Australia?

Each publication investigates a different technical concept. The first one examines the generation of steam by CSP and EfB steam generators with identical parameters (Section 4.4.1) and the second one examines the use of biomass to externally superheat steam from a parabolic trough plant (Section 4.4.2). The first concept is suitable for areas with significant biomass or waste feedstock resources as the concept is based on the EfB/EfW component operating continually throughout the year while the CSP component provides valuable peak capacity during the day and, with TES, the evening and morning periods. The external superheating concept is designed to minimise solid feedstock consumption as it only requires a small energy input to raise the already high steam parameters from the parabolic trough plant. Therefore this concept is ideal for utility scale plants in higher DNI areas where biomass and waste feedstock availability declines as shown in Section 4.2. The concept is specifically designed to improve the economic viability of the currently dominant parabolic trough technology with thermal oil as its efficiencies are constrained by thermal oil degradation at temperatures above 400 °C. Other technologies, such as solar towers, do not need to use biomass or waste feedstocks to reach steam temperatures in excess of 500 °C and high pressures. The use of CSP for air and feedwater heating, as well as cold reheat steam, is not considered in the analyses as these options do not allow a significant solar share in hybrid plants.

4.4.1 Identical steam parameter from CSP and biomass components

This publication analyses various CSP, EfB, and EfW technology pairings for hybrid plants to identify the best ones in regards to their technical, economic and environmental performance.

In addition to the process diagram shown in Figure 2 in this paper, two additional process diagrams for scenarios 16 and 17 + 3 h TES can be found in the appendix.





Available online at www.sciencedirect.com



Solar Energy 99 (2014) 203-214

Solar Energy

www.elsevier.com/locate/solener

Hybridisation optimization of concentrating solar thermal and biomass power generation facilities

Juergen H. Peterseim^{a,*}, Udo Hellwig^b, Amir Tadros^c, Stuart White^d

^a Institute for Sustainable Futures, University of Technology, Sydney, Level 11, UTS Building 10, 235 Jones Street, Ultimo, NSW 2007, Australia

^b ERK Eckrohrkessel GmbH, Germany

^c The Engineering Excellence Group-Mechanical Engineering Leader, Laing O'Rourke, Australia ^d Institute for Sustainable Futures, University of Technology, Sydney, Australia

Received 30 April 2013; received in revised form 10 October 2013; accepted 22 October 2013 Available online 10 December 2013

Communicated by: Associate Editor Michael Epstein

Abstract

Recently, the first concentrating solar power-biomass hybrid power plant commenced operation in Spain and the combination of both energy sources is promising to lower plant investment. This assessment investigates 17 different concentrating solar power-biomass hybrid configurations in regards their technical, economic and environmental performance. The integration of molten salt thermal storage is considered for the best performing hybrid configuration. While thermal storage can increase plant output significantly even 7 h full-load thermal storage plants would generate the majority of the electricity, 70%, from the biomass resource.

Only mature technologies with references >5 MWe are considered in this assessment to ensure that the scenarios are bankable. The concentrating solar power technologies selected are parabolic trough, Fresnel and solar tower while the biomass systems include grate, fluidised bed and gasification with producer gas use in a boiler.

A case study approach based on the annual availability of 100,000 t of wood biomass is taken to compare the different plant configurations but the results are transferable to other locations when updating site and cost conditions. Results show that solar tower–biomass hybrids reach the highest net cycle efficiency, 32.9%, but that Fresnel-biomass hybrids have the lowest specific investment, AU\$ 4.5 m/ MWe. The investment difference between the 17 scenarios is with up to 31% significant. Based on the annual electricity generation CSP– biomass hybrids have an up to 69% lower investment compared to standalone concentrating solar power systems. The scenario with the best technical performance, being solar tower and gasification, is at this point in time not necessarily the best commercial choice, being Fresnel and fluidised bed, as the lower Fresnel investment outweighs the additional electricity generation potential solar towers offer. However, other scenarios with different benefits rank closely.

© 2013 Elsevier Ltd. All rights reserved.

Keywords: Concentrating solar power-biomass hybrid plants; Energy from biomass; Parabolic trough; Fresnel; Solar tower

Abbreviations: CSP, concentrating solar power; PPA, power purchase agreement; PT, parabolic trough; ST, solar tower; F, Fresnel; IRR, internal rate of return on investment; TO, thermal oil; DSG, direct steam generation; MS, molten salts; TS, thermal storage; DNI, direct normal irradiance; AU\$, Australian Dollar (AU\$/US\$ = 0.96).

* Corresponding author. Tel.: +61 2 9514 4878; fax: +61 2 9514 4941. *E-mail address:* JuergenHeinzMartin.Peterseim@uts.edu.au (J.H. Peterseim).

0038-092X/\$ - see front matter © 2013 Elsevier Ltd. All rights reserved. http://dx.doi.org/10.1016/j.solener.2013.10.041

1. Introduction

Concentrating solar power (CSP)–biomass hybrid plants are a well accepted solution for comparatively low cost base-load/dispatchable renewable energy but are a niche market considering the limited areas with a sufficiently high direct normal irradiance, >1700 kW h/m²/ year, and biomass resources. Countries such as Australia, Spain, Italy, Greece, Thailand, India, and Brazil are, amongst others, prime candidates for such projects as these countries have locations that meet both criteria. To maximize the commercial viability and efficient biomass use the power plant concepts should be as efficient as possible considering the economic realities of obtaining finance for such projects.

Recently, the first CSP-biomass hybrid plant commenced operation near Barcelona, Spain (Protermosolar, 2012), which proves that such concepts work technically and are bankable solutions. Despite other CSP-biomass configurations being investigated in the past, including Fresnel and tower systems, no other projects commenced construction at this point in time.

This paper investigates different possible CSP-biomass hybrid configurations with the aim of identifying the best in regards to technical, economic, and environmental performance, such as cycle efficiency, investment, and CO_2 abatement potential. The efficient use of biomass is not only necessary to maximize plant output, but also to optimize usage as biomass has to be purchased competitively against other users, such as pulp and paper industry. Understanding the different CSP-biomass hybrid options will enable project developers to concentrate on the most promising ones therewith increasing chances to successfully implement such projects.

All scenarios investigated in this paper assume that the biomass boiler is operating constantly at full capacity, that the CSP system provides additional capacity during the daytime when electricity demand/prices are typically higher in Australia (Lekovic et al., 2011), and that the CSP and biomass systems can operate independent of each other by providing identical steam flows/parameters to the joint turbine. To minimize plant investment molten salt thermal storage is considered only for one high temperature scenario as its cost are still comparatively high and the biomass component can generate electricity at lower cost during the night and extended cloud coverage. Future work could investigate CSP-biomass hybrids with thermal storage in more detail by considering other storage media, such as pressurized water or steam, and tank configurations, such as single tanks.

2. Current hybrid proposals

First proposals to combine CSP with biomass/waste materials using dish systems were investigated briefly in the 1980s (McDonald, 1986). However, due to technical and financial issues no plants were built. It took more than two decades before the first commercial CSP-biomass hybrid plant, Termosolar Borges 22.5 MWe (Morell, 2012), commenced operation near Lleida, ca. 150 km west of Barcelona, Spain, see Fig. 1. The plant is located further north than any other CSP project in Spain and uses the mature trough technology with thermal oil (Cot et al., 2010). The solar field generates saturated steam at 40 bar and the biomass boilers superheat this steam to 520 °C (Morell, 2012). The steam temperature is well selected to avoid high-temperature corrosion on the superheaters from this type of fuel, wood see Fig. 1. To further optimize the concept its steam pressure could be raised >40 bar (Morell, 2012).

Several other studies investigated the hybridisation of parabolic trough plants with biomass considering capacities from 2 MW_{th} to 107 MWe but no other project has yet commenced construction (Pérez and Torres, 2010; Schnatbaum, 2009; Nixon et al., 2010; California Energy Comission, 2008). These studies concentrate on integrating a biomass fired steam boiler into the CSP plant's watersteam cycle but also consider a biomass fired heater in the thermal oil or molten salt cycle.

Alternatively, Fresnel systems have been investigated for hybridisation with biomass and waste materials (Peterseim et al., 2012; Rojas et al., 2010; Spliethoff et al., 2010; Nixon et al., 2012). The benefit of using Fresnel would be steam temperatures up to 500 °C (Fluri et al., 2012) and subsequent higher conversion efficiencies. However, no reference plants yet exist for this CSP configuration.

Solar towers with direct steam generation (DSG) and molten salts are being investigated in combination with biomass (Peterseim et al., 2013) as are solar towers using a volumetric air receiver (Coelho et al., 2012). Due to the limited reference situation of high-temperature air tower systems and the higher complexity of such a system it is expected that solar towers with DSG or molten salts are easier to finance. Another option to combine high-temper-



Fig. 1. First CSP-biomass plant under construction near Lleida, Spain (left) and biomass fuel (right).

ature air systems with biomass is an integrated gasification combined cycle plant with the CSP component heating the gas turbine air downstream the compressor to 800 °C and syngas combustion to increase the temperature to 1150 °C, similar to SOLGATE project (ORMAT, 2005) but with biomass derived syngas rather than natural gas. The use of the gas turbine exhaust to operate a Rankine cycle would make such a process very efficient. However, with both technologies not being tested individually at commercial scale it is very unlikely to secure finance for such a hybrid concept and it is therefore not considered.

Further bankable concepts discussed include the use of CSP for air and feedwater heating as well as the generation of identical steam parameters as the host plant. However, none of the mature concepts considers CSP steam >430 °C which limits the conversion efficiency of the biomass feedstock.

3. Methods

The assessment aims to identify the best CSP-biomass hybrid plant configuration by investigating the technical, economic and environmental performance of 17 different scenarios. A case study approach is chosen to compare the scenario differences but the assessment is transferable to other locations when adapting site and cost conditions.

3.1. Modeling approach

All scenarios were thermodynamically and economically modeled with Thermoflex version 23.0. The software is well established for Rankine cycle systems and is widely used in academia and industry for actual CSP and conventional power plant modeling. All efficiency information is peak net cycle efficiencies based on $\eta = MWe/(MW_{th} \text{ biomass} + MW_{th} \text{ CSP}).$

Despite higher investment and lower cycle efficiency all 17 scenarios are modeled with air rather than water-cooling as water is typically a scarce commodity in high DNI areas with competing uses from agriculture and horticulture.

Both the biomass and CSP steam generators are supplied with the same feedwater source and generate identical steam flows/parameters, see Fig. 2. This allows the independent operation of both components and ensures acceptable part-load efficiency at night when only the biomass boiler is providing steam to the turbine. Thermal storage (TS) is not considered for all scenarios as in 2011 TS costs were still high, AU\$ 90 kW hth (Hinkley et al., 2011), and without additional incentives, such as a financial reward for capacity value, the biomass plant can provide electricity at lower cost when solar irradiance conditions prevent CSP operation. In case the 2020 TS cost predictions, AU\$ 22/ kW h_{th} (Hinkley et al., 2011), are met the technology would complement CSP-biomass hybrid plants well and the assessment therefore includes, technical, economic and environmental information on 1-7 h molten salt TS configurations. To maximize CSP generation without thermal storage all scenarios are designed with a solar multiple of 1.4.

Investment data derived from the Thermoflex database, which is considered quite accurate as modeling a 50 MWe Andasol type parabolic trough plant with 7.5 h of thermal storage in Spain resulted in the published € 300 m investment (Solar Millennium AG, 2006). Australian conditions such as higher labor and equipment prices are considered in all scenarios. With an assumed plant commissioning in 2016 the Thermoflex solar field investment is lowered by 10% to reflect 2015 pricing based on a reasonable learning curve (IRENA, 2012). Biomass boilers are not a standard Thermoflex component and its database is therefore less reliable. To maximize cost accuracy industry prices were obtained for this equipment (ERK Eckrohrkessel GmbH, 2013). Different to the Termosolar Borges plant this assessment considers only one and not two biomass boilers. Single biomass boilers are available to run >10 MWe steam turbines, achieve low part-load operation and reduce investment.

The economic assumptions are a plant lifetime of 30 years, 91.3% capacity factor for the biomass component, varying capacity factors for the CSP components (trough 18.3%, Fresnel 15.5% and tower 19.2%), commissioning in 2016, 65% debt finance, 8% debt interest rate, 7% discount rate, and biomass price of AU\$ 50 t for forestry residues. A power purchase agreement (PPA) of AU\$ 145 MW h is used, which is significantly lower than the AU\$ 225 MWh a 64 MWe standalone CSP plant would need at Mildura (IT Power, 2012). The modeling includes capital and operational costs as well as annual escalation rates for inflation (3%), fuel (2%), electricity (3%) and water prices (4%). All investments shown are the owners total project cost consisting of the engineering, procurement and construction (EPC) price and 9% owner's soft cost covering permits, project management, legal and finance aspects of a power plant project. Payback periods, net present values and internal rates of return on investment (IRR) derive from the total project cost.

3.2. Technology selection

The aim of this assessment is to evaluate CSP-biomass hybrid concepts that can be financed immediately and therefore includes only CSP and biomass technologies that have >5 MWe reference plants. For CSP this includes parabolic trough with thermal oil (Solar Millennium AG, 2006), DSG (Krüger et al., 2012) and molten salt (Falchetta et al., 2009), solar towers with DSG (Abengoa Solar, 2011) and molten salts (García and Calvo, 2012), and Fresnel with DSG. These technologies are already assessed and found suitable for hybridisation with biomass (Peterseim et al., 2013). Fresnel using molten salts is promising and could reach working fluid temperatures of 550 °C (Smith and Cohn, 2012) but the technology is currently at proto-







Fig. 3. 5 MWe plants using parabolic trough with molten salt (left) (Falchetta et al., 2009) and DSG (middle) (Krüger et al., 2012) as well as 10 MWe biomass gasification plant with syngas combustion in a boiler (right) (Photograph courtesy ERK Eckrohrkessel GmbH, Berlin, Germany).

type scale and therefore not considered. Similarly volumetric air solar towers are being investigated with biomass (Coelho et al., 2012) but reference plants do not yet exist. Some data are provided to compare these concepts with selected CSP technologies.

Biomass conversion technologies include grate and fluidised bed systems (Spliethoff, 2010) as well as gasification units burning the producer gas in a boiler (ERK Eckrohrkessel GmbH, 2013). Grate systems have most references while fluidised bed systems are also mature for woody biomass (Spliethoff, 2010). Gasification references do exist for up to 20 MWe standalone biomass plants (Eckrohrkessel GmbH, 2012) and even larger retrofits to existing boilers (Metso Corporation, 2011). Therefore gasification with syngas firing in a boiler is considered mature for the plant capacity considered in this study.

Gasification systems producing a syngas for open or combined cycle engine/gas turbine operation are excluded as this technology is not considered mature yet for plant capacities >5 MWe. Fig. 3 presents references for the less mature but considered molten salt and DSG parabolic trough systems as well as biomass gasification.

3.3. Case study site selection and biomass fuel

The site selected for this analysis is near Mildura, Australia. The area has a DNI of 2198 kW h/m²/year (Bureau of Meteorology, 2012), which is better than the 1800 kW h/ m²/year of the Termosolar Borges plant in Spain (Morell, 2012), and is home to one of Australia's prime horticulture and agriculture regions. In general the Mildura region is quite similar to the area where the Termosolar Borges plant is operating. Biomass feedstocks from horticulture, forest and urban residue streams as well as straw from agriculture are available and could provide the required energy equivalent of 100,000 t/a wood feedstock at 8.8 MJ/kg (forestry residues with 40% moisture content) for each scenario. Agroforestry could be another option for future fuel supply. The different nature and composition of biomass affects the power station's annual biomass demand and Table 1 provides information on other relevant feedstocks in the Mildura region. Several power plants using multiple biomass feedstocks, such as straw and wood chips, exist and the technology can be considered mature, for example 30 MWe Holstebro plant in Denmark (Andersen et al.,

| Biomass quantities re | quired to | substitute | 100,000 t/ | a of forestr | y residues. |
|-----------------------|-----------|------------|------------|--------------|-------------|
|-----------------------|-----------|------------|------------|--------------|-------------|

| Biomass material | Annual demand (t) | Calorific value (MJ/kg) | Moisture content (%) |
|---------------------|-------------------|----------------------------|-------------------------|
| Straw | 66,200 | 13.3 | 13 |
| Construction timber | 57,100 | 15.4 | 9 |
| Wood pellets | 52,700 | 16.7 | 8.7 |

1992) and 25 MWe Sangüesa plant in Spain (Acciona Energía, 2002).

The maximum mean ambient temperature for Mildura is 23.8 °C (of Meteorology, 2013) and the plant's air cooling system is optimized for this temperature. Despite lower efficiencies air-cooling is considered as water is a precious commodity in the region and a power station would compete for it with other users, e.g. agriculture.

Mildura is in the state of Victoria which has one of the highest retail electricity prices in Australian Energy Market Commission (2011). This maximizes plant revenue and was another criterion for selecting this site. The electricity could be consumed locally but also exported into the national electricity market via existing transmission infrastructure.

4. Water-steam cycle analysis

Several options exist to increase the efficiency of a Rankine cycle systems, such as optimized feedwater heating, blowdown heat recovery and flue gas condensation. This section briefly outlines the effects of these options on the net cycle efficiency and provides the base case scenario (scenario 1) for the following CSP-biomass hybrid assessment.

4.1. Feedwater preheating

Three models are considered using single, double and triple feedwater heating. The comparison is based on a CSP-biomass hybrid plant with steam flow of 83 t/h at 80 bar and 380 °C generating 16.9 MWe, see Table 2. Rais-

Table 2Feedwater heating effects on net plant efficiency.

| Number of feedwater heaters | Feedwater temperature (°C) | Plant net capacity (MWe) | Net plant efficiency (%) |
|--------------------------------|-------------------------------|-----------------------------|-----------------------------|
| 1 | 143 | 16.9 | 28.5 |
| 2 | 162 | 17.1 | 28.9 |
| 3 | 183 | 17.3 | 29.3 |

ing the high pressure feedwater temperature by 20 °C with bleed steam from the turbine increases the net cycle efficiency by 0.4% generating an additional 200 kW. Further increasing feedwater temperature to 180 °C increases the cycle efficiency by another 0.4%.

While utility scale power plants have at least 5 feedwater heater stages this is not practical for smaller plants but technically possible. The cost and additional complexity are the main barriers.

4.2. Blowdown heat recovery

Every boiler has a blowdown system and depending on the water quality and the operation philosophy blowdown quantities in industrial boilers range from 1% to 6%. Utility boiler blowdown rates are significantly lower, e.g. 0.1% reducing the effect of blowdown heat recovery.

With blowdown temperatures' being close to the saturation steam temperature it offers good potential for additional feedwater heating. The following data are based on 1% boiler blowdown as well as the aforementioned 3 stage feedwater concept and steam parameters. Recovering the boiler blowdown at 85 bar and using it in the high-pressure feedwater heater and deaerator reduces the steam turbine bleed slightly, therewith increasing net power generation by 26 kW.

The potential for blowdown heat recovery is not particularly high and therefore the concept is rarely implemented. However, a few installations exist, such as the Augusta pulp and paper plant in the US. This plant increased the boiler feedwater temperature by 10 °C and achieved a simple payback of only 6 months (US Department of Energy 2002).

4.3. Flue gas condensation

Generally, flue gas temperatures should be kept above the dew point to avoid acid corrosion on the boiler's low temperature heating surfaces. However, due to rising fuel costs and higher efficiency expectations some power stations installed flue gas condensers, such as the energy from waste plant in Malmö, Sweden (Friotherm AG, 2008) and the coal fired power station Schwarze Pumpe in Germany (Borsig Steinmüller GmbH, 2007). The concept can increase the cycle efficiency by up to 1% but depends on the power plant configuration and use of the low temperature heat, e.g. feedwater/air preheating or district heating/ cooling. One reason for the limited implementation of flue gas condensers is the comparatively high investment as the small temperature differences between water and flue gas lead to larger heating surfaces which also have to be acid resistant.

Reducing the flue gas exit temperature from 120 °C to 70 °C, Malmö stack temperature (Friotherm AG, 2008), and using this heat to replace bleed steam from the low-pressure turbine generates 204 kW additional power (compared to blowdown scenario) taking the net plant capacity to 17.5 MWe. This equals a plant net efficiency of 29.6%.

Despite its promising potential this assessment does not consider fuel gas condensation as the comparatively high

Table 3 Technical comparison of hybrid options.

| Scenario | CSP technology | Primary CSP working fluid | Biomass technology | Live steam temperature (°C) | Live steam pressure (bar) | Plant net capacity (kWe) | Peak net efficiency (%) |
|---------------|---------------------|------------------------------|-----------------------|--------------------------------|------------------------------|-----------------------------|----------------------------|
| 1 - PT + TO | Parabolic trough | Thermal oil | Grate | 380 | 80 | 17,330 | 29.3 |
| 2 - PT + TO | Parabolic trough | Thermal oil | Fluidised bed | 380 | 80 | 17,430 | 29.5 |
| 3 - PT + DSG | Parabolic trough | DSG | Fluidised bed | 400 | 90 | 17,890 | 30.3 |
| 4 - F + DSG | Fresnel | DSG | Fluidised bed | 400 | 90 | 17,910 | 30.4 |
| 5 - PT + DSG | Parabolic trough | DSG | Fluidised bed | 450 | 100 | 18,530 | 31.5 |
| 6 - F + DSG | Fresnel | DSG | Fluidised bed | 450 | 100 | 18,540 | 31.5 |
| 7 - ST + DSG | Solar tower | DSG | Fluidised bed | 450 | 100 | 18,600 | 31.5 |
| 8 - PT + MS | Parabolic trough | Molten salt | Fluidised bed | 500 | 110 | 19,060 | 32.2 |
| 9 - F + DSG | Fresnel | DSG | Fluidised bed | 500 | 110 | 19,150 | 32.5 |
| 10 - ST + DSG | Solar tower | DSG | Fluidised bed | 500 | 110 | 19,170 | 32.5 |
| 11 - ST + MS | Solar tower | Molten salt | Fluidised bed | 500 | 110 | 19,080 | 32.3 |
| 12 - PT + MS | Parabolic trough | Molten salt | Fluidised bed | 525 | 120 | 19,340 | 32.7 |
| 13 - ST + DSG | Solar tower | DSG | Fluidised bed | 525 | 120 | 19,430 | 33.0 |
| 14 - ST + MS | Solar tower | Molten salt | Fluidised bed | 525 | 120 | 19,370 | 32.8 |
| 15 - PT + MS | Parabolic trough | Molten salt | Gasification | 540 | 130 | 19,400 | 32.8 |
| 16 - ST + DSG | Solar tower | DSG | Gasification | 540 | 130 | 19,540 | 33.2 |
| 17 - ST + MS | Solar tower | Molten salt | Gasification | 540 | 130 | 19,410 | 32.9 |

Table 4 Economic comparison of CSP-biomass hybrid scenarios.

| Scenario | Total plant investment (mAU\$) | Specific plant investment (AU\$/MW) | Specific cost reduction (%) | Years payback | Net Present Value (AU\$m) | Internal rate of return on investment (%) |
|---------------|-----------------------------------|--|--------------------------------|------------------|------------------------------|--|
| 1 - PT + TO | 115 | 6.7 | | 15.8 | 26.0 | 8.2 |
| 2 - PT + TO | 115 | 6.6 | 0 | 15.7 | 26.9 | 8.3 |
| 3 - PT + DSG | 115 | 6.4 | 2 | 15.3 | 31.1 | 8.5 |
| 4 - F + DSG | 85 | 4.8 | 27 | 9.7 | 50.4 | 10.8 |
| 5 - PT + DSG | 115 | 6.2 | 5 | 14.6 | 36.6 | 8.9 |
| 6 - F + DSG | 86 | 4.6 | 29 | 9.1 | 55.4 | 11.2 |
| 7 - ST + DSG | 97 | 5.2 | 20 | 10.5 | 51.7 | 10.4 |
| 8 - PT + MS | 118 | 6.2 | 5 | 14.6 | 38.0 | 8.9 |
| 9 - F + DSG | 86 | 4.5 | 31 | 8.6 | 59.8 | 11.5 |
| 10 - ST + DSG | 97 | 5.1 | 22 | 9.9 | 56.3 | 10.7 |
| 11 - ST + MS | 99 | 5.2 | 21 | 10.4 | 53.8 | 10.4 |
| 12 - PT + MS | 120 | 6.2 | 5 | 14.4 | 39.3 | 9.0 |
| 13 - ST + DSG | 98 | 5.0 | 23 | 9.7 | 57.9 | 10.8 |
| 14 - ST + MS | 100 | 5.2 | 21 | 10.2 | 55.5 | 10.5 |
| 15 - PT + MS | 120 | 6.2 | 5 | 14.3 | 39.8 | 9.1 |
| 16 - ST + DSG | 98 | 5.0 | 23 | 9.6 | 58.8 | 10.9 |
| 17 - ST + MS | 101 | 5.2 | 21 | 10.2 | 55.6 | 10.5 |

Table 5

Renewable energy generation, CO2 abatement potential.

| Scenario | Annual generation potential (MWh) | CO ₂ abatement potential for forestry residues wood (t/a) | CO ₂ abatement potential for straw (t/a) |
|-----------------------|-----------------------------------|--|--|
| 1 - PT + TO | 83,220 | 47,800 | 50,700 |
| 2 - PT + TO | 83,650 | 48,100 | 51,000 |
| 3 - PT + DSG | 85,870 | 49,500 | 52,400 |
| 4 - F + DSG | 83,760 | 48,200 | 51,100 |
| 5 - PT + DSG | 88,940 | 51,500 | 54,400 |
| 6 - F + DSG | 86,730 | 50,000 | 52,900 |
| 7 - ST + DSG | 90,070 | 52,200 | 55,100 |
| 8 - PT + MS | 91,510 | 53,100 | 56,000 |
| 9 - F + DSG | 89,580 | 51,900 | 54,800 |
| 10 - ST + DSG | 92,810 | 53,900 | 56,800 |
| 11 - ST + MS | 92,380 | 53,700 | 56,600 |
| 12 - PT + MS | 92,860 | 54,000 | 56,900 |
| 13 - ST + DSG | 94,090 | 54,800 | 57,700 |
| 14 - ST + MS | 93,790 | 54,600 | 57,500 |
| 15 - PT + MS | 93,130 | 54,100 | 57,000 |
| 16 - ST + DSG | 94,630 | 55,100 | 58,000 |
| 17 - ST + MS | 94,010 | 54,700 | 57,600 |
| 17 - ST + MS + 1 h TS | 99,570 | 58,300 | 61,200 |
| 17 - ST + MS + 3 h TS | 104,520 | 61,400 | 64,300 |
| 17 - ST + MS + 5 h TS | 107,820 | 63,500 | 66,400 |
| 17 - ST + MS + 7 h TS | 110,490 | 65,200 | 68,100 |

investment would reduce the plant's economic viability in Australia's traditionally low cost electricity market. However, the additional capacity given can be added to the scenarios discussed later on.

5. Results

The technical, economic and environmental results shown for the 17 different scenarios are based on triple feedwater heating and blowdown heat recovery. Some information is provided on the impact of molten salt TS and less mature plant concepts. To clearly identify the scenarios and its different results in Tables 3–5 an abbreviation code is used where for example 1-PT+TO stands for scenario 1 using the parabolic trough technology with thermal oil. Each CSP technology is paired with the biomass technology outlined in Table 3, either grate, fluidised bed or gasification.

5.1. Technical analysis

This assessment compares 17 different scenarios with steam parameters from 375 °C at 80 bar to 540 °C at 130 bar, see Table 3. The base case scenario 1 assumes

the integration of biomass into a conventional parabolic trough plant with both the biomass and CSP steam generator generating identical steam parameters. The highest steam parameter scenarios compare solar towers (DSG and molten salt) and parabolic trough (molten salt) with biomass gasification and syngas firing in a boiler.

Scenario 16 using a solar tower with DSG reaches the highest net cycle efficiency, 33.2%, and plant capacity, 19,540 kW, compared to all other scenarios, see Table 3. Scenario 13 ranks second with scenario 17 being third. The efficiency increase from scenario 1 to 16 is 13%. The two other scenarios in this steam parameter category (15 and 17) have slightly lower efficiencies, -0.4% and -0.3%, as the primary molten salt and secondary watersteam cycle leads to slightly higher parasitic losses.

The results for the 525 °C and 500 °C categories have the DSG scenarios 13, 10 and 9, as the best concepts for CSP biomass hybrids. When comparing the annual generation scenarios 13 and 10 perform best in these groups as solar towers have higher capacity factors than Fresnel, see Table 5. It is interesting to notice that the change from parabolic trough with thermal oil (scenario 2) to DSG (scenario 3) with only slightly higher steam parameters increases cycle efficiency by almost 1%. As seen in Table 3 cycle efficiency increases stronger from 375 to 500 °C scenarios, +3.2%, than in the 500–540 °C scenarios, +0.7%, which is in line with expectations as the efficiency curve flattens with higher steam parameters.

The differences between the combustion systems is only marginal (scenarios 1–2 and 14–15) with fluidised bed and gasification systems having lower unburned carbon contents in the ash than grates, <1% compared to <3%, and lower excess air requirements, 20% compared to 25%. The main benefit of the gasification system with syngas firing in a boiler is the high ash retention in the gasifiers and subsequently lower high-temperature corrosion problems on the boiler's superheater banks.



Fig. 4. CSP-biomass hybrid plant investment, NPV and IRR.



Fig. 5. CO₂ abatement potential vs. IRR with and without CSP capacity value.



Fig. 6. Seasonal net electricity generation for a scenario 17 CSP-biomass hybrid plants with different hours of thermal storage.

5.2. Economic analysis

Different to the best technical scenario (16) the commercially most attractive one at this point in time is scenario 9. which uses Fresnel, with an IRR of 11.5% (scenario 16 ranks third with 10.9% IRR), see Table 4. Compared to the base case scenario 1 the IRR increases from 8.2% to 11.5%, a 40% increase. Despite a lower capacity factor and the scenario only ranking 8th in the cycle efficiency criterion the significantly lower specific investment. AU\$ 4.5 m/MWe compared to AU\$ 5 m/MWe for scenario 16, leads to the highest IRR and NPV. The lower Fresnel capacity factor and investments identified are in line with other studies (Morin et al., 2012; Narula and Gleckman, 2012). Only scenario 9 meets the 11.5% IRR criterion typically required by utilities (Sargent & Lundy LLC Consulting Group, 2003). The even lower cycle efficiency scenario 6, also Fresnel, ranks second despite its annual generation being 7900 MWh lower than scenario 16. The low Fresnel investment is a strong incentive for the technology.

The scenarios with the highest annual generation (16, 13 and 17) have, due to the higher investments, slightly lower IRR's (10.9%, 10.8% and 10.5%) which might still be of interest to conservative investors. The mature parabolic troughs with thermal oil (scenarios 1–2) have the lowest IRR (8.2% and 8.3%) and NPV (AU\$26 m and AU\$26.9 m) due to the higher specific investment and lower cycle efficiencies. Therefore reducing the efficiency of the biomass component to match the CSP steam parameters is not recommended, particularly when considering that the biomass generates the majority of annual output, see Fig. 6.

With changing CSP technologies the specific hybrid plant investment decreases by 31% from scenario 1 to 9, which is substantial reduction without encountering higher technology risk as all technologies are demonstrated and supplied by well established companies which provide performance guarantees. Scenarios 6 and 4, also Fresnel, have the second and third lowest specific investment. The realization of a scenario 9 plant would not only directly stimulate the local economy but also indirectly therewith increasing the overall project stimulus to AU\$ 114 m, when assuming 29% international procurement (Caldés et al., 2009) and a value multiplier of 1.86, combined from 1.42 multiplier for biomass plant including fuel procurement (Gan and Smith, 2007) and a 2.3 CSP multiplier (Caldés et al., 2009). That is a significant investment for a rural region such as Mildura.

Adding any 1-7 h molten salt TS configuration to scenario 17, technology with highest IRR and commercially proven TS, reduces IRR and NPV but increases annual electricity generation and CO₂ abatement potential significantly, see Figs. 4 and 5. However, when including an additional value of AU\$ 21 MWh for CSP electricity dispatchability, which has been discussed in the Australian context recently and consists of AU\$ 20 MWh capacity value and AU\$ 1 MWh ancillary services (IT Power, 2012), the IRR increases for 1 h TS to 10.7%, for 3 h TS to 10.6%, for 5 h to 10.3% and for 7 h TS to 10.1%, see Fig. 4. This means that up to 3 h TS the IRR is higher than the 10.5% for scenario 17 without TS but larger TS reduces the IRR due to higher cost and a slowing capacity factor increase, see Fig. 6. If Fresnel references with molten salt and TS would exist, currently being investigated with molten salt temperatures up to 550 °C and 16 h TS (Smith and Cohn, 2012), it would be the least cost TS scenario.

With the biomass providing the majority of the annual electricity its capacity factor should be kept a high as possible and actual plants do reach 91.3% (8000 h/a), e.g. 25 MWe Sangüesa in Spain (Acciona Energía, 2002). However, such high capacity factors require good plant maintenance and fuel logistics which might not be available everywhere and at all times. A lower biomass component capacity factor of 88% would reduce the scenario 9 IRR to 11.1% and to 10.8% when assuming 85%.

This assessment is based on equal finance for all technologies considered. However, the economic results strongly depend on the financiers perceived technology risk. Finance institutions familiar with a particular CSP technology would be able to provide lower cost finance for this technology compared to others therewith altering the investment and IRR results provided in Table 4. These risk factors can vary significantly between financiers and have to be investigated on a project by project basis.

5.3. Environmental analysis

Due to the higher cycle efficiency and best capacity factor the solar tower scenarios 16, 13 and 17 achieve the highest annual renewable electricity generation with up to 94,630 MWh, see Table 5. The annual output of the least performing scenario (1) is almost 14% lower.

The 2010 CO₂ intensity of electricity generation in Australia was 841 kg CO₂/MW h (IEA, 2012) and 2050 modeling for the Australian Government Treasury projects that the national annual electricity demand increases to 400 TWh while CO_2 emission from electricity generation fall to around 68 Mt (ROAM Consulting Pty Ltd, 2011), which equals a CO_2 intensity of 162 kg/MWh. This is not going to be a linear decline (\sim 740 kg CO₂/MWh in 2020, \sim 720 kg CO₂/MWh in 2030 and \sim 500 kg CO₂/MWh in 2040) with the strongest CO_2 reductions from 2040-50. For the 2016-45 plant lifetime considered this leads to an average CO₂ intensity of Australia's electricity sector of \sim 640 kg/MWh. However, it needs to be highlighted that uncertainties around the actual CO2 intensity development exist with electricity demand currently falling rather than increasing and the share of coal fired generation decreasing faster than expected (pitt&sherry, 2012). Biomass production and transport emissions have to be deducted from the CO₂ abatement potential and this assessment considers $54 \text{ kg CO}_2/t$ for plantation softwood residues and $25 \text{ kg CO}_2/\text{t}$ for straw based on a 50 km transport distance (Farine et al., 2011). The forestry residues scenario 16 has the highest annual CO₂ abatement potential, up to 55,100 t, closely followed by scenarios 13 and 17, see Table 5. Scenario 1 has with up to 47,800 t the lowest annual CO₂ abatement potential. Adding TS would significantly increase the CO_2 abatement potential to 65,200 t/a (scenario 17 with 7 h TS) but reduce IRR to 9.6% unless energy dispatchability is valued or TS costs decrease significantly, see Fig. 5. In addition to a higher CO_2 abatement TS would increase the annual biomass conversion efficiency

by reducing steam turbine part-load operation. The CO_2 abatement potential for a straw fired CSP–biomass hybrid is slightly higher due to its lower production and transport emissions.

With the CSP and biomass components sharing a joint steam turbine the turbine operates at part-load when CSP is not delivering its design capacity, therewith reducing cycle efficiency. The part-load steam turbine operation at night and during extended cloud coverage reduces the biomass capacity in scenario 17 from 9.7 MWe with CSP to 9 MWe without CSP contribution, which equals a lost annual biomass output of 2580 MWh (-3.3%) when comparing it to a 19.4 MWe standalone biomass plant. However, the loss through part-load operation has to be compared to the annual use of 100,000 t forestry residues in a standalone biomass plant which has an inherently lower efficiency, 28.6% compared to 32.9% in scenario 16, and there with a lower net plant capacity of 8.8 MWe. Considering this the CSP-biomass hybrid generates and additional 4890 MWh/year. This benefit increases with 1 h TS to 5660 MWh, with 3 h TS to 5720 MWh, with 5 h TS to 5770 MWh and with 7 h TS to 5820 MWh as the steam turbine operates more hours at its design point.

Even when considering TS the majority of the electricity is generated through the biomass component as its capacity factor is significantly higher, see Fig. 6. The plant generation over the year is not constant decreasing up to 16% in winter compared to summer. The CSP summer-winter generation difference is with 57% significantly larger and the 4.4% higher biomass plant output in winter compared to summer, caused by lower ambient temperatures, can only partly offset this. The lower seasonal generation variation of a hybrid plant is beneficial for grid operators and could be further decreased by adding larger TS or extra capacity natural gas burners to the biomass boiler.

From a specific land use land use perspective the Fresnel-biomass scenarios perform best with ~ 1.1 ha/MWe followed by parabolic troughs with ~ 1.3 ha/MWe and solar towers with ~ 2 ha/MWe. The biomass plant requires 2 ha in all scenarios. Adding TS would increase land use up to 3.4 ha/MWe (scenario 17 with 7 h TS). The plant footprint is not a main criterion for utility scale CSP as such plants are usually remote, but it is relevant for distrib-



Fig. 7. Summer (left) and winter (right) electricity generation in the morning and evening hours for a CSP only and hybrid plant.

uted CSP systems as these are typically closer to urbanizations therewith facing higher land prices.

With all plant configurations designed with air-cooling water consumption is already minimized with only mirror cleaning and boiler blowdown being relevant water consumers. While the boiler blowdown depends on the water quality, and is the same percentage for all scenarios (1%), water consumption for mirror cleaning varies with the CSP technology. Cleaning robots are developed to minimize water consumption and it seems likely that Fresnel systems require least cleaning water followed by troughs and tower systems.

5.4. Hybrid vs. stand-alone cost comparison

The hybridization of CSP with biomass, or other energy sources, can reduce the CSP investment of the assessed scenarios by 12% compared to a standalone CSP plant. The specific 2015 investment for the 9.6 MWe CSP share of scenario 9, Fresnel technology, is AU\$ 5.7 MWe while a 9.6 MWe standalone CSP without TS would require a specific investment of AU\$ 6.4 MWe. The lower investment results from sharing plant equipment, such as steam turbine, condenser and auxiliary equipment. Solar tower hybrids could also share plant infrastructure by using the biomass plant's stack to support the receiver. In this model the solar tower height is 60 m which is twice the height required for the stack of the biomass plant. Adapting the stack to support receiver equipment is possible and has been discussed for integrated solar combined cycle (Peterseim et al., 2012) and CSP-biomass hybrid plants (Peterseim et al., 2013). Tower construction is 5% of a solar tower project investment (Hinkley et al., 2011) and not all of that could be saved but 3% are realistic at this plant scale.

Different to comparing the specific investment a comparison of plants with identical annual outputs shows the CSP hybrid benefits more clearly. To meet the annual output a of a scenario 9 CSP-biomass hybrid (AU\$ 86 m) a 66 MWe standalone Fresnel plant would be required with a 2015 investment of AU\$ 280 m. To meet the demand of scenario 16 (AU\$ 98 m) a 56 MWe standalone solar tower costing AU\$ 295 m would be required. TS can reduce the plant capacity and a 3 h TS molten salt solar tower plant would require a capacity of 34 MWe with an associated investment of AU\$ 255 m. While the specific investment reduction of CSP-biomass hybrids is only 12% the investment reduction to generate the same annual electricity output is up to 69%. All three standalone CSP options mentioned would require a PPA in excess of AU\$ 200 MWh to be commercially viable in Australia.

In addition to lower investment the annual electricity generation of a CSP-biomass hybrid plant is slightly higher, 0.4%, compared to a CSP standalone because the hybrid plant does not require the 25% minimum steam turbine load to commence/cease operation as the turbine is already/remains in operation with steam from the biomass boiler. Considering a minimum solar field steam output of 10% means that even small stream flows can be converted into electricity. Fig. 7 shows the electricity generation during the summer/winter morning and evening hours for a scenario 16 plant. It can be observed that the CSP steam in a hybrid plant is used circa 24 min longer in summer and winter than in the CSP only plant. Assuming no additional start-ups and shut-downs during the day this adds up to an additional annualized output of 279 MWh which equals additional revenue of AU\$ 40,425. When considering one additional steam turbine start/stop every second day the additional annual output increases to 418 MWh and AU\$ 60,640 additional revenue.

6. Conclusion

The assessment shows that at the moment the best technical and environmental CSP-biomass hybrid configuration, being solar tower and gasification, is not the best commercial choice, being Fresnel and fluidised bed. While the efficiency differences for the 17 scenarios reach 13% the investment variations are with 31% significantly larger. The results also show that, based on identical annual electricity generation, the CSP-biomass hybrid plant investment is up to 69% lower compared to a standalone CSP plant without thermal storage. This makes CSP-biomass hybrids commercially viable at a power purchase price of AU\$ 145 MWh compared to >AU\$ 200 MWh for standalone CSP systems in Australia while having a more stable annual generation curve.

The integration of thermal storage can increase the annual generation of CSP-biomass hybrid plants up to 17% (7 h TS) but currently requires a capacity value payment to be competitive with a no storage plant. With thermal storage cost expected to decrease significantly by 2020 future CSP-biomass plants are likely to have thermal storage. However, even with extensive thermal storage the majority of the electricity, 70%, still derives from the biomass resource. The first CSP-biomass hybrid in Spain using parabolic troughs will provide financiers and operators with more expertise about such concepts and therewith reduce risk and plant finance.

Despite the benefits CSP-biomass hybrid plants offer it has to be acknowledged that the sites for such plants are limited as only few locations have a sufficiently high DNI and biomass resource. It is therefore a niche solution worth exploring further as its lower cost can pave the way for CSP into traditionally low cost electricity markets, such as but not limited to Australia.

References

- Abengoa Solar, Operating facilities Spain. 2011. <www.abengoasolar.com/corp/web/en/nuestras_plantas/plantas_en_operacion/espana/ index.html> (accessed 09.10.13)
- Acciona Energía. 2002. Biomass plant from straw combustion in Sangüesa. <www.acciona-energia.com/media/219313/ACCIONA_Sanguesa biomass plant_EN.pdf> (accessed 07.10.13)
- Andersen, E.J., Oksen, F., Olesen, J.V., 1992. Power plant Holstebro/ Struer with separately fired superheater. Eckrohrkessel Licensee Conf..

- Office of Industrial Technologies Energy Efficiency and Renewable Energy. 2002. Boiler blowdown heat recovery project reduces steam system energy losses at augusta newsprint, US Department of Energy. <www.nrel.gov/docs/fy02osti/31697.pdf> (accessed 10.10.13)
- Australian Energy Market Commission. 2011. Possible future retail electricity price movements: 1 July 2011 to 30 June 2014, Final Report, Sydney.
- Babcock Borsig Steinmüller GmbH. 2007. Powerise[®] Schwarze Pumpe unit A. <www.bbs.bilfinger.com/fileadmin/power_steinmuller/Mediathek/Referenzblatt_schwarze_Pumpe_Redesign_engl_Web.pdf> (accessed 09.10.13)
- Bureau of Meteorology. 2012. Australian monthly average hourly global and direct normal solar exposure gridded data, Melbourne.
- Bureau of Meteorology. 2013. Climate statistics for Australian locations summary statistics Mildura Airport. www.bom.gov.au/climate/averages/tables/cw_076031.shtml (accessed 09.10.13)
- Caldés, N., Varela, M., Santamaría, M., Sáez, R., 2009. Economic impact of solar thermal electricity deployment in Spain. Energy Policy 37 (5), 1628–1636.
- California Energy Commission, 2008. San Joaquin Solar 1 & 2 Power Plant Licensing Case. Sacramento, California.
- Coelho, B., Schwarzbözl, P., Oliveira, A., Mendes, A., 2012. Biomass and central receiver system (CRS) hybridization: volumetric air CRS and integration of a biomass waste direct burning boiler on steam cycle. Sol. Energy 86 (10), 2912–2922.
- Cot, A., Amettler, A., Vall-Llovera, J., Aguilo, J., Arque, J.M., 2010. Termosolar borges: a thermosolar hybrid plant with biomass. Third Int. Symp. Energy Biomass Waste.
- ERK Eckrohrkessel GmbH. 2012. References biomass. <<u>www.eckrohrkessel.com/Eng/index_e.htm</u>> (accessed 10.10.13)
- ERK Eckrohrkessel GmbH, 2013. Personal Communication with ERK Eckrohrkessel GmbH. Germany, Berlin.
- Falchetta, M., Mazzei, D., Crescenzi, T., Merlo, L., 2009. Design of the archimede 5 MW molten salt parabolic trough solar plant. SolarPaces.
- Farine, D., O'Connell, D.A., John Raison, R., May, B.M., O'Connor, M.H., Crawford, D.F., Herr, A., Taylor, J.A., Jovanovic, T., Campbell, P.K., Dunlop, M.I.A., Rodriguez, L.C., Poole, M.L., Braid, A.L., Kriticos, D., 2011. An assessment of biomass for bioelectricity and biofuel, and for greenhouse gas emission reduction in Australia. GCB Bioenergy.
- Fluri, T.P., Lude, S., Lam, J., Morin, G., Paul, C., Platzer, W.J., 2012. Optimization of live steam properties for a linear Fresnel power plant. In: SolarPaces Conference.
- Friotherm AG. 2008. Waste-to-energy plant Sysav Malmö, enhancing the overal energy efficiency with 2 unitop 28C heat pump units. <www.friotherm.de/webautor-data/41/sysave006_uk.pdf> (accessed 09.10.13)
- Gan, J., Smith, C.T., 2007. Co-benefits of utilizing logging residues for bioenergy production: the case for east Texas, USA. Biomass and Bioenergy 31 (9), 623–630.
- García, E., Calvo, Roberto, 2012. One year operation experience of gemasolar plant. SolarPaces Conf..
- Hinkley, J., Curtin, B., Hayward, J., Wonhaus, A., Boyd, R., Grima, C., Tadros, A., Hall, R., Naicker, K., Mikhail, A., 2011. Concentrating solar power – Drivers and Opportunities for Cost-competitive Electricity. CSIRO & Aurecon Australia Pty Ltd., Sydney, Australia.
- IEA. 2012. CO2 Emissions from fuel combustion highlights, Paris.
- IRENA. 2012. Renewable Energy Technologies: cost analysis series concentrating solar power, Abu Dhabi, UAE 1 (2/5).
- Power, I.T., 2012. Realizing the Potential of Concentrating Solar Power in Australia. Canberra, Australia.
- Krüger, D., Krüger, J., Pandian, Y., O'Connell, B., Feldhoff, J.F., Karthikeyan, R., Hempel, S., Muniasamy, K., Hirsch, T., Eickhoff, M., Hennecke, K., 2012. Experience with direct steam generation at the Kanchanaburi solar thermal power plant. SolarPaces Conf..
- Lekovic, S., Priftakis, P., Donoghue, K., Rigzin, T., Cole, S., Pryor, B., Simic, K., 2011. Electricity Gas Australia. Energy Supply Association of Australia, Melbourne, Australia.

- McDonald, C.F., 1986. A hybrid solar closed-cycle gas turbine combined heat and power plant concept to meet the continuous total energy needs of a small community. J. Heat Recover. Syst. 6 (5), 399–419.
- Metso Corporation. 2010. Metso to deliver world's largest biomass gasification plant to Vaskiluodon Voima in Vaasa. <www.metso.com/ in/india_news.nsf/WebWID/WTB-110614-22577-E2917?OpenDocument> (accessed 25.06.13)
- Morell, D., 2012. CSP BORGES. The World's First CSP plant hybridized with biomass. In: CSP Today USA 2012–6th Concentrated Solar Thermal Power Conference.
- Morin, G., Dersch, J., Platzer, W., Eck, M., Häberle, A., 2012. Comparison of linear Fresnel and parabolic trough collector power plants. Sol. Energy 86 (1), 1–12.
- Narula, M., Gleckman, P., 2012. Central receiver vs linear Fresnel: a comparison of direct molten salt CSP plants. SolarPaces Conf..
- Nixon, J.D., Engineer, Z., Hossain, A.K., Davies, P.A., 2010. A Hybrid Solar–Biomass Power Plant for India. World Renewable Energy Congress XI, Abu Dhabi, UAE.
- Nixon, J.D., Dey, P.K., Davies, P.A., 2012. The feasibility of hybrid solar-biomass power plants in India. Energy 46 (1), 541-554.
- ORMAT, CIEMAT, DLR, SOLUCAR, TUMA. 2005. O. for O.P. of the E. Communities, and E. and S.D. Energy. SOLGATE – Solar Hybrid Gas Turbine Electric Power System. European Commission, Brussels, 2005.
- Pérez, Á., Torres, N., 2010. Solar parabolic trough biomass hybrid plants: a cost – efficient concept suitable for places in low irradiation conditions. SolarPaces Conf..
- Peterseim, J.H., Tadros, A., White, S., Hellwig, U., Klostermann, F., 2012. Concentrated solar power/energy from waste hybrid plants – creating synergies. In: SolarPaces Conference.
- Peterseim, J.H., White, S., Tadros, A., Hellwig, U., 2012. Integrated solar combined cycle plants using solar power towers to optimise plant performance. In: SolarPaces Conference.
- Peterseim, J.H., Tadros, A., White, S., Hellwig, U., Landler, J., Galang, K., 2013. Solar tower-biomass hybrid plants – maximizing plant performance. In: SolarPaces Conference.
- Peterseim, J.H., White, S., Tadros, A., Hellwig, U., 2013. Concentrated solar power hybrid plants, which technologies are best suited for hybridisation?. Renew. Energy 57 520–532.
- pitt&sherry. 2012. Cedex report carbon emissions index.
- Protermosolar. 2012. Abantia and Comsa Emte set up the first facility in the world that combines solar thermal and biomass energy. <www.protermosolar.com/saladeprensa/?q=en/content/abantia-and-comsa-emte-set-first-facility-world-combines-solar-thermal-and-bio-mass-energy> (accessed 03.10.13)
- ROAM Consulting Pty Ltd. 2011. Additional projections of electricity generation in Australia to 2050, Brisbane.
- Rojas, M., Hellwig, U., Peterseim, J.H., Harding, J., 2010. Combining solar thermal power systems with solid fuels in new plants. In: SolarPaces Conference.
- Sargent & Lundy LLC Consulting Group, 2003. Assessment of Parabolic Trough and Power Tower Solar Technology Cost and Performance Forecasts. National Renewable Energy Laboratory, Chicago, Illinois, USA.
- Schnatbaum, L., 2009. Biomass utilization for co firing in parabolic trough power plants. SolarPaces Conf.
- Smith, R., Cohn, L., 2012. Fresnel Meets Salt, Sun & Wind Energy, Bielefeld, Germany, pp. 48–49
- Solar Millennium, AG., 2006. Financing granted to solar millennium AG for andasol 2, the second parabolic trough power plant in Europe. <www.solarmillennium.de/Archiv/Presse/Meldungen/Archiv_2006/ Meldungen_2006_,lang2,51,235.html> (accessed 09.10.13)
- Spliethoff, H., 2010. Power Generation from Solid Fuels, 1st ed. Springer, Heidelberg.
- Spliethoff, H., Kaeding, N., Murer, M., Alonso-Herranz, E., Gohlke, O., 2010. Combining energy from waste and concentrated solar power: new solutions for sustainable energy generation. In: International Solid Waste Association Conference.

4.4.2 External CSP steam superheating with biomass

The following publication investigates the use of various biomass and waste feedstocks to raise steam parameters in a conventional parabolic trough plant, through external steam superheating. The analysis considers the technical, economic and environmental impacts of the different scenarios.

In addition to the process diagram shown in Figure 2 in this paper, two additional process diagrams for other biomass and waste feedstocks, scenarios 1 air-cooled and 3 air-cooled, can be found in the appendix.

Energy Conversion and Management 77 (2014) 784-793

Contents lists available at ScienceDirect

Energy Conversion and Management

journal homepage: www.elsevier.com/locate/enconman

Increasing the efficiency of parabolic trough plants using thermal oil through external superheating with biomass



Juergen H. Peterseim^{a,*}, Amir Tadros^{b,1}, Udo Hellwig^{c,2}, Stuart White^d

^a Institute for Sustainable Futures, University of Technology Sydney, Level 11, UTS Building 10, 235 Jones Street, Ultimo, NSW 2007, Australia ^b Laing O'Rourke, Australia

^c ERK Eckrohrkessel GmbH, Germany

^d Institute for Sustainable Futures, University of Technology, Sydney, Australia

ARTICLE INFO

Article history: Received 6 May 2013 Accepted 8 October 2013

Keywords: Concentrating solar power Solar hybrid plants Parabolic trough Biomass External superheating

ABSTRACT

It is well understood that the cost of concentrating solar power (CSP) will need to decrease quickly to ensure competitiveness with photovoltaic (PV) systems and other forms of power generation. Research and development on CSP plant components is crucial in order to reduce costs but is typically time consuming. New CSP plant concepts combining proven technologies with CSP represent another option that can be implemented quickly.

This paper investigates the use of several biomass materials to externally superheat steam in conventional parabolic trough plants. Currently, parabolic trough plants are easiest to finance and external steam superheating can overcome the lower efficiencies compared to other CSP technologies. Seven scenarios, each air and water cooled, with steam parameters ranging from 380 °C at 100 bar to 540 °C at 130 bar have been modeled, and the results presented here are based on a 50 MWe plant with 7.5 h molten salt thermal storage.

Our results show that the peak solar to electricity net efficiency increases up to 10.5% while the specific investment can decrease immediately from AU\$8.2m/MWe to AU\$6.3m/MWe, a 23.5% reduction. That is significant considering the expected 17–40% CSP cost reduction targets by the end of this decade. The modeling shows that even major fuel and water price changes are significantly less relevant than small changes in the agreed electricity purchase price.

The technical, economic and environmental analysis reveals that external superheating with biomass can provide significant benefits, is able to use a variety of fuels and despite a limited global market, could immediately enable the implementation of several hundred MWe of CSP capacity at lower cost.

© 2013 Elsevier Ltd. All rights reserved.

1. Introduction

Compared to the 98 GWe of worldwide installed photovoltaic (PV) capacity; concentrating solar power (CSP) is lagging behind with only 3 GWe installed capacity at the end of 2012 [1]. This significant difference is expected to remain with predictions of 308 GWe PV compared to 12 GWe CSP capacity by 2018 [1]. Therefore new technologies, innovative power plant concepts, and learning curve advancements are required to increase the CSP global market share

¹ Tel.: +61 (0)2 9455 4822.

² Tel.: +49 (0)30 8977 4623.

and competitiveness against PV. The conversion of the first 500 MWe and potentially second 500 MWe phase of the 1000 MWe Blythe Solar Power Project, US, from parabolic trough to PV is the most outstanding example so far [2]. New technologies such as solar tower and Fresnel plants promise to increase efficiency and reduce cost of electricity but financing such plants is still more complicated than well-proven parabolic trough plants.

Because of its long track record parabolic trough plants had a 94% market share of all CSP technologies at the end of 2011 [3]. This technology is the most mature, has with the 280 MWe gross Solana project the largest CSP plant under construction [4], and is therefore easiest to develop and finance at this point in time. However, new tower and Fresnel plants coming online will alter the market share in their favor and therefore parabolic trough providers have to find ways to increase efficiency and decrease costs. Several options to do this exist including the use of new heat transfer fluids, such as advanced thermal oils, direct steam generation (DSG) or molten salts, as well as enhanced collectors/receiver



Abbreviations: CSP, concentrating solar power; PV, photovoltaic; AU\$, Australian Dollar (AU\$/US\$ = 0.96); CapEx, capital expenditure; IRR, internal rate of return on investment; DNI, direct normal irradiance; BFB, bubbling fluidised bed; AC, aircooling; WC, water-cooling; NPV, water-cooling; StE efficiency, solar to electricity efficiency.

^{*} Corresponding author. Tel.: +61 2 9514 4878.

E-mail addresses: JuergenHeinzMartin.Peterseim@uts.edu.au (J.H. Peterseim), a.tadros@UNSWalumni.com (A. Tadros), uhellwig@eckrohrkessel.com (U. Hellwig).

^{0196-8904/\$ -} see front matter @ 2013 Elsevier Ltd. All rights reserved. http://dx.doi.org/10.1016/j.enconman.2013.10.022

tubes. However, these developments take time, and therefore this paper focuses on external steam superheating, as this concept can immediately increase plant efficiency, thereby reducing the size of the solar field, and subsequently reducing plant investment by up to 23.5%. This is already within the 2020 cost reduction targets of 17–40% [3,5,6].

It is well accepted that CSP hybrids can reduce capital expenditure (CapEx) significantly and several plants in operation worldwide verify its benefits, e.g., Liddell in Australia [7], Next Generation in the USA [8], Kuraymat in Egypt [9] and Termosolar Borges in Spain [10]. Furthermore, several studies investigate the hybridisation of CSP with coal [11,12], gas [13,14] and geothermal [15,16] energy sources. This paper investigates the novel concept of using biomass to externally superheat the steam of a conventional 50 MWe parabolic trough plant with 7.5 h thermal storage from 380 °C up to 540 °C. In principle, external superheaters using natural gas could even generate supercritical steam parameters, but this is not possible with biomass in practice due to the problem of high-temperature corrosion on the superheater tube banks. Different fuels and corresponding maximum steam parameters are compared in terms of plant performance and cost as well as environmental impact. The use of biomass for external superheating makes excellent use of small biomass resources, such as those under 50,000 t/year, which would otherwise be used in small-scale biomass plants with inherently low efficiencies or not be used at all.

CSP-biomass hybrid plants are likely to be a niche market with only limited locations worldwide where both resources are sufficiently abundant. However, several countries meet this condition, including but not limited to, Australia, Greece, Spain, Turkey, India, China and Brazil, which offers the potential to implement CSP in these markets at lower cost and advance the industry along the technology learning curve. Also other fuels can be used for external superheating including biogas and natural gas.

2. Hybrid concepts for CSP and biomass

Hybrid plants not only reduce the cost of CSP but also have the potential to move CSP out of remote/arid into regional/semi-arid/ agricultural regions where biomass and other waste materials are available. Typically, a CSP standalone plant requires a direct normal irradiance (DNI) of >2000 kWh/m²/year but the specific CapEx reduction of CSP-biomass hybrid plants enables locations with DNI levels of only >1700 kWh/m²/year [17]. Locations with low/negative cost solid fuels, such as waste materials, are preferable to maximise financial viability.

2.1. Current concepts

Some proposals to combine CSP with biomass or waste materials using dish systems were discussed briefly in the 1980s [18].

However, due to technical and financial issues no plants were built. It took another 25 years before construction of the first commercial CSP-biomass hybrid plant commenced near Lleida, ca. 150 km west of Barcelona, Spain [19], see Fig. 1. The 22.5 MWe Termosolar Borges plant came on-line in late 2012 [10], is located further north than any other CSP project in Spain and uses the mature parabolic trough technology with thermal oil [20]. Several other studies investigated the hybridisation of parabolic trough plants with biomass for high temperature/pressure turbine steam and feedwater heating [17,21–23] but no other project has yet commenced construction.

Alternatively, several Fresnel hybridisation options with biomass and waste materials have been investigated, such as high temperature/pressure turbine steam, combustion air and feedwater heating [24–27]. The benefit of using Fresnel would be steam temperatures of up to 500 °C [28], lower CapEx expectations and subsequently improved cycle efficiencies and economic viability. However, no reference plants yet exist for this CSP configuration.

Similarly, solar towers with molten salts and DSG are being investigated in combination with biomass [29] as are solar towers using a volumetric air receiver [30]. Due to the poor reference situation of volumetric air receiver tower systems and the higher complexity of such a system it is expected that solar towers with DSG or molten salts are easier to finance. Tower hybrids are promising due to high cycle efficiencies and lower cost energy storage.

2.2. Biomass fired external superheater

The purpose of an external superheater is to increase the cycle efficiency of power plants with inherently low efficiencies. Several factors can limit cycle efficiency including high temperature corrosion issues inside the boiler, as in energy from waste plants, or temperature limitations of the working fluid, as in parabolic trough plants using thermal oil.

The concept is based on forwarding "low" temperature steam from a boiler/heat exchanger to a separately fired heater, the external superheater, which uses a designated fuel to raise the steam temperature to the desired level, e.g., from 380 °C to 540 °C, see Fig. 2. The flue gas exit temperature leaving the superheater section is above the steam inlet temperature and the flue gas therefore still contains a significant amount of energy suitable for preheating boiler feedwater and combustion air.

The Holstebro, Denmark energy from waste/biomass plant, started implementing natural gas fired external superheaters in the 1990s to increase cycle efficiency by further superheating the 420 °C steam from the solid fuel fired boilers to >540 °C [31]. Similarly, some energy from waste plants adjacent to combined cycle gas turbine power stations provide steam at around 400 °C to the heat recovery steam generator for further superheating to >530 °C, e.g., Moerdijk in the Netherlands [32], Mainz in Germany [33] and Bilbao in Spain [34].



Fig. 1. First CSP-biomass plant under construction near Lleida, Spain (left) and its biomass fuel (right).


Fig. 2. External superheater concept using wood chips (scenario 6).

Table 1

Technical and economic inputs into scenario modeling.

| Plant commissioning | 2016 | Power purchase agreement | AU\$200/MWe |
|----------------------------|---------------|--------------------------|-----------------------------|
| Full load hours per year | 3500 [39] | Debt finance | 65% |
| Plant lifetime | 40 Years [39] | Debt interest rate | 8% |
| AC condenser back pressure | 0.12 bar | Discount rate | 7% |
| WC condenser back pressure | 0.08 bar | Water price | AU\$1.8/m ³ [42] |

Table 2

External superheater fuel specifications.

| Biomass fuel | Calorific value (MJ/kg) | Moisture content (%) | Fuel price (AU\$/t) |
|--------------------------|-------------------------|----------------------|---------------------|
| RDF | 14.0 | 22 | 0 |
| Urban wood waste | 15.7 | 10 | 20 |
| Wood waste from forestry | 10.5 | 40 | 50 |
| Stubble | 13.3 | 13 | 60 |

The first parabolic trough plant with an external superheater recently commenced operation near Abu Dhabi in the UAE, 100 MWe Shams One [35]. The natural gas fired external superheater increases the steam temperature from 380 °C to 540 °C and provides around 18% of the plant's total heat input [35].

The external superheater concept as described is well proven with natural gas, however the use of biomass is novel. This configuration has the potential of providing fully renewable energy with low technical risk due to the available experience and expertise with external superheaters and standalone biomass plants generating steam temperatures up to 540 °C.

Grate, fluidised bed combustion and some gasification systems are suitable for the solid fuels considered, as described in Table 2. Grate systems are the most mature [36] but fluidised bed systems are also very well proven at capacities <30 MWth. Several gasification systems burning producer gas in a designated combustion chamber operate successfully. These systems provide benefits in regards to higher dust retention in the gasifiers resulting in lower dust loads passing through the superheater banks [37]. Because of the high combustion efficiency, good reference situation and a variety of suppliers a bubbling fluidised bed (BFB) system has been chosen for the external superheater modeling in this paper. The cost difference between grate and BFB systems is minimal. Grates have a lower investment, but fluidised bed systems require less surplus air, therefore reducing the flue gas volume and consequently the component size downstream of the furnace. Gasification systems are similar in cost to grates while requiring less surplus air.

Table 3

Technical concept comparison.

The main benefit of the proposed concept is the solar field size reduction, which is the most capital intensive individual component of a parabolic trough plant [3,38], without using fossil fuels. This factor, in combination with lower financing cost for a parabolic trough plant would lead to lower electricity generation costs.

3. Plant modeling

The plant modeling is based on the concept outlined in section 2.2 and considers different biomass feedstocks as well as different air and water cooling configurations.

3.1. Methodology and scenarios

The technical and economic modeling work was carried out with Thermoflex version 23.0, which is a well accepted and widely used software in academia and industry.

Parabolic trough plants, without external superheating, are well proven with >50 plants in operation worldwide. To minimise finance risk all scenarios in this assessment are based on a 50 MWe plant capacity with 7.5 h molten salt thermal storage system, such as Andasol I–III, as there is significant expertise with this plant configuration.

Initially, higher temperature storage at 540 °C downstream of the external superheater was considered but this would require more changes to the standard 50 MWe parabolic trough concept

| Scenario | Biomass fuel | Live steam temperature (°C) | Live steam pressure (bar) | Peak StE gross plant efficiency (%) | Peak StE net plant efficiency (%) | Annual biomass demand (t) |
|----------|---------------------|--------------------------------|------------------------------|--|--------------------------------------|------------------------------|
| 1 AC | No ext. SH | 380 | 100 | 27.1 | 24.4 | - |
| 1 WC | No ext. SH | 380 | 100 | 28.0 | 25.0 | |
| 2 AC | RDF | 430 | 105 | 28.0 | 25.2 | 14,270 |
| 2 WC | RDF | 430 | 105 | 28.8 | 25.7 | 13,910 |
| 3 AC | Urban wood waste | 450 | 110 | 28.5 | 25.6 | 16,890 |
| 3 WC | Urban wood waste | 450 | 110 | 29.2 | 26.1 | 16,520 |
| 4 AC | Wood waste | 480 | 110 | 28.8 | 26.0 | 33,980 |
| 4 WC | Wood waste | 480 | 110 | 29.6 | 26.4 | 33,250 |
| 5 AC | Wood waste | 510 | 120 | 29.4 | 26.5 | 42,040 |
| 5 WC | Wood waste | 510 | 120 | 30.2 | 27.0 | 41,020 |
| 6 AC | Wood waste | 540 | 130 | 30.0 | 27.0 | 49,250 |
| 6 WC | Wood waste | 540 | 130 | 30.8 | 27.5 | 47,600 |
| 7 AC | Stubble | 540 | 130 | 29.8 | 26.9 | 39,340 |
| 7 WC | Stubble | 540 | 130 | 30.7 | 27.5 | 38,050 |

and this concept was abandoned to minimise technical/financial risk. However, higher temperature storage would reduce investment requirements and is therefore a subject for further study.

Seven scenarios, both air and water cooled with steam reheating are compared, with steam parameters ranging from 380 °C at 100 bar to 540 °C at 130 bar, as shown in Table 3. The base-case is a parabolic trough plant without external superheating. All other scenarios are based on different biomass feedstocks and consider that the nature and composition of the solid fuel determines the maximum steam parameters without encountering high-temperature corrosion and ash fusion problems on the superheater tubes, mainly chlorine and alkali metal driven.

The economic assumptions are a plant lifetime of a minimum 40 years with 3500 annual full load hours as in the Andasol 3 plant [39], EPC contract awarded in 2015, commissioning in 2016, 65% debt finance, a debt interest rate of 8% and 7% discount rate, see Table 1. The assumed power purchase price is AU\$200/MWh which is 40% lower than the price received by the Andasol 1 plant in Spain (AU\$340/MWh based on AU\$/€exchange rate of 1.26 and €270/MWh [40]) and similar to a 250 MW water-cooled trough plant without storage [3]. The modeling includes capital and operating costs as well as annual escalation rates for inflation (3%), fuel (2%), electricity (3%) and water (4%). All investments shown are the owner's total project cost consisting of EPC price plus 9% owner's soft costs covering permits, project management, legal and finance aspects of a CSP plant. Payback periods and net present values derive from the total project and not only the EPC cost.

To provide confidence that the technical and economic modeling in this assessment is in line with reality, the water-cooled scenario 1 was modeled under Spanish conditions to meet the ϵ 300m for Andasol 2 plant investment [41], 51 ha aperture [40] and 28% gross solar to electricity efficiency [39]. Subsequently, the model was adapted to Australian conditions, which include higher cost factors for equipment (+12%) and labor (+20%), as well as air cooling with a mean maximum ambient temperature of 23.8 °C. The high labor costs are a result of the ongoing mining boom creating skilled labor shortages.

As external superheaters are not a standard Thermoflex component investment assumptions would have higher uncertainties and for this reason indicative prices were obtained from an industry partner with actual external superheater expertise [37].

Scenario 1 (air and water cooled) is the basis for all other scenarios. All models use identical DNI, weather conditions and feedwater heating arrangement.

3.2. Fuels considered

In Australia the use of biomass for biofuels has been investigated and a recent study concluded that biomass has the potential to provide a significantly greater greenhouse gas abatement when used for power generation rather than biofuels. This is due to the high carbon intensity of Australia's electricity sector [43].

Several biomass/waste fuels can be considered as a source for external superheating, including: refuse derived fuel (RDF); wood waste; wood chips; and stubble, as indicated in Table 2. However, to broaden the applicability of the concept other fuels can be considered, such as biogas or natural gas. Typically, RDF fired power plants limit steam temperatures to around 430 °C while wood waste units operate at up to 450 °C. In the past, stubble fired power stations limited steam temperatures to <430 °C due to the chlorine content of the fuel, however new alloy materials, developed for energy from waste plants, enable modern units to operate with steam temperatures of up to 540 °C, e.g., the 25 MWe Sangüesa plant in Spain [44]. Clean biomass, such as wood chips from forestry and horticulture, are suitable for steam temperatures up to 540 °C. This assessment uses the these steam temperatures in its scenarios but

includes 480 °C and 510 °C scenarios for wood chips as several plants exist operating at these steam temperature levels.

3.3. Site information

The site selected for this analysis is in the Mildura region, Australia, which has a population of around 53,000 and is home to one of Australia's prime agriculture and horticulture regions. The solid fuel quantities required for external superheating for a 50 MWe parabolic trough plant, Tables 2 and 3, are available from agriculture, horticulture, forestry and urban waste streams [45]. The assumed fuel prices are shown in Table 2.

With an average annual DNI of 2198 kWh/m²/year [46] the Mildura region has a higher DNI than Lleida in Spain, where the first CSP-biomass plant commenced operation recently [10], but is similar in regard to the agriculture and horticulture sources of biomass. Generally, the DNI levels are similar to southern Spain where there are several 50 MWe parabolic trough plants with 7.5 h storage.

In addition to the good solar and biomass resources Mildura is in the state of Victoria, which has one of the highest retail electricity prices in Australia [47]. This maximises plant revenue and was another criterion for selecting this site. The electricity could be consumed locally but also exported via existing transmission lines into the national electricity market.

The ambient temperatures used in this modeling derive from measurements at Mildura Airport by the Bureau of Meteorology Australia³ and DNI data from the Thermoflex database.

Water is a scarce commodity in the region and a power plant would compete with other uses, such as irrigation water. Therefore air-cooling is the preferred option to minimise water consumption but water-cooling has also been investigated in this study.

4. Results

To provide a comprehensive assessment the different scenarios are analysed in regards to technical, economic and environmental performance as well as plant layout. Air (AC) and water-cooled (WC) concepts were modeled and the results for the different scenarios follow the same trends. The focus of the analysis is on aircooled plants but water-cooled results are also shown.

4.1. Plant performance

With increasing steam parameters the plant solar to electricity (StE) efficiency increases significantly in the air and water cooled scenarios, as shown in Table 3. The efficiency changes shown are almost linear while the parasitic losses remain constant at around 10%. Scenarios 6 ranks best with 10% higher net solar to electricity efficiencies than a standard 50 MWe parabolic trough plant (scenarios 1). At Mildura even a 250 MWe net parabolic trough plant, such as Solana in the US, would not reach the net electric efficiency of scenario 6 AC, 27-25%, despite significant economy-of-scale benefits. The only reason to choose steam temperature of 480/ 510 °C (scenario 4/5) over 540 °C, when using wood chips, is the larger number of reference plants in this temperature range. Despite the same steam parameters of scenarios 6 (wood) and 7 (stubble) the stubble plant's cycle efficiency is slightly lower due to higher flue gas temperature at the stack (160 °C compared to 120 °C), which is caused by the higher chlorine content in fuel.

The efficiency difference between water and air cooling is 2.5%. This is lower than the 5% indicated in other studies [48] but is considered realistic as cooling water temperatures from the Murray

³ www.bom.gov.au/climate/averages/tables/cw_076031.shtml.



Fig. 3. Ambient temperature effect on air and water cooled scenarios 6 over a period of 24 h.

River can reach 23 °C and because the AC scenarios are optimised for the high ambient temperatures in the Mildura region.

To increase the steam parameters larger biomass quantities are required, reaching almost 50,000 t/a in scenarios 6 compared to only 14,300 t/a for scenarios 2. The additional use is justified as it results in higher overall plant efficiencies. The use of relatively small biomass quantities, such as 50,000 t/a, is beneficial as its use in a small standalone biomass plant, fuel sufficient for a 4 MWe capacity when assuming 8000 h operation/a, would have an inherently lower gross efficiency of maximum 24%. Using the 50,000 t/a wood chips in a scenario 6 AC plant generates 47.7 GWh compared to 31.3 GWh in a standalone biomass plant, an increase of 52%. The benefits of using smaller biomass quantities are even stronger, up to 80% more MWh/year, as efficiencies fall rapidly at plant capacities of 2.5 MWe (scenarios 4) or only 1.2 MWe (scenarios 2). Considering a scenario 1 plant and increasing the temperature/pressure of the same steam flow from 380 °C/ 100 bar to 540 °C/130 bar would increase the 50 MWe plant capacity to 55 MWe while only marginally increasing the investment for the external superheater, AU\$11m or AU\$2.2m/MWe. For these reasons the use of biomass for external superheating of "low" temperature steam sources, including but not limited to parabolic trough plants using thermal oil, should be encouraged even though this is likely to be a niche market.

The high temperature flue gas leaving the external superheater at 395 °C contains sufficient energy to further preheat the boiler feedwater by 13 °C, +6% compared to scenario 1, and preheat combustion air from ambient temperature to 260 °C.

Ambient temperatures in the Mildura region vary significantly, with a minimum mean temperature of 10.3 °C and a mean maximum temperature of 23.8 °C. It is not uncommon that temperatures reach/exceed 40 °C for a few consecutive days [49] and 66 days/year above 30 °C is a statistical average. These temperature changes particularly affect the efficiency of the AC scenarios (11% difference between 16 °C and 40 °C) while the WC scenarios remain more stable (5% difference between 16 °C and 40 °C), as shown in Fig. 3. This equals a capacity loss of 5.7 MWe (AC) and 2.6 MWe (WC).

4.2. Economic analysis

Air-cooling increases plant investment by an average of 4.4%, which is in line with expectations and previous studies [48]. The AU\$411m investment for scenario 1 AC is slightly higher than other studies indicate for a 50 MWe parabolic trough with 7.5 h thermal storage [3]. The difference derives from particularly high labor costs in Australia, the inclusion of the 9% owner's soft costs and allowance for exchange rate fluctuations.

As with plant performance the scenarios 6 perform best, WC better than AC, in regards to investment, payback net present value (NPV) and internal rate of return on investment (IRR), as shown in Table 4. The investment cost reductions reach 23.5% when compar-

ing scenarios 1 AC with 6 AC. This is significant considering that these reductions are possible today and not in 2020 at which time cost reductions of 17–40% are expected through continuous deployment and plant optimisation [3,5,6]. By increasing plant efficiency the specific investment of scenario 6 AC, AU\$6.3m/MWe, is lower than the economy-of-scale cost reduction potential a 250 MWe parabolic trough plant on this site would offer, AU\$6.5m/MWe. The reason for this is the specific solar field size reduction, m^2/MWe .

The investment reductions as well as NPV and IRR increases are almost linear and the only exceptions are scenarios 6 and 7, which have identical steam parameters. The investments of scenario 7 plants are marginally higher, 1%, due to the use of stubble which require higher quality alloys in the external superheater and higher temperature resistant baghouse and flue gas ducting, due to the different fuel composition. The slightly higher investment and fuel cost accordingly result in a lower payback, NPV and IRR.

The benefits of external superheating reflect well in the NPV and IRR which improve by 42% (NPV) from scenarios 1 AC to 6 AC and 18% (IRR) respectively. CSP plant costs are strongly affected by financing cost and reducing CapEx improves the NPV significantly despite additional costs for the biomass supply over the plant's lifetime.

The fuel price does not affect the economic viability as much as expected as doubling the fuel price in scenario 6 AC only reduces the NPV to AU\$147m, see Fig. 4. The water price is even less relevant for the economic viability as doubling the price reduces the NPV of the WC scenarios by less than 0.1%. The most significant impact results from changes in the electricity price, as shown in Fig. 4, in which a 15% increase from AU\$200/MWh to AU\$230/MWh, raises the NPV of scenario 6 AC 43%, from AU\$164m to AU\$235m.

A risk to the plant economic viability, in particular for air cooled CSP plants, is the significant mean minimum and maximum ambient temperature difference of 13.5 °C in the Mildura region. Expectations of rising global average temperatures with longer and more frequent "heat wave" periods requires a design that, despite higher investment, has a sufficiently sized condenser to ensure maximum plant performance. With the current price of cooling water being relatively low but supply competition with other users a wet/dry hybrid cooling concept could be considered using water only during above 30 °C ambient temperature days.

Having the same plant capacity and thermal storage arrangement as the Andasol installations in Spain it would be expected to create the same employment of 40 annual operation and maintenance jobs [40]. Assuming that the direct and indirect employment for the external superheater operation, supply chain and fuel transport is similar to a 5 MWe biomass plant [50] the external superheater concept could create up to 30 additional full-time equivalent jobs, 11 direct and 19 indirect. This is significant for a rural region that predominantly relies on agriculture and tourism for employment creation.

| Table 4 | |
|----------|-------------|
| Economic | comparison. |

| Scenario | Total plant investment (mAU\$) | Specific plant investment (AU\$/MW) | Specific cost reduction to scenario 1 (%) | Years payback | Net present value (mAU\$) | IRR (%) |
|----------|-----------------------------------|--|---|------------------|------------------------------|---------|
| 1 AC | 411 | 8.2 | | 15.1 | 116 | 8.6 |
| 1 WC | 392 | 7.9 | | 14.6 | 132 | 9.0 |
| 2 AC | 384 | 7.7 | 6.7 | 14.5 | 135 | 9.1 |
| 2 WC | 366 | 7.3 | 6.7 | 14.0 | 149 | 9.4 |
| 3 AC | 371 | 7.4 | 14.3 | 14.3 | 139 | 9.2 |
| 3 WC | 353 | 7.1 | 10.0 | 13.8 | 154 | 9.6 |
| 4 AC | 348 | 7.0 | 15.4 | 13.9 | 148 | 9.5 |
| 4 WC | 335 | 6.7 | 14.8 | 13.4 | 160 | 9.8 |
| 5 AC | 333 | 6.7 | 19.1 | 13.5 | 156 | 9.8 |
| 5 WC | 320 | 6.4 | 18.4 | 13.1 | 167 | 10.1 |
| 6 AC | 315 | 6.3 | 23.5 | 13.1 | 164 | 10.1 |
| 6 WC | 305 | 6.1 | 22.2 | 12.8 | 175 | 10.4 |
| 7 AC | 318 | 6.4 | 22.9 | 13.3 | 160 | 10.0 |
| 7 WC | 309 | 6.2 | 21.3 | 13.0 | 166 | 10.2 |



Fig. 4. Power purchase and fuel price impact on the net present value of scenarios 1 and 6.

4.3. Environmental performance

All WC scenarios perform better than AC in regards to technical and economic data but when considering the environmental impact the significant cooling water demand is a disadvantage in a region that requires water for existing agriculture and horticulture. Increasing the plant efficiency reduces the full-load cooling water requirements from 3.1 m³/MWh (scenario 1 WC) to 2.5 m³/MWh (scenario 6 WC) but that still adds up to 440 Ml/year, see Table 5. AC systems require ca. 90% less water with only mirror washing and blowdown being relevant consumers.

The solar field footprint decreases significantly from scenario 1 AC to 6 AC, -34%, but the difference between the individual AC/WC scenarios is with 3.5% significantly smaller, see Table 5. Increasing land use efficiency from 3.6 ha/MW to 2.4 ha/MW, scenario 1 AC to 6 AC, has economic benefits but these are limited due to relatively low land prices in the Mildura region.

The specific land use efficiencies of all AC/WC external superheater scenarios is better than the 3.1 ha/MW (WC) and 3.3 ha/MW (AC) required for a 250 MWe plant with 7.5 h of thermal storage on this site.

Despite all scenarios having the same annual electricity output of 175 GWh their greenhouse gas abatement potential varies slightly due to the different biomass harvest and transport emissions. Scenario 1 plants could abate up to $147,200 \text{ t } \text{CO}_2/\text{year}$ (based on the 2010 Australian generation mix with a carbon intensity of 841 kg CO₂/MWh [51]) while scenario 7 plants (stubble) could abate up to $146.200 \text{ t } \text{CO}_2/\text{year}$ (25 kg CO₂/t for 50 km transport) and scenario 6 plants (wood waste) up to 144,500 t CO₂/year (54 kg CO₂/t for 50 km transport) [43]. The slightly lower greenhouse gas abatement potential caused by biomass harvest and transport emissions is insignificant compared to the CapEx reduction potential.

4.4. Plant layout

The layout of the power island does not change in the different scenarios with the steam turbine, heat exchangers, molten salt tanks, condenser, biomass fuel storage, external superheaters, flue gas cleaning with stack, control room and auxiliaries arranged in the centre of the solar field. The most obvious changes to a standard parabolic trough plant are the solar field size reduction of up to 34% and the addition of the two external superheaters with fuel storage. Fig. 5 visualises the impact of the different AC scenarios on the overall plant footprint and highlights the main components.

Due to more frequent truck movements to supply the external superheaters with fuel a wider than usual access road is required including a spacious turning point in front of the fuel storage. Placing the fuel storage outside the solar field and installing long conveyers to transport the fuel to the external superheaters is not recommended as conveyers are a weak point in biomass plants, e.g. plant shutdown due to blockages.

5. Regions for technology implementation

Compared to standalone CSP plants the global potential for CSPbiomass hybrids is lower as only some regions worldwide have both energy sources in sufficiently high abundance. However, these sites have the potential to realise CSP plants today at costs expected in 2020, therewith fast-tracking CSP deployment and potentially avoiding the lock-in of future greenhouse gas emissions from new fossil plants.

When overlaying maps with the global annual DNI [52] and the overall index of land suitability for cultivation [53] several locations in DNI areas >1800 kWh/m²/year seem principally suitable for CSP-biomass hybrids in Australia, Asia, China, India, Africa, southern Europe as well as south, central and north America, see Fig. 6. A minimum DNI of >1800 kWh/m²/year is chosen as this is the DNI level of the 22.5 MWe Termosolar Borges plant site. To avoid competition with food production only areas with a land suitability index of 0.6–1 are considered as these areas could grow biomass for energy purposes in parallel with food production. This

| Scenario | Solar field and power block footprint (ha) | Land use efficiency (ha/MW) | Cooling water consumption (Ml/year) |
|----------|--|-----------------------------|-------------------------------------|
| 1 AC | 179 | 3.6 | 537 |
| 1 WC | 169 | 3.4 | |
| 2 AC | 157 | 3.1 | 504 |
| 2 WC | 151 | 3.0 | |
| 3 AC | 147 | 2.9 | 491 |
| 3 WC | 143 | 2.9 | |
| 4 AC | 139 | 2.8 | 477 |
| 4 WC | 134 | 2.7 | |
| 5 AC | 126 | 2.5 | 458 |
| 5 WC | 124 | 2.5 | |
| 6 AC | 119 | 2.4 | 440 |
| 6 WC | 116 | 2.3 | |
| 7 AC | 119 | 2.4 | 441 |
| 7 WC | 116 | 2.3 | |

Table 5Water and land use comparison.



Fig. 5. Photomontage comparing the impact of scenarios 1–7 AC on the footprint of a 50 MWe parabolic trough plant with 7.5 h thermal storage; scenario 1 = blue, scenario 2 = orange, scenario 3 = red, scenario 4 = green, scenario 5 = yellow and scenario 6 and 7 = purple border around solar field. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)



Fig. 6. Potential regions for CSP-biomass hybrids plants worldwide.

is achieved by using agriculture residues, such as straw, and agroforestry materials, such as wind breaks, or urban wood waste streams, such as construction and demolition timber. Identifying the global potential for CSP-biomass hybrid plants would require detailed geospatial mapping including multiple constraints, such as urbanisation, alternative land uses or land clearing limitations, which is not part of this paper but of interest for future research. This paper only selects some example locations from Fig. 6 to provide a high level overview.

According to a recent bioelectricity assessment [45] Australia has existing biomass resources in suitable DNI regions: in Queensland, including 2.6 mt of bagasse around Mackay; in New South Wales, including 1.9 mt of stubble around Griffith; in Victoria, including 0.7 mt stubble around Mildura; and in South Australia, including 1 mt stubble west of Port Augusta. With <50,000 t/a biomass required to externally superheat the steam of a 50 MWe CSP plant a fraction of the sites would be sufficient to implement several hundred MWe capacity. This is a significant potential in a traditionally low cost electricity country where stand-alone CSP plants struggle despite some significant government programs. As an example of the difficulty of stand-alone CSP plants competing the recently abandoned 250 MWe SolarDawn project had AU\$464m of government funds allocated out of a total investment of AU\$1.2b [54]. However, lower cost CSP-fossil fuel hybrid plants are being built, such as Kogan Creek [55] or Liddell [7]. External superheating with biomass could accelerate CSP deployment without fossil fuels and one such plant is being investigated in Queensland [24,56].

Further promising potential exists in California, USA), where several biomass plants operate [57] in suitable DNI areas and a 107 MWe CSP-biomass hybrid plant was already proposed [23]. Further potential exist on the western end of the wheat belt where stubble could be used for external superheating.

Similarly, Mexico, Brazil and Argentina have suitable sites for CSP-biomass hybrids as do some parts of South Africa and Ethiopia, however competition with land required for food production has to be considered very carefully in these regions. Many of these countries have weak electricity grids where smaller CSP-biomass hybrid plants could provide network benefits. Also industrial and off-grid CSP applications are relevant and present a future growth market [58]. Smaller plant capacities result in a higher specific investment and the hybridisation with biomass can help offset some of the specifically higher cost while increasing the plant's capacity factor.

In Europe the first CSP-biomass hybrid plant already commenced operation in Spain [10] and further sites could be considered in Spain but also southern France, Italy, Croatia, Greece and Turkey. In Asia, India and China seem to have the most suitable locations for CSP-biomass hybrids with large areas where high DNI and land suitability for cultivation overlap. With many of these areas being remote distributed CSP-biomass plants, similar to South America and Africa, provide promising applications.

6. Discussion

When considering an external superheater concept a few points have to be considered very carefully to ensure reliable plant operation. Starting with the fuel supply larger roads are required to accommodate the more frequent truck movements. Increased dust particles are likely from these truck movements and on-site fuel processing but these can be countered with paved roads, covering the freight and enclosed fuel processing equipment. The fuel storage should have a capacity for 4 days full-load operation and needs to be kept under negative pressure using the air as combustion air for the external superheaters.

The external superheaters require a special design to ensure that the temperatures in the furnace remain well below 1000 °C to avoid high-temperature corrosion and ash fusion issues. Furthermore refractory in the furnace should be kept to a minimum as this limits start-up times. However, solutions for this exist such as a water cooled furnace. The material selection is crucial, particularly for the superheaters but also for economiser and air heaters, to reduce plant down-time and maintenance. As the external superheaters need to start/stop/cycle frequently a fluidised bed is a suitable firing system as it operates well at low part-loads. To ensure quick start-up and back-up, in case of biomass supply problems, 100% capacity natural gas burners are required. Fuel supply issues, such as conveyer blockages, are common in biomass plants and the gas back-up would ensure continuous operations. Similarly, two external superheaters are suggested to provide redundancy.

Gasification systems firing producer gas in the external superheaters are worth considering as the technology retains a large ash fraction, due to low gas velocities in the gasifiers, which limits high-temperature corrosion and ash fusion issues in the external superheater. Several plants operate successfully with single fuel references reaching steam temperature of 525 °C [37].

Biomass considered for use in external superheaters should derive from the vicinity of the plant. The low calorific value of biomass does not justify, economically and environmentally, transport distances of more than 50 km unless the fuel is converted to pellets or briquettes on the harvest site. Therefore regions with strong agriculture, horticulture or forestry are prime candidates for the external superheater concept. In addition to the fuels outlined in Table 2 further materials can be considered broadening the applicability of the external superheater concept. Traditional biomass includes olive pits, nut shells, etc. but such material is very location specific and therefore of limited availability. An innovative potential feedstock is algae which require significantly less land to grow and grows quickly. Algae were not considered in this assessment as there are no production facilities at the required scale available yet. Furthermore biogas and landfill gas as well as other gaseous or liquid fuels are suitable for external superheaters but have to be analysed carefully to avoid high-temperature corrosion issues. With the biomass quantities and calorific values given in this paper it is possible to calculate the thermal input and therefore the quantities of other fuels than the ones outlined in Table 2.

A well organised fuel supply infrastructure is very important to ensure material availability at all times. This requires a close operation with ideally more than one supplier and long-term fuel supply contracts. The plant should not be designed to the maximum biomass availability in the region but a conservative estimate as seasonal changes can affect the biomass harvest significantly.

7. Conclusions

Externally superheating the live steam of a parabolic trough plant with biomass has the potential to increase the peak solar to electricity efficiency by up to 10.5%, thereby pushing it above 30% (gross) and into the spheres of modern solar tower plants. Other fuels are suitable for external superheating too but the use of biomass allows fully renewable electricity generation. It needs to be acknowledged that suitable locations for CSP-biomass hybrid plants are limited but as outlined in the paper even small biomass quantities, such as those <15,000 t/year, can provide material benefits.

The increased efficiency can lead to immediate CSP cost reductions of up to 23.5%, particularly driven by the smaller solar field, and a specific plant investment of only AU\$6.1m/MWe for a 50 MWe parabolic trough plant with 7.5 h thermal storage. This is significant and has the potential to strengthen CSP in the increasingly competitive electricity market. Despite the CSP-biomass hybrid potential being limited the concept offers the installation of CSP systems in a variety of locations worldwide at lower cost. This would advance the industry along the learning curve while parallel research and development on new working fluids, optimised receivers, etc. provides the technologies for more competitive CSP power generation in the future.

References

- IEA. In: Renewable energy medium-term market report 2013 market trends and projections to 2018, Paris; 2013.
- [2] Beetz B. Whole 1 GW Blythe project to convert to PV", PV Magazine; 2012. Available: www.pv-magazine.com/news/details/beitrag/whole-1-gw-blythe-project-to-convert-to-pv_100007604/#axzzA5Oyy1]v. [accessed 04.10.13].
- [3] IRENA. Renewable energy technologies: cost analysis series, vol. 1(2/5). Abu Dhabi, UAE: Concentrating Solar Power; 2012.
- [4] Abengoa Solar. Plants under construction; 2013. Available: <www.abengo asolar.com/corp/web/en/nuestras_plantas/plantas_en_construccion/>. [access ed 04.10.13].
- [5] Hinkley J, Curtin B, Hayward J, Wonhaus A, Boyd R, Grima C, et al. Concentrating solar power – drivers and opportunities for cost-competitive electricity. Sydney, Australia: CSIRO & Aurecon Australia Pty Ltd.; 2011.
- [6] IEA. Technology roadmap concentrating solar power. Paris: OECD Publishing; 2010.
- [7] Macquarie Generation. Macquarie generation annual report; 2012.
- [8] Florida Power & Light Company. FPL's martin next generation solar energy center; 2010. Available: <www.fpl.com/environment/solar/pdf/Martin.pdf>. [accessed 04.10.13].
- [9] Brakmann G, Mohammad F, Dolejsi M, Wiemann M. Construction of the ISCC Kuraymat. In: SolarPACES conference; 2009.
- [10] Protermosolar. Abantia and Comsa Emte set up the first facility in the world that combines solar thermal and biomass energy; 2012. Available: http://www.protermosolar.com/saladeprensa/?q=en/content/abantia-and-comsaemte-set-first-facility-world-combines-solar-thermal-and-biomass-energy> [accessed 03.10.13].
- [11] Ozturk M, Dincer I. Thermodynamic assessment of an integrated solar power tower and coal gasification system for multi-generation purposes. Energy Convers Manage 2013;76:1061–72.
- [12] Popov D. An option for solar thermal repowering of fossil fuel fired power plants. Sol Energy 2011;85(2):344–9.
- [13] Allani Y, Favrat D, von Spakovsky MR. CO₂ mitigation through the use of hybrid solar-combined cycles. Energy Convers Manage 1997;38(Suppl. 1):S661–7.
- [14] Peterseim JH, Tadros A, Hellwig U, White S. Integrated solar combined cycle plants using solar towers with thermal storage to increase plant performance. In: ASME, Power; 2013.
- [15] Astolfi M, Xodo L, Romano MC, Macchi E. Technical and economical analysis of a solar-geothermal hybrid plant based on an Organic Rankine Cycle. Geothermics 2011;40(1):58–68.
- [16] Zhou C, Doroodchi E, Moghtaderi B. An in-depth assessment of hybrid solargeothermal power generation. Energy Convers Manage 2013;74:88–101.
- [17] Pérez Á, Torres N. Solar parabolic trough biomass hybrid plants: a cost efficient concept suitable for places in low irradiation conditions. In: SolarPaces conference; 2010.
- [18] McDonald CF. A hybrid solar closed-cycle gas turbine combined heat and power plant concept to meet the continuous total energy needs of a small community. J Heat Recover Syst 1986;6(5):399–419.
- [19] Abantia. The installation of the solar field Termosolar Borges, the first plant to combine thermal and biomass, has started; 2011. Available: <www.abantia.com/noticias-prensa/en_noticias/28/the-installation-of-thesolar-field-termosolar-borges-the-first-plant-to-combine-thermal-andbiomass-has-started>. [accessed 07.10.13].
- [20] Cot A, Amettler A, Vall-Llovera J, Aguilo J, Arque JM. Termosolar Borges: a thermosolar hybrid plant with biomass. In: Third international symposium on energy from biomass and waste; 2010.
- [21] Schnatbaum L. Biomass utilization for co firing in parabolic trough power plants. In: SolarPACES conference; 2009.
- [22] Nixon JD, Dey PK, Davies PA. Which is the best solar thermal collection technology for electricity generation in north-west India? Evaluation of options using the analytical hierarchy process. Energy 2010;35(12):5230–40.
- [23] California Energy Comission. San Joaquin solar 1 & 2 power plant licensing case. California: Sacramento; 2008.
- [24] Peterseim JH, Tadros A, White S, Hellwig U, Klostermann F. Concentrated solar power/energy from waste hybrid plants – creating synergies. In: SolarPACES conference; 2012.
- [25] Rojas M, Hellwig U, Peterseim JH, Harding J. Combining solar thermal power systems with solid fuels in new plants. In: SolarPACES conference; 2010.
- [26] Spliethoff H, Kaeding N, Murer M, Alonso-Herranz E, Gohlke O. Combining energy from waste and concentrated solar power: new solutions for sustainable energy generation. In: International solid waste association conference; 2010.

- [27] Kaeding N. Hybridkraftwerk Solar/Abfall: Erstellung und Beurteilung von Konzepten. Munich, Germany: Technische Universitaet Muenchen; 2010.
- [28] Fluri TP, Lude S, Lam J, Morin G, Paul C, Platzer WJ. Optimization of live steam properties for a linear fresnel power plant. In: SolarPACES conference; 2012.
- [29] Peterseim JH, Tadros A, White S, Hellwig U, Landler J, Galang K. Solar towerbiomass hybrid plants – maximizing plant performance. In: SolarPACES conference; 2013.
- [30] Coelho B, Schwarzbözl P, Oliveira A, Mendes A. Biomass and central receiver system (CRS) hybridization: volumetric air CRS and integration of a biomass waste direct burning boiler on steam cycle. Sol Energy 2012;86(10):2912–22.
- [31] Andersen EJ, Oksen F, Olesen JV. Power plant Holstebro/Struer with separately fired superheater. In: Eckrohrkessel licensee conference; 1992.
- [32] Toebes J, Beker M. External steam input and process steam output at the Moerdijk (Netherlands) combined-cycle plant. VGB POWERTECH, vol. 8, no. 7. Essen, Germany; 1998. p. 21–7.
- [33] Entsorgungsgesellschaft Mainz mbH. The Mainz Waste-to-energy plant: exceptionally reliable and environmentally sound waste disposal; 2008. Available: www.mhkw-mainz.de/downloads/EGM_infobro3_eng_20.pdf>, [accessed 07.10.13].
- [34] Ribeiro SG, Kimberlin T. High efficiency waste to energy power plants combining municipal solid waste and natural gas or ethanol. In: 18th annual north american waste-to-energy conference; 2010.
- [35] Goebel O, Luque F. Shams one 100 MW CSP plant in Abu Dhabi update on project status. In: SolarPACES conference; 2012.
- [36] Spliethoff H. Power generation from solid fuels. 1st ed. Heidelberg: Springer; 2010.
- [37] ERK Eckrohrkessel GmbH. Personal communication with ERK Eckrohrkessel GmbH. Germany: Berlin; 2013.
- [38] Power IT. Realising the potential of concentrating solar power in Australia. Australia: Canberra; 2012.
- [39] Flagsol GmbH. Andasol Projects, Spain; 2012. Available: http://www.flagsol.com/flagsol/english/aaareferences-alt/andasol-projects-spain-alt/andasol-projects-spain.html. [accessed 04.10.13].
- [40] NREL. Andasol-1; 2011. Available: http://www.nrel.gov/csp/solarpaces/project_detail.cfm/projectID=3. [accessed 03.10.13].
- [41] Solar Millennium AG. Financing granted to Solar Millennium AG for Andasol 2, the second parabolic trough power plant in Europe; 2006. Available: <www.solarmillennium.de/Archiv/Presse/Meldungen/Archiv_2006/ Meldungen_2006_,lang2,51,235.html>. [accessed 01.05.13].
- [42] Kaczan D, Qureshi ME, Connor J. A summary of water trade and price data for the southern murray-darling basin. Canberra; 2011.
- [43] Farine D, O'Connell DA, John Raison R, May BM, O'Connor MH, Crawford DF, Herr A, Taylor JA, Jovanovic T, Campbell PK, Dunlop MIA, Rodriguez LC, Poole ML, Braid AL, Kriticos D. An assessment of biomass for bioelectricity and biofuel, and for greenhouse gas emission reduction in Australia. GCB Bioenergy; 2011.
- [44] Acciona Energía. Biomass plant from straw combustion in Sangüesa; 2002. Available: http://www.acciona-energia.com/media/219313/ACCIONA_Sanguesabiomassplant_EN.pdf. [accessed 07.10.13].
- [45] Crawford D, Jovanovic T, Connor MO, Herr A, Raison J, Baynes T. AEMO 100% renewable energy study – potential for electricity generation in Australia from biomass in 2010, 2030 and 2050. Australia: Newcastle; 2012.
- [46] Bureau of Meteorology. Australian monthly average hourly global and direct normal solar exposure gridded data, Melbourne; 2012.
- [47] Australian Energy Market Commission. Possible future retail electricity price movements: 1 July 2011 to 30 June 2014. In: Final report, Sydney; 2011.
- [48] Turchi CS, Wagner MJ, Kutscher CF, Laboratory NRE. Water use in parabolic trough power plants: summary results from WorleyParsons' analyses. Colorado, US: Golden; 2010.
- [49] Bureau of Meteorology. Special climate statement 43 extreme January heat, Melbourne; 2013.
- [50] Thornley P. Rogers J. Huang Y. Quantification of employment from biomass power plants. Renew Energy 2008;33(8):1922–7.
- [51] IEA. CO₂ emissions from fuel combustion 2012 highlights, Paris; 2012.
- [52] Trieb F, Schillings C, O'Sullivan M, Pregger T, Hoyer-Klick C. Global potential of concentrating solar power. In: SolarPaces conference; 2009.
- [53] Ramankutty N, Foley Ja, Norman J, McSweeney K. The global distribution of cultivable lands: current patterns and sensitivity to possible climate change. Glob Ecol Biogeogr 2002;11(5):377–92.
- [54] Kelly J. Darkness falls on \$1.2bn SolarDawn project, The Australian, Sydney, Australia; 13-November-2012. p. 4.
- [55] CS Energy. Work starts on Kogan Creek solar boost project; 2011. Available: http://kogansolarboost.com.au/news/work-starts-on-kogan-creek-solar-boost-project/>, [accessed 04.10.13].
- [56] Ipswich City Council. Ipswich supports pioneering eco-precinct concept; 2012. Available: http://www.ipswich.qld.gov.au/about_council/media/view_release/?id=3657&action=viewMediaRelease. [accessed 04.10.13].
- [57] Mayhead GJ, Tittmann P. Outlook: uncertain future for California's biomass power plants. Calif Agric 2012;66(1):6.
- [58] Rawlins J, Ashcroft M. Small-scale concentrated solar power a review of current activity and potential to accelerate deployment, London; 2013.

4.4.3 Future CSP–EfB and CSP–EfW hybrid plants

The CSP, EfB and EfW technologies analysed in the previous sections represent the current state-of-the-art technologies and, like other reports, this research assumes that these remain the dominant ones for the foreseeable future. However, new CSP technologies, and new thermochemical conversion technologies for solid fuels and materials will emerge over time and affect the cycle efficiency and cost of future CSP hybrid plants and of Rankine cycle power plants in general. New thermochemical conversion technologies, such as advanced gasification, paired with new tube materials, such as advanced copper-nickel alloys and ceramics, would allow significantly higher steam parameters than the current state-of-the-art 540 °C at 130 bar without significant high-temperature corrosion and ash fusion problems on the boiler's superheater tube banks. Also, transdisciplinary research introducing ideas into the engineering field from research in biology, such as algae as an energy feedstock, or ideas from nanotechnology, such as nano-coatings to improve component reliability, is likely to change the current technology palette significantly.

The impact of higher steam parameters of up to 700 °C and 350 bar on the efficiency of current Rankine cycle systems is relatively easy to estimate and Figure 32 provides an example for a 100 MWe air-cooled solar tower-biomass combustion-type hybrid plant at Mildura, Australia. However, the efficiency curve provided is only indicative as innovations in turbines, condensers, auxiliary equipment and working fluids are likely but predicting their combined impact on plant efficiency is difficult. Therefore, the future efficiency of such systems could be significantly higher than shown in the example.

Similarly, high-level cycle efficiency estimates of future hybrid plants using CSP to gasify biomass feedstocks, as described in Section 2.5.3.2, are possible. Assuming that such plants would consist of a Brayton cycle with a bottoming Rankine cycle system their net efficiency could reach up to 60% and potentially more if suitable fuel cells were available. However, these estimates again assume the introduction of future innovations, including in gasification efficiencies, syngas cleaning, syngas compression, and improvements to gas turbines and the aforementioned Rankine cycle.

Generally, higher Rankine cycle efficiencies in CSP plants lead to lower specific plant costs due to the solar field size reduction. This is relevant as the solar field is the most expensive plant component (IRENA 2012). However, the combined cost impact of higher efficiency technologies and learning from technology deployment is a challenge as various factors, such as cost trajectories for new materials, are inherently uncertain. Figure 32 provides an example of a 100 MWe CSP–EfB hybrid plant at Mildura. It is apparent that the range of possible cost impacts widens with rising steam parameters. At current material costs, a plant with live steam parameters of 700 °C and 350 bar would be 30% more expensive than a standard power plant, but assuming future cost reductions for currently exotic materials, such as Alloy 740 and 740H (Patel et al. 2013), through deployment and ongoing R&D has the potential to lower future plant cost by 10%. Additionally, ongoing learnings from technology deployment, new construction and finance methods, supply chain improvements, and R&D could contribute to further CSP cost reductions of up to 40% by 2020 (IRENA 2012).



Figure 32: Potential efficiency increase (black line) and range of cost impact (red dotted lines) based on future steam parameters for a 100 MWe (net) CSP–EfB hybrid plant with air cooling at Mildura, Australia¹²

Ongoing global R&D will further improve the efficiency of future power plants and in light of the multitude of factors which can affect cost, a reliable estimate of CSP–EfB and CSP– EfW hybrid plant efficiencies and costs over the next several decades is a complicated task and the large number of uncertainties make any prediction very unreliable. This one of the reasons why this research project concentrated on the use of current technologies that have a maturity that is sufficiently high to obtain finance. Obtaining finance is crucial for near-term CSP deployment in Australia and elsewhere. Technology trajectories are required to convince governments and financiers of the benefits a technology such as CSP can provide. However, basing one's investment decisions on predictions about the future efficiency and cost of CSP decades ahead might be an unwise use of resources as competing

¹² Data derived from Thermoflex, version 23, modelling based on an altitude of 50 m, ambient temperature of 25 °C, single steam reheating, air-cooling and a design DNI of 800 W/m².

technologies, such as PV with battery storage, might mature more quickly than currently expected, thereby making CSP commercially uncompetitive. Long-term predictions can also raise false expectations and cause a slowdown in near-term technology deployment as investors and other stakeholders delay committing to projects on the expectation that lowcost renewable energy technologies will become available in the future. Hence, a fine balance between the learning opportunities derived from technology deployment and further R&D is required, and technology trajectories for the next decade should be based on trends that can be observed currently, such as current R&D efforts in thermochemical conversion technologies and materials.

4.5 Implementation barriers

Renewable sources contributed 10% of Australia's total electricity generation in 2010–11 (Bureau of Resources and Energy Economics 2013). This is significantly lower than the global average of 20% (International Energy Agency 2013) despite several excellent resources, such as solar irradiance, plentiful wind and geothermal sites. Considering that hydro is the dominant source of renewable electricity in Australia, and that most hydro installations have been in operation for decades it becomes apparent that growth in new renewable electricity generation has been small and therefore a set of barriers must be limiting its uptake.

Several studies have identified renewable energy barriers worldwide (Moomaw et al. 2011; Painuly 2001; Verbruggen et al. 2010) and for Australia (Effendi & Courvisanos 2012; Martin & Rice 2012) but this study focuses specifically on CSP-only and hybrid plants. Multiple barrier definitions are available and this thesis uses the definition from Wilkins (2002) which states that barriers are "factors that impede the adoption of a new technology" (Wilkins 2002, page 120). At times barriers are also referred to as market failures (Wilkins 2002), but this terminology refers more specifically to a lower market penetration of a certain technology than would be most economically efficient (Gillingham & Sweeney 2012). Barriers to renewable energy, as well as to energy efficiency and low carbon technologies, can have a social, technical, environmental, economic or policy background. They can cover more than one of the aforementioned categories and are often interrelated. For example, policy uncertainty or technical maturity influence finance costs. To accelerate renewable energy implementation a clear understanding of the current barriers is essential. Some barriers (such as higher LCOE for renewables due to the failure to consider externalities in the pricing of electricity from fossil fuels) are universal. Other (such as policy settings) are country-specific.

Changing the current fossil fuel dominated electricity market to a more sustainable one is crucial to limit anthropogenic climate change and its negative effects on Australia (Reisinger et al. 2014). Renewable energy technologies can achieve this and recent studies investigating 100% renewable electricity scenarios for Australia have concluded that CSP is an important technology for achieving this goal (Elliston, Diesendorf & MacGill 2012; Elliston, MacGill & Diesendorf 2013; Wright et al. 2011) but currently there are no CSP-only plants in Australia. Two significant projects were offered state/federal funding: AU\$ 464m for the 250 MWe SolarDawn project and AU\$ 60m for the 40 MWe SolarOasis project, had their offers withdrawn as neither project developer was able to secure the remaining funding they needed (Edwards 2013; Kelly 2012). CSP hybrids have been more successful:

one plant is already in operation at the Liddell power station (Novatec Solar GmbH 2012b), another one is under construction at the Kogan Creek power station (CS Energy 2011), and some, such as Collinsville and Swanbank, are being investigated at different levels of detail (Australian Renewable Energy Agency 2013a; Peterseim et al. 2012b). From these observations it can be concluded that hybrids seem to have lower implementation barriers and this study aims to identify the key barriers for CSP hybrid plants in Australia and to ascertain the extent to which they differ from barriers for CSP-only plants. This knowledge is needed to make predictions about the potential to accelerate CSP uptake in the current Australian environment.

The development, design, construction and operation of CSP plants and power plants in general requires people with various skills, and some of these people will have different views on specific aspects of a project. For example researchers, engineers and financiers are likely to have different perspectives on risks. To take account of these variations in perspective, people with different backgrounds were invited to the workshop, including researchers, plant owners/operators, consultants, technology providers and government representatives. Plant financiers did not attend but power plant owners/operators have financial expertise. The participant breakdown is shown in Figure 33 and the results which follow are based on their individual ratings.



Figure 33: Participant breakdown for implementation barrier ranking

4.5.1 Significant Australian barriers to CSP

Implementation barriers are multi-faceted and cover a variety of aspects, such as social, technical, environmental, economic and policy, and they often influence each other. A clear understanding of the different barriers is required to accelerate CSP-only and hybrid implementation in Australia and this assessment considers a range of barrier categories and individual barriers. The same barriers are likely to exist in different markets worldwide.

4.5.1.1 Social barriers

Several social barriers exist for CSP installations and this assessment compares eight, five of them were presented to the workshop and three were added by the workshop participants. <u>Consumer fear of higher energy prices</u>: Since2008 years electricity prices in Australia increased significantly (see Figure 5 on page 8) and naturally people are concerned with the higher cost of living, not only the cost of electricity. However, the reasons for the price increases have to be communicated transparently. In Australia, for example, 50% of the price increases were driven by significant investments in distribution and transmission infrastructure, while the large-scale renewable energy target is responsible for 7% of the increase (Australian Energy Market Commission 2011). Information about the historic and current electricity price composition, as well as future predictions, is available but its transparent dissemination is key to gaining, maintaining and increasing public support for renewable energy.

Lack of confidence in the technology: This barrier was added by workshop participants. In Australia the CSP track record is poor, with several high profile projects being abandoned in the four three years, such as the 250 MWe SolarDawn (Kelly 2012) project, the 40 MWe SolarOasis project (Edwards 2013) and the 10 MWe Cloncurry project (Michael 2010). On the other hand PV projects, such as the 10 MWe Greenough River Solar Farm in Western Australia (Verve Energy 2013), the 102 MWe Nyngan project and the 53 MW Broken Hill project in New South Wales, have gone ahead (Hobson & Winsbury 2013). Only one small 18.3 MWth CSP hybrid plant is currently in operation at the coal fired Liddell power station (AREVA Solar 2012b; Novatec Solar GmbH 2012a) with a significantly larger 44 MWe equivalent project under construction at the Kogan Creek power station (CS Energy 2011). However, worldwide around 50 commercial CSP plants with capacities from 5 to 280 MWe are operating successfully and these provide confidence that the technology would also work in Australia.

<u>Lack of accessible data</u>: This barrier was added by workshop participants. A variety of data is available for CSP systems in the public domain, especially in journal and conference publications, but some of them require subscriptions and are therefore not easily accessible for everyone who is interested in this field. However, several easily accessible and high quality reports with global and country-specific information exist on the internet – for example on the International Renewable Energy Agency's website (IRENA 2012) and on the Australian Solar Thermal Electricity Association's website (Lovegrove et al. 2012). Lack of, or poor quality of, information, education and awareness: Information about the need to move to a low carbon future and the provision of this information to the general public is vital for implementing renewable energy and energy efficiency systems. A vast body of literature is available in the public domain but not all of it can be understood by people who do not work in the energy sector. On the other hand the complexity of the field means that there is a limit how much the information can be simplified without neglecting relevant facts. Education programs can increase awareness about renewable energy and are available through several institutions, ranging from day care centres, schools, universities, private and public organisations, to internet-based sources. Providing information rather than estimates and this not only raises awareness of the technologies but also influences other social barriers, such as technology acceptance, fear of electricity prices and fear of change. Similarly, Australian case studies would help familiarise the community with the technology.

<u>Fear of change</u>: This barrier was added by workshop participants. The current centralised power generation regime in Australia is long established and therefore well understood. Many renewable energy and distributed generation plants challenge this current regime and introduce not only new generation assets, small base-load and intermittent generators, but also new ways to transport energy, such as smart grids. These changes can cause uncertainty for individuals, organisations and communities in regards to employment, new infrastructure, and cost of living. The resistance to technical change in organisations has been investigated and some observers have concluded that communication of the changes and their benefits is a key to avoiding speculation, half-truths and uncertainty (Cunningham, Farquharson & Hull 1991).

<u>Technology acceptance</u>: Some renewable energy technologies face opposition for a variety of real or perceived reasons. In the case of wind farms this includes concerns over low-frequency noise, landscape change and visual impact on amenity, and merely providing facts does not solve the problem as people make judgments based on their own perspectives and circumstances. However, the negativity displayed in the media and by some interest groups towards certain technologies such as wind can be quite different to the actual level of community acceptance (Hall, Ashworth & Shaw 2012). Extensive community engagement is required to understand and address local concerns and increase technology acceptance. Due to the limited reference plants and hence the limited local experience with CSP, the technology can encounter similar acceptance problems and lawsuits have already been brought against project developers in the USA claiming that

"the impacts to Native American culture and the environment are extraordinary" (Woody 2011). Some lawsuits have successfully obtained injunctions halting CSP projects and have even caused developers to withdraw their proposals.

<u>Impact on visual amenity</u>: This barrier was added by workshop participants. Visual impact is not only an environmental barrier, see 4.5.1.3, but also has social implications as it clearly shows the presence of a renewable energy plant and reminds people constantly about this facility. It needs to be mentioned though that parabolic trough, Fresnel and dish plants do not have tall structures and therefore they involve a minimal visual impact. Solar tower systems on the other hand are, due to tower heights of over 100 m, clearly visible from the distance. People's perceptions of the landscape change and the visual impact on amenity varies significantly and the wind industry provides a good example where responses to the presence of wind turbines range from enthusiastic acceptance to strong rejection. In Queensland one solar tower project was withdrawn in 2010 due to solar glare issues (Michael 2010), which shows that site selection and technology selection have to be considered carefully.

<u>Consumer lack of interest in renewable energy:</u> Uncertainty about the state of the global economy and about the future prosperity of individual countries can cause consumer priorities to shift, with the result that employment and cost of living concerns may rank higher than concerns about climate change. A recent USA survey found that the economy and job creation are the concerns that rank highest for most people, with environmental protection ranking 12th and climate change ranking only 21st (Dimock, Doherty & Kohut 2013). While in 2013 the majority of Americans (52%) believed environmental protection should be a priority the issue was considered more significant in 2001 (63%), before the global economic crisis. Similarly, in Australia employment concerns come to the fore when people discuss renewable energy, climate change and carbon pricing, particularly in fossil fuel dominated areas, such as the Latrobe Valley in Victoria (Carey 2011). Therefore, short-term public concerns have to be addressed carefully when communicating the scientifically agreed need for renewable energy implementation and its socio-economic benefits.

4.5.1.2 Technical barriers

Technical barriers cover a wide field ranging from network access to technology maturity and codes and standards. This assessment compares seven barriers, six of which were prepared for the workshop and one of which was added by the workshop participants.

<u>Network access, capacity and availability:</u> The Australian transmission network is designed to transport electricity from centralised facilities to consumers and does not necessarily cater for the addition of many distributed generators located close to demand centres. The connection costs for new CSP plants without TES varies from 8% to 11% of the total investment (Rutovitz et al., 2013) which makes it a high cost item for project developers. However, costs can be lowered by adding CSP plants to grid-constrained locations where they can defer or even remove the need for network augmentation (Rutovitz et al., 2013). In addition to connection costs, network availability is a constraint with almost no transmission lines for utility-scale plants in the best direct normal irradiance regions of Australia (Lovegrove et al. 2012). However, smaller systems, of less than 30 MWe, have better access to the existing network as they can connect to lower capacity transmission lines (Lovegrove et al. 2012).

<u>Technology maturity</u>: Compared to other renewable and fossil fuel technologies the estimated global CSP deployment of 4 GWe in 2013 (International Energy Agency 2013) shows that the technology is not as widely deployed as other renewable alternatives. However, significant differences exist between the different CSP technologies, and parabolic trough plants are the most mature with commercial operating experience since the 1980s. Solar towers are maturing quickly with various 20–130 MWe plants in operation (BrightSource Energy Inc. 2013; García & Roberto Calvo 2012). Also, Fresnel systems operate successfully at 30MWe scales (Novatec Solar GmbH 2012c) and a 2x 125 MWe unit is under construction (AREVA Solar 2012a). The continuous operation and commissioning of further plants will provide better information about tower and Fresnel systems and will enable a more realistic technology comparison based on actual performance rather than modelled data.

<u>Lack of accessible technical data</u>: This barrier was added by workshop participants. Little detailed technical data is available about the operational performance of individual CSP plants as this information is regarded as commercially sensitive by the plant owners. A multitude of assessments have investigated the technical and economic performances of CSP plants and sophisticated software tools exist for project developers to model plant performance at specific locations, such as the System Advisory Model by NREL¹³ or Thermoflex by Thermoflow. With no CSP-only plants in Australia all information is based on models and actual detailed information, including compliance with Australian standards, is not available.

<u>Technology complexity</u>: Due to the spread of CSP systems over large areas, the significant number of reflectors, and the use of TES equipment the technology is more complex than

¹³ https://sam.nrel.gov, accessed 04 April 2014

current coal and gas plants, PV installations and wind generators. However, the solar field components are standardised units, the power blocks are similar to those used in other Rankine cycle plants and the TES system has been proven in many CSP references with up to 15 h full-load storage capacities (García & Roberto Calvo 2012). Also, high degrees of automation, such as solar tracking, reduce operational risk and costs.

<u>Generation intermittency/lack of energy storage</u>: PV and wind generators only generate electricity during times of sufficient solar irradiance and wind speed, while CSP systems can store energy during the day and shift output to higher demand times. One 20 MWe plant with 15 h TES in Spain demonstrated 24-hour operation (García & Roberto Calvo 2012), several other operational plants have 7.5 h of TES, and a new system will commence construction this year with 17.5 h thermal storage (Abengoa Solar 2014a). Having the ability to store energy and being flexible in plant size gives CSP a unique advantage as currently no other renewable technology in Australia can combine high capacity factors with utility scale. PV and wind have utility scale while biomass and geothermal plants have high capacity factors but none have both. Therefore, several studies that have investigated high penetration renewable electricity scenarios in Australia which allocate a significant future generation capacity to CSP (Elliston, Diesendorf & MacGill 2012; Wright et al. 2011).

Lack of standards and codes: The international ASME and Australian Standards cover all relevant aspects of designing a CSP plants according to local requirements, such as pressure vessel, piping and boiler codes. These codes must have already been applied to the two CSP hybrid plants in Australia and can be used for future CSP-only plants too. With little CSP expertise in Australia, approving a compliant design will take slightly longer as engineers will have to apply codes and standards to unfamiliar components. However, they are familiar with requirements for components such as pressure vessels and control systems for significantly more complex systems, such as gas processing plants, and should be able to apply this knowledge quickly.

<u>Resource assessments/knowledge of best locations</u>: Knowing where to build CSP plants is essential to maximise economic viability and several reports have identified various prospective regions for CSP plants across Australia. While hybrid plants are discussed in the Australian context (Evans & Peck 2011; Lovegrove et al. 2012; Meehan 2013; Peterseim et al. 2014d) few analyses have identified potential regions and specific opportunities for CSP hybrids in detail (Meehan 2013; Peterseim et al. 2014d). The combination of two or more energy sources adds another layer of complexity to site identification for hybrid plants but modern geospatial software is capable of doing this, as demonstrated by Peterseim et al. (2014b) for CSP–EfB and CSP–EfW hybrid plants.

4.5.1.3 Environmental barriers

With CSP systems having to comply with the same environmental approval processes that other generators, they too can encounter environmental barriers. This assessment considers four different barriers.

<u>Water consumption:</u> Typically, water is a scarce commodity in high DNI areas and a power station would compete for it with other users, such as agriculture, or it can encounter significant limitations, such as legislatively regulated minimum environmental flows. Using dry- rather than wet-cooling systems can reduce water consumption by more than 90% (Turchi et al. 2010) and modern mirror cleaning robots can significantly reduce cleaning water requirements (Novatec Solar GmbH 2011; Vicente et al. 2012), which is a small water user anyway (Turchi et al. 2010). Water scarcity is apparent in Australia as it is one of the few countries with air-cooled utility scale coal fired power plants such as the Kogan Creek power station (CS Energy 2013).

<u>Impact on visual amenity</u>: Typically, CSP plants are in remote locations with limited population centres, flora and fauna. The plant components for line focusing and dish systems are no higher than for other industrial facilities and only tower systems have a more significant visual impact through the tower structure itself and the highly illuminated receiver. Most other components are similar to other Rankine cycle systems. The benefit of CSP systems is that they can be located in remote locations as substitutes for power plants and fuel production systems close to urban centres. This eases visual amenity problems (EASAC 2011).

<u>Plant footprint/land use:</u> In comparison to other renewable developments such as biomass plants, and fossil energy systems such as coal plants, the physical footprint of a CSP power plant is significantly larger. However, this is different when one includes fuel production, such as biomass plantation and coal mining, and CSP compares well with land requirements of 17 m²/MWh/year compared to 550 m²/MWh/year for biomass and 60 m²/MWh/year for coal (EASAC 2011). The plant footprint varies significantly between different CSP technologies (Müller-Steinhagen & Trieb 2004) but the generally large power plant footprint can, amongst other problems, lead to compacting of larger areas during construction, cutting dispersion routes for animals and potential isolation of regional animal populations during construction and operation (EASAC 2011). The local flora and fauna in desert and semi-desert environments needs longer to recover from disturbances and therefore careful consideration has to be given to site selection and plant construction. Several CSP projects have already been halted and withdrawn due to community concerns about the environmental impact on fragile landscapes (Woody 2011). Birds could be affected if they mistake the large reflective areas for air or water. They may sustain heat shocks or burning damage in the concentrated light beams, or they may collide with the solar tower structure. However, birds rarely collide with CSP plants. Casualties have been documented, but these are very limited with only two recorded bird deaths between 2007 and 2011 at a monitored tower plant in Spain (EASAC 2011).

<u>Inefficient use of back-up fuels</u>: All CSP plants have a back-up system for start-up, emergency or part-load load operation. These systems are not intended to operate over long periods and some countries limit their contribution to marginal energy inputs. Natural gas is the predominant back-up fuel and using this fuel in a modern CSP plant allows conversion efficiencies of around 40% while a combined cycle gas turbine plant can achieve 60%. The natural gas consumption in back-up systems should therefore be minimised as designated natural gas plants could generate up to 50% more electricity from the same amount of fuel. Hybrid plants with larger shares of fossil or renewable fuels take this into consideration and are designed for maximum efficiency for both types of fuels. Integrated solar combined cycle plants are an example of this.

4.5.1.4 Economic barriers

Despite significant investment increases in renewable energy systems in recent years they still face economic barriers and this assessment considers six, with five prepared for the workshop and one added by the workshop participants.

<u>High project capital/financing cost and typically large projects:</u> Despite significant cost reductions through CSP deployment, innovation and associated learning the specific cost is still comparatively high in Australia e.g. 100 MWe in a 2,400 kWh/m²/year DNI location without TES AU\$ 4.65/MWe and AU\$ 7.35/MWe with 5h TES (Lovegrove et al. 2012). By 2020 costs could decrease by up to 40% (IRENA, 2012) but this would require continuous deployment and technological improvements. The solar tower systems that commenced operations recently, and the parabolic trough plants with molten salt will provide actual information on the economic impact of technological improvements, particularly in regards to high-temperature TES. To obtain economies-of-scale large plant capacities are required and current plant capacities reach 280 MWe, for example in the Solana plant (Abengoa Solar 2013b). The trend towards large capacities, coupled with comparatively high investments, quickly leads to hundred million or even billion dollar project budgets. Inherently, this complicates project finances and adds a risk premium. Hybrid plants have

an advantage in this regard as even small capacities can be added to existing plants, such as only 9.3 MWth at the Liddell project (Novatec Solar GmbH 2012b).

Lack of financial incentives: Despite the current existence of a carbon and renewable energy certificate price in Australia there is still a cost gap for CSP plants which varies according to location from AU\$ 110-115+/MWh for 50-250 MWe plants connected to the NEM and from AU\$ 10-50+/MWh for 1-10 MWe plants in mini/off grid locations (Lovegrove et al. 2012). This clearly shows that in Australia further incentives are required to implement CSP systems and the lack of CSP-only plants proves this. Plants in mini/off grid locations are much closer to market viability and the lower specific investment for hybrid plants relative to CSP-only plants has resulted in two CSP-coal hybrids projects. Further incentives to stimulate deployment include feed-in tariffs, loan guarantees or government grants.

Lower cost renewable energy alternatives: Currently, other forms of renewable energy, such as wind or PV, are lower cost options with more capacity deployed and therefore more learning experience. Both technologies are further advanced and it is therefore likely that future cost reductions will be lower than for CSP, assuming continuous CSP deployment. The cost structure is different when comparing technologies based on their annual generation and dispatchability, since intermittent renewables currently rely on high cost battery storage. Considering that CSP has mature thermal storage its costs are expected to be lower than PV with battery storage in 2025 (AT Kearney & ESTELA 2010). With larger proportions of intermittent renewable energy connected to the electricity grid, dispatchable power is essential to ensure stable grid operation and this provides a price premium for technologies that can dispatch electricity during high demand times. CSP with TES can do this and various high renewable energy scenarios therefore include significant CSP contributions in the future (Elliston, Diesendorf & MacGill 2012; Wright et al. 2011).

<u>Abundant and subsidised fossil fuel resources:</u> Australia not only has abundant coal, gas and uranium but also solar, wind and geothermal resources. While financial incentives exist for renewable energy and were estimated to be between AU\$ 317-334m in 2005-06, these were significantly lower than the AU\$ 9.3–10.1b for fossil fuel energy and transport subsidies for same period (Riedy 2007). More recent information indicates that by 2011 fossil fuel subsidies increased to AU\$ 12.2b with AU\$ 1.1b spent on clean energy programs (Morton 2011). This development is not confined to Australia as in 2011 global fossil fuel subsidies were estimated to be US\$ 523b, an increase of nearly 30% compared to 2010 and six times the support provided to renewables (International Energy Agency 2012b). Correcting this mismatch could boost renewable deployment by providing a level playing field for all power generation technologies.

<u>Lack of externality costing</u>: This barrier was added by workshop participants. Currently, Australia has a carbon emission price of AU\$ 24.15/t internalising some of the environmental costs incurred through the use of fossil fuels. However, recent developments aiming to abandon this mechanism (The Parliament of the Commonwealth of Australia 2013b) or linking the scheme to Europe's lower cost, AU\$ 6-10/t, emission trading market (Peatling 2013) will harm the business case for renewable energy and slow project deployment.

<u>Pre-existing investment in existing equipment and infrastructure:</u> Companies that invested in generation infrastructure need to operate their plants for the lifetime of the plant to obtain the expected return on investment. This particularly affects recently commissioned power stations which have not amortised themselves yet, such as the 750 MWe Kogan Creek coal fired power station commissioned in 2007 (CS Energy 2013). One way to future proof such assets and reduce their environmental footprint is to retrofit or even convert them to renewable energy and the 44 MWe Kogan Creek Solar Boost project is one good example of this (CS Energy 2011). However, the majority of Australia's coal fired power station fleet was built in the 1970s and 1980s and considering a design life of 40 years these plants are about to reach/have reached the end of their design life. This provides a unique opportunity to increase renewable energy capacity significantly over the next few years but also poses a risk as the construction of new fossil plants would lock in such generation for the next decades, leading to a lower renewable energy uptake or stranded assets.

4.5.1.5 Policy barriers

Worldwide, governments try to stimulate the uptake of renewable energies and reduce carbon emissions. A variety of measures were introduced but to significantly increase private capital the right policy settings are required. This assessment considers seven barriers in the policy space. Six of them were prepared for the workshop and one was added by the workshop participants.

<u>Renewable energy policy uncertainty</u>: Australia has an unconditional 5% greenhouse gas reduction target for 2020 (Department of Climate Change and Energy Efficiency 2012) but the government's and opposition's strategies to reach it are different. The previous government introduced a carbon pricing mechanism (Australian Government 2011) but the newly elected one plans to repeal this mechanism and instead implement a direct action policy (Commonwealth of Australia 2013). Only some details are available about the direct action policy and adding to the current uncertainty is another review of the 45 TWh RET (Hunt & Macfarlane 2014) after it was left unchanged in 2012 (Climate Change Authority 2012). Also, there have been discussions about lowering the RET to accommodate the current decrease in annual electricity demand (Fitzgibbon 2012). Such policy uncertainty is not only an Australian problem as other countries, for example Spain, have reduced or are planning to reduce renewable energy support (Crespo 2013). This is particularly significant for CSP as Spain has been its strongest growth market in recent years (IRENA 2012).

<u>Lack of government policy focus</u>: Traditionally, the coal, oil and gas industries are strong in Australia and in the context of slowing economic growth individual multi-billion dollar projects might be more appealing to government decision-makers than smaller renewable energy projects. While renewable energy projects can create significant new employment (Caldés et al. 2009; Lucas et al. 2011; Wei, Patadia & Kammen 2010) these benefits have to be communicated to decision-makers to ensure that these facts are considered when policies are drafted. Schläpfer (2009) argues that:

policy makers in Australia appear to believe unreservedly that we can only find solutions to the technical challenges associated with fossil fuel and nuclear technologies and equally believe that we cannot find similar solutions to the technical challenges associated with renewable energy technologies (Schläpfer 2009, p. 459).

One observation supporting a bias towards fossil fuels could be seen in the 2009–10 budget where AU\$ 2b of the AU\$ 3.5b clean energy initiative was allocated to carbon capture and storage, and AU\$ 1.5b went to CSP demonstration projects (Australian Government 2009).

<u>Market concentration</u>: In Australia three energy retailers provide 75% of the electricity in the NEM which complicates power purchase agreements (PPA) and market access for small project developers (Hannam 2013). The current renewable energy target does require all retailers to provide a total of 41 TWh from renewable energy sources by 2020. This amount plus the 4 TWh allocated for small renewable energy generators would be 20% of the previously expected annual electricity generation (Climate Change Authority 2012) but there are no earlier required milestones. Not having earlier milestones poses the threat that project development efforts are being delayed towards the end of this decade. This would leave smaller developers without a viable business case for the intervening period and creates a boom and bust cycle for the industry.

<u>Lack of coordination at the state and national levels</u>: Different financial incentives exist at the federal and state levels in Australia, such as the current federal carbon pricing mechanism and PV feed-in tariffs in Queensland (Queensland Competition Authority 2013).

This increases project development complexity and uncertainty as state legislation can be changed more quickly than federal legislation. The incentives also vary with plant size. There are feed-in tariffs available for household PV installations (Queensland Competition Authority 2013) but there is nothing for large scale systems, such as CSP. Without continuous reviews the large scale fraction of the renewable energy target could provide a valuable incentive for CSP but the current renewable energy certificate price is insufficient.

<u>Regulators not having sufficient familiarity with the technology:</u> This barrier was added by workshop participants. Regulators and policy makers are very familiar with fossil fuel projects in Australia as many of them have been realised over the last five decades. However, they are significantly less experienced with renewable energy projects given that renewable energy only contributed 10% of total electricity output in 2010-11 and two thirds of this came from existing hydro assets (Bureau of Resources and Energy Economics 2013). With further renewable energy systems being deployed in the future, regulators and policy makers will gain more experience and can in the meantime draw on overseas experience in all forms of renewable energies, particularly from Spain and USA for CSP.

<u>Complex plant compliance processes:</u> In Australia large and small generators have to comply with similar legislations which makes the compliance costs for a GW-size plant a less significant part of the overall project budget but a more significant one for smaller renewable energy plants. In Queensland for example, renewable energy projects have to comply with the Environment Conservation and Biodiversity Conservation Act (1999), the Integrated Planning Act (1997), the Environmental Protection Act (1994), the Electricity Act (1994) and further land-use regulations and approvals (Martin & Rice 2012). All generators, including renewable energy projects, have to ensure a minimal environmental impact and need to be assessed accordingly but streamlining the process is important as compliance costs on an AU\$ per MWe basis are higher for smaller plants. Queensland's 'Green Door' initiative aims to deliver this through joint state and municipal council case management teams assessing ecological, economic and community wellbeing aspects (Queensland Government 2011). Such initiatives are required to support project developers.

<u>Lack of a skilled workforce</u>: The Australian unemployment rate was 5.9% in January 2014 (Australian Bureau of Statistics 2014). This is comparatively low compared to other countries and with CSP plants typically being remote, their owners are likely to struggle for qualified operators as the mining industry attracts people to work in remote locations with higher wages. Training residents in high DNI urban centres, such as Roma in Queensland or Whyalla in South Australia, is key to providing qualified operators and diversifying employment opportunities in these locations. People are likely to accept a local jobs with a

slightly lower remuneration rather than fly-in fly-out jobs that separate them from their families. The workforce required to design CSP plants is available in Australia even though some components, such as reflector or receiver details, would have to be designed overseas. Highly qualified process and civil engineers are available and their knowledge from power, oil and gas, and mining projects can be applied to CSP projects.

4.5.2 Rating results

The participants' ratings for the barriers for CSP-only and hybrid plants vary significantly and Figure 34 shows the results. All barriers selected were considered significant with no barrier ranked 1 (not important) and the lowest barrier ranking being 3.8 of a maximum of 9 (extremely important). While some participants rated individual barriers with the highest score of 9 no barrier reached such a high rating on average, which is why the barrier importance scale in Figure 34 ends at the rating of 8.

This section outlines the barrier ranking results, compares the individual barriers and barrier categories, and shows the rating differences between CSP-only and hybrid plants.



Figure 34: Barrier rating results for CSP-only (orange) and CSP hybrid plants (green)

4.5.2.1 CSP-only plants

The results for the individual barrier ratings are shown in Figure 34 and the category ratings with the participant group differences are shown in Figure 35.

The top five barriers identified for CSP-only plants in Australia are: 1) high project capital cost (average rating of 7.8 out of a maximum of 9), 2) lack of financial incentives (7.68), 3) renewable energy policy uncertainty (7.65), 4) lower cost renewable alternatives (7.3), and 5) abundant and subsidised fossil fuels (7.1) (see Figure 34). Pre-existing investment in equipment and infrastructure is considered the least important economic barrier (6.2). Four of the five top barriers are economic and Figure 35 shows that despite rating differences between the workshop groups all consider economics as the main barrier category. Based on the group ratings, owners/operators rated economic barriers as the most important (7.9) followed by technology providers (7.6) and government (7.5). Only researchers ranked the economics category significantly lower (6.5).

The second-most important category of barriers is policy with renewable energy policy uncertainty (7.68), lack of government policy focus (7.0) and market concentration (7.0) being the most significant. The technology providers (7.0), owner/operators (6.9) and government representatives (6.8) considered the policy category to be more significant than the group average (6.6). Lack of skilled workforce (5.3) was the least important policy barrier but was rated more significant by the owner/operator group (6.0).

On average technical barriers emerged as the third-most important barrier category with network access, capacity and availability (6.53) as well as technology maturity (6.45) considered to be the most important followed by lack of accessible technical data (6.1). Resource assessments and knowledge of the best locations was the least important technical barrier (4.9). Owners/operators rated the technical barrier category significantly more important (6.7) than the group average (5.7) with technology providers ranking it least significant (5.3).

Social barriers were rated the second-least important barrier category (5.5) with significant rating differences for the individual barriers. Fear of higher electricity prices was considered highly important (6.9), followed by lack of confidence in the technology (6.2) and lack of accessible data (5.9). Impact on visual amenity and consumer lack of interest in renewable energy were among the least important barriers with ratings of 4.9 and 4.3 respectively. Plant owners/operators and technology providers were the participant group that ranked the social barrier category highest (6.1) while the government group considered them significantly less important (5.1).

Not unexpectedly the environmental barriers were considered least important and all workshop groups rated them low (4.9 to 5.1). Water consumption was considered to be the most significant environmental barrier (5.8) and the inefficient use of back-up fuels least important (4.3).



Figure 35: Barrier category ratings for CSP-only plants

4.5.2.2 CSP hybrid plants

The barrier category ranking for CSP hybrids in economic, policy, technical, social and environmental categories was similar to CSP-only plants and on average all barriers were considered less important for CSP hybrids (see Figure 34 and Figure 36). The top five barriers for CSP hybrids are: 1) policy uncertainty (6.9), 2) lack of financial incentives (6.7), 3) abundant and subsidised fossil fuels (6.6), 4) lack of government policy focus (6.5), and 5) high project capital cost (6.4). While four of the top five barriers are identical to CSP-only plants, their ratings were different and the barrier 'lack of government policy focus' (6.5) replaced the barrier 'lower cost renewable alternatives' (6.3). This demonstrates that the cost benefits CSP hybrids offer are recognised by a wider group of people.

Identical to CSP-only plants the economic barrier category was on average ranked most important (6.4) with the owner/operator group rating this category above average (7.1) and researchers below average (6.0). All groups considered the economic barriers less significant for CSP hybrids than for CSP-only plants.

Renewable energy policy uncertainty is considered the most important barrier (6.9) followed by lack of government policy focus (6.5) and lack of coordination at the state and federal levels (6.4). The lack of a skilled workforce was again considered the least significant policy barrier (4.8) and was again rated significantly higher by the owner/operator group (6.0). Of all groups the owners/operators considered the policy barrier category most

important (6.7) with an almost identical rating from the government group (6.6). Only the consultants ranked it significantly lower (5.5).

The technical barriers were the third-most important category (5.0) and owners/operators rated it significantly more important (6.5) than the other groups (4.7 to 5.4). Technology maturity was the most significant technical barrier (5.8) followed by lack of accessible technical data (5.6), and technology complexity (5.3). Generation intermittency/lack of energy storage was not only rated the least important technical barrier but also the least important barrier of all (3.8).

Of all participants the owner/operator and technology provider groups considered the social barrier category significantly more significant (5.8) than others (4.6 to 4.7) which was similar to CSP-only plant rating. Also similar to the results for CSP-only plants, fear of higher electricity prices (6.1), lack of confidence in the technology (5.8), and lack of accessible data (5.8) ranked most important within the social barrier group.

The environmental barrier category was also least significant for CSP hybrid plants with the government group rating it more important (4.9) than the other groups (4.2 to 4.5). The most important barrier in this category was again water consumption (4.9) while the impact on visual amenity was considered the least important environmental barrier (4.2).



Figure 36: Barrier category ratings for CSP hybrid plants

4.5.2.3 CSP-only versus hybrid rating differences

On average all implementation barriers were considered less significant for CSP hybrids than for CSP-only plants but rating differences varied from marginal to significant. Table 2 shows these differences and due to the different backgrounds of the workshop participants/interviewees barrier rating differences can be observed. Technology providers (Δ TP) for example ranked most barrier categories lower for hybrids than the other groups while owners/operators (Δ O) also ranked them lower but to a smaller extent.

Table 2: Ranking differences between CSP-only and hybrid plants showing total average and group averages for researcher (ΔR), owners/operators (ΔO), consultants (ΔC), technology provider (ΔTP) and government (ΔG)

| Barrier | Barrier | | A Total | | 40 | AC | ATD | AG |
|----------|--|---------|---------|------|------|------|------|------|
| category | | A TOTAL | Δ TOtal | Δn | 70 | 20 | | 20 |
| | Consumer fear of higher energy prices | -0.8 | -12% | -1.2 | -0.8 | -1.2 | -0.3 | -0.2 |
| | Lack of confidence in the technology | -0.5 | -7% | -0.8 | 0 | -0.3 | -0.2 | -1 |
| | Lack of accessible data | -0.1 | -2% | -0.1 | 0.3 | -0.2 | -0.7 | 0.3 |
| | Lack/quality of information, education and | -0.5 | -9% | -0.3 | -03 | -11 | 0 | -0.7 |
| 0 | awareness | | 570 | 0.5 | 0.5 | 1.1 | 0 | -0.7 |
| oci | Technology acceptance | -0.8 | -16% | -1.5 | -0.5 | -0.6 | -0.6 | -0.5 |
| S | Fear of change | | -10% | -1.1 | -0.3 | -1 | 0.5 | 0 |
| | Impact on visual amenity | -0.7 | -14% | -1.3 | -0.3 | -0.5 | -0.8 | -0.3 |
| | Consumer lack of interest in renewable energy | -0.3 | -8% | -0.2 | -0.3 | -0.8 | 0 | -0.3 |
| _ | Group average | -0.3 | -10% | -0.5 | -0.2 | -0.4 | -0.1 | -0.2 |
| | Network access, capacity and availability | -1.6 | -24% | -2.1 | -0.5 | -1.8 | -0.7 | -2 |
| | Technology maturity | -0.6 | -10% | -1.2 | 0.3 | -0.3 | -0.9 | -0.2 |
| | Lack of accessible technical data | -0.5 | -9% | -0.8 | 0 | 0.8 | 0 | 0 |
| | Technology complexity compared to other | -0.3 | -5% | -0.8 | 13 | 0.1 | -0.2 | -0.7 |
| ica | renewables, coal and gas | 0.5 | 570 | 0.0 | 1.5 | 0.1 | 0.2 | 0.7 |
| Techn | Generation intermittency/lack of energy storage | -1.3 | -25% | -1.7 | -2.8 | -1.1 | -1 | -0.2 |
| | Resource assessments/knowledge of best locations | -0.1 | -2% | -0.2 | 0.3 | 0 | -0.1 | -0.5 |
| | Lack of standards and codes | -0.1 | -2% | 0 | 0.3 | 0.1 | -0.7 | 0.2 |
| | Group average | -0.2 | -11% | -0.3 | -0.3 | -0.1 | -0.3 | -0.1 |
| Ital | Water consumption (cooling & cleaning) | -0.9 | -16% | -1.5 | -0.3 | -0.6 | -1.1 | -0.2 |
| ner | Impact on visual amenity Plant footprint/land use | | -16% | -1 | -0.3 | -0.7 | -1.2 | -0.3 |
| our | | | -14% | -0.6 | -0.5 | -0.7 | -0.7 | -0.8 |
| wir | Inefficient use of back-up fuels | -0.1 | -2% | 0.4 | -0.5 | -1.1 | 0.3 | 0.5 |
| <u> </u> | Group average | -0.6 | -12% | -0.7 | -0.4 | -0.8 | -0.7 | -0.2 |
| | High project capital/financing cost and typically large projects | -1.5 | -19% | -1 | -1.3 | -1.7 | -1.8 | -1.6 |
| | Lack of financial incentives | -1 | -12% | -1 | -1 | -1 | -0.9 | -0.8 |
| ji. | Lower cost renewable energy alternatives | -1 | -14% | -1 | -1 | -0.4 | -1.1 | -0.8 |
| Louo | Abundant and subsidised fossil fuel resources | -0.5 | -7% | -0.1 | -0.8 | -0.2 | -1.5 | -0.3 |
| ŭ | Lack of externality costing | -0.6 | -8% | -0.4 | -0.5 | -1 | -0.8 | 0 |
| 2 | Pre-existing investment in existing equipment and | -0.4 | -6% | 0.5 | -0.3 | -0.5 | -11 | .0.8 |
| | infrastructure | -0.4 | -070 | 0.5 | -0.5 | -0.5 | -1.1 | -0.8 |
| | Group average | -0.8 | -11% | -0.5 | -0.8 | -0.8 | -1.2 | -0.7 |
| | Renewable energy policy uncertainty, e.g. carbon pricing, renewable energy target | -0.8 | -10% | -1.2 | -1 | -0.5 | -0.8 | -0.3 |
| | Lack of government policy focus | -0.5 | -7% | -0.5 | 0 | -1 | -0.4 | 0 |
| | Market concentration limits market and PPA | | 4004 | | | | | 0.7 |
| <u>5</u> | access for smaller developers | -0.7 | -10% | -0.4 | -0.5 | -1.1 | -0.8 | -0.7 |
| Pol | Lack of coordination at state/national level | -0.3 | -4% | -0.2 | -0.5 | -0.6 | -0.3 | 0.2 |
| | Regulators not having sufficient familiarity | -0.3 | -5% | -0.4 | 0 | 0 | -0.5 | -0.7 |
| | Complex plant compliance processes | -0.2 | -3% | -0.2 | 0.8 | -0.2 | -0.7 | 0.2 |
| | Lack of skilled workforce | -0.4 | -8% | -0.1 | 0 | -0.7 | -1 | 0.2 |
| | Group average | -0.4 | -7% | -0.4 | -0.2 | -0.6 | -0.6 | -0.2 |

The highest barrier rating reductions between CSP hybrid and CSP-only plants can be observed for generation intermittency/lack of energy storage (25% less significant), network access, capacity and availability (24% less significant), and high project capital/financing costs (19% less significant) while the barriers with the smallest ranking differences were lack of codes/standards (1.5% less significant), inefficient use of back-up fuels (1.6% less significant), and resource assessment/knowledge of best sites (2.4% less significant).

The most significant ranking differences between the groups were for generation intermittency/lack of energy storage (ΔO 42% less significant compared to ΔG 4% less significant), technology complexity (ΔR 17% less significant compared to ΔO 19% more significant), and lack of accessible technical data (ΔR 12% less significant compared to ΔO 15% more significant). All three barriers are technical and the differences in the levels of researchers, technical expertise between government representatives and owners/operators are often large. Also owners/operators have to operate the plant over its lifetime while researchers are on the design side and therefore not exposed to financial and operational risks, which is a likely reason why they rated complexity lower. The barriers the participant groups ranked most closely are lack of financial incentives (ΔG 11% less significant compared to ΔR and ΔC 13% less significant), plant footprint/land use (ΔTP 12% less significant compared to ΔG 17% less significant), and lower cost renewable energy alternatives (ΔC 6% less significant compared to ΔTP 14% less significant). The responses for economic barriers show that economic barriers are uniformly considered important while the land use rating is likely to derive from the generally good land availability in Australia. However, this might be different in more densely populated regions, such as Europe.

4.5.3 Discussion

It is not unexpected that the economic and policy barriers were considered most important as they are key aspects in long-term decision-making. They are the most country-specific barriers because they are affected by federal/state legislation, electricity prices and other localised influences while the other categories are more universally applicable with overseas experience available on technical, social and environmental matters. A 2011 study on barriers to demand management (Dunstan, Ross & Ghiotto 2011) showed similar barrier ranking patterns with a lack of coordination at the state and federal levels ranked high, lack of information ranked average and consumer lack of interest in saving energy ranked low. This supports the trends identified in this study as barriers to energy efficiency and renewables are often similar.

The author believes that the reason the social barriers was considered most important by owners/operators and technology providers derive from their project development expertise and experience with community concerns. Similarly, the 'owners/operators' comparatively high technical barrier rating probably derives from knowing actual operational problems over a power station's lifetime. Some specific barriers are significantly lower for hybrid plants than CSP-only plants – for example generation intermittency/lack of energy storage, network access, capacity and availability, and high project capital/financing costs. Hybrid plants are likely to benefit most from improved energy dispatchability, lower cost and joint use of plant equipment. The minimal CSP hybrid to CSP-only rating differences for lack of codes/standards, inefficient use of back-up fuels, and resource assessment/knowledge of best sites are likely to be based on available codes/standards for all Rankine cycle systems, the small use of back-up fuels in CSP-only plants and maximum fuel efficiency expectations for hybrids considering rising fuel prices in Australia, as well as several available studies investigating suitable regions for CSP plants. While investment in pre-existing infrastructure was rated as important it is not in the top ten barriers. The reason for this is probably the significant average age of Australian power stations particularly coal-powered plants. Many units were built in the 1970s and '80s and are about to reach the end of their lifetimes. They will require replacement in the foreseeable future and therefore have a comparatively low asset value.

Due to the significantly lower economic and policy barriers for CSP hybrids they provide a good opportunity to break the detrimental CSP implementation cycle in Australia and elsewhere (see Figure 37) by reducing cost, thereby providing a higher incentive to build CSP systems and subsequently increase learning. The experiences gained with CSP hybrids are likely to raise confidence in the technology and investment in CSP and this will enable the construction of future CSP-only plants, thus breaking the vicious cycle.



Figure 37: Detrimental CSP implementation cycle with intervention option. Adapted from Effendi and Courvisanos (2012)

While on average all hybrid barriers ranked lower than average, this is likely to vary depending on whether the fuel used in the hybrid plant is coal, biomass or waste. It is also likely to depend on the solar share, and in retrofit scenarios the host plant's age and remaining lifetime. Fully renewable CSP hybrids with biomass or geothermal components are likely to be more easily accepted by the community than hybrids using a waste or coal

feedstock. In a CSP-EfW scenario community concern might arise over the use of recyclable materials for power generation and the resulting emissions. Both aspects can be addressed by pointing to the use of waste materials downstream of a recycling process (Gendebien et al. 2003), and by highlighting the advances in modern flue gas cleaning technologies (Psomopoulos, Bourka & Themelis 2009). Hybridisation with coal plants has to be considered carefully to avoid "greenwashing". Typically, coal fired power plants have a design lifetime of 40 years and units nearing the end of their lifetime should not be retrofitted with small commercial CSP plants to enable lifetime extensions. Two different examples exist in Australia at the Liddell and Kogan Creek coal fired power stations and these are discussed in Section 4.1. Despite the potential of CSP retrofits to significantly reduce economic barriers to implementation, such proposals have to be considered on a project basis to avoid negative environmental impacts and community opposition.

4.5.4 Conclusions

Currently, several key barriers limit the uptake of CSP in Australia and this research found that for both CSP-only and hybrid plants the economic and policy barriers are the most significant barriers, followed by technical, social and environmental barriers. Four of the top five barriers were identical for CSP-only and hybrid plants. They are: high project capital/financing cost, lack of financial incentives, policy uncertainty, and abundant and subsidised fossil fuels. However, there were significant rating differences in regards to generation intermittency/lack of energy storage, network access, capacity and availability, and high project capital/financing costs. For these barriers, the ratings for hybrids were significantly lower.

Based on the current policy and economic settings, CSP hybrids, both new plants and retrofits, seem a good option, not only for addressing technical and economic barriers, but also to help overcome significant policy, social and environmental barriers in Australia. The findings are supported by the fact that two hybrid projects are being operated/built in Australia while CSP-only plants have not progressed to the construction stage despite the selected offer of significant government support. The construction of CSP hybrids can provide technology providers, financiers, owners and operators with local experience and advance knowledge about the technologies. However, hybrid proposals, particularly retrofits to fossil fuel plants, have to be assessed on a project by project basis to ensure that they provide an overall environmental benefit or technology advancement and are not used to extend the operational lifetime of an aging generation asset.

4.6 Case studies

As mentioned in Section 3.2.4, two case studies at two sites in Australia were analysed to identify the technical, economic, environmental, and socio-economic benefits of CSP–EfB and CSP–EfW hybrid plants. Both case studies address the cost differences between CSP–EfB and CSP–EfW hybrids and CSP-only plants, the share of electricity generation from the individual energy sources and the plants' greenhouse gas mitigation potentials.

4.6.1 Swanbank, CSP-multiple feedstock hybrid

This case study is a 35.5 MWe CSP–EfB and CSP–EfW hybrid plant at the Swanbank landfill site in the Ipswich local government area in Queensland. The plant proposal is able to convert five different feedstocks very efficiently to electricity and to also potentially provide process heating and cooling. Section 4.6.1.1 provides a more detailed concept description. The Swanbank landfill is now owned and operated by Remondis Australia Pty. Ltd. The site was previously owned and operated by Thiess Services Pty. Ltd. and the case study was supported by this company financially and with detailed information about the site and feedstock quantities. Figure 38 shows the site with the proposed power station and existing infrastructure. The power plant proposal is a key element in a complementary visioning study on the economic and socio-ecological renewal of Swanbank (Baumann et al. 2012), which is endorsed by the local government (Ipswich City Council 2012).



Figure 38: Swanbank site¹⁴ with the proposed CSP–EfB and CSP–EfW hybrid plant (CSP = yellow and EfB/EfW = green), the existing coal fired Swanbank B power plant and the existing gas fired Swanbank E power plant (red squares), and landfill (blue polygon)

¹⁴ Map derived from Google Earth, version 7.1.2.2041

The architecture proposal for this power plant was produced by Elena Vanz, a PhD candidate in architecture and urban design at Melbourne School of Design at the University of Melbourne. She was given component dimensions and constraints in regards to component arrangements by the author and iteratively produced the plant layouts and renderings provided in the following conference publication. The author thanks her for her cooperation in this project and for her valuable insights into architecture and urban design. The case study was presented at the SolarPACES 2012 conference¹⁵ in Marrakech, Morocco, (Peterseim et al. 2012a) and published in the conference proceedings but unlike the other papers in this thesis, it was not peer-reviewed. In addition to the conference paper, Sections 4.6.1.2 to 4.6.1.5 provide further analyses of the environmental, economic and socio-economic benefits of this project as well as a synopsis of the plant's contribution to an economic and socio-ecological renewal at Swanbank.

4.6.1.1 SolarPACES 2012 conference paper

This conference paper outlines a highly efficient CSP-multiple feedstock hybrid plant concept for Swanbank capable of using of five energy sources in a single power plant. It also explains the technology selection process, discusses the power plant siting and provides an analysis of the proposed project's economic viability.

A more detailed process diagram than the one shown in Figure 3 in the conference paper is provided in the appendix.

¹⁵ www.solarpaces2012.org, accessed 04 April 2014



CONCENTRATING SOLAR POWER / ENERGY FROM WASTE HYBRID PLANTS - CREATING SYNERGIES

<u>Juergen H. Peterseim¹</u>, Stuart White², Udo Hellwig³, Amir Tadros⁴, and Frank Klostermann⁵

¹PhD Candidate, Institute for Sustainable Futures, University of Technology Sydney, Level 11, UTS Building 10, 235 Jones Street, Ultimo NSW 2007, Australia, Phone: +61 (0)2 9514 4636, E-Mail: JuergenHeinzMartin.Peterseim@uts.edu.au

² Prof. Dr., Director, Institute for Sustainable Futures, University of Technology Sydney ³ Prof. Dr., Managing Director, ERK Eckrohrkessel GmbH

⁴ Dr., Senior Engineer, Aurecon Australia Pty Ltd

⁵ Manager Market Development and Engineering, Thiess Services, Australia

Abstract

Tightening environmental legislation and landfill restrictions require waste management companies to increase their environmental sustainability. Several options exist to meet these targets, such as increased reuse and recycling. Modern Energy from Waste (EfW) plants are another option when they do not compete with recycling efforts; refuse derived fuels (RDF) fulfill this criterion. This paper investigates the hybridisation of EfW with concentrating solar power (CSP) plants focussing on CSP technology selection, integration concepts, synergies and suitable locations. A case study in Queensland Australia is provided to place the concept in a real world context.

For CSP technologies a high direct normal irradiance (DNI) is essential, typically >2,000kWh/m²/a for a stand-alone plant. However, for CSP/EfW hybrid plants lower DNI levels, >1,700kWh/m²/a, are acceptable as hybrid plants use some plant components jointly and can therefore lower the specific capital investment. In addition the hybrid technology allows the CSP asset to move closer to load centers, reduce new network costs and ensure fuel availability. A DNI >1,700kWh/m²/a is still significant but many countries, such as Australia, Greece, Turkey, Saudi Arabia and India, fit this criterion and are future growth markets for EfW and CSP systems.

Several CSP technologies are available for the hybridisation with EfW, such as parabolic trough, Fresnel, solar towers or dishes. Identifying the ideal technology is crucial and a variety of criteria have to be taken into consideration, such as land & water use, technology maturity and cost. CSP technologies in this paper were evaluated for feedwater heating, reheat steam and superheated steam meeting steam turbine requirements. Steam parameter considered range from 270-430°C.

CSP and EfW plants share similarities in terms of steam temperatures and capital investment. Steam temperatures of mature CSP technologies reach 440°C, which matches the steam temperatures of EfW plants well. Additionally, both technologies have high capital requirements and enabling them to share equipment, such as steam turbine, condenser, building infrastructure etc, will lead to specific cost reductions and make the hybrid plant concept more competitive.

Keywords: Energy from Waste, concentrating solar power, hybrid plants, Fresnel, multicriteria decision making


1. Introduction

With electricity prices and demand typically being higher during the day in Australia [1] a CSP component attached to an EfW plant can provide additional capacity during these times while the EfW facility provides base-load power. This is particularly interesting for locations with high daytime ambient temperatures as they negatively affect the condenser performance which leads to a reduction in generation output. The hybrid system benefits from this configuration by providing more electricity to the grid during these economically attractive hours.

When hybridizing EfW plants with CSP a high DNI is essential. Typically, hybrid plants are viable in lower DNI areas than stand-alone CSP plants which enables them to move closer to load centers, avoid network costs and ensure fuel availability. The first Energy from Biomass (EfB)/CSP hybrid plant worldwide currently under construction in Spain proves these hybridization benefits, as it is located further north than any other CSP plant in Spain [2], [3].

For Australia, and other countries with a high DNI, the hybridisation of EfW with CSP technologies is promising to comply with landfill diversion targets and better align the capacity of power generation assets with demand profiles. EfW/CSP hybrid plants are likely to be niche solutions as several fuel resources have to be in one location and EfW plants typically have a smaller capacity, \leq 35MWe.

2. CSP hybrid plant benefits

The main benefit of CSP hybrid over stand-alone CSP plants are immediate LCOE reductions of up to 28% [4]. Such reductions would reduce/eliminate the need for government incentives, allow plant suppliers and financiers to gain expertise, and are likely to accelerate the construction of CSP systems. The comparatively high LCOE of CSP is the key reason for its small contribution to global electricity supply.

Typically, renewable energy sources such as wind and solar suffer from poor capacity factors, 20-30% [5], compared to conventional fossil fuel plants, except for CSP with currently high cost thermal storage, AU\$90/kWhth [6]. Hybrid plants have the ability to reliably provide electricity during the night, extended cloud coverage or DNI fluctuations, without thermal storage which is a significant benefit in terms of plant investment and complexity. Thermal storage systems could potentially be retrofitted at a later stage when costs reach the expected AU\$22/kWhth by the end of this decade [6]. Additionally, CSP hybrid plants can follow daily electricity demand with the host plant operating constantly at design point and the CSP component satisfying the higher electricity demand during the day when electricity prices are economically more attractive.

Typically, CSP plants require a DNI >2,000kWh/m2/year to be commercially viable. Due to the joint use of equipment, such as steam turbine, condenser etc, CSP/EfB hybrid plants can be considered for DNI areas >1,700kWh/m2/year [7]. The first CSP/EfB hybrid plant near Barcelona verifies this assumption as it is the CSP installation furthest north in Spain. CSP/EfW hybrid plants could be build in even lower DNI areas as the low/negative fuel price for waste materials has a positive effect on the plants economic performance. DNI levels of >1,500kWh/m2/year are considered acceptable. Moving CSP out of arid/semi desert regions closer to agricultural/urban regions expands potential CSP sites and enables



access of back-up fuel resources, e.g. agricultural and urban waste materials.

Currently, many power plant operators in Australia and elsewhere are not familiar with CSP systems and are therefore likely to favor technologies, renewable or fossil, they know over CSP when deciding on new generation assets. CSP hybrid plants would allow them to use current staff while simultaneously up-skilling them to confidentially operate initially smaller but subsequently larger and larger CSP installations.

With CSP/EfW & EfB hybrid plants unlikely to exceed plant capacities of 60MWe such systems can be considered distributed generators that could be placed close to demand centers. This not only reduces transmission losses but also offers the chance to avoid/defer investment in transmission and distribution infrastructure, which are main drivers of current electricity prices rises. In Australia distribution infrastructure is expected to be responsible for 42% of the total electricity price increase from 2011-12 to 2013-14 and transmission 8% [8]. Moving CSP closer to load centers also increases chances for highly efficient combined heat and power applications.

3. Energy from Waste/biomass hybrid concepts

Principally, the CSP component of a hybrid plant can provide steam at different qualities. Low-temperature options include feedwater heating, mid-range temperature options include saturated steam into the high pressure boiler drum or steam into the cold reheat line, and the high-temperature option is superheated steam to the joint steam turbine, see Figure 1.



Figure 1: CSP integration options into an EfW plant; 1 = feedwater heating, 2 = cold reheat line, 3 = high pressure/temperature turbine steam

First concepts to pair biomass and waste materials with CSP were investigated at a high level in the mid 1980's with paraboloidal dish systems [9] but no plants were built. It took another 25 years before construction of the first commercial CSP/EfB hybrid plant, 25MWe [3], commenced ca. 150km west of Barcelona, Spain [2]. To minimise risk the plant uses the mature parabolic trough technology with thermal oil [3]. The disadvantage of having the biomass system in the thermal oil loop (see Figure 2) is the lower steam temperature of



375°C compared to 450°C, which is technically possible with forestry and agricultural waste materials. Other parabolic trough concepts integrate the biomass system into the secondary water-steam cycle to increase plant efficiency (see Figure 2), e.g. 107MWe hybrid plant proposal at San Joaquin, US [10].

Currently, no CSP/EfW hybrid plants are under construction anywhere in the world but EfW steam temperatures, typically 380-440°C [11], match well with current CSP technologies, such as parabolic trough, Fresnel or solar tower. Some studies investigate the integration of Fresnel [11], [12] and parabolic trough systems in EfW plant [11], [13]. Concepts discussed include the use of CSP for air and feedwater heating [11] as well as the generation of identical steam parameters as the host plant using Fresnel [12] and parabolic trough systems [11], [14]. External EfW steam superheating using Fresnel is being investigated [14] but seems unlikely to be a reliable option for high-temperature steam supply. All of the concepts consider CSP steam temperatures <430°C.

The hybridisation of EfW with paraboloidal dish systems was briefly discussed in the past [9] while solar towers are currently being investigated for construction and demolition timber as well as RDF [15].

Some EfW plants are hybridized with combined cycle gas turbine (CCGT) plants. The EfW plants Moerdijk in the Netherlands [16], Mainz in Germany [17] and Bilbao in Spain [18] provide steam to the heat recovery steam generators of adjacent CCGT plants for further superheating. An EfW plant in Måbjerg, Denmark has taken a different approach using natural gas to further superheat the steam [19]. All these plants raise the final steam temperature from <430°C to >520°C, therewith increasing the overall conversion efficiency.



Figure 2: Simplified hybrid concept for the Termosolar project, Spain (left) and San Joaquin proposal, US (right)

4. Swanbank case study, Queensland, Australia

The Swanbank landfill at Ipswich, Queensland is owned and operated by Thiess Services Pty Ltd. The site is ideal for a new power station as it is an industrial zoned area with a long tradition in power generation, recently decommissioned 480MWe coal fired and one operating 385MWe CCGT power station.

With Ipswich being one of the fastest growing communities in Australia the proposed power plant could provide new local industries and residential areas in its vicinity with renewable and low-carbon intensity electricity, heating and cooling while creating long-term high value employment.

The landfill has the capability to ensure long term supply of wood waste, refuse derived fuel (RDF), landfill- and biogas for a 35.5MWe net power plant, EfW contribution 30.7MWe and



CSP contribution 4.8MWe. Fuel availability is a significant benefit as the reliable supply from local sources is the main criteria for an EfW and EfB power station. Using the aforementioned fuels would defer ca. 150,000t/a waste from landfill without sacrificing recycling efforts as only solid materials downstream a recycling process are used.

The proposed plant is modeled with Thermoflex Version 22.0.1, can generate up to 252,800MWh per year, see Table 1, and follow daily demand with its main fuels. It also has the potential to provide extra steam from the solar field during the day to increase the plant output when electricity demand/prices are high.

| Peak net annual power output | 252,800MWh |
|---|------------|
| Net power output wood waste and RDF component | 245,600MWh |
| Peak net power output of the CSP component | 7,200MWh |

Table 1: Maximum annual power plant electricity output

4.1. Plant concept

The Swanbank hybrid is designed to maximize plant efficiency. Due to a novel plant hybrid concept the power station achieves an electric net efficiency of 33.6% which is significantly higher than the 30% of other modern EfW plants, such as the significantly larger, 66MWe, EfW plant in Amsterdam, Netherlands [20].

The simplified technical concept of the Swanbank CSP/EfW hybrid plant is outlined in Figure 3. Upon waste arrival the fuel is sorted in the material recovery facility. Recyclable materials and waste destined for landfill leave the facility while the organic rich fraction, wood waste and RDF are suitable power plant feedstocks. Two third of the solid material used in the power station is wood waste, 12.5t/h, and the remaining third RDF, 6.25t/h. Up to 2,600m³/h of bio- and landfill gas are required during peak capacity operation.

Wood waste and RDF are supplied to the boiler which is generating steam at 430°C and 90bar. The steam temperature is chosen to minimise high temperature corrosion issues inside the main boiler. The solar field is generating identical steam parameters as the main boiler and both steam flows are combined before entering an external superheater. The organic rich fraction of the waste material is digested in Thiess Services proprietary biocell technology [21] and the biogas, in combination with available landfill gas, is fired into an external superheater to further raise the steam temperature from 430°C to 530°C. External superheating of EfW steam has been realised at the EfW plant in Måbjerg, Denmark using natural gas [19].

The combined high pressure/temperature steam flow, up to 117t/h, enters one steam turbine. The turbine exit steam is condensed, using a water cooled condenser, and pumped back into the solid fuel boiler and solar field, thus closing the thermodynamic cycle. Flue gases are cleaned according to Australian emission limits using scrubbing and baghouse filter systems.





Figure 3: Simplified concept of a CSP / solid fuel hybrid plant with external superheating

4.2. Technology selection

Several CSP technologies are available for hybridisation with an EfW facility and in this assessment we considered the following:

- Parabolic trough; thermal oil (TO), direct steam generation (DSG) and molten salt (MS)
- Solar tower; molten salt (MS), direct steam generation (DSG) and air (A)
- Fresnel; saturated (Sat. St.) and superheated (Sup. St.) steam
- Paraboloidal dish; direct steam generation (DSG).

In July 2011 we organised a workshop at the University of Technology Sydney with 49 industry professionals with different expertise in the energy business (plant operators, technology provider, financiers and researcher) to identify the best CSP technologies to integrate into, amongst other, wood waste and RDF plants. The following steam temperatures scenarios were investigated:

- Live steam at 430°C to steam turbine,
- Steam at 300°C into the cold reheat line, and
- Steam at 270°C for feedwater heating.

To identify the best CSP technology for the Swanbank project we used the Analytical Hierarchy Process (AHP) as it allows the decomposition of a complex problem into several sub-problems, such as land use with levelised cost of electricity (LCOE), and provides a comprehensive and rational decision making framework [22]. The method is widely used in the research and industry world, including assessments comparing fossil fuels with renewable sources [23] and different CSP standalone technologies with each other [24].

The problem decomposition takes place by identifying criteria (main- and sub-criteria) relevant to the problem and organizing them in different levels of hierarchy. The AHP can use precise criteria data (quantitative information) as well as the personal judgments (qualitative information). Subsequently, quantitative and qualitative information can be



merged to calculate the total score for each option. Four main criteria groups (feasibility, risk reduction, environmental impact reduction and LCOE) with several sub-criteria, such as land and cleaning water use, site gradient tolerance, technology maturity, peak efficiency, complexity, were chosen to cover the relevant aspects of the Swanbank multi-criteria decision problem.

We identified quantitative data for all criteria from literature as well as own calculations/modeling. These quantitative data were merged with qualitative data derived from the participant's individual rating of the main/sub-criteria importance. CSP technologies with the highest total score are the preferred options. To accommodate uncertainties in the input data $a \pm 10\%$ sensitivity is applied to all results. As seen in Figure 4 not all CSP technologies can achieve the steam temperatures required for the different scenarios. CSP technologies unable to produce the desired steam temperature were excluded from the assessment.

For the integration of a CSP component into the high pressure/temperature steam cycle of a wood waste/RDF host plant the Fresnel technology with superheated steam scores best, see Figure 4, and is therefore the chosen technology for the Swanbank project. The reasons for the good score of Fresnel systems providing superheated steam include the low cleaning water requirements through robotic cleaning of the flat mirror panels, and the compact solar field minimizing land use.

Fresnel (superheated steam) and parabolic troughs (thermal oil) systems would be the preferred options for cold reheat steam, while parabolic troughs (thermal oil) score best for feedwater heating followed by Fresnel (saturated steam). However, these options are not considered in the Swanbank case study as they would reduce the CSP contribution to the overall plant output compared to high pressure/temperature turbine steam.



Figure 4: CSP technology selection for Swanbank hybrid plant



4.3. Plant siting and layout

As mentioned earlier CSP stand-alone plants typically require a DNI of >2,000kWh/m²/a. However, through the joint use of equipment the site with a DNI of only 1,890kWh/m²/a is still suitable for a CSP hybrid plant.

Space is constrained at Swanbank with the only possible site for a power plant south-west of the currently active landfill. A benefit of the location is the proximity to the current and new landfill which reduces material transport. The selected site is not level yet but earthworks to do this are not significant.

With the CSP component requiring the largest area its footprint is the limiting factor for the energy contribution. By arranging the EfW facility in the south, stretching from east to west, the area north of it is maximised for the CSP field, see Figure 5. To accommodate two Fresnel fields they have to be located north and south of the access road to the landfill. The fuel-exhaust gas flow of the power station is east to west, starting with the fuel processing facility, fuel storage, boiler, gas cleaning and stack. The steam turbine, auxiliary, workshop and cooling tower buildings are located south of the plant.



Figure 5: Plant layout of the Swanbank hybrid power station





Figure 6: Artist impressions of the proposed Swanbank hybrid power station¹⁶

4.4. Economic analysis

The economic modeling is carried out with Thermoflex Version 22.0.1 and based on a plant life of 25 years. It includes capital as well as operational expenditures, e.g. personnel, fuel, water and residue disposal costs, as well as escalation rates for inflation, fuel, electricity, water and CO_2 prices. All assumptions are based on the plant commencing operation in 2017.

In the base case scenario the solid fuels are considered to have zero fuel cost while the assumed landfill and biogas price is AU\$5/GJ. Depending on the future market developments for green waste and construction and demolition timber modeling was carried out for AU\$-10, AU\$10 and AU\$20 per ton of solid fuel, Table 2.

The base case wholesale electricity price scenario assumes \$45/MWh. With electricity prices currently increasing scenarios were modeled for electricity prices ranging from \$30-\$70/MWh, see Table 2.

The base case scenario is a renewable energy certificate (REC) price of \$35/MWh. Due to fluctuations in the REC market scenarios were modeled for REC prices ranging from \$30-\$50/MWh, see Table 2.

The total investment for the power station is expected to be around AU\$150-160m or 4.2-4.5m/MWe net. This price includes the fuel processing and storage facilities as well as grid connection. Considering the additional investment for the solar component the investment is in line with other EfW & EfB installations.

The levelised cost of electricity of the new installation is expected to be between AU\$80-120/MWh, see Table 2. The final investment strongly depends on detailed negotiations with EPC plant contractors, expected CSP cost reductions in the next 3-4 years as well as fuel, carbon and renewable energy certificate pricing.

¹⁶ Architecture proposal by Elena Vanz, PhD candidate in architecture and urban design at Melbourne School of Design, University of Melbourne



The modeling considered electricity generation only but the supply of process heat/cold to adjacent industries would strengthen the economic case and reduce the payback times of the different scenarios by up to 25%.

Except for scenario 1, see Table 2, the power station has a payback within its operational life but the scenario 2 and the base case scenario are not particularly attractive to institutional investors without other financial incentives. It is obvious that the electricity price agreed in a power purchase agreement has a significant impact on the plant's commercial viability. The fuel prices are relevant too but to a significantly lesser extent.

| Scenarios | Solid fuel price in \$/t | Electricity in \$/MWh ¹⁷ | Payback in years |
|---------------|-----------------------------|--|---------------------|
| Scenario 1 | 20.00 | 60.00 | >25.0 |
| Scenario 2 | 10.00 | 100.00 | 14.5 |
| Base scenario | 0.00 | 80.00 | 15.1 |
| Scenario 3 | 0.00 | 100.00 | 13.3 |
| Scenario 4 | 0.00 | 120.00 | 10.6 |
| Scenario 5 | -10.00 | 100.00 | 12.3 |
| Scenario 6 | -10.00 | 120.00 | 9.9 |

 Table 2: Economic viability of the hybrid power plant for different fuel and electricity price scenarios

5. Conclusion

The hybridization of CSP with non-conventional fuels is likely to be a niche market compared to natural gas or coal hybrid systems but allows, subject to waste material composition, renewable base-load power generation. Only waste materials downstream a recycling process should be considered for such plant concepts.

All the individual components required for the Swanbank CSP/EfW hybrid project are proven with reference plants in operation using wood waste and RDF fired boilers, Fresnel systems, external steam superheating, and bio- / landfill gas combustion. The combination of the individual components is new but manageable with experienced project partners and modern power plant engineering tools.

The Swanbank site is ideal for such a concept as the landfill ensures fuel supply over the operational lifetime of the plant, the CSP system provides additional power during high electricity demand/price times and staff from the recently decommissioned coal fired power station could be recruited to operate the new facility. Due to the joint use of plant equipment the LCOE are competitive compared to other forms of renewable energy and the concept demonstration could trigger the development of similar projects in Australia and overseas.

¹⁷ Includes wholesale and renewable energy certificate prices



Acknowledgements

The authors would like to thank Thiess Services Pty Ltd for providing information for this case study and their excellent cooperation.

References

- [1] S. Lekovic, P. Priftakis, K. Donoghue, T. Rigzin, S. Cole, B. Pryor, and K. Simic, "electricity gas australia 2011," Energy Supply Association of Australia, Melbourne, Australia, 2011.
- [2] Abantia, "The installation of the solar field Termosolar Borges, the first plant to combine thermal and biomass, has started," 2011. [Online]. Available: www.abantia.com/noticias-prensa/en_noticias/28/the-installation-of-the-solar-field-termosolar-borges-the-first-plant-to-combine-thermal-and-biomass-has-started.
- [3] A. Cot, A. Amettler, J. Vall-Llovera, J. Aguilo, and J. M. Arque, "Termosolar Borges: A Thermosolar Hybrid Plant with Biomass," in Third International Symposium on Energy from Biomass and Waste, 2010.
- [4] C. S. Turchi and M. Erbes, "Gas Turbine / Solar Parabolic Trough Hybrid Designs," ASME TurboExpo 2011. Vancouver, Canada, 2011.
- [5] P. Hearps and D. McConnel, "Renewable Energy Technology Cost Review," Melbourne Energy Institute, Melbourne, Australia, 2011.
- [6] J. Hinkley, B. Curtin, J. Hayward, A. Wonhaus, R. Boyd, C. Grima, A. Tadros, R. Hall, K. Naicker, and A. Mikhail, "Concentrating solar power drivers and opportunities for cost-competitive electricity," CSIRO & Aurecon Australia Pty Ltd, Sydney, Australia, 2011.
- [7] Á. Pérez and N. Torres, "Solar Parabolic Trough Biomass Hybrid Plants: A Cost Efficient Concept Suitable For Places In Low Irradiation Conditions," in SolarPaces Conference 2010, 2010.
- [8] Australian Energy Market Commission, "Possible Future Retail Electricity Price Movements: 1 July 2011 to 30 June 2014, Final Report," Sydney, 2011.
- [9] C. F. McDonald, "A hybrid solar closed-cycle gas turbine combined heat and power plant concept to meet the continuous total energy needs of a small community," Journal of Heat Recovery Systems, vol. 6, no. 5, pp. 399–419, 1986.
- [10] California Energy Comission, "San Joaquin Solar 1 & 2 Power Plant Licensing Case." Sacramento, California, 2008.
- [11] H. Spliethoff, N. Kaeding, M. Murer, E. Alonso-Herranz, and O. Gohlke, "Combining Energy from Waste and Concentrated Solar Power: New Solutions for Sustainable Energy Generation," in International Solid Waste Association Conference 2010, 2010.
- [12] M. Rojas, U. Hellwig, J. H. Peterseim, and J. Harding, "Combining solar thermal power systems with solid fuels in new plants," in SolarPACES, 2010.
- [13] L. Schnatbaum-Laumann, "Biomass Utilization for Co Firing in Parabolic Trough Power Plants," SolarPACES 2009. Berlin, Germany, 2009.



- [14] N. Kaeding, "Hybridkraftwerk Solar/Abfall: Erstellung und Beurteilung von Konzepten," Technische Universitaet Muenchen, Munich, Germany, 2010.
- [15] J. H. Peterseim, "Biomass/CSP Hybrid Plants," in Bioenergy Australia 2011 Conference, 2011.
- [16] J. Toebes and M. Beker, "External steam input and process steam output at the Moerdijk (Netherlands) combined-cycle plant," VGB POWERTECH, vol. 8, no. 7, Essen, Germany, pp. 21–27, 1998.
- [17] Entsorgungsgesellschaft Mainz mbH, "The Mainz Waste-to-Energy Plant: Exceptionally reliable and environmentally sound waste disposal," 2008. [Online]. Available: www.mhkw-mainz.de/downloads/EGM_infobro3_eng_20.pdf.
- [18] S. G. Ribeiro and T. Kimberlin, "High Efficiency Waste to Energy Power Plants Combining Municipal Solid Waste and Natural Gas or Ethanol," in 18th Annual North American Waste-to-Energy Conference, 2010.
- [19] E. J. Andersen, F. Oksen, and J. V. Olesen, "Power plant Holstebro / Struer with separately fired superheater," in ERK licensee conference, 1992, p. 20.
- [20] City of Amsterdam Afval Energie Bedrijf and C. of A. Afval Energie Bedrijf, "Value from Waste; Waste Fired Power Plant - The new standard for recovery of sustainable energy, metals and building materials from urban waste," City of Amsterdam, Netherlands, 2006.
- [21] W. P. Clarke, "Reusable Landfill Biocells for Rapidly Degrading the Organic Fraction of Municipal Solid Waste," in EcoForum 2012, 2012.
- [22] E. Triantaphyllou, Multi-Criteria Decision Making Methods: A Comparative Study. Dordrecht, The Netherlands: Kluwer Academic Publishers, 2000.
- [23] B. A. Akash, R. Mamlook, and M. S. Mohsen, "Multi-criteria selection of electric power plants using analytical hierarchy process," Electric Power Systems Research, vol. 52, no. 1, pp. 29–35, 1999.
- [24] J. D. Nixon, P. K. Dey, and P. A. Davies, "Which is the best solar thermal collection technology for electricity generation in north-west India? Evaluation of options using the analytical hierarchy process," Energy, vol. 35, no. 12, pp. 5230–5240, 2010.

4.6.1.2 Environmental analysis

The Swanbank hybrid plant proposal is designed to use predominantly renewable energy sources, such as wood waste, landfill gas, biogas and CSP, and the only feedstock with a fossil content is RDF. The renewable content of RDF ranges from 40-60% (Arias-Garcia, Gleeson & Ltd 2009; BGS e.V. 2011) and renewbale materials include wood, cardboard and paper waste. The following assessment is based on fossil fuel CO_2 emissions from the RDF combustion and preparation of 652 kg CO₂/t RDF (Arias-Garcia, Gleeson & Ltd 2009) which, when combined with the RDF consumption of 50,000 t/year and electricity generation of 252,800 MWh, leads to a carbon intensity for the Swanbank proposal of 129 kg CO_2/MWh . This is significantly lower than the 2010 average carbon intensity of Australia's electricity generation of 841 kg CO₂/MWh (International Energy Agency 2012a) and the expected average carbon intensity from 2017–42 of around 673 kg CO₂/MWh¹⁸ (ROAM Consulting Pty Ltd 2011). Comparing the carbon intensity at Swanbank with the ROAM Consulting (2011) average carbon intensity leads to a CO_2 abatement potential of 3.4 million tonnes over the plant's 25-year operational life. It is worth mentioning that the 673 kg CO_2/MWh seems high but according to the aforementioned report the emissions intensity of gridsourced electricity is expected to stay above 500 kg CO_2 /MWh until around 2038.

Energy recovery from wood waste, the biogenic RDF portion and the use of biogas from the organic rich fraction can reduce potent future landfill gas emissions by increasing the current landfill gas capture rates from 50–65% to above 98%¹⁹ when the organic materials are processed in anaerobic digesters. The total organic carbon in the post-combustion products, ash and slag, would be below 3% before landfill, which meets a requirement introduced in Germany that contributed to only 1% of municipal waste going to landfill (European Environmental Agency 2009). When considering avoided landfill gas emissions the aforementioned carbon intensity of 129 kg CO₂/MWh would be lower or even negative, which is in line with other EfB/EfW life cycle assessments (Sathaye et al. 2011).

Considering that annually 150,000 t of predominantly renewable energy feedstock is currently being landfilled at Swanbank, and in view of the availability of other renewable sources such as landfill gas and solar irradiance, the use of these resources represents a cost effective means of diverting waste from landfill and generating renewable electricity without sacrificing recycling efforts.

¹⁸ 2017-2042 carbon intensity calculated with data from the ROAM Consulting (2011) report

¹⁹ Personal communication with Frank Klostermann, formerly Manager Engineering and Strategy at Thiess Services Pty. Ltd., during the feasibility study.

4.6.1.3 Hybrid versus CSP-only cost comparison

The DNI at Swanbank is 1,890 kWh/m²/a (Bureau of Meteorology 2012) which is sufficiently high enough to operate a CSP-hybrid and in fact better than the 1,800 kWh/ m^2 /a at the site of the 22.5 MWe Termosolar Borges CSP-EfB hybrid plant in Spain (Morell 2012). One of the main benefits of a CSP hybrid at this site is the cost reduction through the joint use of plant infrastructure such as the steam turbine and condenser. The specific cost for the 4.8 MWe CSP component at Swanbank, including its share of the use of the steam turbine, cooling system, buildings, roads, control system and other infrastructure, is AU\$ 5m/MWe (net). A 4.8 MWe CSP-only plant on the same site would have a specific cost of AU\$ 8.3m/MWe (net) and therefore the CSP hybridisation at Swanbank leads to a 40% cost reduction. This is significant and mainly a result of the comparatively small CSP system (4.8 MWe net) benefiting from the much larger EfB/EfW component (30.7 MWe net). This, for example, leads to a steam turbine cost, including installation, of AU\$ 3.4m in the 4.8 MWe hybrid scenario compared to AU\$ 7.4m for the CSP-only scenario. Also the Fresnel field itself would be larger and more capital intensive in a CSP-only scenario due to the lower net cycle efficiency of 29.1% (with steam parameters of 500°C and 90 bar) compared to the 33.6% Swanbank hybrid scenario. The cycle efficiency difference results from the economies of-scale that the larger power plant equipment offers and the slightly higher steam temperature due to external steam superheating to 530°C at 90 bar.

The significant cost difference between the CSP hybrid and CSP-only scenarios at Swanbank allows the construction of a CSP plant which would otherwise not be economically viable. This supports the assumption that CSP hybrid plants can fast track CSP implementation in Australia and it can enable technology suppliers, operators and financiers to advance their local understanding of CSP, gain practical experience, and progress the knowledge of technology with minimal dependence on government incentives.

4.6.1.4 Socio-economic benefits

The economic uncertainties currently affecting the world make the social acceptance of any energy project very much related to the potential net impact that such an activity will have on stimulating the economy as well as creating new employment opportunities. In fact, the latter is one of the reasons frequently given for encouraging renewable energy deployment. Several studies have analysed the socio-economic drivers and benefits of renewable energy projects (Caldés et al. 2009; Deloitte 2011; Domac, Richards & Risovic 2005; Wei, Patadia & Kammen 2010) and comparing the employment of 0.11 job years/GWh for coal and gas fired generation with 0.21 job years/GWh for bioenergy and 0.23 job years/GWh for CSP

(Wei, Patadia & Kammen 2010) shows a significantly higher job creation from the renewable alternatives. This is one of the reasons why the Swanbank proposal has the potential to generate a very positive socio-economic net impact for the Ipswich local government area in regards to employment generation as well as economic development.

To assess the economic and employment stimuli of a project one must consider not only its effects on sectors directly affected by a project, but also its impact on sectors that supply goods and services to the directly affected sectors. One must also consider the induced effects – that is, the impact on the economy in general due to the economic activity and spending of direct and indirect employees. One of the most widely used methodologies to quantify the total direct and indirect effects of any given project on the economy is the input-output methodology (Ten Raa 2006). This methodology is a tool to systematically gather information about the productive relations between the different sectors in any given country or regional economy. Besides estimating the associated direct and indirect effects on the economy and on employment creation, input-output models are used to estimate the multiplier effect that a certain investment generates in the economy.

Operating the proposed Swanbank power station would require about 30 staff, an estimate which is in line with another EfB employment study by Thornley et al. (2008), and would additionally secure current employment on the landfill through business diversification, for example for a material recovery facility in addition to the landfill operation. Operator positions could be filled with personnel from the recently decommissioned Swanbank B coal fired power station. This would not only train these personnel to operate a renewable energy asset and future proof their careers, but it would also de-risk the initial phase of plant operation due to their relevant expertise with thermal power plants. Based on the literature quantifying the employment from EfB power plants (Thornley, Rogers & Huang 2008) the indirect and induced employment an EfB plant creates can be 10 times higher than its direct employment. This includes employment in feedstock growth, harvesting, transport, and equipment supply and maintenance. However, these numbers cannot be transferred entirely to the Swanbank proposal as the biomass and waste materials are already going to landfill through existing transport infrastructure. Also, feedstock cultivation will not be required. Considering these factors the additional employment could reach a multiplier factor of five through the additional operation of a material recovery facility and limited transport and supply change increases caused by the need for additional residue and recycling material transport. In a growing council area where many people commute elsewhere for work, this presents an opportunity worth pursuing.

Based on the more conservative investment estimate of AU\$ 160m for the Swanbank power plant, and assuming an Australian-based engineering procurement and construction (EPC) company, 58% or AU\$ 93m of the investment is expected to remain in the Australian economy (see Table 3). The network connection is not included in this cost but could be significant depending on the accessibility of existing infrastructure at the adjacent Swanbank E combined cycle power plant (see Figure 38 on page 160). Of this AU\$ 93m, a large percentage would remain in the Queensland and Ipswich economies in investments related mainly to plant construction, equipment installation and some component manufacturing. Only 42% or AU\$ 67m of the plant investment would flow overseas for the manufacture of major plant items, such as boiler, steam turbine and condenser systems.

| Cost broakdown | Investment, | Australian content, | Overseas content, |
|--------------------------|-------------|---------------------|-------------------|
| | AU\$m | AU\$m (%) | AU\$m (%) |
| Turbo generator set | 25 | 5 (20%) | 20 (80%) |
| Boiler system | 36 | 18 (50%) | 18 (50%) |
| Flue gas cleaning | 26 | 16 (60%) | 10 (40%) |
| Solar Field | 15 | 10 (65%) | 5 (35%) |
| Condenser/cooling system | 5 | 3 (50%) | 3 (50%) |
| Balance of Plant | 10 | 7 (70%) | 3 (30%) |
| Fuel processing facility | 15 | 12 (80%) | 3 (20%) |
| Buildings and roads | 20 | 19 (95%) | 1 (5%) |
| Electricals & controls | 8 | 4 (50%) | 4 (50%) |
| Total | 160 | 93 (58%) | 67 (42%) |

Table 3: Swanbank power plant cost breakdown and investment distribution

To determine the multiplier effect of the Swanbank hybrid plant the individual contributions from the main energy sources have to be considered. For biomass electricity generation a multiplier effect of 1.32 has been published (Gan & Smith 2007), which excludes feedstock procurement, as in this case study the materials are already transported to the current landfill. For CSP a multiplier effect of 2.3 was used in another study (Caldés et al. 2009). Combining these two values proportionally to the 35.5 MWe plant capacity, 30.7 MWe from EfB and EfW and 4.8 MWe from CSP, results in a combined multiplier effect for Swanbank of 1.45. Based on this multiplier the direct plant investment would lead to an overall financial stimulus to the Australian economy of AU\$ 135m.

4.6.1.5 Economic and socio-ecological renewal at Swanbank

In addition to the techno-economic analysis of the Swanbank power plant proposal the author was part of a research team which prepared a report on a vision for the economic and socio-ecological renewal at Swanbank (Baumann et al. 2012).

Ipswich is one of the fastest growing regions in Australia with substantial new commercial and residential developments around the current industrial site. The power plant proposal will help integrate the current industrial area into this new environment by creating employment opportunities, providing low-carbon intensity energy, and enhancing waste management. Figure 39 shows how the proposed power station can be developed in the area to contribute to the transformation of the site into an eco-industry park with an integrated waste and energy management system.





Adjacent to the proposed power station site sufficient land is available to co-locate new industries and provide them with electricity and potentially significant quantities of process heat and/or energy for cooling. In addition to industry consumers, which typically require high value process steam, commercial and residential developments could be supplied with electricity as well as lower grade heat and/or cooling through a district heating/cooling network. A particular benefit would be the low and stable cost of energy given that the

energy sources, biomass and waste, are already available and largely independent of world oil and gas prices, as is CSP. This has the potential to be a major driver to attract further commercial and industrial developments, create additional employment and help diversify the business profile of the area. A diverse local economy would also lead to residents spending and re-spending more money in Ipswich rather than in other council areas as they would have access to various services in their local area. This in turn would strengthen the local economy, finance additional investments and ultimately create further business opportunities, employment and tax revenue.

The waste streams arising from existing and new residential, commercial and potentially industrial developments could be sent to the landfill for recycling and energy recovery. With this circular economy concept the different local stakeholders would complement and mutually benefit each other, for example by ensuring stable waste disposal costs and feedstock for stable energy costs. In particular, energy intensive industries, such as food processing, would benefit from stable energy prices as these industries have experienced electricity price increases and now expect significant rises in natural gas pricing.

To separate the industrial facilities from the residential developments buffer zones as well as commercial areas would be required, with the proposed power station located in the centre of the industrial area (see Figure 39). In close proximity to existing and potentially new industrial facilities dense commercial developments could emerge, thinning out towards existing and new residential developments. Buffer zones, including areas created by the conversion of the old landfill site into a green area, could be expected to include parks for recreational purposes.

Currently, most people in the Ipswich area work outside their council area and with the city's continued growth commutes are bound to increase with all their associated negative side effects, such as pollution, health impacts, and infrastructure construction and maintenance. It is well accepted that mixed land use enables sustainable transport, with its related environmental benefits, and this positively influences individuals' quality of life as well as the efficiency of businesses (Banister 2008). Creating local employment through mixed land use at Swanbank would enable better public transport due to increased demand and increased available tax income. It would also promote active transport – walking and cycling as the distance to the new residential developments would be less than 3 km and existing residential areas are less than 6 km away. Active transport has health benefits and the potential to reduce automobile trips, which have a particularly high carbon intensity (Sallis et al. 2004). Cycling and walking routes could be easily integrated into the new developments and green buffer zones.

4.6.2 Griffith, CSP-single feedstock hybrid

This case study of a proposed 30 MWe CSP–EfB hybrid plant at Griffith, New South Wales, is based on results from a previous biomass resource assessment (Herr et al. 2012). The suitability of the area for such a plant was confirmed by the CSP–EfB resource assessment (see Section 4.2) which identified a cumulative potential of more than 500 MWe for CSP–EfB hybrid plants in the Griffith region. As the resource assessment does not identify specific sites for CSP hybrid plants but only prospective areas, a meeting was organised by the author with Nicola James, Economic Development Coordinator at Griffith City local government area, on 30 May 2013 to visit three potential sites in the area. The site in close proximity to the Tharbogang substation near Griffith (see Figure 40) was subsequently chosen by the author as the substation is scheduled to be upgraded, the area is earmarked for industrial development, and the block of land for the site is owned by the council.

The architecture proposal for this power plant was produced by Kinneth Galang, Master's candidate at the University of Technology, Sydney's School of Architecture, under the supervision of Juliet Landler, a senior lecturer at the school. Galang was supplied with dimensions and constraints in regards to component arrangements by the author and he iteratively produced the plant layouts and renderings for the peer-reviewed publication which is included in Section 4.6.2.1. The paper was presented at the SolarPACES 2013 conference²⁰ in Las Vegas, USA. The author thanks Kinneth Galang and Juliet Landler for their assistance in this project and for the valuable discussions about power plant architecture.



Figure 40: Potential site for the CSP–EfB hybrid plant near Griffith²¹

²⁰ www.solarpaces2013.solarpaces.org, accessed 04 April 2014

²¹ Map derived from Google Earth, version 7.1.2.2041

4.6.2.1 SolarPACES 2013 conference paper

This conference paper analyses the technical, economic and environmental benefits a CSP– EfB hybrid plant in Griffith could provide and outlines the cost differences between a hybrid plant and a CSP-only plant. The plant is designed to use straw which is expected to be delivered to the plant in bales. The straw bales would be derived from the collected stubble on the fields.

A higher quality process diagram than Figure 3 is provided in the appendix.





Available online at www.sciencedirect.com



Energy Procedia 49 (2014) 1197 - 1206



SolarPACES 2013

Solar tower-biomass hybrid plants - maximizing plant performance

J.H. Peterseim^a*, A. Tadros^b, S. White^c, U. Hellwig^d, J. Landler^e and Kinneth Galang^f

^a PhD Candidate, Institute for Sustainable Futures, University of Technology Sydney, Level 11, UTS Building 10, 235 Jones Street, Ultimo NSW

2007, Australia, Phone: +61 2 9514 4636, E-Mail: JuergenHeinzMartin.Peterseim@uts.edu.au

^bDr., Group-Mechanical Engineering Leader, Laing O'Rourke Australia, Australia

^c Prof. Dr., Director, Institute for Sustainable Futures, University of Technology Sydney, Australia

^d Prof. Dr., Managing Director, ERK Eckrohrkessel GmbH, Germany

^e Senior Lecturer, School of Architecture, University of Technology Sydney, Australia

^f Master candidate, School of Architecture, University of Technology Sydney, Australia

Abstract

Concentrating solar power (CSP)-biomass hybrids plants are becoming increasingly interesting as a low cost option to provide dispatchable renewable energy since the first reference plant commenced operation late 2012, 22.5MWe Termosolar Borges in Spain. The development of such project is a complex task with not only one but two energy sources required to make the project successful. The availability of several studies but only one reference plant worldwide is proof of that.

This paper investigates the hybridisation of a biomass power plant with a molten salt solar tower system. The benefit of this combination is a high cycle efficiency as both the steam generators can provide steam at 525°C and 120bar to the steam turbine. A case study approach is used to provide technical, economic and environmental benefits of a 30MWe CSP-biomass plant with 3h thermal storage in Griffith, New South Wales. At this site such a plant could provide annually 160,300MWh of electricity with an annual average electricity price of AU\$155/MWh. Compared to a standalone CSP plant with 15h of thermal storage the hybrid plant investment is 43% lower, providing a possibility to fast-track CSP implementation in countries where CSP is struggling to enter the market due to low wholesale electricity prices, such as Australia.

© 2013 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/3.0/). Selection and peer review by the scientific conference committee of SolarPACES 2013 under responsibility of PSE AG.

Final manuscript published as received without editorial corrections.

Keywords: Solar tower; energy from biomass; hybrid plant

* Corresponding author: Tel.: +61 (0)2 9514 4878; fax: +61 (0)2 9514 4941 E-mail address: JuergenHeinzMartin.Peterseim@uts.edu.au

1876-6102 © 2013 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/3.0/).

Selection and peer review by the scientific conference committee of SolarPACES 2013 under responsibility of PSE AG. Final manuscript published as received without editorial corrections. doi:10.1016/j.egypro.2014.03.129

1. Introduction

CSP-biomass hybrid plants are a well-accepted option to decrease the investment and levelised cost of electricity of CSP plants while increasing power dispatchability. The first reference plant in Spain [1] proves the concept and is likely to enable further installation in high direct normal irradiance (DNI) locations where biomass is also available, such as Australia, India, Greece or Spain.

Several CSP-biomass concepts have been investigated in the past but only the 22.5MWe Termosolar Borges project in northern Spain was built [1]. This shows that the development of such projects is complicated as in addition to all regular project development considerations, such as land availability and network access, not only one but two energy resources have to be abundant in this particular location. While this is a challenge modern geospatial modeling software can help to identify the best sites and allows project developers to not only get a good understanding of the energy resources but also land use, water availability, network access etc.

This paper is based on preliminary geospatial work for CSP-biomass hybrid plants in Australia and outlines the benefits of this concept in a case study in Griffith, New South Wales. The paper provides a technical, economic and environmental analysis for this site and while not all the results are fully transferrable to other sites the concept could be deployed in several regions in Australia and overseas.

2. Current CSP-biomass hybrid concepts

Some early efforts to combine CSP with biomass or waste materials were discussed briefly in the 1980's with dish systems [2]. However, due to technical and financial issues no plants were built. It took another 25 years before construction of the first commercial CSP-biomass hybrid plant commenced near Lleida, ca. 150km west of Barcelona in Spain, see Figure 1. The 22.5MWe Termosolar Borges plant came online late 2012 [1], is located further north than any other CSP project in Spain and uses the mature parabolic trough technology with thermal oil [3]. Several other studies investigated the hybridisation of parabolic trough plants with biomass [4–6] but no other project has commenced construction yet.

Alternatively, Fresnel has been investigated for hybridisation with biomass and waste materials [7–9]. The benefit of using Fresnel would be steam temperatures of up to 500°C [10] and subsequently higher conversion efficiencies. However, no reference plants exist yet.

In addition to live turbine steam other CSP-biomass concepts included the use of CSP for air and feedwater heating as well as external steam superheating from an energy from waste plant [9]. However, none of the concepts considers CSP steam temperatures >430°C which limits the cycle efficiency.



Figure 1: First CSP-biomass plant under construction near Lleida, Spain (left) and biomass fuel (right)

3. Solar tower-biomass hybrid plants

Solar tower systems with direct steam generation and molten salts have the potential to provide steam parameters identical to a biomass plant, >500°C and >100bar, and are according to a recent CSP hybrid assessment the most suitable for hybridisation with biomass [11]. Without thermal storage a direct steam generation would be most favorable while a thermal storage concept would favor a molten salt system. In addition to solar towers with molten salts and DSG systems using a volumetric air receiver are being investigated [12]. However, due to the limited reference situation and the higher complexity of such systems they are more complicated to finance.

In addition to the cost sharing opportunities of hybrid plants in general, such as joint use of steam turbine and condenser, solar tower biomass hybrid plants have the potential to use the required stack to support the receiver. This concept has been investigated for integrated solar combined cycle plants [13] and is also suitable for solar tower-biomass hybrids.

To have a significant solar share in a CSP-biomass hybrid both steam generators have to be able to provide high temperature/pressure steam to the turbine rather than using CSP for feedwater heating or reheat steam (steam reheating is not common in smaller power plants). A CSP-biomass hybrid example is shown in Figure 3.

4. Case study

The case study is based on a CSP-biomass hybrid plant with a total capacity of 30MWe, 15MWe biomass and 15MWe CSP. The CSP component will be modeled as a molten salt and direct steam generation solar tower. Biomass and the aforementioned solar tower references exist for steam temperatures of 525°C at 120bar which provides financiers with investment certainty. Technical reliability is highly important to secure project finance and turn projects into actual power plant installations.

4.1. Methods

Thermoflex version 23.0.1[†] was used for the techno-economic assessment. The software is widely used in academia and industry to model actual solar tower, parabolic trough and Fresnel plants. The software's cost database is used to identify the investment for the different power plant components. With biomass boilers not being a standard Thermoflex component the authors' industry expertise was used to ensure equipment cost accuracy. With an assumed plant commissioning in 2016 the solar field investment is lowered by 10% to reflect 2015 pricing based on a reasonable learning curve [14].

The economic assumptions are a plant lifetime of 30 years, commissioning in 2016, 65% debt finance, a debt interest rate of 8%, a 7% discount rate and an internal rates of return on investment of 11%. The modeling includes capital and operational costs as well as escalation rates for inflation, fuel, electricity and water prices. The investment shown are the owners total project cost consisting of the engineering, procurement and construction price and 9% owner's soft cost covering permits, project management, legal and finance aspects of a power plant project. Network connection is also included. Payback times and net present values derive from the total project cost.

From the different CSP technologies available a solar tower was selected as according to a recent multi-criteria assessment for CSP hybrid plants [11] it is the preferred technology for >500°C hybrid systems. Without thermal storage a direct steam generation tower scores best but due to the integration of 3h thermal storage a molten salt system is preferred as its thermal storage is proven and commercially available from a variety of suppliers.

A water tube boiler with a vibrating grate is selected for the biomass boiler as this is the dominant technology for straw. References exist up to 540°C [15] which matches the CSP parameter well and allows high efficiency power generation.

[†]www.thermoflow.com

The technology selection for a hybrid plant is crucial to obtain finance. This assessment requires the CSP and biomass components to have at least 15MWe reference plants in operation. Both technologies fulfill this criterion with commercial plants in operation, e.g. 20MWe Gemasolar solar tower [16] and 25MWe Sanguesa straw fired power plant [15].

4.2. Site

Griffith in central New South Wales is chosen for the case study as it is a region with strong agriculture and horticulture industries and a DNI of 2,150kWh/m²/year [17]. From a DNI perspective Griffith is significantly better than Lleida in Spain, 1,800kWh/m²/year [18], where the first CSP biomass plant commenced operation recently.

A recent study [19] indicates sufficient stubble in this region to provide the 86,000t required annually to fuel a 15MWe biomass plant continuously. The maximum biomass potential in the Mildura region is even higher but the annual biomass yield varies and a conservative approach is important to ensure continuous biomass supply.

The Tharbogang substation near Griffith would require little upgrades to accommodate the output of a 30MWe power plant. The CSP-biomass hybrid plant could follow the daily demand well by providing base-load as well as peak capacity during the day and evening, see Figure 2. This optimises revenue by covering the high price periods.



Figure 2: Ideal CSP-biomass generation profile for a 30MWe plant (left) and NSW average weekdays system demand and spot price year ending 2010 (right) [20]

4.3. Technical analysis

Figure 3 shows the process diagram of the proposed hybrid plant with its main components, such as furnace and boiler heating surfaces (1-8), flue gas cleaning (14), stack (18), solar tower with 3h thermal storage (34), molten salt/steam heat exchanger (35-37), steam turbine (19-21, 53, 57), air-cooled condenser (22), feedwater heating system (25-26, 30, 52, 61), and auxiliary equipment.

At full load each steam generator provides 59t/h of steam at 525°C and 120bar to the joint steam turbine generating a net output of 30MWe. Due to the high steam parameters and feedwater heating the plant reaches a net cycle efficiency of 33.4% at full load and 30.2% with only the biomass boiler in operation, see Table 1.

The annual plant output was modeled for the years 2006-10 and the CSP component reached an average annual capacity factor of 29.1% resulting in an average plant output of 160,300MWh. During this period the annual DNI changed resulting in the lowest annual generation in 2010 and highest in 2006, see Table 1. In winter the DNI output decreases significantly while the biomass output increases slightly due to lower ambient temperatures. However, the biomass increase cannot offset the CSP shortfall.

To operate the biomass plant for 8,000h/year a continuous straw supply of 10.74t/h is required. The onsite storage capacity for straw bales is 4 days of full-load operation with the remaining material being stored on the fields where it was harvested. Baling is essential to avoid the material degrading during outdoor storage.



Figure 3: Process diagram for 30MWe solar tower-biomass hybrid plant in Griffith

| Total plant capacity | 30 | MWe |
|---|---------|-----|
| CSP contribution to steam turbine capacity | 15 | MWe |
| Biomass contribution to steam turbine capacity | 15 | MWe |
| CSP contribution to annual generation (2006-10) | 38,300 | MWh |
| Biomass contribution to annual generation (2006-10) | 122,000 | MWh |
| Total annual average generation (2006-10) | 160,300 | MWh |
| Minimum annual generation (2010) | 154,100 | MWh |
| Maximum annual generation (2006) | 164,400 | MWh |
| Peak electric plant efficiency (gross) | 36.1 | % |
| Net electric plant efficiency | 33.4 | % |
| Parasitic losses | 7.5 | % |
| Net electric plant efficiency biomass only | 30.2 | % |
| Annual straw consumption at 15.6MJ/kg | 86,000 | t |
| Plant footprint | 74 | ha |

Table 1: Technical data summary

4.4. Economic analysis

The economic data for the 30MWe CSP-biomass hybrid at Griffith are shown in Table 2. It should be noted that the grid connection, AU\$17m, is a significant addition to the plant investment and adds AU\$10/MWh to the electricity price. To meet industry typical internal rates of return (IRR) of around 11% an annual average electricity price of AU\$155/MWh is required assuming a biomass price of AU\$100/t.

The low specific investment for the biomass components, AU\$4.2m/MWe, reduces the higher CSP specific investment, AU\$7mMWe, to an average of AU\$5.6m/MWe while increasing the plant's capacity factor.

Generating the same annual electricity output as the CSP-biomass hybrid would require 26MWe CSP standalone plant with 15h thermal storage (solar multiple of 2.8 and 70% capacity factor). In 2015 such a plant would have a total invest of AU\$292m incl. network connection or AU\$11.2/MWe. Comparing this with the CSP hybrid results in a 43% cost reduction. The CSP standalone plant would require an annual average electricity price of AU\$205/MWh to achieve same IRR.

| Table 2. Economic data summary | | | |
|-----------------------------------|-------|-----------|--|
| Plant investment | 151 | AU\$m | |
| Network connection charge | 17 | AU\$m | |
| Specific investment | 5.6 | AU\$m/MWe | |
| Specific CSP investment | 7.0 | AU\$m/MWe | |
| CSP percentage of investment | 62 | % | |
| Biomass percentage of investment | 38 | % | |
| Biomass price | 100 | AU\$/t | |
| Net present value | 103.7 | AU\$m | |
| Internal rate of return | 10.9 | % | |
| Payback | 9.5 | years | |
| Required annual electricity price | 155.0 | AU\$/MWh | |

Table 2: Economic data summary

The proposed power station would create 30 positions for plant operators and engineers. Based on literature [21] the additional employment from a biomass power plant can be 10 times higher than the direct employment, e.g. fuel harvesting and logistics. With the Griffith proposal being half CSP and half biomass a factor of 5 has to be assumed for the maximum additional indirect/induced employment. This is still significant for a rural region.

A benefit worth mentioning is the opportunity for local farmers to diversify their businesses as they currently receive annually varying revenues for their crops. A power station would stabilize their revenue stream by making use of a product that is currently considered waste and is, at an expense, burnt in the field.

4.5. Environmental analysis

Both CSP and the biomass are renewable energies which can offset the use of fossils fuels elsewhere in the electricity network. Based on future annual electricity demand and CO_2 emissions trajectories [22] the average annual carbon intensity of Australia's electricity mix between 2016-45 could be 604kg CO_2/MWh . This value seems high but the emissions intensity is expected to stay above 500kg CO_2/MWh until around 2038. Based on this assumption the CSP-biomass hybrid plant could abate up to 2.2 million tonnes CO_2 over its 30 year lifetime. It has to be mentioned that long-term future electricity and CO_2 emissions trajectories contain a high degree of uncertainty, which can be observed currently by a falling rather than increasing electricity demand in Australia.

In addition to avoided CO_2 emissions the use of straw in a power station has further benefits as it can avoid straw burning in the field with its associated particle emissions and subsequent public health impact. The ash collected from the power station could be brought back to the field as it still contains valuable compounds, such as potassium.

4.6. Plant layout and architecture

As the plant stack and receiving tower climbs approximately 70m above the surrounding planar river valley landscape and the conspicuous reflective mirrored heliostats occupy a significant area (72ha), the plant will serve as a significant landmark for the region. Thoughtful consideration to its architecture is essential. Therefore, for the purposes of this study, devising design concepts for the plant was proposed as an assignment for masters students enrolled in subject Advanced Environmental Design at University Technology Sydney (UTS) during the 2013 fall semester. Master student Kinneth Galang developed the preliminary conceptual scheme described below.

For maximum solar power generation, 1,480 heliostats are evenly placed around the central stack in an elliptical shape measuring approximately 1km x 0.9km located on a flat plot of land to the southwest of the intersection of Harward Road and Kidman Way. Walter Burley Griffin used the circle several times in designing Griffith's masterplan and these circles are still prominently evident in city road network today, a continuation of the circular layout was initially considered for the field of heliostats. However as the elliptical layout increases optical efficiency, this variation was determined the superior solution. One road bisects the ellipse to provide a secure main entrance to the central plant. The road then encircles the central plant providing serviceable access to its six main components: fuel storage, the boiler, thermal storage, flue gas cleaning, condenser and the steam turbine, see Figure 4. Of these, the combined stack/receiving tower is the most visible rising above the surrounding buildings and area with a height of 70m, see Figure 5. The fuel storage with its area of approximately 3,100m² occupies the most space. While several different configurations of the various components were considered, keeping the stack on the central axis and then clustering the other elements nearby so that distances between the processes could be kept minimal resulted in the most satisfactory solution.

In deference to the agrarian nature of the surroundings, metal siding typical of agricultural buildings was considered most appropriate cladding for the operations. The variation of the building's overall form and detailing of this metal cladding however seek to create a dynamic more consistent with electricity generation. In deference to the local regenerative fuel sources, strawbale infill walls were considered a possibility as a secondary construction system in the more public and visible spaces. The plant complex would include a small public lobby to showcase and explain the technologies being used at the plant. This showroom would be housed next to the main office rooms for the plant and located at the end of the central corridor. In keeping with the circular vocabulary used in the public areas of Griffin, and circular landscaped pool would surround this circular portion of the plant, with the water serving the dual purpose of being the required fire reserve.



Figure 4: Plan view of the CSP-biomass hybrid plant (heliostat arrangement only schematic)



Figure 5: Cross section view of the CSP-biomass hybrid plant; 1. Stack/solar tower, 2. Condenser, 3. Molten salt tanks, 4. Offices, 5. Biomass boiler, 6. Building for auxiliary equipment, 7. Fuel storage, 8. Reception

4.7. Molten salt biomass heater alternative

With 3h thermal storage integrated in the CSP hybrid plant the option of a biomass fired molten salt heater rather than a steam boiler is worth investigating as such a plant configuration could optimise electricity dispatchability with pricing. Currently, there are no references for biomass fired molten salt heaters but significant industry expertise is available for biomass and waste fired thermal oil heaters, which could be used for the design of a molten salt unit, such as fluid circulation, drainability and solidification. To simplify molten salt draining a heater design with vertically arranged heating surfaces, as in the Energy from Waste plant Bamberg in Germany [23], would be ideal. The technical risk of a molten salt heater would be slightly higher compared to a conventional steam boiler but the benefit of shifting electricity output to higher price periods, such as morning peak, makes the concept interesting. A biomass fired molten salt heater would have a lower investment than a steam boiler due to features such as lower working fluid pressure and no drum. However, the molten salt/steam heat exchanger would have to be designed for the total plant capacity and not only the CSP flow which is likely to offset the heater savings.

Typically, electricity prices are lower during the night than the day with peaks in the morning and evening. The evening peak could be covered with CSP charging the thermal storage but the morning period could not. However, with a molten salt biomass heater the plant could operate the steam turbine at minimum load, 7.5MWe, during low electricity price times at night, simultaneously charge the thermal storage and dispatch 30MWe as soon as electricity

prices recover in the morning. The biomass fired molten salt unit could also charge the thermal storage during lower DNI winter days to ensure appropriate thermal storage levels to cover the evening peak.

Figure 6 provides an example based on the New South Wales wholesale electricity prices for the 23/07/2012 and the plant's operational strategy to maximize revenue. By diverting thermal energy from biomass to the thermal storage during the 0-5AM low electricity price period and only operating the plant at minimum 7.5MWe load the thermal storage could be charged to 91%. With electricity prices increasing significantly after 5:30AM the thermal storage can be discharged until CSP is able to provide its 15MWe share at 9:30AM. In addition to CSP charging the thermal storage to 48% during the day a biomass fired molten salt heater could further charge the thermal storage from 2-3:30PM to 100%, which allows longer electricity dispatch during the economically attractive evening times. Having the plant online earlier at full load for the morning peak and longer for the evening peak increases the daily revenue by 6.1% compared to a biomass fired steam boiler operating continuously at full load.

With electricity prices fluctuating during the year the economic benefits of a biomass fired molten salt heater are lower at other times. On the 23/01/2013 the New South Wales electricity price was stable at a low level and a molten salt biomass concept could have only capitalized on the small price differences during the night and early morning by charging the TS from 02-4:30AM to 50% and dispatching at slightly higher prices from 5-7:30AM, see Figure 7. The small electricity price differences result in a daily revenue increase of only 0.3%.



Figure 6: Example of daily electricity generation with molten salt biomass heater in winter (23/07/2012); Peak electricity price at 6:30PM was AU\$291.66/MWh



Figure 7: Example of daily electricity generation with molten salt biomass heater in summer (23/01/2013)

5. Conclusion

The combination of a biomass and solar tower energy system is beneficial to maximise the cycle efficiency with proven technologies, which financiers can see operating successfully elsewhere in the world. By combining these two energy sources in Griffith, New South Wales, a power plant could provide lower cost, AU\$155/MWh, dispatchable renewable electricity, base-load with additional capacity during the day and evening, as well as additional benefits by avoiding the burning agricultural residues in the field.

Reducing the investment by 43% with a CSP-biomass hybrid plant, compared to standalone CSP, has the potential to fast-track CSP implementation in Australia and familiarize financiers and operators with the different CSP technologies available.

References

- Protermosolar, "Abantia and Comsa Emte set up the first facility in the world that combines solar thermal and biomass energy," 2012.
 [Online]. Available: http://www.protermosolar.com/saladeprensa/?q=en/content/abantia-and-comsa-emte-set-first-facility-world-combines-solar-thermal-and-biomass-energy. [Accessed: 28-Jun-2013].
- [2] C. F. McDonald, "A hybrid solar closed-cycle gas turbine combined heat and power plant concept to meet the continuous total energy needs of a small community," Journal of Heat Recovery Systems, vol. 6, no. 5, pp. 399–419, 1986.
- [3] A. Cot, A. Amettler, J. Vall-Llovera, J. Aguilo, and J. M. Arque, "Termosolar Borges: A Thermosolar Hybrid Plant with Biomass," in Third International Symposium on Energy from Biomass and Waste, 2010.
- [4] Á. Pérez and N. Torres, "Solar Parabolic Trough Biomass Hybrid Plants: A Cost Efficient Concept Suitable For Places In Low Irradiation Conditions," in SolarPaces Conference, 2010.
- [5] L. Schnatbaum, "Biomass Utilization for Co Firing in Parabolic Trough Power Plants," in SolarPACES Conference, 2009.
- [6] California Energy Comission, "San Joaquin Solar 1 & 2 Power Plant Licensing Case." Sacramento, California, 2008.
- [7] J. H. Peterseim, A. Tadros, S. White, U. Hellwig, and F. Klostermann, "Concentrated solar power / Energy from Waste hybrid plants creating synergies," in SolarPACES Conference, 2012.
- [8] M. Rojas, U. Hellwig, J. H. Peterseim, and J. Harding, "Combining solar thermal power systems with solid fuels in new plants," in SolarPACES Conference, 2010.
- [9] N. Kaeding, "Hybridkraftwerk Solar/Abfall: Erstellung und Beurteilung von Konzepten," Technische Universitaet Muenchen, Munich, Germany, 2010.
- [10] T. P. Fluri, S. Lude, J. Lam, G. Morin, C. Paul, and W. J. Platzer, "Optimization of Live Steam Properties for a Linear Fresnel Power Plant," in SolarPACES Conference, 2012.
- [11] J. H. Peterseim, S. White, A. Tadros, and U. Hellwig, "Concentrated solar power hybrid plants, which technologies are best suited for hybridisation?," Renewable Energy, vol. 57, pp. 520–532, Sep. 2013.
- [12] B. Coelho, P. Schwarzbözl, A. Oliveira, and A. Mendes, "Biomass and central receiver system (CRS) hybridization: Volumetric air CRS and integration of a biomass waste direct burning boiler on steam cycle," Solar Energy, vol. 86, no. 10, pp. 2912–2922, Oct. 2012.
- [13] J. H. Peterseim, A. Tadros, U. Hellwig, and S. White, "Integrated solar combined cycle plants using solar towers with thermal storage to increase plant performance," in ASME Power, 2013.
- [14] IRENA, "Renewable Energy Technologies: Cost analysis Series Volume 1 Issue 2/5 Concentrating Solar Power," Abu Dhabi, UAE, 2012.
- [15] Acciona Energía, "Biomass plant from straw combustion in Sangüesa," 2002. [Online]. Available: http://www.accionaenergia.com/media/219313/ACCIONA_Sanguesa biomass plant_EN.pdf. [Accessed: 17-Jun-2013].
- [16] E. García and Roberto Calvo, "One Year Operation Experience of Gemasolar Plant," in SolarPACES Conference, 2012.
- [17] Bureau of Meteorology, "Australian Monthly Average Hourly Global and Direct Normal Solar Exposure Gridded Data," Melbourne, 2012.
 [18] D. Morell, "CSP BORGES The World's First CSP plant hybridized with biomass," in CSP Today USA 2012 6th Concentrated Solar
- Thermal Power Conference, 2012.
 [19] A. Herr, D. O'Connell, M. Dunlop, M. Unkovich, P. Poulton, and M. Poole, "Second harvest–Is there sufficient stubble for biofuel production in Australia?," GCB Bioenergy, vol. 4, no. 6, pp. 654–660, 2012.
- [20] S. Lekovic, P. Priftakis, K. Donoghue, T. Rigzin, S. Cole, B. Pryor, and K. Simic, "Electricity Gas Australia 2011," Energy Supply Association of Australia, Melbourne, Australia, 2011.
- [21] P. Thornley, J. Rogers, and Y. Huang, "Quantification of employment from biomass power plants," Renewable Energy, vol. 33, no. 8, pp. 1922–1927, Aug. 2008.
- [22] ROAM Consulting Pty Ltd, "Additional Projections of Electricity Generation in Australia to 2050," Brisbane, 2011.
- [23] J. H. Peterseim, U. Hellwig, and K. Endrullat, "Parallel flow boiler designs to minimise erosion and corrosion from dust loaded flue gases," in ASME Power, 2013.

4.6.2.2 Socio-economic benefits

In line with the higher employment creation of renewable electricity generation compared to fossil fuel fired generation as outlined by Wei et al. (2010) the CSP–EfB hybrid plant proposal at Griffith would lead to a higher employment generation than a coal, natural gas or even wind alternative.

To assess the socio-economic benefits of the Griffith CSP–EfB power station proposal the Input-Output methodology (Ten Raa 2006) was again used. Based on the investment estimate of AU\$ 168m, consisting of power plant and network connection costs, 63% or AU\$ 105m of the investment is expected to remain in the Australian economy (see Table 4). The cost breakdown assumes plant construction by an Australian-based EPC company. Compared to the Swanbank case study the Australian content is slightly higher, at 63% compared to 58%, due to the larger CSP component, which would be expected to be predominantly manufactured close to the Griffith site. Of the AU\$ 105m, a significant percentage is expected to remain the New South Wales and Griffith economies. This would primarily be related to plant construction, equipment installation and some component manufacturing. Overseas manufacturing of major plant items, such as the boiler, steam turbine and condenser system, makes up the remaining 37% or AU\$ 63m.

| Cost Breakdown | Investment, | Australian content, | Overseas content, |
|-------------------------|-------------|---------------------|-------------------|
| | AU\$m | AU\$m (%) | AU\$m (%) |
| Turbo generator set | 17 | 3 (20%) | 14 (80%) |
| Boiler system | 8 | 4 (50%) | 4 (50%) |
| Flue gas cleaning | 3 | 2 (60%) | 1 (40%) |
| Solar Field | 75 | 49 (65%) | 26 (35%) |
| Air cooled condenser | 10 | 5 (50%) | 5 (50%) |
| Balance of plant | 6 | 4 (70%) | 2 (30%) |
| Fuel storage facility | 6 | 5 (80%) | 1 (20%) |
| Buildings and roads | 18 | 17 (95%) | 1 (5%) |
| Electricals & controls | 8 | 4 (50%) | 4 (50%) |
| Total plant cost | 151 | 93 (62%) | 58 (38%) |
| Network connection cost | 17 | 12 (70%) | 5 (30%) |
| Total investment | 168 | 105 (63%) | 63 (37%) |

Table 4: Griffith power plant cost breakdown and investment distribution

To assess the overall stimulus of the Griffith plant investment to the Australian economy multiplier factors were used as in the Swanbank case study. A higher value than the biomass electricity multiplier of 1.32 for Swanbank had to be used for Griffith as the setting up of a feedstock supply chain has to be included. A multiplier effect of 1.42 has been used for biomass electricity generation facilities with feedstock procurement (Gan & Smith 2007). Combining this value with the previously used multiplier of 2.3 for CSP (Caldés et al. 2009), and considering the equal 15 MWe EfB and 15 MWe CSP contribution to 30 MWe plant capacity, leads to a combined multiplier effect of 1.86 for the Griffith proposal. This is a 28% higher value than in the Swanbank case study due to the larger CSP component at Griffith and the inclusion of a new feedstock supply chain. Based on the multiplier the direct plant investment would lead to an overall financial stimulus to the Australian economy of AU\$ 196m.

The new direct and indirect employment opportunities of the Griffith CSP–EfB hybrid proposal (see Section 4.6.2.1), as well as the local, regional and national financial stimulus the development would provide all support the view that such a plant has the potential to strengthen regional employment, use local renewable energy sources and diversify business opportunities for farmers, both in Griffith and in other regional centres across Australia.

4.7 CSP hybrids as a pathway to a low carbon future

To answer the research question regarding what role CSP–EfB and CSP–EfW hybrid plants can have in Australia's transition to a low-carbon energy future, an analysis of the CSP hybrid niche in the context of the Australian electricity generation market is required. Until recently, this market has been very stable with continuously growing demand and stable market conditions. Coal fired generation has provided the majority of electricity, established market players and qualified personnel. To assess whether CSP–EfB and CSP– EfW hybrid plants are a niche innovation able to support Australia's transition to a low carbon future the two key criteria, mentioned in Section 3.3 about the nature of interaction and the timing of innovation developments have to be addressed.

According to Geels & Schot (2007) the interaction between the niche innovations and landscape developments can reinforce the existing regime, stabilising it and discentivising transitions, or this interaction can disrupt the status quo, applying pressure and incentivising transition. In addition to impact of the niche–landscape interaction on the regime, niche innovations can have a symbiotic relationship with the existing regime that allows their easy adoption as a "competence-enhancing add-on in the existing regime to solve problems and improve performance" (Geels & Schot 2007, p. 406). Alternatively, niche innovations aiming to replace the existing regime have a competitive relationship with it, complicating their implementation. Looking at the current Australian electricity generation market using this multi-level perspective the following pressures and relationships emerge:

- Landscape: Landscape pressures on the current electricity regime include: environmental issues such as increased frequency and intensity of disasters; internationally growing political acceptance of the need to respond to anthropogenically induced climate change; more ambitious international greenhouse gas reduction targets such as the 20% European target for 2020 (European Commission 2014); a global upward trend for oil and gas prices; global deployment of renewable energy technologies; rising environmental awareness; and a growing desire to provide a sustainable planet for future generations. The landscape not only applies these pressures to the regime but also to the niches where actors invent suitable solutions.
- Regime: Pressures within the current Australian electricity regime include government policies such as the current Renewable Energy Target (Climate Change Authority 2012)and the carbon pricing mechanism (Australian Government 2011); the unconditional 5% emission reduction target by 2020; scientific consent that greenhouse gas emission reductions are required (Cook et al. 2013); public demand

for lower greenhouse gas emissions; falling electricity demand in recent years; the growing proportion of Australia's energy output provided by intermittent renewable energy generation; the aging of existing generation assets; and rising natural gas and electricity prices. To address these pressures the regime can draw on mature niche innovations, such as various renewable energy technologies, and provide incentives to the niches, such as R&D and technology demonstration grants, to provide further innovations.

Niche: Incentivised by the aforementioned landscape and regime pressures, various actors in Australia have developed/provided niche innovations ranging from specific energy technologies, including CSP hybrids and intelligent grid systems, to new business models, including energy contracting and community finance. Some of these innovations have been tried and tested overseas, giving them a competitive advantage. They include CSP hybrids and CSP-only plants but also various other innovations, such as intelligent grid systems, and incentives to encourage energy efficiency, community finance, and behaviour change.

In regards to the competitiveness or symbiosis of CSP, EfB and EfW hybrids, some differentiations are necessary. CSP technologies are competitive innovations because in a low carbon/renewable future they aim to replace parts of the regime. New CSP EfB and EfW hybrid plants can be both competitive and symbiotic as they can replace existing generation assets in the regime but can also enhance the regime by solving its problem with the current high cost of CSP. CSP retrofits to existing plants can be considered symbiotic as they are the least-cost means of implementing CSP in the Australian electricity generation market.

From the literature review and the nature of the landscape-regime-niche interactions, it can be assumed that the Australian electricity generation market will undergo a transition in the next two decades. However, is the timing/ maturity of CSP–EfB and CSP–EfW hybrid plants right to take advantage of the window of opportunity that results from the landscape and regime pressures? While technology maturity is debatable, with researchers, technology suppliers, plant operators and financiers all having different perspectives, four criteria have been introduced by Geels & Schot (2007) as reasonable indicators of maturity:

- a) learning processes have stabilised in a dominant design,
- b) powerful actors have joined the support network
- c) price/performance improvements have improved and there are strong expectations of further improvement (e.g. learning curves), and

d) the innovation is used in market niches, which cumulatively amount to more than 5% market share (Geels & Schot 2007, p. 405).

Currently, the dominant design with a market share in deployed capacity of 94% of existing CSP plants (IRENA 2012) is the parabolic trough technology. However, in new projects solar tower systems are being deployed at a similar utility scale worldwide. Fresnel is a third design that is also being deployed but with fewer references. Dishes have stabilised in their design principles but are not yet being deployed at a commercial scale. A technology selection process has occurred over the last three decades and it can be concluded that currently one dominant design exists with two new alternatives, solar towers and Fresnel systems, with towers maturing more rapidly than Fresnel systems. The situation for EfB and EfW technologies is similar: combustion systems are currently the dominant design with others maturing. While there is no one dominant design as suggested in criterion 1 of Geels and Schot's list, the availability of few stabilised/mature technologies for CSP–EfB and CSP–EfW hybrid plants provides confidence that the concept can be successfully implemented in the electricity market. The successful CSP–EfB Termosolar project in Spain supports this assumption.

In the past decade powerful actors have joined the CSP industry. They include: well-known EPC contractors such as Alstom, Abengoa, Bechtel and Areva; financiers such as the World Bank, HSBC, Deutsche Bank, and KfW; plant operators such as RWE, NextEra, Florida Power & Light Co., and NRG Energy; research institutes such as NREL in the US, DLR in Germany and CSIRO in Australia; and other institutions such as the International Renewable Energy Agency, the US Department of Energy and the International Energy Agency. The EfB and EfW industry has a longer deployment history, with various powerful actors, including Babcock & Wilcox, Alstom, Hitachi and JFE Engineering. The backing of CSP–EfB and CSP–EfW through such players is demonstrated by the significant Spanish construction company Abantia building the Termosolar plant. With these and other powerful actors engaged in CSP, EfB and EfW projects, Geels and Schot's second criterion is satisfied.

Since the deployment of the first CSP plants in the 1980s the specific costs for CSP fell from AU\$ 34m/MWe to around AU\$ 6.5m/MWe in 2010 (Hayward, Graham & Campbell 2011) and a recent analysis of various CSP studies (Lovegrove et al. 2012) identified further cost reductions of 10–20% per doubling of cumulative deployment. Through the hybridisation of CSP with EfB and EfW, systems costs can be improved further (see Sections 4.4 and 4.6), which satisfies criterion 3.

CSP is a niche technology used in various markets. The dominant market is electricity generation but other plants operate in markets such as process heating and enhanced oil

recovery. In 2014 CSP is expected to provide 0.3% of the world's renewable electricity generation and bioenergy is expected to provide 7.9% (International Energy Agency 2013)²² Despite CSP being used in various markets it does not fulfil criterion 4, as deployment of commercial plants on a broad scale only started in 2006 in Spain and therefore the technology is not as advanced as others, such as hydro, PV or wind. Also, unlike PV, CSP does not have a small-scale market where new technologies can be tested at lower cost and risk.

In summary this research concludes that CSP technologies for hybridisation with EfB and EfW seem to have sufficient technical maturity to enter the Australian electricity generation market as Geels and Schot's four criteria are fully or partially met. Despite the CSP market share being below the mentioned threshold, the availability of utility scale plants of up to 280 MWe, provides proof that financiers are confident enough to make significant investments in such plants, e.g. US\$ 2b for the Solana project (NREL 2014d). The availability of local CSP and CSP hybrid projects at utility scale, including the Termosolar CSP–EfB hybrid plant, is in line with niche management observations shown in Figure 26 on page 53 where at a global level a technology is still a niche but various commercial projects at a local scale exist and demonstrate the innovation's benefits (Geels & Raven 2006). Despite CSP–EfB and CSP–EfW hybrid plants meeting most of Geels and Schot's four criteria a high technical maturity of a certain technology alone is no guarantee for its implementation as other factors, such as policy and social barriers, are highly relevant too, see implementation barrier section 4.5. However, it is a key requirement to convince private and public financiers to engage with and deploy generation assets.

Based on the aforementioned observations it becomes clear that moderate pressures exist within the current Australian electricity generation market which incentivise a transition process by allowing the adoption of sufficiently developed symbiotic innovations. Despite the potential for CSP–EfB and CSP–EfW hybrid plants to replace some regime actors, they are currently considered a symbiotic innovation as they:

- a) enable a lower-cost CSP uptake compared to CSP-only plants (Sections 4.4 and 4.6),
- b) can generate low-cost dispatchable renewable electricity (Sections 4.4 and 4.6),
- c) encounter lower implementation barriers more than CSP-only plants (Section 4.5),
- d) have sufficient resources in various Australian states (Section4.2),
- e) can enable the uptake of other CSP hybrid plants (Section 4.1),

²² Percentages are derived from the annual generation figures provided by International Energy Agency (2013)

- f) are sufficiently developed (Sections 4.3, 4.4 and 4.6), and
- g) can therefore help the Australian Government to meet its renewable energy goals at lower cost while enabling the uptake of a strategically relevant energy technology.

In the narrow context of implementing CSP in the Australian electricity generation market, five features of the SNM steps governments can use to influence a transition (Kemp, Schot & Hoogma 1998) can be identified:

- 1. Technology choice: The technological features of CSP hybrid and CSP-only plants, particularly mature energy storage, make them, despite currently being more expensive than wind or PV, an attractive option for the federal and state governments to contribute to a stable low carbon electricity generation market in the future. Despite the absence of specific funding pools for CSP, the technology has been supported in the past through funding for research organisations such as ASTRI and big dish research (CSIRO 2013; Lovegrove, Burgess & Pye 2011), for feasibility studies such as the CSP retrofit and Collinsville hybrid study (Australian Renewable Energy Agency 2013a; Meehan 2013), and for demonstration projects such as CSP retrofits to Liddell and Kogan Creek power stations (CS Energy 2011; Novatec Solar GmbH 2012b). This demonstrates that CSP is a desired technology choice.
- 2. Experiment selection: In addition to small prototype testing of CSP, such as experimental tower in Newcastle (CSIRO 2012) or the single big dish in Canberra (Lovegrove, Burgess & Pye 2011), the testing of CSP at megawatt scale is possible at low risk through its integration into one of the various existing power plants in Australia. A CSP retrofit can utilise existing power plant infrastructure and simultaneously demonstrate its capabilities in daily commercial operation.
- 3. Experiment set-up: The initial CSP retrofit at Liddell (Mills, Lièvre & Morrison 2003) can be considered an experimental set up. The selected technology provider enjoyed protection through government funding but had to compete with other potential suppliers to reach that stage. Some early selection pressure is vital to allow the best technologies to emerge.
- 4. Experimental scale up: Subsequent projects, such as the recent CSP expansion at Liddell (Novatec Solar GmbH 2012b) and the CSP retrofit to the Kogan Creek power station (CS Energy 2011), can be considered experimental scale ups as they are based on some results from the original experiment at Liddell. The SolarDawn proposal could also be seen as an experimental scale-up but the capacity of 250
MW and the expected investment of AU\$1.2b (Kelly 2012) is very significant which, despite significant government funding, may have contributed to the abandonment of the project. The 40 MWe SolarOasis dish proposal (Regional Development Australia 2011) might be another such example. Hence, the technology scale up has to be significant to prove commercial operation but also needs to be within risk limits for financiers, owners and operators. In Australia the current RET and the carbon pricing mechanism support CSP and other projects, and are drivers contributing to the deployment of commercial scale projects.

5. The breakdown of protection by means of policy: At this point the breakdown of the CSP protection would be detrimental for CSP as the technology has not really started in Australia yet. Once the international and national learning experiences and research efforts lead to lower costs and a CSP uptake, federal and state governments could wind back protection by reducing funding pools for studies and projects, or by introducing of price premiums that are technology and time dependent. However, without a commercial CSP-only plant and with only one operational CSP hybrid, Australia has not reached that stage yet.

Based on these observations some SNM practices to implement CSP can in fact be observed in the Australian electricity generation market and it becomes apparent that CSP hybrid projects have progressed to construction and operation in this context while CSP-only projects have not. Therefore hybridisation seems to be a viable way forward to implement CSP in Australia and CSP–EfB and CSP–EfW can do this at low cost and low carbon emissions from the non-CSP component.

Niche technologies slowly enter a regime through incremental deployment and then upscale in capacity. Based on the current renewable energy deployment in Australia, some types of renewable energy are closer to entering the Australian electricity generation market than others but all can be considered niche technologies due to their relatively small overall contribution to annual electricity generation. Using the multi-level perspective from Geels & Schot (2007), and considering that currently hydro is Australia's dominant renewable energy contributor, it can be seen that hydro is the technology closest to full regime implementation, followed by wind, bioenergy and PV (Bureau of Resources and Energy Economics 2013). Therefore a future niche technology uptake as shown in Figure 41 is likely.

A future uptake as outlined in Figure 41 would fit with an analysis on the possible Australian energy mix by 2049–50 (Syed 2012) where renewable energy technologies such as wind and solar enter the Australian energy market later than hydro but eventually surpass it due

to better resource availability. Similarly, geothermal is expected to grow its market share slowly from currently being non-existent to 8% of Australia's electricity mix in 2049–50. This can be explained through a later technology uptake by the regime due the fact that it is less developed in Australia at this time. Based on Figure 41 CSP is a technology that will experience a later uptake. This is unfortunate as CSP, despite currently involving higher costs than wind and PV, is of significance for Australia's renewable energy future due to the excellent solar resource, the flexible plant capacity range and the proven energy storage capability, which can balance intermittent renewable technologies, such as wind and PV. It is therefore in the country's interest to provide ongoing niche protection for CSP through R&D and project implementation funding.



Figure 41: Multi-level perspective for different renewable energy technologies in the Australian electricity generation market; Blue = hydro, grey = wind, green = biomass, yellow = PV, orange = CSP, and red = others; Adapted from Geels & Schott (2007)

To fast-track the CSP uptake and take advantage of the currently moderate pressures within the Australian electricity generation market, several social, technical, economic, environmental and policy barrier reductions are required. As shown earlier CSP–EfB and CSP–EfW hybrid plants can provide these barrier reductions immediately. Figure 42 outlines how the various benefits of CSP–EfB and CSP–EfW hybrid plants could fast-track implementation and in the Australian context the CSP retrofits to the Liddell and Kogan

Creek coal fired power stations support this assumption. These two CSP hybrid projects have progressed towards construction (CS Energy 2011; Novatec Solar GmbH 2012b) while the two significant CSP-only proposals in South Australia and Queensland were unfortunately not able to secure project finance despite significant government grants (Edwards 2013; Kelly 2012).

The accelerated CSP uptake through hybridisation, not only with EfB and EfW systems but with other energy sources as well, would enable technology providers, financiers, plant operators, government agencies, project developers and researchers in Australia to gain local experience with CSP, increase technology confidence, and demonstrate the efficacy of less mature CSP technologies with low financial risk by adding smaller units to existing or new power plants.



Figure 42: Multi-level perspective for CSP-only and hybrid technologies in the Australian electricity generation market. Adapted from Geels & Schott (2007)

But how could the CSP uptake through hybridisation occur in Australia? The reconfiguration pathway described in Sections 2.6 and 3.3 outlines a plausible process. The pathway provides guidance on how moderate landscape pressure allows the regime to implement some sufficiently developed symbiotic innovations to solve its problems. Initially, the regime remains mostly unchanged, as in the transformation pathway, but over time regime actors explore further combinations and implement more innovations. By doing so, they adjust the basic architecture of the regime. Currently the Australian electricity market is concentrated, with three energy retailers providing the majority of the electricity in the NEM (Hannam 2013), which limits competition. This situation is not unusual and has been acknowledged by transition management researchers who have stated that this poses a threat as

large incumbent firms will probably not be the initial leaders of sustainability transitions [however] their involvement might accelerate the breakthrough of environmental innovations if they support these innovations. ... This would, however, require a strategic reorientation of incumbents who presently still defend existing systems and regimes (Geels 2011, p. 25).

Considering the current situation in the Australian electricity market, a more detailed prospective pathway than the one shown in Figure 42, is needed to accelerate CSP uptake through hybridisation. Such a pathway is shown in Figure 43. The figure is simplified, showing a hypothetical power station scenario and only the technology aspect of the regime. Other aspects of the regime, such as market user preferences, industry, policy, science and culture, are not shown as this would unnecessarily complicate the graph. Initially, the lowest cost CSP hybrid options, being CSP retrofits to existing power stations, are implemented and this process has already started in Australia with the CSP add-ons to the Liddell and Kogan Creek coal fired power stations (CS Energy 2011; Novatec Solar GmbH 2012b). In Figure 43 Liddell power station is shown as 1 and Kogan Creek power station as 2. The Liddell power station is an old unit with a limited remaining lifetime and over time the plant will disappear. It is unlikely that the attached CSP system will remain operational as it is too small to be converted to a standalone plant. While the decommissioning of the CSP plant at Liddell in the foreseeable future might seem a retrograde step, it has served its purpose of demonstrating the technology in a commercial context. In fact the technology for the first stage of the project (Mills, Lièvre & Morrison 2003) was subsequently improved and applied to much larger CSP projects worldwide, for example at the 5 MWe Kimberlina plant in the USA (NREL 2014b), at the 44 MWe equivalent Kogan Creek Solar Boost (CS Energy 2011) and at the 2x 125 MWe Reliance power plant in India (AREVA Solar 2012a). The Kogan Creek coal fired power station is the most modern power station in Australia and is likely to remain in operation with the CSP add-on that is currently under construction for the next several decades as shown with 2 in Figure 43. A third option for an add-on that is capable of eventually replacing an existing coal unit, see 3 in Figure 43, is the retrofit of a solar tower to an existing power plant as such a system could provide steam parameters

identical to a coal, gas or oil fired units from the 1980s. Obviously, detailed technical issues such as steam turbine lifetime have to be considered on a project-by-project basis.



Figure 43: Possible reconfiguration pathway for the implementation of CSP technologies in the Australian electricity generation market; Red squares = CSP add-ons to existing power plants, green triangles = new CSP hybrid plants, and yellow pentagons = new CSP-only plants. Adapted from Geels & Schott (2007).

According to the reconfiguration pathway the first adoption of symbiotic innovations leaves the regime mostly unchanged but the exploration of further combinations leads to the implementation of more innovations and a subsequent adjustment of the regime's basic architecture. The CSP retrofit at Liddell left the Australian electricity generation market basically unchanged but the CSP retrofit to the Kogan Creek power station can be considered an exploration of further combinations and an implementation of more innovations as the plant integrates the solar steam at higher steam parameters into the Rankine cycle systems and is based on design improvements from the initial demonstration plant at Liddell. However, possible CSP retrofits to existing power stations are limited to 16 opportunities in Australia (Meehan 2013) and only retrofitting CSP to fossil fuel assets carries the risk of extended fossil fuel lock-in. The fossil fuel lock-in has been discussed in the context of carbon capture and storage which identified its pairing with biomass as a sustainable long-term path for the technology (Vergragt, Markusson & Karlsson 2011). Similarly, future CSP-fossil fuel hybrids have to be assessed based on their net greenhouse gas abatement potential. However, with the local CSP experiences gained from first CSP retrofits, regime actors aiming to replace aging infrastructure could explore and implement new CSP hybrids as the next low-cost CSP options (see 4 and 5 in Figure 43). Current proposals for such replacements are the 30 MWe CSP-natural gas study at Collinsville (Australian Renewable Energy Agency 2013a) and the proposed 35.5 MWe CSP-EfB and CSP-EfW hybrid at Swanbank (Ipswich City Council 2012; Peterseim et al. 2012a). The benefits of CSP-EfB and CSP-EfW hybrids over CSP-natural natural gas hybrids include fully renewable power generation, apart from the fossil fuel content in RDF, qualification for more renewable energy certificates, relatively stable fuel prices when using waste materials, and more employment through the feedstock supply chain. At this point in time significant potential exists for CSP–EfB and CSP–EfW hybrid plants using current agricultural and forestry production systems in Australia and first demonstration units, shown as 4 in Figure 43, can pave the way for the deployment of various plants with different capacities across Australia, shown as 5 in Figure 43. Ongoing plant deployment would give component suppliers the confidence to invest in efficient manufacturing infrastructure, upskill staff and lower future production costs rather than going through the uncertain boom and bust cycles that occasional projects create. The cumulative learning from these new CSP-EfB and CSP–EfW, as well as other, hybrid plants can lower costs and increase confidence as well as awareness of CSP in the Australian electricity generation market. Additionally, the learning from CSP–EfB and CSP–EfW hybrid plants in Australia could be transferred to other parts of the world where good solar, biomass and waste feedstock resources are available in the same locations. This would strengthen the awareness of CSP internationally as a reliable renewable energy technology and simultaneously create export opportunities for local companies. The cumulative deployment of, and associated learning with, CSP hybrids in Australia and overseas, as shown in Figure 26 on page 53, would lower CSP costs continuously to a point where actors in the Australian electricity generation market are capable of financing commercial CSP-only plants with thermal storage in locations where CSP–EfB and CSP–EfW hybrid plants cannot be built due to resource constraints, shown as 6 in Figure 43.

The outlined hypothetical CSP implementation pathway describes a plausible transition but transitions are "*so complex with so many uncontrollable factors playing a role that attempts to steer or guide their course will always have an element of tentativeness*" (Elzen, Geels & Green 2004, p. 298). Therefore continuous monitoring, feedback, and adjustments of the transition process are necessary to maximise the likelihood of reaching the desired goal of a sustainable electricity generation market.

Based on the currently moderate landscape and regime pressure in the Australian electricity generation market, its transition from the current fossil fuel dominance to one where CSP plants play a significant role will require the adoption of various symbiotic CSP hybrid technologies. This trend has already started with the construction of two low-cost CSP retrofits to existing coal power plants. The limited retrofit opportunities will lead to new CSP hybrids being the next low-cost CSP option, and depending on the plant location various energy sources exist for CSP hybrids including EfB, EfW and natural gas. The CSP-EfB and CSP–EfW potential in Australia is significant and therefore such hybrids can play a relevant role, but to accelerate CSP deployment and reduce costs as much as possible, additional hybrid configurations have to be embraced considering the net environmental benefits and learning experiences provided by such plants. This highlights that a single promising technical niche innovation, such as CSP-EfB and CSP-EfW hybrids, is not able to commence a transition process but requires interactions and support from the regime and landscape levels, as per transition theory (Schot & Geels 2008). At the regime level this includes current government policies, such as the RET and the carbon pricing mechanism, as well as innovative plant operators and financiers. At the landscape level geopolitical pressure, such as a potential climate agreement at the 2015 UN climate summit in Paris, could accelerate the implementation of sufficiently mature niche innovations, such as CSP-EfB and CSP–EfW hybrid plants.

5 FUTURE RESEARCH

This research is considered to be a comprehensive starting point for CSP–EfB and CSP–EfW hybrid plants in Australia and internationally, and provides the basis for further analysis on this specific combination of energy sources but also other CSP hybrid configurations.

Research building on this PhD thesis could include the investigation of specific policy settings and tools to overcome the barriers identified in Section 4.5. Identifying the optimal policy settings is crucial for the deployment of renewable energy and other technologies and research in this area could identify which policies are required to continue the implementation of CSP hybrids and how these differ to the ones required for CSP-only plants. This work would have to be accompanied by techno-economic analyses to understand the plant cost as well as legislative aspects at the regional, state and federal levels.

Based on the CSP hybrid concepts in Section 4.4, future research could build on this analysis to incorporate new CSP, TES, EfB and EfW technologies and their impacts on hybrid plant efficiency and economic viability. Also, a detailed analysis of the optimal energy storage size based on technology cost improvements and future biomass/RDF feedstock prices would be relevant to better understand future hybrid concepts. New cooling concepts also provide scope for future research including the analysis of air–water hybrid cooling to maximise annual plant output. All these factors would also affect the greenhouse gas abatement potential of CSP–EfB and CSP–EfW hybrids.

The development of CSP technologies is ongoing and the results for the multi-criteria CSP suitability assessment for hybridisation, Section 4.3, are likely to change in the future. Updating this analysis with future CSP data would be interesting not only to identify the most suitable CSP technologies for hybridisation but also the trends that occurred through deployment, research and development.

Future economic modelling could be improved by using actual rather than predicted cost reductions. With additional CSP plant deployment, more accurate CSP costs will be available and these could improve the current modelling. Also of relevance is a more detailed understanding of the construction cost differences for CSP hybrids across Australia. A first analysis identified cost differences for CSP-only plants (Lovegrove et al. 2012) but more detailed work is required and similar work is yet to be done for hybrids. This area of research would require cooperation with an experienced construction company and a more localised geospatial analysis than the one in Section 4.2 to identify the best CSP–EfB and CSP–EfW hybrid plant sites within the identified regions in regards to resource availability

and plant construction costs. A future geospatial analysis would have to investigate prospective sites on a local level, not only considering the solid feedstocks and DNI resources available but also additional constraints such as land ownership, other land users, site gradients and transport infrastructure.

Section 4.1 touches on the hybridisation of CSP with natural gas, coal, geothermal power and wind, and each field has scope for further research. Due to the expected long-term growth of natural gas fired generation and resources projects that strongly depend on increasingly expensive natural gas, CSP has the potential to be a valuable fuel saver in areas with sufficiently high solar irradiance. Currently, most CSP hybrid references are with natural gas and in addition to fuel and greenhouse gas savings the operational risk of such plants would be low as the natural gas fired component can ensure redundancy for energy dependent industries at all times. Figure 44 shows that in principle, various mine sites were in good to excellent DNI areas in 2010 and future research could analyse these and new developments in regards to their suitability for the implementation of CSP–natural gas hybrids for process heat and electricity generation. Such research should also analyse the net environmental benefits of new CSP hybrid plants and CSP retrofits to existing natural gas plants.



Figure 44: Overlay of DNI (PIRSA Spatial Information Services 2009) with mine sites in Australia (Geoscience Australia 2010)

The high energy cost and high solar irradiance would in some cases increase the prospect of implementing the first CSP plants without government support or with minimal government support. Significant research has been done on retrofitting CSP to coal plants and further research should focus on the plant age and net greenhouse gas savings of such concepts, bearing in mind that that they might extend the operational life of the coal fired unit. In Australia CSP, geothermal and wind resources are available in the some locations and further research into integration concepts, cost and greenhouse gas reductions is required to understand their potential benefits in more detail.

The transition management framework used in this research could also be used to analyse the impact renewable energy plants could have on Australia's transition to a low carbon future. Further research could include a detailed analysis of the timeframe of niche adoptions in the regime and the role of the different regime actors, such as researcher, financiers, operators and technology providers, in the adoption process.

Also of relevance for future research could be the identification of transition pathways that enable a quicker uptake of CSP, and other renewable energy sources in the Australian electricity market, including the identification of the necessary intervention, monitoring and adjustment processes.

6 CONCLUSIONS

Currently, some other renewables, such as PV and wind, are more cost competitive than CSP but CSP is of strategic importance for Australia because the country has one of the best solar resources in the world and because CSP can balance the intermittent renewable generation capacity that is currently being deployed (see Section 2.2). To enable the development of CSP in Australia its hybridisation with EfB and EfW plants and other energy sources offers, among other benefits, an immediate cost reduction pathway (see Section 2.4). However, to deploy such plants in an economically successful manner requires a detailed understanding of the current electricity market (see Section 2.1), energy transition processes and drivers (see Sections 2.6, 3.3 and 4.7), the available technologies (see Sections 2.2, 2.3, 3.2.1.1 and 4.3), technical concepts (see Sections 2.5, 3.2.2.1, 4.1, 4.4 and 4.6), resources (see Sections 2.2.3, 3.2.2.2 and 4.2) and implementation barriers (see Sections 3.2.1.2 and 4.5).

To analyse hybrid plants a categorisation of light, medium and strong hybrids was introduced in Section 4.1 which classifies projects according to the degree of interconnection of the CSP with the additional power generation component such as an EfB or EfW steam system. This categorisation also indicates the level of benefit a CSP hybrid plant can provide, as strong hybrids are able to use more equipment and more capital intensive equipment jointly. Strong hybrids therefore create greater cost synergies than light hybrids that only share a few plant components. The same section expands beyond CSP–EfB and CSP–EfW hybrids and provides a high-level analysis of prospective regions for CSP hybrids with natural gas, geothermal and wind in Australia. All these hybrid combinations have combined resources in Australia, providing a basis for potential deployment in the future.

Separate resource analyses for dedicated CSP and EfB plants were available but to prove that CSP–EfB and CSP–EfW hybrid plants are a viable option in Australia a detailed resource analysis was required to not only identify prospective regions that have sufficiently high DNI and biomass/waste feedstocks, but also to assess the annual electricity output and greenhouse gas abatement potential. Section 4.2 demonstrates that in Australia the states of Western Australia and New South Wales have the highest potential for CSP–EfB and CSP– EfW hybrids but good resources also exist in Victoria, South Australia and Queensland. Most of the identified regions could support 5-40 MWe plants but in proximity to Perth, Brisbane, Adelaide and Canberra in particular, large CSP-multiple feedstock hybrid plants with capacities of up to 270 MWe are possible. The analysis also shows that stubble is the most promising feedstock for CSP–EfB hybrids and such plants could be located across large areas of Australia while wood waste is the dominant fuel for CSP-EfW hybrids and mostly occurs close to population centres. Using all the feedstocks in the identified areas would suffice to generate up to 30.8 TWh of electricity in CSP-single feedstock hybrid plants or 33.5 TWh of electricity in CSP-multiple feedstock hybrid plants. This equals up to 74% of the current 45 TWh renewable energy target and adding mature thermal energy storage can further raise the total electricity potential, for example the figure rises to 40.3 TWh when adding 7.5 h thermal energy storage to all plants. Based on this electricity generation potential and Australia's 2010 average carbon intensity for electricity generation of 841 kg CO₂/MWh, the annual greenhouse gas abatement potential is 27 million tonnes or 4.8% of all Australian 2009–10 emissions. These facts show that the resources for CSP–EfB and CSP–EfW hybrid plants from current agricultural and forestry production systems within proximity to existing transmission infrastructure are significant and worth investigating further.

Given that there is a good resource potential for CSP–EfB and CSP–EfW hybrid plants in Australia, an understanding of the most promising CSP technologies for hybridisation is important. In Section 4.3 various currently available CSP technologies were analysed using a multi-criteria decision-making approach that incorporated quantitative as well as qualitative information. Because of factors relating to the biomass/waste feedstock and the point of CSP integration (such as feedwater heating, reheat steam line or direct turbine steam) solar tower and Fresnel systems with direct steam generation rank best at this time for hybridisation at high steam parameters. This is due to a variety of factors including high steam parameter capability, cost, plant complexity, and good reference situation for smaller plants. The results of the technology ranking process developed and implemented as part of this research matches the actual CSP selection for various hybrid plants already in operation/under construction worldwide and provides a fast but scientifically sound decision-making tool for developers of future hybrid projects. To remain usable in the future, and to solve location-specific technology selection problems, the ranking process is designed flexibly to allow the incorporation of new and site-specific CSP data.

To gain a detailed understanding of the most suitable CSP technologies for hybridisation with EfB and EfW plants, a comparative analysis of the different hybrid concepts was essential to identify the best available at this time. Previous research is available on a few selected CSP–EfB and CSP–EfW hybrid technologies and concepts but in Section 4.4 various scenarios were analysed for two fundamentally different high solar share concepts: a) projects in which the CSP, EfB and EfW steam generators provide identical high pressure/temperature steam to a joint turbine; and b) projects in which biomass and waste feedstocks are used to externally superheat the steam from a conventional parabolic trough plant. The first concept is suitable for areas with significant biomass/waste feedstock resources and allows the independent operation of the CSP, EfB and EfW steam generators while the second concept aims to lower the cost of the mature parabolic trough technology through the use of small quantities of biomass and waste feedstock. However, this does require the simultaneous operation of the CSP, EfB and EfW steam systems. The modelling results for the first concept show that solar tower hybrids reach the highest net cycle efficiency (32.9%), but Fresnel hybrids have the lowest specific investment (AU\$ 4.5m/MWe) and they therefore realise the highest internal rate of return. The analysis also reveals that the cost differences between the 17 analysed scenarios vary by up to 31%, which is significant and underlines that the right technology pairing is highly important. CSP hybrids have a 12% lower specific cost than a same capacity CSP-only plants but when comparing plant cost based on the same annual electricity generation, CSP-EfB and CSP–EfW hybrids can be 69% less capital intensive than CSP-only plants as the EfB/EfW component is significantly less capital intensive than solar field or thermal energy storage equipment. The analysis of seven scenarios for the biomass/waste feedstock-fired external superheater concept, applied to a 50 MWe plant with 7.5 h thermal energy storage, shows that depending on the feedstock used, the peak solar to electricity net efficiency increases by up to 10.5% while the specific investment decreases immediately from AU\$ 8.2m/MWe to AU\$ 6.3m/MWe, a 23.5% reduction. This is significant considering that the CSP industry aims to achieve a 17–40% cost reduction by the end of this decade. Through hybridisation this target could be met immediately.

Despite the significant cost reduction potential through hybridisation, the capital requirement is still a substantial barrier to CSP deployment. However, other social, technical, environmental and policy barriers exist, and knowing them and their importance is important for identifying a pathway for CSP in Australia. Various barriers are identified and analysed in Section 4.5 with a novel assessment of the rating differences between the main barriers for CSP hybrid and CSP-only plants. Four of the top five barriers are identical for CSP-only and hybrid plants. These are: high project capital cost, lack of financial incentives, policy uncertainty, and abundant and subsidised fossil fuels, but on average these barriers are ranked less important for hybrids than for CSP-only plants. While the availability of lower cost renewable alternatives is one of the top five barriers for CSP-only plants it was not rated in the top five barriers for hybrids and replaced by the lack of government policy focus. Workshop participants rated some barriers as being significantly

less important for hybrid plants than for CSP-only plants. These barriers included: generation intermittency/lack of energy storage (25% less important), network access, capacity and availability (24% less important), and high project capital/financing costs (19% less important). The barriers with the smallest ranking differences included the lack of codes/standards (1.5% less important for hybrids), inefficient use of back-up fuels (1.6% less important), and resource assessment/knowledge of best sites (2.4% less important). These results support the assumption that CSP hybrids provide more than technical and economic benefits as they also lower relevant policy, social and environmental barriers.

The findings from the various research components have been used in two case studies, (see Section 4.6) to assess the viability of CSP-EfB and CSP-EfW hybrid plants in real business case scenarios. The first case study at the Swanbank landfill in Queensland demonstrates that a 35.5 MWe plant, with 30.7 MWe from EfB and EfW and 4.8 MWe from CSP, could operate highly efficiently on multiple feedstocks, producing not only renewable electricity but also enabling waste diversion from landfill. The proposed plant could deploy the CSP system at 40% lower cost than a CSP-only plant, generate 252,800 MWh of electricity annually with a low carbon intensity of $129 \text{ kg CO}_2/\text{MWh}$, abate up to 3.4 million tonnes of CO_2 over its lifetime, and be economically viable within an electricity price range of AU\$ 100–120/MWh. The second case study at Griffith in New South Wales investigates a 30 MWe CSP-straw hybrid plant with 3h thermal energy storage, 15 MWe from CSP and 15 MWe from straw. Compared to a CSP-only plant with identical annual generation, the proposal could lower CSP costs by 43%, generate on average 160,300 MWh of electricity annually, abate up to 2.2 million tonnes of CO_2 over its lifetime, and be economically viable with an electricity price of AU\$ 155/MWh. Both plants would provide significant socioeconomic benefits through higher employment than a fossil fuel fired power generation alternative and stimulation of the local economies, for example through increased local manufacturing and installation of plant components.

With all this information supporting the various benefits of CSP–EfB and CSP–EfW hybrid plants, their potential role and pathway in Australia's transition to a low carbon electricity future had to be analysed (see Section 4.7). The current electricity market is changing due to pressures at various levels, including environmental problems, higher international greenhouse gas reduction targets, current government policies to incentivise renewable energy, and falling electricity demand in recent years. While policy uncertainty is a key concern changes in the electricity market are unlikely to stop due to parallel developments, such as residential roof-top photovoltaic systems and rising natural gas prices. CSP is a strategically relevant technology for Australia's low carbon electricity future and CSP hybrids have the potential to accelerate CSP uptake in the near term. Transition management is a suitable framework to analyse their role and several of its features can be observed in regards to CSP. The technology has been/is being supported through government funding for research, through programs such as the Australian Solar Thermal Research Initiative, feasibility studies such as Collinsville hybrid study, and first experimental plants such as solar towers in Newcastle and a big dish in Canberra. First demonstration plants, such as the original CSP retrofit to Liddell power station, were built and experiment scale-ups can be observed, such as the recent CSP expansion at the Liddell power station and the CSP retrofit to the Kogan Creek power station. While the deployment of CSP-only plants has unfortunately not yet occurred, some CSP hybrid projects have progressed and provide local learning experiences that will be relevant to the future CSP development in Australia. With limited market pressures, a reconfiguration pathway is very suitable for assessing the potential future deployment of CSP in Australia. On this pathway, regime actors adopt symbiotic innovations to solve local problems, and over time, explore and implement further combinations which change the regime. Current and new low-cost CSP retrofits to existing power plants, and the deployment of new CSP hybrid plants, not only with EfB and EfW but with other energy sources as well, have the potential to progressively replace aging electricity generation infrastructure and build local learning and investment confidence to a point where costs have decreased to the extent that it will be feasible to deploy CSP-only plants with a range of generation and energy storage capacities. In Australia CSP–EfB and CSP–EfW hybrid plants, in combination with other renewable energy, energy efficiency and fossil fuel technologies, have the technical, economic and resource potential to transition the electricity market towards a significantly lower carbon intensity and a more sustainable future. The local learnings obtained from current hybrid installations and plants that will be built in the near future will contribute, in combination with growing international experience, to lower technology costs and the progressive deployment of CSP in low-cost wholesale electricity markets. This applies to the Australian market and it is also a model for other countries worldwide.

REFERENCES

The references below derive from chapters 1-4. However, the publications referred to in

chapter 4 contain additional references and not all of them are listed here. The access dates

to online references in the publications can differ from the access dates in chapters 1-4.

ABC 2013, 'Forbes solar demonstration model get tick of approval from council', 23 September, Sydney, viewed 4 April 2014, http://www.abc.net.au/news/2013-09-23/forbes-solar/4974124>.

Abengoa 2013, Integrated Solar Combined Cycle, Seville.

Abengoa Solar 2013a, 'Abengoa completes tower construction of the first solar tower in South Africa, Khi Solar One', 27 August, viewed 4 April 2014, <http://www.abengoa.com/web/en/noticias_y_publicaciones/noticias/historico/2013/08_ agosto/abg_20130826.html>.

— 2013b, 'Abengoa's Solana, the US's first large-scale solar plant with thermal energy storage system, begins commercial operation', 9 October, viewed 4 April 2014,
http://www.abengoa.com/web/en/noticias_y_publicaciones/noticias/historico/2013/10_octubre/abg_20131009.html.

 2014a, 'Abengoa to develop South America's largest solar-thermal plant in Chile', 9 January, viewed 4 April 2014,

<http://www.abengoasolar.com/web/en/acerca_de_nosotros/sala_de_prensa/noticias/20 14/solar_20140109.html?q=chile>.

— 2014b, Operating facilities - Spain, Seville, Spain, viewed 4 April 2014, <www.abengoasolar.com/corp/web/en/nuestras_plantas/plantas_en_operacion/espana/in dex.html>.

Adelaide Brighton 2007, *Alt fuels and raw materials*, Adelaide, viewed 4 April 2014, <www.adbri.com.au/altfuelsandrawmaterials.html>.

Akash, BA, Mamlook, R & Mohsen, MS 1999, 'Multi-criteria selection of electric power plants using analytical hierarchy process', *Electric Power Systems Research*, vol. 52, no. 1, pp. 29–35.

Alrobaei, H 2008, 'Novel integrated gas turbine solar cogeneration power plant', *Desalination*, vol. 220, no. 1-3, pp. 574–587.

Andersen, EJ, Oksen, F & Olesen, JV 1992, 'Power plant Holstebro / Struer with separately fired superheater', in *Eckrohrkessel licensee conference*, ERK Eckrohrkessel GmbH, Bamberg.

Arena, U 2012, 'Process and technological aspects of municipal solid waste gasification. A review.', *Waste management*, vol. 32, no. 4, pp. 625–39.

AREVA Solar 2011, 'Australia: AREVA Part of Consortium Selected as Preferred Project for Solar Flagships Program', 21 June, Paris, viewed 4 April 2014, <http://www.areva.com/EN/news-8934/australia-areva-part-of-consortium-selected-aspreferred-project-for-solar-flagships-program.html>.

 — 2012a, 'India: AREVA awarded contract by Reliance to build Asia's largest concentrated solar power installation', 11 April, Paris, viewed 4 April 2014, <www.areva.com/EN/news-9299/india-areva-awarded-contract-by-reliance-to-build-asia-s-largest-concentrated-solarpower-installation.html>. - 2012b, Liddell Solar Thermal Station.

— 2012c, Sundt Solar Boost Project, viewed 4 April 2014, <http://www.areva.com/arevasolar/liblocal/docs/TEP-Sundt_Solar_Boost_ Project-Keyfact.pdf>.

2014, AREVA commissions molten salt energy storage demonstration, viewed 30 July 2014, http://www.areva.com/EN/news-10195/solar-areva-commissions-molten-salt-energy-storage-demonstration.html#>.

Arias-Garcia, A, Gleeson, J & Ltd, RPSP and D 2009, *Solid Recovered Fuel Regional Assessments Coventry and Derbyshire*, Department for Environment, Food and Rural Affairs, UK, Bristol.

Astolfi, M, Xodo, L, Romano, MC & Macchi, E 2011, 'Technical and economical analysis of a solar–geothermal hybrid plant based on an Organic Rankine Cycle', *Geothermics*, vol. 40, no. 1, pp. 58–68.

AT Kearney & ESTELA 2010, Solar Thermal Electricity 2025, Duesseldorf, Germany.

Australian Bureau of Agricultural and Resource Economics 2011, *Energy in Australia 2011*, Canberra, Australia.

Australian Bureau of Statistics 2014, *Labour Force, Australia, Jan 2014*, viewed 4 March 2014, <http://www.abs.gov.au/ausstats/abs@.nsf/mf/6202.0>.

Australian Coal Assocation 2014, *COAL21*, viewed 4 April 2014, <www.australiancoal.com.au/coal-21.html>.

Australian Energy Market Commission 2011, *Possible Future Retail Electricity Price Movements: 1 July 2011 to 30 June 2014, Final Report*, Sydney.

- 2013, Residential Electricity Price Trends, Sydney.

Australian Energy Market Operator 2014, Supply Demand Snapshot February 2014.

Australian Government 2009, *Budget paper no.2 2009-10, Part 2: Expense Measures*, Canberra.

— 2011, Clean Energy Act 2011, Canberra.

Australian Petroleum Production and Exploration Association 2010, *Climate change policy principles*, Canberra .

Australian Renewable Energy Agency 2013a, 'Feasibility study into conversion of Collinsville Power Station from coal to hybrid solar thermal/gas', 7 January, Canberra, viewed 4 April 2014, <http://arena.gov.au/project/feasibility-study-into-conversion-of-collinsville-powerstation-from-coal-to-hybrid-solar-thermalgas/>.

— 2013b, Solar energy projects, viewed 4 April 2014, <http://arena.gov.au/projects/solar/>.

- 2014, About ARENA, viewed 4 April 2014, http://arena.gov.au/about-arena/>.

Ávila-Marín, AL 2011, 'Volumetric receivers in Solar Thermal Power Plants with Central Receiver System technology: A review', *Solar Energy*, vol. 85, no. 5, pp. 891–910.

Baban, SMJ & Parry, T 2001, 'Developing and applying a GIS-assisted approach to locating wind farms in the UK', *Renewable Energy*, vol. 24, no. 1, pp. 59–71.

Babcock & Wilcox Vølund A/S 2008, *Multifuel-fired Combined Heat and Power Plant Måbjergværket Måbjerg, Denmark*, Esbjerg, Denmark, viewed 4 April 2014,

<www.volund.dk/Multi_fuel_energy/~/media/Downloads/Brochures - MULTI/Maabjerg - Holstebro - Denmark.ashx>.

Baffes, J 2013, 'A framework for analyzing the interplay among food, fuels, and biofuels', *Global Food Security*, vol. 2, Elsevier, no. 2, pp. 110–116.

Baghernejad, A & Yaghoubi, M 2010, 'Exergy analysis of an integrated solar combined cycle system', *Renewable Energy*, vol. 35, Elsevier Ltd, no. 10, pp. 2157–2164.

Bahl, C, Bauer, T, Laing, D, Lehmann, D & Steinmann, W-D 2011, 'Thermal energy storage for direct steam generation', *Solar Energy*, vol. 85, no. 4, pp. 627–633.

Bakos, GC & Tsechelidou, C 2013, 'Solar aided power generation of a 300 MW lignite fired power plant combined with line-focus parabolic trough collectors field', *Renewable Energy*, vol. 60, pp. 540–547.

Banister, D 2008, 'The sustainable mobility paradigm', *Transport Policy*, vol. 15, no. 2, pp. 73–80.

Barnett, J & Adger, WN 2007, 'Climate change, human security and violent conflict', *Political Geography*, vol. 26, no. 6, pp. 639–655, viewed 10 July 2014, http://linkinghub.elsevier.com/retrieve/pii/S096262980700039X.

Baumann, C, Asker, S, Giurco, D, Peterseim, JH & White, S 2012, *ECO-INDUSTRIAL TRANSITION' A vision for economic and socio-ecological renewal at Swanbank*, Prepared for Thiess Services Pty Ltd by the Institute for Sustainable Futures, University of Technology, Sydney, Australia.

Beetz, B 2012, 'Whole 1 GW Blythe project to convert to PV', *PV Magazine*, 3 July, viewed 4 April 2014, <www.pv-magazine.com/news/details/beitrag/whole-1-gw-blythe-project-to-convert-to-pv_100007604/#axz2A5Oyy1Jv>.

Bellac, AH & Destefanis, R 1999, *Solar power enhanced combustion turbine power plant and methods, U.S. patent 6000211*, US 6000211, The United States Patent and Trademark Office, U.S.

BGS e.V. 2011, Mitverbrennung von gütegesicherten Sekundärbrennstoffen und energetische Verwertung (co-combustion of quality controlled secondary recovered fuels and energetic use), Münster.

Bianchini, A, Pellegrini, M & Saccani, C 2013, 'Solar steam reforming of natural gas integrated with a gas turbine power plant', *Solar Energy*, vol. 96, pp. 46–55.

Biomass Power & Thermal Magazine 2011, 'Bagasse Power', 25 January, viewed 4 April 2014, http://biomassmagazine.com/articles/5247/bagasse-power.

Boisvert, E 2013, '\$200m Broken Hill solar plant approved', *ABC News*, 2 April, Sydney, viewed 4 April 2014, http://www.abc.net.au/news/2013-04-02/24200m-solar-plant-approved-by-state-government/4604438>.

Brakmann, G, Mohammad, F, Dolejsi, M & Wiemann, M 2009, 'Construction Of The ISCC Kuraymat', in *SolarPACES Conference*, Berlin, Germany.

BrightSource Energy Inc. 2013, *Ivanpah Overview*, Oakland, California, US, viewed 4 April 2014, <http://ivanpahsolar.com/about>.

Bryan, BA, Ward, J & Hobbs, T 2008, 'An assessment of the economic and environmental potential of biomass production in an agricultural region', *Land Use Policy*, vol. 25, no. 4, pp. 533–549.

Bryden, J, Riahi, L & Zissler, R 2013, *MENA renewables status report*, United Arab Emirates Ministry of Foreign Affairs' Directorate of Energy & Climate Change, IRENA & REN21, Paris.

Bureau of Meteorology 2012, Australian Monthly Average Hourly Global and Direct Normal Solar Exposure Gridded Data, Melbourne.

Bureau of Resources and Energy Economics 2013, Energy in Australia 2013, Canberra.

Byrnes, L, Brown, C, Foster, J & Wagner, LD 2013, 'Australian renewable energy policy: Barriers and challenges', *Renewable Energy*, vol. 60, pp. 711–721.

Caldés, N, Varela, M, Santamaría, M & Sáez, R 2009, 'Economic impact of solar thermal electricity deployment in Spain', *Energy Policy*, vol. 37, no. 5, pp. 1628–36.

California Energy Comission 2008, *San Joaquin Solar 1 & 2 Power Plant Licensing Case*, Sacramento, California.

Cameron, P & Crompton, G 2008, *Solar Power Plant Pre-feasibility Study*, Prepared by Parsons Brinckerhoff for ActewAGL and ACT Government, Brisbane.

Carey, A 2011, 'Jobs, not climate change, priority for Latrobe Valley', *The Age*, 7 July, Melbourne.

Cau, G, Cocco, D & Tola, V 2012, 'Performance and cost assessment of Integrated Solar Combined Cycle Systems (ISCCSs) using CO 2 as heat transfer fluid', *Solar Energy*, vol. 86, pp. 2975–85.

Chang, YH, Chen, WC & Chang, N Bin 1998, 'Comparative evaluation of RDF and MSW incineration', *Journal of Hazardous Materials*, vol. 58, no. 1-3, pp. 33–45.

Charabi, Y & Gastli, A 2010, 'GIS assessment of large CSP plant in Duqum, Oman', *Renewable and Sustainable Energy Reviews*, vol. 14, no. 2, pp. 835–841.

Clean Energy Finance Corporation 2014, *About us*, viewed 4 April 2014, https://www.cleanenergyfinancecorp.com.au/about-us.aspx>.

Clifton, J & Boruff, B 2010, *Site Options for Concentrated Solar Power Generation in the Wheatbelt*, Prepared by the University of Western Australia for the Wheatbelt Development Commision, Perth.

Climate Change Authority 2012, Renewable energy target review, Canberra.

 — 2013, Reducing Australia's greenhouse gas emissions - Targets and Progress review -Final report, Melbourne.

Coelho, B, Domingues, P, Schwarzboezl, P, Hennecke, K, Oliveira, A & Mendes, A 2011, 'Biomass and Central Receiver Solar Plant (CRS) Hybridisation', in *SolarPACES Conference*, Granda, Spain.

Commonwealth of Australia 2013, Emissions Reduction Fund Green Paper, Canberra.

Cook, J, Nuccitelli, D, Green, S a, Richardson, M, Winkler, B, Painting, R, Way, R, Jacobs, P & Skuce, A 2013, 'Quantifying the consensus on anthropogenic global warming in the scientific literature', *Environmental Research Letters*, vol. 8, no. 2, p. 024024.

Cot, A, Amettler, A, Vall-Llovera, J, Aguilo, J & Arque, JM 2010, 'Termosolar Borges: A Thermosolar Hybrid Plant with Biomass', in *Third International Symposium on Energy from Biomass and Waste*, Venice, Italy.

Cox, J, Downie, A, Jenkins, A, Hickey, M, Lines-Kelly, R, McClintock, A, Powell, J, Singh, BP & Van Zwieten, L 2012, *Biochar in horticulture: Prospects for the use of biochar in Australian horticulture*, NSW Department of Primary Industries, Sydney.

Crespo, L 2013, 'Spain: Dramatic Impact of Retroactive Regulatory Changes', in *ESTELA Summer Workshop*, Protermo Solar, Brussels.

Creswell, JW 2009, *Research design: qualitative, quantitative, and mixed methods approaches*, 3rd edn, SAGE Publications, Thousand Oaks, California.

CS Energy 2011, 'World's largest solar integration with a coal-fired power station gets go ahead', 13 April, Brisbane, viewed 4 April 2014, <http://kogansolarboost.com.au/wp-content/uploads/2011/04/Media-2011-MR11-019-Kogan-Creek-Solar-Boost-FINAL-Project-to-proceed-CS-Energy-media-13-April-2011.pdf>.

— 2013, Kogan Creek power station, CS Energy, Brisbane, viewed 4 April 2014,
http://www.csenergy.com.au/userfiles/file/Kogan Creek Power Station fact sheet - April 2013.pdf>.

CSIRO 2012, Australia's largest solar thermal research hub, viewed 4 April 2014, http://www.csiro.au/Outcomes/Energy/Renewables-and-Smart-Systems/Solar-Brayton-Cycle.aspx.

— 2013, Australian solar thermal research initiative, viewed 4 April 2014, <http://www.csiro.au/en/Organisation-Structure/Flagships/Energy-Flagship/ASTRI.aspx>.

CSP World 2013, 'Australia's Vast Solar gets ok for 1.2 MW demonstration CSP plant', 24 September, viewed 4 April 2014, http://www.csp-

world.com/news/20130924/001185/australias-vast-solar-gets-ok-12-mw-demonstration-csp-plant>.

— 2014, Liddell Solar Thermal Station, viewed 31 July 2014, http://www.csp-world.com/cspworldmap/liddell-solar-thermal-station.

Cunningham, JB, Farquharson, J & Hull, D 1991, 'A profile of the human fears of technological change', *Technological Forecasting and Social Change*, vol. 40, no. 4, pp. 355–370.

Dawson, L & Schlyter, P 2012, 'Less is more: Strategic scale site suitability for concentrated solar thermal power in Western Australia', *Energy Policy*, vol. 47, pp. 91–101.

Deloitte 2011, *Macroeconomic impact of the Solar Thermal Electricity Industry in Spain*, Prepared by Deloitte for Protermosolar, Madrid.

Denholm, P & Mehos, M 2011, Enabling Greater Penetration of Solar Power via the Use of CSP with Thermal Energy Storage, National Renewable Energy Laboratory, Golden, Colorado.

Department of Climate Change and Energy Efficiency 2012, *Australia's emissions projections*, Canberra.

Dersch, J, Geyer, M, Herrmann, U, Jones, SA, Kelly, B, Kistner, R, Ortmanns, W, Pitz-Paal, R & Price, H 2004, 'Trough integration into power plants - a study on the performance and economy of integrated solar combined cycle systems', *Energy*, vol. 29, no. 5-6, pp. 947–959.

Dimock, M, Doherty, C & Kohut, A 2013, *Public's Policy Priorities: 1994-2013*, Pew Research Center, Washington.

Domac, J, Richards, K & Risovic, S 2005, 'Socio-economic drivers in implementing bioenergy projects', *Biomass and Bioenergy*, vol. 28, no. 2, pp. 97–106.

Dopita, M & Williamson, R 2009, *Australia's renewable energy future*, Australian Academy of Science, Canberra, Australia.

Van Driel, H & Schot, J 2005, 'Radical Innovation as a Multilevel Process: Introducing Floating Grain Elevators in the Port of Rotterdam', *Technology and Culture*, vol. 46, no. 1, pp. 51–76.

Dunstan, C, Ross, K & Ghiotto, N 2011, *Barriers to demand management: A survey of stakeholder perceptions*, Prepared for the Australian Alliance to Save Energy by the Institute for Sustainable Futures, University of Technology, Sydney.

EASAC 2011, *Concentrating solar power: its potential contribution to a sustainable energy future*, German Academy of Sciences Leopoldina, Halle, Germany.

ecoprog 2013, *Waste to Energy - The World Market for Waste Incineration Plants 2013/2014*, Cologne.

Edwards, V 2013, 'Axing of \$60m solar grant floors Whyalla', *The Australian*, 11 June, Sydney, Australia.

Effendi, P & Courvisanos, J 2012, 'Political aspects of innovation: Examining renewable energy in Australia', *Renewable Energy*, vol. 38, no. 1, pp. 245–252.

Elliston, B, Diesendorf, M & MacGill, I 2012, 'Simulations of scenarios with 100% renewable electricity in the Australian National Electricity Market', *Energy Policy*, vol. 45, pp. 606–613.

Elliston, B, MacGill, I & Diesendorf, M 2013, 'Least cost 100% renewable electricity scenarios in the Australian National Electricity Market', *Energy Policy*, vol. 59, pp. 270–282.

Elzen, B, Geels, FW & Green, K 2004, *System Innovation and the Transition to Sustainability: Theory, Evidence and Policy*, Edward Elgar Publishing Limited, Cheltenham.

EnBW Energie Baden-Württemberg AG 2009, 'EnBW versorgt Europas größte Papiermaschine für Wellpappenrohpapier langfristig mit Dampf und Strom (EnBW to provide long-term steam and electricity supply to Europe's largest paper machine for containerboard)', 15 January, Karlsruhe, Germany, viewed 4 April 2014, <http://www.enbw.com/unternehmen/presse/pressemitteilungen/pressedetailseite_10098.html>.

ENEA-Press and Media Relations Office 2012, 'Roma Capitale, ENELGreen Power and ENEA sign an agreement to build a concentrated solar power and biomass plant', 13 April, Rome, viewed 4 April 2014, http://www.enel.com/en-GB/media/press_releases/roma-capitale-enel-green-power-and-enea-sign-an-agreement-to-build-a-concentrated-solar-power-and-biomass-plant/r/1651545.

Entsorgungsgesellschaft Mainz mbH 2007, *The Mainz Waste-to-Energy Plant: Exceptionally reliable and environmentally sound waste disposal*, Mainz, Germany, viewed 4 April 2014, <www.mhkw-mainz.de/downloads/EGM_infobro3_eng_20.pdf>.

Environment Protection and Heritage Council 2010, *National Waste Report 2010*, Canberra, Australia.

Environmental Protection Agency NSW 2013, *NSW Enery from waste - draft policy statement for public consultation*, Sydney.

Environmental Protection Agency Victoria 2013, *Energy from waste - Publication 1549*, Melbourne.

Environmental Protection Authority 2013, 'Pilbara Waste to Energy facility recommended for approval', 8 April, Perth, viewed 4 April 2014,

<http://www.epa.wa.gov.au/News/mediaStmnts/Pages/PilbaraWastetoEnergyfacilityreco mmendedforapproval.aspx?pageID=323&url=News/mediaStmnts>. eSolar Inc. 2013, *Sierra SunTower - A New Blueprint for Solar Energy*, Burbank, California, US, viewed 4 April 2014, http://www.esolar.com/wp-content/uploads/2013/10/sierra fact sheet.pdf>.

European Commission 2014, *The 2020 climate and energy package*, viewed 4 April 2014, <http://ec.europa.eu/clima/policies/package/>.

European Environment Agency 2007, Europe's environment - The fourth assessment, Chapter 6: Sustainable consumption and production, Copenhagen.

European Environmental Agency 2009, *Diverting waste from landfill - Effectiveness of waste management policies in the European Union*, European Environment Agency, Copenhagen.

European Parliament and Council 2010, *Directive 2010/75/EU on industrial emissions* (*integrated pollution prevention and control*), European Parliament and Council, Brussels.

Europlasma Group 2014, *Hazardous waste treatment: Asbestos waste vitrification*, viewed 4 April 2014, http://www.europlasma.com/en/hazardous-waste.html>.

Evans & Peck 2011, Assessment of the potential for renewable energy projects and systems in the Mid West, Perth.

Falchetta, M, Mazzei, D, Crescenzi, T & Merlo, L 2009, 'Design of the Archimede 5MW Molten Salt Parabolic Trough Solar Plant', in *Solar PACES Conference*, Berlin, Germany.

Fanning, M 2013, Investigation into the performance (environmental and health) of waste to energy technologies internationally, Stage One - Review of Legislative and Regulatory Frameworks for Waste to Energy Plants, Perth.

Farine, D, O'Connell, DA, John Raison, R, May, BM, O'Connor, MH, Crawford, DF, Herr, A, Taylor, JA, Jovanovic, T, Campbell, PK, Dunlop, MIA, Rodriguez, LC, Poole, ML, Braid, AL & Kriticos, D 2011, 'An assessment of biomass for bioelectricity and biofuel, and for greenhouse gas emission reduction in Australia', *GCB Bioenergy*, vol. 4, no. 2, pp. 148–175.

Feeney, K, Kua, D, Balding, J & Goodhand, R 2010, *Pre-Feasibility Study for a Solar Power Precinct*, Prepared by AECOM Australia Pty Ltd for the NSW Department of Environment, Climate Change and Water, Sydney.

Fitzgibbon, J 2012, 'Cut the ideology and reduce renewable energy target', *The Australian*, 12 November, Sydney.

Fleisher, CS & Bensoussan, BE 2003, *Strategic and Competitive Analysis: Methods and Techniques for Analyzing Business Competition*, Prentice Hall.

Florida Power & Light Company 2010, *FPL's Martin Next Generation Solar Energy Center*, viewed 4 April 2014, <www.fpl.com/environment/solar/pdf/Martin.pdf>.

Foxon, TJ, Hammond, GP & Pearson, PJG 2010, 'Developing transition pathways for a low carbon electricity system in the UK', *Technological Forecasting and Social Change*, vol. 77, no. 8, pp. 1203–1213.

Franchini, G, Perdichizzi, A, Ravelli, S & Barigozzi, G 2013, 'A comparative study between parabolic trough and solar tower technologies in Solar Rankine Cycle and Integrated Solar Combined Cycle plants', *Solar Energy*, vol. 98, pp. 302–314.

Frazier, E 2013, 'Solar Augmentation Using Parabolic Trough', in *Eskom Solar Augmentation Conference*, Pretoria.

Frei, C, Whitney, R, Schiffer, H-W, Rose, K, Rieser, DA, Al-Qahtani, A & Thomas, P 2013, World Energy Scenarios - Composing energy futures to 2050, World Energy Council, London. Gan, J & Smith, CT 2007, 'Co-benefits of utilizing logging residues for bioenergy production: The case for East Texas, USA', *Biomass and Bioenergy*, vol. 31, no. 9, pp. 623–630.

García, E & Roberto Calvo 2012, 'One Year Operation Experience of Gemasolar Plant', in *SolarPACES Conference*, PSE AG, Marrakech.

Garcia-Perez, M, Lewis, T & Kruger, CE 2010, *Methods for producing biochar and advanced biofuels in Washington state - Part 1: Literature review of pyrolysis reactors*, Department of Biological Systems Engineering and the Center for Sustaining Agriculture and Natural Resources, Washington State University, Pullmann.

Gastli, A, Charabi, Y & Zekri, S 2010, 'GIS-based assessment of combined CSP electric power and seawater desalination plant for Duqum-Oman', *Renewable and Sustainable Energy Reviews*, vol. 14, no. 2, pp. 821–827.

GDF SUEZ 2013, 'GDF SUEZ opens one of the world's largest biomass units in Poland', 3 June, viewed 4 April 2014, http://www.gdfsuez.com/en/news/gdf-suez-opens-one-of-the-worlds-largest-biomass-units-in-poland/.

Geels, F & Raven, R 2006, 'Non-linearity and Expectations in Niche-Development Trajectories: Ups and Downs in Dutch Biogas Development (1973–2003)', *Technology Analysis & Strategic Management*, vol. 18, no. 3-4, pp. 375–392.

Geels, FW 2002, 'Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study', *Research Policy*, vol. 31, no. 8-9, pp. 1257–1274.

- 2011, 'The multi-level perspective on sustainability transitions: Responses to seven criticisms', *Environmental Innovation and Societal Transitions*, vol. 1, no. 1, pp. 24–40.

Geels, FW & Schot, J 2007, 'Typology of sociotechnical transition pathways', *Research Policy*, vol. 36, no. 3, pp. 399–417.

Gendebien, A, Leavens, A, Blackmore, K, Godley, A, Lewin, K, Whiting, KJ, Davis, R, Giegrich, J, Fehrenbach, H, Gromke, U, del Bufalo, N & Hogg, D 2003, *Refuse derived fuel, current practice and perspectives*, European Commission, Brussels.

Geoscience Australia 2010, Australian operating mines, Canberra.

German Federal Environment Agency 2008, *Waste Incineration and Waste Prevention: not a contradiction in terms*, Dessau, Germany.

Gerring, J 2007, *Case study research: principles and practices*, Cambridge University Press, Cambridge.

Gillingham, K & Sweeney, J 2012, 'Barriers to Implementing Low Carbon Technologies', *Climate Change Economics*, vol. 03, no. 04, p. 21.

Goebel, O & Luque, F 2012, 'Shams One 100 MW CSP Plant in Abu Dhabi - Update on Project Status', in *SolarPACES Conference*, Marrakech.

Gohlke, O 2008, 'Verbesserungspotenziale der Energieeffizienz (potentials to improve energy efficiency)', in *Thermische Abfallbehandlung, 13. Fachtagung*, Schriftenreihe des Fachgebietes Abfalltechnik, Universität Kassel, München.

Gohlke, O & Spliethoff, H 2007, 'Innovative approaches to increase efficiency in EfW plants - potential and limitations', in *ISWA conference*, Amsterdam.

Gordon, J 2001, *Solar energy: the state of the art: ISES position papers*, Earthscan/James & James, London.

Gou, C, Cai, R & Hong, H 2007, 'A novel hybrid oxy-fuel power cycle utilizing solar thermal energy', *Energy*, vol. 32, no. 9, pp. 1707–1714.

Granatstein, DL 2003, *Case study on waste-fuelled gasification project Greve in Chianti, Italy*, Prepared by Natural Resources Canada/CANMET Energy Technology Centre for IEA Bioenergy Task 36,.

GTM Research & Solar Energy Industries Association 2013, U.S. Solar market insight - Q2 2013.

Hall, N, Ashworth, P & Shaw, H 2012, *Exploring community acceptance of rural wind farms in Australia: a snapshot*, CSIRO, Canberra.

Hannam, P 2013, 'Renewables future at mercy of big power retailers', *The Sydney Morning Herald*, 2 April, Sydney.

Harvey, J 2012, 'Munmorah Power Station to close after 45 years of operation', *Delta Electricity*, 3 July, Sydney, viewed 4 April 2014,

<http://www.de.com.au/ArticleDocuments/15/Delta Electricity Munmorah Power Station Closure Media Statement FINAL.pdf.aspx>.

Hayward, JA, Graham, PW & Campbell, PK 2011, *Projections of the future costs of electricity generation technologies: An application of CSIRO's Global and Local Learning Model (GALLM)*, CSIRO, Newcastle, Australia.

Hearps, P & McConnel, D 2011, *Renewable Energy Technology Cost Review*, Melbourne Energy Institute, Melbourne, Australia.

Hellweg, E 1983, STEP 100 einhundert: Dt.-Austral. Sonnenkraftwerk in Meekatharra, West-Australien (STEP 100: german-australian solar power plant at Meekatharra, Western Australia), 1st edn, Fachinformationszentrum Energie, Physik, Mathematik, Karlsruhe, Germany.

Hepworth, A 2014, 'Gas prices force switch to coal for power stations', *The Australian*, Sydney.

Herr, A, O'Connell, D, Dunlop, M, Unkovich, M, Poulton, P & Poole, M 2012, 'Second harvest–Is there sufficient stubble for biofuel production in Australia?', *GCB Bioenergy*, vol. 4, no. 6, pp. 654–660.

Hinkley, J, Curtin, B, Hayward, J, Wonhaus, A, Boyd, R, Grima, C, Tadros, A, Hall, R, Naicker, K & Mikhail, A 2011, *Concentrating solar power - drivers and opportunities for cost-competitive electricity*, CSIRO & Aurecon Australia Pty Ltd, Sydney.

Ho, C, Mehos, M & Wagner, M 2014, 'Probabilistic Analysis of Power Tower Systems to Achieve SunShot Goals', *Energy Procedia*, vol. 49, PSE AG, Las Vegas, In Press, no. SolarPACES 2013 conference special, pp. 1410–1419.

Hobson, J & Winsbury, K 2013, 'AGL to proceed with Australia's largest solar projects', AGL, 31 July, Sydney, viewed 4 April 2014, http://www.agl.com.au/about-agl/how-we-source-energy/renewable-energy/nyngan-solar-plant/the-project.

Horn, M, Führing, H & Rheinländer, J 2004, 'Economic analysis of integrated solar combined cycle power plants: A sample case: The economic feasibility of an ISCCS power plant in Egypt', *Energy*, vol. 29, no. 5-6, pp. 935–945.

Hossain, J, Sinha, V & Kishore, VVN 2011, 'A GIS based assessment of potential for windfarms in India', *Renewable Energy*, vol. 36, no. 12, pp. 3257–3267.

Hu, E, Yang, Y, Nishimura, A, Yilmaz, F & Kouzani, A 2010, 'Solar thermal aided power generation', *Applied Energy*, vol. 87, no. 9, pp. 2881–2885.

Huang, Q, Chi, Y & Themelis, NJ 2013, 'A Rapidly Emerging WTE Technology: Circulating Fluid Bed Combustion', in *International Thermal Treatment Technologies*, San Antonio.

Hunt, G 2014, 'Climate Change Authority report', *Department of the Environment*, 27 February, Canberra, viewed 4 April 2014,

http://www.environment.gov.au/minister/hunt/2014/pubs/mr20140227.pdf>.

Hunt, G & Macfarlane, I 2014, 'Review of the Renewable Energy Target', *Department of the Environment and Department of Industry*, 17 February, Canberra, viewed 4 April 2014, http://www.environment.gov.au/minister/hunt/2014/pubs/mr20140217.pdf>.

Hyslop, P 2013, 'Tarong mothballing the right response to today's market', *The Courier-Mail*, 17 January, Brisbane.

International Energy Agency 2012a, CO2 Emissions from Fuel Combustion 2012 - Highlights, Paris.

- 2012b, World Energy Outlook 2012, Paris.

- 2013, Renewable Energy Medium-Term Market Report 2013 - Market Trends and Projections to 2018, Paris.

Ipswich City Council 2012, 'Ipswich supports pioneering eco-precinct concept', 7 August, Ipswich City Council, Ipswich, viewed 4 April 2014,

<http://www.ipswich.qld.gov.au/about_council/media/view_release/?id=3657&action=vie wMediaRelease>.

IRENA 2012, Renewable Energy Technologies: Cost analysis Series Volume 1 Issue 2/5 Concentrating Solar Power, International Renewable Energy Agency, Abu Dhabi.

Jackson, E 2012, Global Climate Leadership Review 2012, The Climate Institute, Sydney.

Jamel, MS, Abd Rahman, A & Shamsuddin, AH 2013, 'Advances in the integration of solar thermal energy with conventional and non-conventional power plants', *Renewable and Sustainable Energy Reviews*, vol. 20, pp. 71–81.

Janke, JR 2010, 'Multicriteria GIS modeling of wind and solar farms in Colorado', *Renewable Energy*, vol. 35, no. 10, pp. 2228–2234.

Jeafatura del Estado 2012, *Royal Decree 1/2012*, BOE núm. 24 de 28 de Enero de 2012, Gobierno de Espana, Madrid.

JFE Engineering 2011, Operational Results of Fukuyama Recycle Power, Yokohama.

- 2013, *APP-Albano Power Plant (Italy)*, viewed 4 April 2014, <http://www.jfeeng.co.jp/en/foreign/int07.html>.

Jonker, JGG & Faaij, APC 2013, 'Techno-economic assessment of micro-algae as feedstock for renewable bio-energy production', *Applied Energy*, vol. 102, pp. 461–475.

Kaeding, N 2010, 'Hybridkraftwerk Solar/Abfall: Erstellung und Beurteilung von Konzepten', *Lehrstuhl fuer Energiesysteme*, vol. Diplom Ing, Technische Universitaet Muenchen, Munich, Germany.

Kaltschmitt, M 1998, 'Nachwachsende Energieträger: Grundlagen, Verfahren, ökologische Bilanzierung (renewable energy sources: basics, technologies, ecologic accounting)', *Umweltwissenschaften und Schadstoff-Forschung*, vol. 10, Springer-Verlag, no. 3. Kane, M, Favrat, D, Ziegler, K & Allani, Y 2000, 'Thermoeconomic Analysis of Advanced Solar-Fossil Combined Power Plants', *International Journal of Thermodynamics*, vol. 3, no. 4, pp. 191–198.

Kaneff, S 1991, *The White Cliffs project: overview for the period 1979-89*, Office of Energy New South Wales, Sydney, Australia.

Kelly, B, Herrmann, U & Hale, MJ 2001, 'Optimization studies for integrated solar combined cycle systems', in *Solar Forum 2001*, Washington, DC.

Kelly, J 2012, 'Darkness falls on \$1.2bn SolarDawn project', *The Australian*, 13 November, Sydney, Australia, p. 4.

Kemp, R & Loorbach, D 2003, 'Governance for Sustainability Through Transition Management', in *EAEPE 2003 Conference*, Maastricht.

Kemp, R, Schot, J & Hoogma, R 1998, 'Regime shifts to sustainability through processes of niche formation: The approach of strategic niche management', *Technology Analysis & Strategic Management*, vol. 10, no. 2, pp. 175–198.

Kern, F & Smith, A 2008, 'Restructuring energy systems for sustainability? Energy transition policy in the Netherlands', *Energy Policy*, vol. 36, no. 11, pp. 4093–4103.

Kitney, D 2011, 'Anthony Pratt seeks \$300m from government to flick switch on waste-topower scheme', *The Australian*, 23 November, The Australian, Sydney.

Koll, G, Schwarzbozl, P, Hennecke, K, Hartz, T, Schmitz, M & Hoffschmidt, B 2009, 'The solar tower Jülich - a research and demonstration plant for central receiver systems', in *SolarPACES Conference*, Berlin, Germany.

Kruesi, M, Jovanovic, ZR, dos Santos, EC, Yoon, HC & Steinfeld, A 2013, 'Solar-driven steambased gasification of sugarcane bagasse in a combined drop-tube and fixed-bed reactor – Thermodynamic, kinetic, and experimental analyses', *Biomass and Bioenergy*, vol. 52, pp. 173–183.

Kruesi, M, Jovanovic, ZR & Steinfeld, A 2014, 'A two-zone solar-driven gasifier concept: Reactor design and experimental evaluation with bagasse particles', *Fuel*, vol. 117, pp. 680– 687.

Krüger, D, Krüger, J, Pandian, Y, O'Connell, B, Feldhoff, JF, Karthikeyan, R, Hempel, S, Muniasamy, K, Hirsch, T, Eickhoff, M & Hennecke, K 2012, 'Experience with Direct Steam Generation at the Kanchanaburi Solar Thermal Power Plant', in *SolarPACES Conference*, Marrakech.

Kuravi, S, Trahan, J, Goswami, DY, Rahman, MM & Stefanakos, EK 2013, 'Thermal energy storage technologies and systems for concentrating solar power plants', *Progress in Energy and Combustion Science*, vol. 39, no. 4, pp. 285–319.

Kuyumcu, OC, Solarglu, UZD, Akar, S & Serin, O 2013, 'Hybrid Geothermal and Solar Thermal Power Plant Case Study: Gumuskoy GEPP', *GHC Bulletin*, Klamath Falls, Oregon.

Lachman, D a. 2014, 'A combination of existing concepts and approaches to take on energy system transitions – The Republic of Panama as a case-study', *Sustainable Energy Technologies and Assessments*, vol. 5, pp. 84–94.

Lahoda, J, Arndt, O & Hanstein, W 2006, 'Biomass lookig for efficient utilisation - the reheat concept', in *Power Gen Europe*, Cologne, Germany.

Laing, D, Eck, M, Hempel, M, Johnson, M, Steinmann, W-D, Meyer-Gruenfeld, M & Eickhoff, M 2012, 'High Temperature PCM Storage for DSG Solar Thermal Power Plants Tested in Various Operating Modes of Water/Steam Flow', in *Solar PACES Conference*, Marrakech.

Lamers, F, Fleck, E, Pelloni, L & Kamuk, B 2013, *Alternative Waste Conversion Technologies - White paper*, Prepared for ISWA by ISWA Working Group on Energy Recovery, Vienna.

Larson, RW & West, RE 1996, *Implementation of Solar Thermal Technology*, MIT Press, Boston, p. 995.

Lempp, P 2013, Biomass co-firing - Technology brief, IEA-ETSAP and IRENA, Abu Dhabi.

Leviston, Z, Price, J, Malkin, S & McCrea, R 2014, *Fourth annual survey of Australian attitudes to climate change: Interim report*, Perth, Australia.

Li, H, Yan, J & Campana, PE 2012, 'Feasibility of integrating solar energy into a power plant with amine-based chemical absorption for CO2 capture', *International Journal of Greenhouse Gas Control*, vol. 9, pp. 272–280.

Lichty, P, Perkins, C, Woodruff, B, Bingham, C & Weimer, A 2010, 'Rapid High Temperature Solar Thermal Biomass Gasification in a Prototype Cavity Reactor', *Journal of Solar Energy Engineering*, vol. 132, no. 1, p. 011012.

Livshits, M & Kribus, A 2012, 'Solar hybrid steam injection gas turbine (STIG) cycle', *Solar Energy*, vol. 86, no. 1, pp. 190–199.

Lloyd Energy Systems 2011, Lake Cargelligo solar thermal project, Advanced Electricity Storage Technologies Program - Final public report, Report prepared for the Department of Resources, Energy and Tourism.

Lluna, CH 2011, *Solar thermal electricity in China in 2011 and future outlook*, Prepared for the European Solar Thermal Electricity Association, Brussels.

Lokuge, C & Loftus, G 2010, *Pre-Feasibility Assessment of a Thermal Conversion Facility for the Australian Capital Territory*, Department of the Environment, Chlimate Change, Energy and Water, Sydney.

Loorbach, D 2007, 'Transition Management - New mode of governance for sustainable development', Erasmus University Rotterdam.

- 2010, 'Transition Management for Sustainable Development: A Prescriptive, Complexity-Based Governance Framework', *Governance*, vol. 23, no. 1, pp. 161–183.

Lovegrove, K, Burgess, G & Pye, J 2011, 'A new 500 m2 paraboloidal dish solar concentrator', *Solar Energy*, vol. 85, no. 4, pp. 620–626.

Lovegrove, K & Luzzi, A 2002, *Solar Thermal Power Systems*, 3rd edn, vol. 15, Encyclopedia of Physical Science and Technology, Academic Press, San Diego.

Lovegrove, K, Watt, M, Passey, R, Pollock, G, Wyder, J & Dowse, J 2012, *Realising the potential of Concentrating Solar Power in Australia*, Prepared by IT Power for the Australian Solar Institute, Canberra, Australia.

Lucas, H, Ferroukhi, R, Wichmann, J & Ghazal-Aswad, N 2011, *Renewable Energy Jobs: Status, Prospects & Policies*, International Renewable Energy Agency, Abu Dhabi.

Macquarie Generation 2012, Macquarie Generation Annual Report, Newcastle, Australia.

Macromonitor 2013, Australian construction outlook - forecasting service utilities, Sydney.

Manning, P 2011, 'Melbourne to get first biochar plant', *The Age*, 6 September, Melbourne.

Markard, J, Raven, R & Truffer, B 2012, 'Sustainability transitions: An emerging field of research and its prospects', *Research Policy*, vol. 41, no. 6, pp. 955–967.

Martin, NJ & Rice, JL 2012, 'Developing renewable energy supply in Queensland, Australia: A study of the barriers, targets, policies and actions', *Renewable Energy*, vol. 44, pp. 119–127.

McDonald, CF 1986, 'A hybrid solar closed-cycle gas turbine combined heat and power plant concept to meet the continuous total energy needs of a small community', *Journal of Heat Recovery Systems*, vol. 6, no. 5, pp. 399–419.

McEvilly, G, Abeysuriya, S & Dix, S 2011, *Facilitating the Adoption of Biomass Co-firing for Power Generation*, Rural Industries Research and Development Corporation, Canberra.

McGregor, J, Dave, N, Hinkley, J, Do, T & Khare, S 2013, 'Techno-economic assessment of solar assisted post combustion capture of carbon dioxide', in *3rd International Conference on Energy Process Engineering – Transition to Renewable Energy Systems*, Frankfurt am Main.

Mckenzie, MM & Howes, M 2006, 'Remote Renewable Energy in Australia : Barriers to Uptake and the Community Engagement Imperative', in *Australasian Political Studies Association conference*, Newcastle.

Meehan, R 2013, *Hybridisation of Fossil Fuel Energy Generation in Australia - Public report*, Prepared by Parsons Brinckerhoff for the Australian Renewable Energy Agency (ARENA), Melbourne.

Meinecke, W, Bohn, M & Becker, M 1995, *Solar Energy Concentrating Systems: Applications and Technologies*, Müller Verlag, Heidelberg, Germany.

Metso Corporation 2011, 'World's largest biomass gasification plant inaugurated', 11 March, viewed 4 April 2014,

<http://www.metso.com/news/newsdocuments.nsf/web3newsdoc/4950FCC1037AC5A2C2 257B2B003E89E8?OpenDocument&ch=ChMetsoPaperWebEng&#.Uz3vM6iSznE>.

Michael, P 2010, 'Queensland solar power project in Cloncurry stalls because of bright light of panels', *The Courier-Mail*, 12 August, Brisbane.

Mills, DR, Lièvre, P & Morrison, G 2003, 'Solar Preheating of the Liddell Coal-fired Powerplant', in *Destination Renewables- ANZSES*, Melbourne, pp. 600–4.

Ministerio de Economia 2004, *Royal Decree 436/2004*, BOE núm. 75 de 27 de Marzo de 2004, Gobierno de Espana, Madrid.

Ministry of New and Renewable Energy 2011, *Jawaharlal Nehru National Solar Mission*, Government of India, New Delhi.

Mokhtar, M, Ali, MT, Khalilpour, R, Abbas, A, Shah, N, Hajaj, A Al, Armstrong, P, Chiesa, M & Sgouridis, S 2012, 'Solar-assisted Post-combustion Carbon Capture feasibility study', *Applied Energy*, vol. 92, pp. 668–676.

Montes, MJ, Rovira, A, Muñoz, M & Martínez-Val, JM 2011, 'Performance analysis of an Integrated Solar Combined Cycle using Direct Steam Generation in parabolic trough collectors', *Applied Energy*, vol. 88, no. 9, pp. 3228–3238.

Moomaw, W, Yamba, F, Kamimoto, M, Maurice, L, Nyboer, J, Urama, K & Weir, T 2011, Introduction. In IPCC Special Report on Renewable Energy Sources and Climate Change Mitigation, Cambridge, United Kingdom and New York, USA. Morell, D 2012, 'CSP BORGES The World's First CSP plant hybridized with biomass', in CSP Today USA 2012 - 6th Concentrated Solar Thermal Power Conference, CSP Today, Las Vegas.

Morin, G, Mertins, M, Kirchberger, J & Selig, M 2011, 'SUPERNOVA – Construction, control & performance of steam superheating Linear Fresnel collector', in *SolarPACES Conference 2011*, Granada Spain.

Morton, A 2011, 'Billions spent on fossil fuel incentives', *Sydney Morning Herald*, 1 March, Sydney.

Müller-Steinhagen, H & Trieb, F 2004, 'Concentrating solar power: A review of the technology', *Ingenia*, The Royal Academy of Engineering, London, UK, no. 18, pp. 43–50.

National Energy Technology Laboratory 2013, *PROJECT NUMBER: DE-SC00008355 - Hybrid Solar Coal Gasifier*, viewed 4 April 2014, http://www.netl.doe.gov/research/coal/energy-systems/gasification/gas-optimization/project-number-de-sc00008355.

Nelson, G & Larsen, G 2013, 'Concentrated Solar and Geothermal Hybrid Power Project', *GHC Bulletin*, Klamath Falls, Oregon.

New York Times 1916, 'American Inventor Uses Egypt's Sun for Power', 2 July, New York.

Nezammahalleh, H, Farhadi, F & Tanhaemami, M 2010, 'Conceptual design and technoeconomic assessment of integrated solar combined cycle system with DSG technology', *Solar Energy*, vol. 84, no. 9, pp. 1696–1705.

Ng, YC & Lipiński, W 2012, 'Thermodynamic analyses of solar thermal gasification of coal for hybrid solar-fossil power and fuel production', *Energy*, vol. 44, no. 1, pp. 720–731.

Nithyanandam, K, Pitchumani, R & Mathur, A 2014, 'Analysis of a latent thermocline storage system with encapsulated phase change materials for concentrating solar power', *Applied Energy*, vol. 113, pp. 1446–1460.

Nixon, JD, Dey, PK & Davies, P a. 2012, 'The feasibility of hybrid solar-biomass power plants in India', *Energy*, vol. 46, no. 1, pp. 541–554.

Nixon, JD, Dey, PK & Davies, PA 2010, 'Which is the best solar thermal collection technology for electricity generation in north-west India? Evaluation of options using the analytical hierarchy process', *Energy*, vol. 35, no. 12, pp. 5230–5240.

Nordås, R & Gleditsch, NP 2007, 'Climate change and conflict', *Political Geography*, vol. 26, no. 6, pp. 627–638, viewed 11 July 2014, http://linkinghub.elsevier.com/retrieve/pii/S0962629807000856>.

Novatec Solar GmbH 2011, Personal communication between Juergen H. Peterseim and Novatec on 20 May 2011 regarding water consumption of Fresnel plants, Karlsruhe, Germany.

— 2012a, Liddell in Australia, Karlsruhe, Germany, viewed 4 April 2014, <http://www.novatecsolar.com/80-1-Liddell.html>.

2012b, 'Novatec Solar's Australian Fuel-Saver Commences Operation', 24 October,
Karlsruhe, viewed 4 April 2014, http://www.novatecsolar.com/104-1-Liddell-eng.html.

— 2012c, Puerto Errado 2 in Spain, Liestal, Switzerland, viewed 4 April 2014, <http://www.novatecsolar.com/56-1-PE-2.html>.

— 2013, 'Novatec Solar starts construction of DMS demo-collector', 30 October, Karlsruhe, viewed 4 April 2014, http://www.novatecsolar.com/121-1-DMS-Demo-E.html.

Noyes, GW 1990, 'The Effect of Component Efficiency and Operating Conditions on the 50kW Dish Stirling System in Riyadh, Saudia Arabia', *Journal of Solar Energy Engineering*, vol. 112, no. 4, pp. 244–246.

NREL 2014a, *Ivanpah Solar Electric Generating Station*, viewed 4 April 2014, <www.nrel.gov/csp/solarpaces/project_detail.cfm/projectID=62>.

— 2014b, Kimberlina Solar Thermal Power Plant, viewed 4 April 2014, <http://www.nrel.gov/csp/solarpaces/project_detail.cfm/projectID=37>.

— 2014c, Kogan Creek Solar Boost, viewed 4 April 2014, <http://www.nrel.gov/csp/solarpaces/project_detail.cfm/projectID=243>.

— 2014d, Solana Generating Station, viewed 4 April 2014, <http://www.nrel.gov/csp/solarpaces/project_detail.cfm/projectID=23>.

O'Connell, D, Braid, A, Raison, J, Handberg, K, Cowie, A, Rodriguez, L & George, B 2009, Sustainable production of bioenergy: a review of global bioenergy sustainability frameworks and assessment systems., Rural Industries Research and Development Corporation, Canberra, Australia.

ORMAT, CIEMAT, DLR, SOLUCAR, TUMA, Communities, O for OP of the E & Energy, E and SD 2005, *SOLGATE - Solar hybrid gas turbine electric power system*, European Commission, Brussels, no. Contract ENK5-CT-2000-00333.

Ottosson, S 2003, 'Participation action research-: A key to improved knowledge of management', *Technovation*, vol. 23, no. 2, pp. 87–94.

Ozturk, M & Dincer, I 2013, 'Thermodynamic assessment of an integrated solar power tower and coal gasification system for multi-generation purposes', *Energy Conversion and Management*, vol. 76, pp. 1061–1072.

Pacific Pyrolysis 2010, *Demonstration Facility*, viewed 4 April 2014, http://pacificpyrolysis.com/technology.html#demonstration>.

Painuly, J. 2001, 'Barriers to renewable energy penetration; a framework for analysis', *Renewable Energy*, vol. 24, no. 1, pp. 73–89.

Papachristos, G, Sofianos, A & Adamides, E 2013, 'System interactions in socio-technical transitions: Extending the multi-level perspective', *Environmental Innovation and Societal Transitions*, vol. 7, pp. 53–69.

Parsons Brinckerhoff Australia Pty Ltd 2010, *Queensland Concentrated Solar Power, Pre-feasibility Report December 2009*, Queensland Government, Office of Clean Energy, Department of Employment, Economic Development and Industry, Brisbane.

Patel, SJ, DeBarbadillo, JJ, Baker, B a. & Gollihue, RD 2013, 'Nickel Base Superalloys for Next Generation Coal Fired AUSC Power Plants', *Procedia Engineering*, vol. 55, pp. 246–252.

Peatling, S 2013, 'Kevin Rudd confirms government to scrap fixed carbon price', *The Sydney Morning Herald*, 14 July, Sydney.

Pérez, Á & Torres, N 2010, 'Solar Parabolic Trough – Biomass Hybrid Plants: A Cost – Efficient Concept Suitable For Places In Low Irradiation Conditions', in *SolarPaces Conference*, Perpignan, France.

Peterseim, JH, Hellwig, U, Tadros, A & White, S 2014a, 'Hybridisation optimization of concentrating solar thermal and biomass power generation facilities', *Solar Energy*, vol. 99, pp. 203–214.

Peterseim, JH, Herr, A, Miller, S, White, S & O'Connell, DA 2014b, 'Concentrating solar power / alternative fuel hybrid plants : Annual electricity potential and ideal areas in Australia', *Energy*, vol. 68, pp. 698–711.

Peterseim, JH, Tadros, A, Hellwig, U & White, S 2013a, 'Integrated solar combined cycle plants using solar towers with thermal storage to increase plant performance', in *ASME Power Conference*, Boston.

Peterseim, JH, Tadros, A, White, S, Hellwig, U & Klostermann, F 2012a, 'Concentrated solar power / Energy from Waste hybrid plants - creating synergies', in *SolarPACES Conference*, Marrakech.

Peterseim, JH, Tadros, A, White, S, Hellwig, U, Landler, J & Galang, K 2014c, 'Solar towerbiomass hybrid plants – maximizing plant performance', *Energy Procedia*, vol. 49, no. SolarPACES 2013 conference special, pp. 1197–1206.

Peterseim, JH, White, S, Hellwig, U, Tadros, A & Vanz, E 2012b, *Pre-feasibility study for a multi-fuel / concentrated solar power hybrid plant at Swanbank, QLD*, Prepared for Thiess Services Pty Ltd, Sydney, Unpublished report.

Peterseim, JH, White, S, Tadros, A & Hellwig, U 2012c, 'Integrated Solar Combined Cycle plants using solar power towers to optimise plant performance', in *SolarPACES Conference*, Marrakech.

- 2013b, 'Concentrated solar power hybrid plants, which technologies are best suited for hybridisation?', *Renewable Energy*, vol. 57, pp. 520–532.

 2014d, 'Concentrating solar power hybrid plants - enabling cost effective synergies', Renewable Energy, vol. 67, pp. 178–185.

Piatkowski, N, Wieckert, C, Weimer, AW & Steinfeld, A 2011, 'Solar-driven gasification of carbonaceous feedstock—a review', *Energy & Environmental Science*, vol. 4, no. 1, p. 73.

Pierce, W, Gauché, P, von Backström, T, Brent, AC & Tadros, A 2013, 'A comparison of solar aided power generation (SAPG) and stand-alone concentrating solar power (CSP): A South African case study', *Applied Thermal Engineering*, vol. 61, no. 2, pp. 657–662.

PIRSA Spatial Information Services 2009, *Concentrating solar resource based on 1998-2007 average direct beam hourly irradiance data*, Government of South Australia, Primary Industries and Resources South Australia, Adelaide.

pitt&sherry 2014, carbon emissions index cedex[®] - Electricity emissions update data to 31 December 2013, Canberra.

Popov, D 2011, 'An option for solar thermal repowering of fossil fuel fired power plants', *Solar Energy*, vol. 85, no. 2, pp. 344–349.

— 2014, 'Innovative Solar Augmentation of Gas Turbine Combined Cycle Plants', *Applied Thermal Engineering*, vol. 64, no. 1-2, pp. 40–50.

Protermosolar 2012, 'Abantia and Comsa Emte set up the first facility in the world that combines solar thermal and biomass energy', 13 December, viewed 4 April 2014, http://www.protermosolar.com/saladeprensa/?q=en/content/abantia-and-comsa-emte-set-first-facility-world-combines-solar-thermal-and-biomass-energy.

Psomopoulos, CS, Bourka, A & Themelis, NJ 2009, 'Waste-to-energy: A review of the status and benefits in USA.', *Waste Management*, vol. 29, no. 5, pp. 1718–24.

Qadir, A, Mokhtar, M, Khalilpour, R, Milani, D, Vassallo, A, Chiesa, M & Abbas, A 2013, 'Potential for solar-assisted post-combustion carbon capture in Australia', *Applied Energy*, vol. 111, no. July 2012, pp. 175–185.

Queensland Competition Authority 2013, *Estimating a Fair and Reasonable Solar Feed-in Tariff for Queensland*, Brisbane.

Queensland Government 2011, *Opening the Green Door for sustainable development projects*, Brisbane.

Quero, M, Korzynietz, R, Ebert, M, Jiménez, AA, Río, A del & Brioso, JA 2013, 'Solugas – Operation Experience of the First Solar Hybrid Gas Turbine System at MW Scale', in *SolarPACES Conference*, Las Vegas, In Press.

Ten Raa, T 2006, *The Economics of Input-Output Analysis*, Cambridge University Press, Cambridge.

Ragheb, M 2011, *Solar Thermal Power and Energy Storage Historical Perspective*, Global Solar Thermal Energy Council, University of Illinois at Urbana-Champaign, Urbana, IL.

Rashid, N, Cui, Y-F, Saif Ur Rehman, M & Han, J-I 2013, 'Enhanced electricity generation by using algae biomass and activated sludge in microbial fuel cell.', *The Science of the total environment*, vol. 456-457, pp. 91–4.

Rawlins, J & Ashcroft, M 2013, *Small-scale Concentrated Solar Power - A review of current activity and potential to accelerate deployment*, The Carbon Trust, London.

Reece, A 2013, 'New Earth ACT plant begins operations', *Resource Magazine*, 19 June, Bristol, viewed 4 April 2014,

<http://www.resource.uk.com/article/News/New_Earth_ACT_plant_begins_operations-3230#.UyvVOKiSznE>.

Regional Development Australia 2011, *Whyalla Solar Thermal Concentration Plant*, Port Lincoln, Australia, viewed 4 April 2014, <www.eyreregion.com.au/doing-business/major-projects/whyalla-solar-thermal-concentration-plant>.

Reisinger, A, Kitching, R, Chiew, F, Newton, P, Schuster, S, Tait, A & Whetton, P 2014, *Climate Change 2014: Impacts, Adaptation, and Vulnerability, Chapter 25. Australasia*, Paris.

RepuTex 2013, *Coal generation: Moving out of the crosshairs, Outlook for Fuel mix across the NEM FY13-20*, Melbourne.

Resurreccion, EP, Colosi, LM, White, M a & Clarens, AF 2012, 'Comparison of algae cultivation methods for bioenergy production using a combined life cycle assessment and life cycle costing approach.', *Bioresource technology*, vol. 126, pp. 298–306.

Reuveny, R 2007, 'Climate change-induced migration and violent conflict', *Political Geography*, vol. 26, no. 6, pp. 656–673, viewed 10 July 2014, http://linkinghub.elsevier.com/retrieve/pii/S0962629807000601.

Riedy, C 2007, *Energy and Transport Subsidies in Australia*, Prepared for for Greenpeace Australia Pacific by the Institute for Sustainable Futures, Sydney.

Rip, A & Kemp, RPM 1998, 'Technological Change', in *Human Choice and Climate Change*, Chapter 6, Batelle Press, Washington, DC, pp. 327–399.

ROAM Consulting Pty Ltd 2011, Additional Projections of Electricity Generation in Australia to 2050, Brisbane.

Rojas, M, Hellwig, U, Peterseim, JH & Harding, J 2010, 'Combining solar thermal power systems with solid fuels in new plants', in *SolarPACES Conference*, Perpignan, France.

Romero, M, Marcos, MJ, Osuna, R, Fernandez, V, Mendoza, S & Esteban, L 2002, 'SOLBIO: Solar-Biomass Hybrid Tower Power Plant', in *7th World Renewable Energy Congress VII*, Cologne, Germany.

Rotmans, J, Kemp, R & van Asselt, M 2001, 'More evolution than revolution: transition management in public policy', *Foresight*, vol. 3, no. 1, pp. 15–31.

Rotmans, J & Loorbach, D 2008, 'Transition management: reflexive governance of societal complexity through searching, learning and experimenting', in *Managing the Transition to Renewable Energy - Theory and Practice from Local, Regional and Macro Perspective*, Edward Elga Publishing Limited, pp. 15–46.

Rotter, VS, Lehmann, A, Marzi, T, Mohle, E, Schingnitz, D & Hoffmann, G 2011, 'New techniques for the characterization of refuse-derived fuels and solid recovered fuels', *Waste Management & Research*, vol. 29, no. 2, pp. 229–236.

Rutovitz, J, Peterseim, J, Elliston, B, Harris, S, Mohr, S, Lovegrove, K, Want, A, Langham, E & MacGill, I 2013, *Breaking the solar gridlock. Potential benefits of installing concentrating solar thermal power at constrained locations in the NEM*, Prepared for the Australian Solar Thermal Energy Association (AUSTELA) by the Institute for Sustainable Futures, UTS, Sydney.

Saade, E, Bingham, C, Clough, DE & Weimer, AW 2012, 'Dynamics of a solar-thermal transport-tube reactor', *Chemical Engineering Journal*, vol. 213, pp. 272–285.

Saaty, TL & Ozdemir, MS 2003, 'Why the magic number seven plus or minus two', *Mathematical and Computer Modelling*, vol. 38, no. 3-4, pp. 233–244.

Sallis, JF, Frank, LD, Saelens, BE & Kraft, MK 2004, 'Active transportation and physical activity: opportunities for collaboration on transportation and public health research', *Transportation Research Part A: Policy and Practice*, vol. 38, no. 4, pp. 249–268.

Sathaye, J, Lucon, O, Rahman, A, Christensen, J, Denton, F, Fujino, J, Heath, G, Mirza, M, Rudnick, H, Schlaepfer, A & Shmakin, A 2011, *Renewable Energy in the Context of Sustainable Development. In IPCC Special Report on Renewable Energy Sources and Climate Change Mitigation*, in O Edenhofer, R Pichs-Madruga, Y Sokona, K Seyboth, P Matschoss, S Kadner, T Zwickel, P Eickemeier, G Hansen, S Schlömer & C von Stechow (eds), *IPCC Special Report on Renewable Energy Sources and Climate change Mitigation*, Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

Saunders, M 2012, 'Ballina Council to build a biochar plant', *ABC Rural*, 7 June, Melbourne, viewed 4 April 2014, http://www.abc.net.au/site-archive/rural/news/content/201206/s3520032.htm.

SBC Energy Institute 2013, Concentrating Solar Power, Gravenhage, Netherlands.

Schläpfer, A 2009, 'Hidden biases in Australian energy policy', *Renewable Energy*, vol. 34, no. 2, pp. 456–460.

Schnatbaum, L 2009, 'Biomass Utilization for Co Firing in Parabolic Trough Power Plants', in *SolarPACES Conference*, Berlin, Germany.

Schot, J & Geels, FW 2008, 'Strategic niche management and sustainable innovation journeys: theory, findings, research agenda, and policy', *Technology Analysis & Strategic Management*, vol. 20, no. 5, pp. 537–554.

Sempra U.S. Gas & Power 2012, *Mesquite Solar 1*, viewed 4 April 2014, http://www.semprausgp.com/_/downloads/pdfs/FactSht_MesquiteSolar1.pdf>.

SENER 2014, Gemasolar plant in Seville, Spain, viewed 4 April 2014, <http://www.sener-power-

process.com/ENERGIA/ProjectsI/gemasolar/en?pag=0&agno=&area=&categoria=ENERGIA_ SOLAR&totalperpage=5>.

Service, RF 2009, 'Sunlight in Your Tank', Science, vol. 326, pp. 1472–75.

Sharp, A, Gauchat, A, Li, C, Nielsen, E, Doran, J, Fairman, J, Rogowsky, M, Cumming, S, Vance, S, O'Reilly, S, Frith, S & Endelman, T 2004, *Histories of Green Square*, School of History, The University of New South Wales, Sydney.

Silex Systems Ltd 2013, 'Australia's Largest Concentrated Photovoltaic (CPV) Solar Plant Opens, Solar Systems to officially open its 1.5MW facility in Mildura, Victoria', 17 July, viewed 4 April 2013, http://www.silex.com.au/downloads/asxannouncements/1237713>.

Siros, F, Philibert, C, Moullec, Y Le, Tusseau, M & Bonnelle, D 2012, 'The Value of Hybridizing CSP', in *SolarPACES Conference*, Marrakech.

Siva Reddy, V, Kaushik, SC & Tyagi, SK 2012, 'Exergetic analysis of solar concentrator aided natural gas fired combined cycle power plant', *Renewable Energy*, vol. 39, no. 1, pp. 114–125.

Skumanich, A 2011, 'CSP at a crossroads: The first solar electric power plants are still proving their worth after three decades, so why aren't we seeing more CSP reach the development stage?', *Renewable Energy Focus*, vol. 12, no. 1, pp. 52–55.

SkyFuel 2012, *SkyFuel's Parabolic Trough to Save Fuel in The Gas City*, SkyFuel, viewed 4 April 2014,

<http://www.skyfuel.com/downloads/press_releases/10412PressReleaseMedicineHat(AMa son).pdf>.

Smith, A, Stirling, A & Berkhout, F 2005, 'The governance of sustainable socio-technical transitions', *Research Policy*, vol. 34, no. 10, pp. 1491–1510.

Sniezek, JA & Henry, RA 1989, 'Accuracy and confidence in group judgment', *Organizational Behavior and Human Decision Processes*, vol. 43, no. 1, pp. 1–28.

Sohi, S, Lopez-Capel, E, Krull, E & Bol, R 2009, *Biochar, climate change and soil: A review to guide future research*, CSIRO Land and Water Science Report 05/09, Glen Osmond, SA.

Solar Millennium AG 2006, 'Financing granted to Solar Millennium AG for Andasol 2, the second parabolic trough power plant in Europe', 28 December, viewed 4 April 2014, <www.solarmillennium.de/Archiv/Presse/Meldungen/Archiv_2006/Meldungen_2006_,lang 2,51,235.html>.

Solarlite CSP Technology GmbH 2014, *TRESERT Phitanulok, Thailand - TriGeneration* (*electricity, heat, refrigeration*), Marrakech, viewed 4 April 2014, http://solarlite-csp.com/en/reference/tse-2-suphanburi-provinz-thailand/>.

SolarReserve LLC 2012, *Crescent Dunes*, Santa Monica, viewed 4 April 2014, <www.solarreserve.com/what-we-do/csp-projects/crescent-dunes>.

Solúcar 2006, 10 MW Solar Thermal Power Plant for Southern Spain, European Commission.

Spectrum Magazine 2009, 'First-of-its-kind at Skive', vol. 1, Graz, no. 20.

Spliethoff, H 2010, *Power generation from solid fuels*, *Power Systems*, 1st edn, Springer, Heidelberg.

Spliethoff, H, Kaeding, N, Murer, M, Alonso-Herranz, E & Gohlke, O 2010, 'Combining Energy from Waste and Concentrated Solar Power: New Solutions for Sustainable Energy Generation', in *International Solid Waste Association Conference 2010*, Hamburg.

Stein, W, Edwards, J, Hinkley, J & Sattler, C 2009, 'Natural Gas: Solar-Thermal Steam Reforming', in *Encyclopedia of Electrochemical Power Sources*, Elsevier, Amsterdam, pp. 300–312.

Stevenson, W 2007, 'Emissions from Large and Small MWC Units at MACT Compliance', *Unites States Environmental Protection Agency*, Research Triangle Park NC.

Stinnesbeck, L 1914, 'Sonnenkraftmaschinen (solar power machines)', *Keller's Monatsblätter*, vol. 3, No.1, Bergstadtverlag, Breslau, pp. 171–174.

Stolpe, GW 1986, 'Solar plant 1: The formula for success', *Alternative Sources Energy* (United States).

Stucley, C, Schuck, S, Sims, R, Bland, J, Marino, B, Borowitzka, M, Abadi, A, Bartle, J, Giles, R & Thomas, Q 2012, *Bioenergy in Australia*, Bioenergy Australia, Sydney.

Sumio, Y, Masuto, S & Fumihiro, M 2004, *Thermoselect Waste Gasification and Reforming Process*, JFE, Technical report No.3.

SunShot Initiative 2012, SunShot Vision Study, U.S. Department of Energy, Washington, DC.

Suresh, MVJJ, Reddy, KS & Kolar, AK 2010, '4-E (Energy, Exergy, Environment, and Economic) analysis of solar thermal aided coal-fired power plants', *Energy for Sustainable Development*, vol. 14, no. 4, pp. 267–279.

Syed, A 2012, *Australian Energy Projections to 2049–50*, Bureau of Resources and Energy Economics, Canberra.

Tadros, A, Boyd, R, Benyon, P & Grima, C 2009, 'Options for Biomass Utilization in Coal Fired Power Stations', *Journal of Thermal Science and Engineering Applications*, vol. 1, no. 4, pp. 041008–1.

Teschner, N, McDonald, A, Foxon, TJ & Paavola, J 2012, 'Integrated transitions toward sustainability: The case of water and energy policies in Israel', *Technological Forecasting and Social Change*, vol. 79, no. 3, pp. 457–468.

The Federal Ministry for the Environment Nature Conservation and Nuclear Safety 2012, Act on granting priority to renewable energy sources (Renewable Energy Sources Act – EEG), Berlin, Germany.

The Parliament of the Commonwealth of Australia 2013a, *Clean Energy Finance Corporation* (Abolition) Bill 2013, Canberra.

- 2013b, Clean Energy Legislation (Carbon Tax Repeal) Bill 2013, Canberra.

Themelis, N 2006, 'Energy recovery from Global Waste-to-Energy', *Summer review issue of Waste Management World*.

Thornley, P, Rogers, J & Huang, Y 2008, 'Quantification of employment from biomass power plants', *Renewable Energy*, vol. 33, no. 8, pp. 1922–1927.

Toebes, J & Beker, M 1998, 'External steam input and process steam output at the Moerdijk (Netherlands) combined-cycle plant', *VGB POWERTECH*, Essen, Germany, no. 7, pp. 21–27.

Triantaphyllou, E 2000, *Multi-Criteria Decision Making Methods: A Comparative Study*, Kluwer Academic Publishers, Dordrecht, The Netherlands.

Trieb, F, Schillings, C, O'Sullivan, M, Pregger, T & Hoyer-Klick, C 2009, 'Global Potential of Concentrating Solar Power', in *SolarPACES Conference*, Berlin, Germany.

Turchi, CS & Erbes, M 2011, 'Gas Turbine / Solar Parabolic Trough Hybrid Designs', in *ASME TurboExpo 2011*, ASME, Vancouver, Canada.

Turchi, CS, Langle, N, Bedilion, R & Libby, C 2011, *Solar-Augment Potential of U.S. Fossil-Fired Power Plants*, National Renewable Energy Laboratory, Golden, Colorado.

Turchi, CS & Ma, Z 2014, 'Co-located gas turbine/solar thermal hybrid designs for power production', *Renewable Energy*, vol. 64, pp. 172–179.

Turchi, CS, Wagner, MJ, Kutscher, CF & Laboratory, NRE 2010, *Water Use in Parabolic Trough Power Plants: Summary Results from WorleyParsons' Analyses*, Golden, Colorado.

U.S. Department of Energy 2013a, *Loan programs office*, viewed 4 April 2014, <https://lpo.energy.gov/>.

— 2013b, 'Sacramento Utility to Launch Concentrating Solar Power-Natural Gas Project', 31
October, viewed 4 April 2014, http://energy.gov/articles/sacramento-utility-launch-concentrating-solar-power-natural-gas-project.

Ugolini, D, Zachary, J & Park, HJ 2009, 'Fossil Fuels + Solar Energy = The Future of Electricity Generation', *POWER Magazine*.

Vehlow, J 2004, *Biogenic Waste to Energy – an Overview*, Forschungszentrum Karlsruhe, IEA Bioenergy Task 36, Karlsruhe, Germany.

Verbong, G & Geels, F 2007, 'The ongoing energy transition: Lessons from a socio-technical, multi-level analysis of the Dutch electricity system (1960–2004)', *Energy Policy*, vol. 35, no. 2, pp. 1025–1037.

Verbruggen, A, Fischedick, M, Moomaw, W, Weir, T, Nadaï, A, Nilsson, LJ, Nyboer, J & Sathaye, J 2010, 'Renewable energy costs, potentials, barriers: Conceptual issues', *Energy Policy*, vol. 38, no. 2, pp. 850–861.

Vergragt, PJ, Markusson, N & Karlsson, H 2011, 'Carbon capture and storage, bio-energy with carbon capture and storage, and the escape from the fossil-fuel lock-in', *Global Environmental Change*, vol. 21, no. 2, pp. 282–292.

Verve Energy 2013, *Greenough River Solar Farm*, viewed 4 April 2014, http://generation.synergy.net.au/generating-electricity/sustainable-portfolio/greenough-river-solar-farm.

Viana, H, Cohen, WB, Lopes, D & Aranha, J 2010, 'Assessment of forest biomass for use as energy. GIS-based analysis of geographical availability and locations of wood-fired power plants in Portugal', *Applied Energy*, vol. 87, no. 8, pp. 2551–2560.

Vicente, CJ, Tejerina, P, Massabé, F, Marco, M, Candini, L, Villa, JR & 6 2012, 'PARIS – A Parabolic Trough Autonomous Robotic Cleaning System', in *SolarPACES Conference*, Marrakech.

Visy Industries Australia 2012, *Clean Energy*, Melbourne, viewed 4 April 2014, <www.visy.com.au/sustainability/clean-energy>.

Warren, K, Gandy, S, Davis, G, Read, A, Fitzgerald, J & Holdaway, E 2013, *Waste to energy background paper*, Prepared by Ricardo-AEA for for Zero Waste SA, Harwell, Didcot.

Waste Management World 2012, '12 MW plasma gasification facility completed in France', 2 July, Tulsa, Oklahoma, viewed 4 April 2014, http://www.waste-management-world.com/articles/2012/07/12-mw-plasma-gasification-facility-completed-in-france.html>.
Watson, A & Peterseim, JH 2010, 'Energy from Process Engineered Fuel (PEF)', *Energy Generation*, pp. 44–45.

Wei, M, Patadia, S & Kammen, DM 2010, 'Putting renewables and energy efficiency to work: How many jobs can the clean energy industry generate in the US?', *Energy Policy*, vol. 38, no. 2, pp. 919–931.

Wilkins, G 2002, *Technology Transfer for Renewable Energy: Overcoming Barriers in Developing Countries*, Earthscan Publications Ltd, London.

Willis, B 2014, 'India seeks World Bank loan to kick-start 4GW "ultra mega" PV project', *PV Magazine*, 10 February, viewed 4 April 2014, <http://www.pv-tech.org/news/india_seeks_world_bank_loan_to_kick_start_4gw_ultra_mega_pv_project>.

Woodruff, BR & Weimer, AW 2013, 'A novel technique for measuring the kinetics of high-temperature gasification of biomass char with steam', *Fuel*, vol. 103, pp. 749–757.

Woody, T 2011, 'Solar Energy Faces Tests On Greenness', *New York Times*, 23 February, San Francisco.

Wright, M, Hearps, P, Bolton, D, Brait, P, Bramwell, J & Campbell, R 2011, *Zero Carbon Australia Stationary Energy Plan*, Beyond Zero Emissions and The University of Melbourne, Melbourne.

Xcel Energy Inc 2010, *The Colorado Integrated Solar Project*, Docket No. 09A-015E, Denver, USA.

Yin, H, Ding, J & Yang, X 2014, 'Experimental research on thermal characteristics of a hybrid thermocline heat storage system', *Applied Thermal Engineering*, vol. 62, no. 1, pp. 293–301.

Yin, RK 2009, *Case study research: design and methods*, 4. edn, vol. 4., SAGE Publications, Thousand Oaks, California.

Zedtwitz, P v & Steinfeld, A 2003, 'The solar thermal gasification of coal — energy conversion efficiency and CO2 mitigation potential', *Energy*, vol. 28, no. 5, pp. 441–456.

Zhou, C, Doroodchi, E & Moghtaderi, B 2013, *An in-depth assessment of hybrid solar– geothermal power generation, Energy Conversion and Management*, vol. 74, pp. 88–101. APPENDIX

| | SITY OF ILOGY SYDNEY | | |
|---|--|---|---|
| CSP implementation barriers STEEP analysis | | Participant's background Technology provider | |
| | | | |
| | | Researcher | |
| | | Govern | ıment |
| | | 1 | Barrier rating 5 Moderately important |
| Not important | Moderately important | Very important |] |
| Barrier category | Barrier | Standalone CSP | CSP hybrids |
| | | Standalone Col | |
| | Lack/guality of information, education and awareness | | |
| | Consumer lack of interest in PE | | |
| | Consumer fact of higher energy prices | | |
| | | | |
| | 2 | | |
| | 2 | | |
| | <u>,</u> | | |
| Technical | Technology maturity | | |
| | Technology complexity compared to other RE, coal and gas | | |
| | Generation intermittency/lack of energy storage | | |
| | Network access, capacity and availability | | |
| | Lack of standards and codes | | |
| | Resource assessments/knowledge of best locations | | |
| | 5 | | |
| | 6 | | |
| | 7 | | |
| | 8 | | |
| Ecological | Plant footprint/land use | | |
| | Water consumption (cooling & cleaning) | 1 | |
| | Visual impact | | |
| | Inefficient use of back-up fuels | | |
| | 9 | | |
| | 10 | | |
| | 11 | | |
| | 12 | | |
| Economic | Lack of financial incentives/recognition of externalities | | |
| | High project capital/financing cost and typically large projects | | |
| | Pre-existing investment in existing equipment and infrastructure | | |
| | Abundant and subsidised fossil fuel resources | | |
| | Lower cost renewable energy alternatives | | |
| | 13 | | |
| | 14 | | |
| | 15 | | |
| | 16 | | |
| olicy | Renewable energy policy uncertainty, e.g. carbon pricing, RET | | |
| 5 | Market concentration limits market and PPA access for smaller developers | | |

| Market concentration limits market and PPA access for smaller developers | |
|--|--|
| Lack of coordination at state/national level | |
| Complex plant compliance processes | |
| Lack of government policy focus | |
| Lack of skilled workforce | |
| 17 | |
| 18 | |
| 90 | |
| 20 | |



Research and Innovation Office City Campus Building 1 Level 14 Room 14.31 PO Box 123 Broadway NSW 2007 Australia T +61 2 9514 9681 F +61 2 9514 1244 www.uts.edu.au UTS CRICOS PROVIDER CODE 00099F

5 July 2011

Professor Stuart White Institute for Sustainable Futures CB10.11 UNIVERSITY OF TECHNOLOGY, SYDNEY

Dear Stuart,

UTS HREC 2011-208 – WHITE, Professor Stuart, HELLWIG, Professor Dr. Udo, TADROS, Dr. Amir (for PETERSEIM, Mr Juergen, PhD student) – "Carbon emission reduction potential for electricity and process heat generation via nonconventional fuels"

Thank you for your response to my email dated 22/06/11. Your response satisfactorily addresses the concerns and questions raised by the Committee, and I am pleased to inform you that ethics clearance is now granted.

Your clearance number is UTS HREC REF NO. 2011-208A

Please note that the ethical conduct of research is an on-going process. The *National Statement on Ethical Conduct in Research Involving Humans* requires us to obtain a report about the progress of the research, and in particular about any changes to the research which may have ethical implications. This report form must be completed at least annually, and at the end of the project (if it takes more than a year). The Ethics Secretariat will contact you when it is time to complete your first report.

I also refer you to the AVCC guidelines relating to the storage of data, which require that data be kept for a minimum of 5 years after publication of research. However, in NSW, longer retention requirements are required for research on human subjects with potential long-term effects, research with long-term environmental effects, or research considered of national or international significance, importance, or controversy. If the data from this research project falls into one of these categories, contact University Records for advice on long-term retention.

If you have any queries about your ethics clearance, or require any amendments to your research in the future, please do not hesitate to contact the Ethics Secretariat at the Research and Innovation Office, on 02 9514 9772.

Yours sincerely,

Production Note:

Professor Marion Haas Chairperson UTS Human Research Ethics Committee

THINK.CHANGE.DO

Section 3.4: Ethics clearance from the UTS research ethics committee



Section 4.4.1: Scenario 16 using a solar tower with direct steam generation



Section 4.4.1: Scenario 17 using a solar tower with molten salt and 3h thermal energy storage



Section 4.4.2: Base case scenario without external steam superheating (Scenario 1 AC)



Section 4.4.2: External superheater concept using RDF (Scenario 3 AC)



Section 4.6.1.1: Process diagram for 35.5MWe (net) Swanbank CSP-multiple feedstock hybrid plant case study



Section 4.6.2.1: Process diagram for 30MWe (net) Griffith CSP-EfB case study